VERIFONE SYSTEMS, INC. Form S-4 December 21, 2010 Table of Contents

As filed with the Securities and Exchange Commission on December 21, 2010

Registration No.

UNITED STATES

SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

FORM S-4

REGISTRATION STATEMENT

UNDER

THE SECURITIES ACT OF 1933

VeriFone Systems, Inc.

(Exact Name of Registrant as Specified in its Charter)

Delaware (State or Other Jurisdiction of 3578 (Primary Standard Industrial 04-3692546 (I.R.S. Employer

Identification Number)

Incorporation or Organization)

Classification Code Number)

Table of Contents

2099 Gateway Place, Suite 600

San Jose, California 95110

(408) 232-7800

(Address, Including Zip Code, and Telephone Number, Including Area Code, of Registrant s Principal Executive Offices)

Albert Y. Liu

Senior Vice President and General Counsel

VeriFone Systems, Inc.

2099 Gateway Place, Suite 600

San Jose, California 95110

(408) 232-7800

(Name, Address, Including Zip Code, and Telephone Number, Including Area Code, of Agent for Service)

Copies to:

Scott D. Miller Sarah P. Payne Sullivan & Cromwell LLP 1870 Embarcadero Road Palo Alto, California 94303 (650) 461-5600 Steven D. Pidgeon

Richard S. Millard DLA Piper LLP (US)

2525 East Camelback Rd. Suite 1000

Phoenix, Arizona 85016

(480) 606-5100

APPROXIMATE DATE OF COMMENCEMENT OF PROPOSED SALE OF THE SECURITIES TO THE PUBLIC: As soon as possible after this Registration Statement is declared effective.

If the securities being registered on this Form are being offered in connection with the formation of a holding company and there is compliance with General Instruction G, check the following box.

Table of Contents

If this Form is filed to register additional securities for an offering pursuant to Rule 462(b) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering.

If this Form is a post-effective amendment filed pursuant to Rule 462(d) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering.

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of large accelerated filer, accelerated filer and smaller reporting company in Rule 12b-2 of the Exchange Act.

Large accelerated filer xAccelerated filer "Non-accelerated filer " (Do not check if a smaller reporting company)Smaller reporting company "If applicable, place an X in the box to designate the appropriate rule provision relied upon in conducting this transaction:

Exchange Act Rule 13e-4(i) (Cross-Border Issuer Tender Offer) "

Exchange Act Rule 14d-1(d) (Cross-Border Third-Party Tender Offer) "

CALCULATION OF REGISTRATION FEE

Title of Each Class of Securities to be Registered Common Stock, \$0.01 par value Amount to be
Registered(1)Proposed Maximum
Offering Per Share16,593,522 sharesNot applicable

Proposed Maximum Aggregate Offering Price(2) H \$662,745,268.68

Amount of Registration Fee(2) \$47,253.74

- (1) This Registration Statement relates to the maximum number of shares of common stock, par value \$0.01 per share (VeriFone Common Stock), of VeriFone Systems, Inc. (VeriFone) issuable to holders of common stock, par value \$0.001 per share (Hypercom Common Stock), of Hypercom Corporation (Hypercom) in the proposed merger of Honey Acquisition Co., a wholly owned subsidiary of VeriFone, with and into Hypercom. The amount of VeriFone Common Stock to be registered has been determined by multiplying the maximum number of shares of Hypercom Common Stock that may be outstanding immediately prior to the completion of the transaction by 0.23 (the number of shares of VeriFone Common Stock to be issued for each outstanding share of Hypercom Common Stock).
- (2) Estimated solely for purposes of calculating the registration fee required by the Securities Act of 1933, as amended, and computed pursuant to Rule 457(c) and Rule 457(f) based on (i) \$39.94, the average of the high and low per share prices of VeriFone Common Stock on the New York Stock Exchange on December 20, 2010 and (ii) the maximum number of shares of VeriFone Common Stock to be received by holders of Hypercom Common Stock.

The registrant hereby amends this registration statement on such date or dates as may be necessary to delay its effective date until the registrant shall file a further amendment which specifically states that this registration statement shall thereafter become effective in accordance with Section 8(a) of the Securities Act of 1933 or until the registration statement shall become effective on such date as the Commission, acting pursuant to said Section 8(a), may determine.

The information in this proxy statement/prospectus is not complete and may be changed. VeriFone may not sell these securities until the registration statement filed with the Securities and Exchange Commission is effective. This proxy statement/prospectus is not an offer to sell these securities and it is not soliciting an offer to buy these securities in any state where the offer or sale is not permitted.

SUBJECT TO COMPLETION. DATED DECEMBER 21, 2010

MERGER PROPOSED YOUR VOTE IS VERY IMPORTANT

Dear Stockholders:

We are writing to you today about the proposed merger of Hypercom Corporation (which we refer to as Hypercom) with a subsidiary of VeriFone Systems, Inc. (which we refer to as VeriFone).

If the merger is completed, Hypercom stockholders will receive 0.23 shares of VeriFone common stock for each share of Hypercom common stock they own.

VeriFone common stock is traded on the New York Stock Exchange under the trading symbol PAY. Based on the closing price of VeriFone common stock on , 2010 of \$, the record date for holders of Hypercom common stock to vote at the special meeting, the value of the per share consideration to be received by Hypercom stockholders would be \$ per share of Hypercom common stock. The implied value of the merger consideration will fluctuate as the market price of the VeriFone common stock fluctuates.

Hypercom is holding a special meeting of stockholders (which we refer to as the special meeting) in order to obtain the approval necessary to complete the merger as more fully described in this proxy statement/prospectus. The merger cannot be completed unless the holders of a majority of the outstanding shares of Hypercom common stock entitled to vote at the special meeting approve and adopt the merger agreement and approve the merger. At the special meeting, Hypercom stockholders will be asked to vote on the merger described in the attached proxy statement/prospectus. Under Delaware law, holders of Hypercom common stock are not entitled to statutory dissenters rights.

Only holders of record of Hypercom common stock at the close of business on , 2010 are entitled to attend and vote at the special meeting or any adjournment thereof.

The Hypercom board of directors has (1) reviewed and considered the terms and conditions of the merger agreement, (2) unanimously determined that the merger is fair to, and in the best interests of, Hypercom and its stockholders, considering the fairness opinion of UBS Securities LLC and such other factors as the board of directors has deemed appropriate and (3) unanimously approved the merger agreement, the merger and all of the transactions contemplated by the merger agreement. **The Hypercom board of directors unanimously recommends that Hypercom stockholders vote FOR the proposal to approve and adopt the merger agreement and approve the merger.**

The merger will not be completed unless the stockholders of Hypercom approve the proposal related to the merger. Your vote is very important, regardless of the number of shares you own. Whether or not you plan to attend the special meeting, please vote all proxy cards that you receive as soon as possible to ensure that your shares are represented at the special meeting.

The attached proxy statement/prospectus provides you with detailed information about VeriFone, Hypercom, the merger agreement and the merger. We encourage you to read the entire proxy statement/prospectus carefully, including the <u>Risk Factors</u> section beginning on page 26.

Yours sincerely,

/s/ Norman Stout Norman Stout

Chairman of the Board

Hypercom Corporation

Neither the Securities and Exchange Commission nor any state securities commission has approved or disapproved these securities or determined if the accompanying proxy statement/prospectus is accurate or adequate. Any representation to the contrary is a criminal offense.

This proxy statement/prospectus is dated

, 2010, and is first being mailed to Hypercom stockholders on or about , 2010.

Sources of Additional Information

This proxy statement/prospectus incorporates important business and financial information about VeriFone and Hypercom from documents that are not included in or delivered with this proxy statement/prospectus. Documents relating to VeriFone incorporated by reference are available from VeriFone without charge, excluding all exhibits unless VeriFone has specifically incorporated by reference an exhibit in this proxy statement/prospectus. Documents relating to Hypercom incorporated by reference are available from Hypercom without charge, excluding all exhibits unless Hypercom has specifically incorporated by reference are available from Hypercom without charge, excluding all exhibits unless Hypercom has specifically incorporated by reference an exhibit in this proxy statement/prospectus. You may obtain documents incorporated by reference in this proxy statement/prospectus by requesting them in writing or by telephone or e-mail from the applicable company at the following addresses and telephone numbers:

VeriFone Systems, Inc.	Hypercom Corporation
Attention: Investor Relations	Attention: Investor Relations
2099 Gateway Place, Suite 600	8888 East Raintree Drive, Suite 300
San Jose, California 95110	Scottsdale, Arizona, 85260
ir@verifone.com	stsujita@hypercom.com
(408) 232-7979 If you would like to request documents from VeriFone or Hypercom, please do so by before the special meeting.	(480) 642-5000 , in order to ensure that you receive them

If you have any questions about the special meeting, the merger or this proxy statement/prospectus or need additional copies of this proxy statement/prospectus or the documents incorporated by reference into this proxy statement/prospectus, please send your request in writing or by telephone to Innisfree M&A Incorporated, Hypercom s information agent and proxy solicitor, at the following address and telephone numbers:

Innisfree M&A Incorporated

501 Madison Avenue, 20th Floor

New York, New York 10022

(888) 750-5834 (toll-free)

Banks & Brokers Call Collect:

(212) 750-5833

For a more detailed description of the information incorporated by reference into this proxy statement/prospectus and how you may obtain it, see Where You Can Find More Information on page 117.

ABOUT THIS DOCUMENT

This document, which forms part of a registration statement on Form S-4 filed with the SEC by VeriFone (File No.), constitutes a prospectus of VeriFone under Section 5 of the Securities Act of 1933, as amended, which we refer to as the Securities Act, with respect to the VeriFone common stock to be issued to Hypercom stockholders as required by the merger agreement. This document also constitutes a proxy statement of Hypercom under Section 14(a) of the Securities Exchange Act of 1934, as amended, which we refer to as the Exchange Act. It also constitutes a notice of meeting with respect to the special meeting of Hypercom stockholders, at which Hypercom stockholders will be asked to consider and vote upon a proposal to approve and adopt the merger agreement and approve the merger.

VeriFone has supplied all information contained or incorporated by reference in this proxy statement/prospectus relating to VeriFone and Hypercom has supplied all such information relating to Hypercom. VeriFone and Hypercom have not authorized anyone to provide you with information that is different from what is contained in this proxy statement/prospectus.

HYPERCOM CORPORATION

8888 EAST RAINTREE DRIVE, SUITE 300

SCOTTSDALE, ARIZONA, 85260

(480) 642-5000

NOTICE OF SPECIAL MEETING OF STOCKHOLDERS

TO BE HELD ON , 2011

To all Hypercom stockholders:

Notice is hereby given that a special meeting of stockholders of Hypercom Corporation, a Delaware corporation, will be held at the principal executive offices of Hypercom located at 8888 East Raintree Drive, Suite 300, Scottsdale, Arizona 85260 on , , 2011 at 9:00 a.m. Arizona time for the following purposes:

- To approve and adopt the Agreement and Plan of Merger, dated as of November 17, 2010, by and among Hypercom, VeriFone Systems, Inc., a Delaware corporation, and Honey Acquisition Co., a Delaware corporation and a wholly owned subsidiary of VeriFone, and approve the merger contemplated by the merger agreement;
- 2. To approve the adjournment of the special meeting of stockholders of Hypercom, if necessary, for any purpose, including to solicit additional proxies if there are not sufficient votes to approve and adopt the merger agreement and approve the merger at the time of the special meeting of stockholders of Hypercom; and
- 3. To conduct any other business that properly comes before the special meeting of stockholders of Hypercom or any adjournment or postponement of such special meeting.

Only stockholders of record on the close of business on , 2010, the record date, are entitled to vote at the special meeting or any postponement or adjournment of the meeting. You are cordially invited to the meeting.

The presence of a majority of the shares entitled to vote, present in person or represented by proxy, will constitute a quorum at a meeting of stockholders.

By order of the Board of Directors of Hypercom Corporation

/s/ Norman Stout

Norman Stout Chairman of the Board

Scottsdale, Arizona

, 2010

REGARDLESS OF THE NUMBER OF SHARES OF HYPERCOM COMMON STOCK YOU OWN OR WHETHER YOU PLAN TO ATTEND THE MEETING, IT IS IMPORTANT THAT YOUR SHARES BE REPRESENTED AND VOTED. THE MERGER CANNOT BE COMPLETED UNLESS THERE IS A QUORUM PRESENT OR REPRESENTED AT THE SPECIAL MEETING AND THE HOLDERS OF A MAJORITY OF THE OUTSTANDING SHARES OF HYPERCOM COMMON STOCK ENTITLED TO VOTE AT THE SPECIAL MEETING APPROVE AND ADOPT THE MERGER AGREEMENT AND APPROVE THE MERGER. THEREFORE, WE URGE YOU TO COMPLETE, SIGN, DATE AND RETURN THE ACCOMPANYING PROXY CARD IN THE ENCLOSED PRE-ADDRESSED, POSTAGE-PAID ENVELOPE. RETURNING THE PROXY CARD DOES NOT DEPRIVE YOU OF YOUR RIGHT TO ATTEND THE MEETING AND TO VOTE YOUR SHARES IN PERSON. YOUR VOTE IS VERY IMPORTANT.

YOUR BOARD OF DIRECTORS UNANIMOUSLY RECOMMENDS THAT YOU VOTE FOR APPROVAL AND ADOPTION OF THE MERGER AGREEMENT AND APPROVAL OF THE MERGER.

PLEASE DO NOT RETURN YOUR HYPERCOM SHARE CERTIFICATES WITH YOUR ENCLOSED PROXY.

TABLE OF CONTENTS

	Page
Questions and Answers about the Merger	1
Questions and Answers about the Special Meeting	3
Summary	6
Summary of VeriFone Selected Financial Data	15
Summary of Hypercom Selected Financial Data	17
Selected Unaudited Pro Forma Condensed Combined Financial Data	20
Comparative Historical and Pro Forma Per Share Data	22
Comparative Per Share Market Price Data	23
Dividend Data	25
Risk Factors	26
Risks Related to the Merger	26
Risks Related to the Combined Company after the Merger	32
Risks Related to the Combined Company s Industry	48
Risks Related to VeriFone s Capital Structure	49
Cautionary Statement Concerning Forward-Looking Statements	53
Information about the Special Meeting	54
General: Date: Time and Place	54
Purpose of the Special Meeting	54
Record Date; Voting Power	54
Recommendation of the Board of Directors of Hypercom Concerning the Merger	54
Method of Voting; Record Date; Stock Entitled to Vote; Quorum	54
Adjournment and Postponement	55
Required Vote	55
Share Ownership of Hypercom Directors and Executive Officers	55
How to Vote	55
To Attend the Special Meeting	56
Contact for Questions and Assistance in Voting	57
Other Matters	57
The Companies	58
VeriFone	58
Hypercom	58
The Merger	59
General	59
Merger Consideration	59
	59
Schedule of Important Dates Background of the Merger	59
VeriFone s Reasons for the Merger	59 69
Hypercom s Reasons for the Merger; Recommendation of the Hypercom Board of Directors	09 70
<u>Opinion of Hypercom</u> s Financial Advisor	70
Material U.S. Federal Income Tax Consequences	72
	70
Accounting Treatment of the Merger	
Regulatory Matters	78
Federal Securities Laws Consequences	79 70
New York Stock Exchange Listing	79 70
Litigation Related to the Merger	79 70
No Dissenters or Appraisal Rights	79
Executives: Executive Compensation: Stock Ownership of Directors, Executive Officers and Five Percent Stockholders	80
Interests of Certain Persons in the Merger	80
Common Stock	80
Hypercom Stock Options, Restricted Stock and Hypercom Warrants	80

	Page
Change of Control Plan	81
Employee Benefits	82
Support Agreement	82
Indemnification and Insurance	82
The Merger Agreement	83
Structure of the Merger	83
Completion and Effectiveness of the Merger	83
Conversion of Hypercom Common Stock in the Merger	83
Exchange Procedures	84
Fractional Shares	84
Dividends and Distributions	84
Transfer of Ownership and Lost Stock Certificates	84
Hypercom s Representations and Warranties	84
VeriFone s and Honey Acquisition Co. s Representations and Warranties	85
Hypercom s Conduct of Business Before Completion of the Merger	86
No Solicitation of Acquisition Proposals by Hypercom	88
Changes in Hypercom s Board of Directors Recommendation	89
Solicitation of Votes	89
Effect on Hypercom Stock Options and Hypercom Restricted Stock Awards	90
Effect on Hypercom Warrants	90
Employee Benefits	90
Conditions to Completion of the Merger	91
Determination of Material Adverse Effect	93
Reasonable Best Efforts Covenant	94
Additional Covenants	94
Waiver and Amendment of the Merger Agreement	95
Termination of the Merger Agreement	96
Termination Fee Payable by Hypercom	97
Termination Fee Payable by VeriFone	98
Indemnification and Insurance	98
Third-Party Beneficiaries	98
Other Material Agreements Relating to the Merger	99
Support Agreement	99
Unaudited Pro Forma Condensed Combined Financial Information	100
Description of VeriFone Capital Stock	109
Common Stock	109
Preferred Stock	109
No Stockholder Rights Plan	109
Transfer Agent and Registrar	109
Comparison of Rights of VeriFone Stockholders and Hypercom Stockholders	110
Validity of Common Stock	116
Experts	116
Future Stockholder Proposals and Director Nominations	116
Delivery of Documents to Stockholders Sharing an Address	117
Where You Can Find More Information	117
VeriFone SEC Filings	118
Hypercom SEC Filings	119
Annexes	
A Agreement and Plan of Merger	A-1
<u>B</u> Support Agreement	B-1
C Opinion of Hypercom s Financial Advisor	C-1

Questions and Answers about the Merger

The following are some of the questions that you, as a stockholder of Hypercom, may have, and answers to those questions. These questions and answers, as well as the following summary, are not meant to be a substitute for the information contained in the remainder of this proxy statement/prospectus, and this information is qualified in its entirety by the more detailed descriptions and explanations contained elsewhere in this proxy statement/prospectus. We urge you to read this proxy statement/prospectus in its entirety prior to making any decision.

Q: Why am I receiving this proxy statement/prospectus?

A: VeriFone and Hypercom have entered into a merger agreement. Upon completion of the merger, Hypercom will become a wholly owned subsidiary of VeriFone. In order to complete the merger, Hypercom stockholders must approve the proposal relating to the merger at the special meeting. The Hypercom proposal to approve and adopt the merger agreement and approve the merger must be approved by holders of a majority of the outstanding shares of Hypercom common stock entitled to vote at the special meeting.

We have included in this proxy statement/prospectus important information about the merger, the merger agreement and the special meeting of Hypercom stockholders. You should read this information carefully and in its entirety. We have also attached a copy of the merger agreement as Annex A. The enclosed voting materials allow you to vote your shares without attending the applicable special meeting. **Your vote is very important and we encourage you to vote your proxy as soon as possible.**

Q: What do I need to do now?

A: Read and consider the information contained in this proxy statement/prospectus carefully, and then please vote your shares as soon as possible by returning your proxy so that your shares may be represented at the special meeting. Your proxy card or voting instruction card must be received, or you must vote using the telephone or Internet if available, no later than , Eastern time, on , 2011 in order for your shares to be voted at the special meeting, unless you attend and vote at the special meeting.

Q: What will Hypercom stockholders receive in the merger?

A: If the merger is completed, Hypercom stockholders will receive 0.23 shares of VeriFone common stock for each share of Hypercom common stock they own.

The exchange ratio will not change even if the market prices of Hypercom common stock or VeriFone common stock fluctuate. Therefore, the market value of the VeriFone common stock that Hypercom stockholders will receive if the merger is completed will fluctuate up or down with fluctuations in the market price of VeriFone common stock.

Q: What is the aggregate consideration to be paid by VeriFone for all of the outstanding shares of Hypercom common stock?

A: Based on the number of shares of Hypercom common stock outstanding as of the record date, VeriFone will issue in the aggregate an estimated shares of VeriFone common stock. Assuming the exercise of all Hypercom options and warrants outstanding as of the record date, VeriFone would issue an additional shares of VeriFone common stock in the merger.

Q: When do you expect the merger to be completed?

A: We are working towards completing the merger as quickly as reasonably possible. Several conditions must be satisfied or waived before the merger is completed. We hope to complete the merger promptly after obtaining certain regulatory approvals, and the satisfaction or waiver of the other conditions to the merger. We currently expect the merger to close in the second half of 2011. See the section of this proxy statement/prospectus titled The Merger Agreement Conditions to Completion of the Merger for a summary description of these conditions.

Q: What happens if the merger is not completed?

A: If the merger agreement is not adopted by Hypercom stockholders or if the merger is not completed for any other reason, you will not receive any payment for your shares of Hypercom common stock in connection with the merger. Instead, Hypercom will remain an independent public company and Hypercom common stock will continue to be listed and traded on the NYSE. If the merger agreement is terminated under specified circumstances, Hypercom may be required to pay VeriFone a termination fee of \$12.2 million, or VeriFone may be required to pay Hypercom a termination fee of \$28.4 million or \$30.4 million, as described under The Merger Agreement Termination Fee Payable by Hypercom and Termination Fee Payable by VeriFone.

Q: Are Hypercom stockholders entitled to dissenters or appraisal rights?

A: No. Under the Delaware General Corporation Law, holders of Hypercom common stock are not entitled to statutory dissenters or appraisal rights in connection with the merger.

Q: Will Hypercom stockholders be able to trade any VeriFone common stock that they receive in the merger?

A: Yes. The VeriFone common stock that Hypercom stockholders will receive will be freely tradable, unless held by an affiliate of VeriFone. VeriFone s common stock is listed on the New York Stock Exchange under the symbol PAY.

Q: What will happen to unexercised Hypercom stock options, restricted stock and Hypercom warrants?

A: Under the terms of the merger agreement, prior to the closing of the merger, Hypercom will take actions to fully vest all Hypercom stock options and restricted stock awards that were outstanding as of November 17, 2010. In addition, certain senior officers are entitled to vesting of any equity awards held upon termination of employment without cause or resignation for good cause occurring prior to the merger or within 12 months thereafter. See Interests of Certain Persons in the Merger Change of Control Plan. Following the merger, each outstanding stock option under Hypercom s stock option plans will be converted into an option to purchase a number of shares of VeriFone common stock equal to the product of (i) the number of shares of Hypercom common stock subject to such option immediately prior to the effective time of the merger and (ii) 0.23, subject to certain adjustments that may be required to comply with U.S. tax law. The exercise price of each option will equal the exercise price prior to the merger, divided by 0.23. Each option will otherwise continue to be governed by the same terms and conditions as applicable under Hypercom s stock option plan. Holders of restricted stock will receive the same merger consideration as other holders of Hypercom common stock: 0.23 shares of VeriFone common stock for each share of Hypercom common stock, with any unvested stock issued after November 17, 2010 and not subject to acceleration by the terms of the applicable award or under the Hypercom Change of Control Plan continuing to be subject to vesting restrictions.

At the effective time of the merger, each outstanding warrant to purchase shares of Hypercom common stock will be converted into a warrant to acquire a number of shares of VeriFone common stock equal to the number of shares of Hypercom common stock subject to such warrant multiplied by 0.23. The exercise price of each warrant will equal the exercise price prior to the merger, divided by 0.23. Following the effective time of the merger, each Hypercom warrant will continue to be governed by the same terms and conditions as were applicable under such warrant immediately prior to the effective time of the merger.

Q: Are there any risks related to the proposed transaction or any risks related to owning VeriFone common stock?

A: Yes. You should carefully review the section entitled Risk Factors beginning on page 26.

Table of Contents

Q: Will the rights of Hypercom stockholders change as a result of the merger?

A: Yes. When the merger is complete, Hypercom stockholders will become VeriFone stockholders and their rights as stockholders will be governed by VeriFone s certificate of incorporation and bylaws. There are numerous differences between the rights and responsibilities of a stockholder of Hypercom and the rights and responsibilities of a stockholder of VeriFone. Please see Comparison of Rights of VeriFone Stockholders and Hypercom Stockholders beginning on page 110 for a discussion of the different rights associated with VeriFone common stock.

Questions and Answers about the Special Meeting

Q: When and where will the special meeting be held?

A: The special meeting is scheduled to be held at 8888 East Raintree Drive, Suite 300, Scottsdale, Arizona 85260, at 9:00 a.m., Arizona time, on , , 2011.

Q: Who is entitled to vote at the special meeting?

A: Hypercom has fixed , 2010 as the record date for the special meeting. If you are a Hypercom stockholder at the close of business on the record date, you are entitled to vote on matters that come before the special meeting. However, a Hypercom stockholder may only vote his or her shares if he or she is present in person or is represented by proxy at the special meeting.

Q: What are the recommendations of the Hypercom board of directors?

A: The Hypercom board of directors has approved the merger agreement, the merger and the related transactions contemplated by the merger agreement and determined that these transactions are fair to and in the best interests of its stockholders.

The Hypercom board of directors recommends that Hypercom stockholders vote FOR the proposal to approve and adopt the merger agreement and approve the merger and FOR the adjournment of the special meeting, if necessary, for any purpose, including to solicit additional proxies in favor of the approval and adoption of the merger agreement and approval of the merger. See The Merger Hypercom's Reasons for the Merger; Recommendation of the Hypercom Board of Directors on page 70.

Q: How can I vote?

A: If you are entitled to vote at the special meeting, you can vote in person at the special meeting, or you can vote by proxy before the special meeting. Even if you plan to attend the special meeting, we encourage you to vote your shares by proxy as soon as possible. After carefully reading and considering the information contained in this proxy statement/prospectus, please submit your proxy in accordance with the instructions set forth on the enclosed proxy card. For detailed information, please see Information about the Special Meeting How to Vote beginning on page 55.

The vote required to approve and adopt the merger agreement and approve the merger at the special meeting is a majority of the outstanding shares of Hypercom common stock entitled to vote.

Q: What happens if a Hypercom stockholder does not indicate how to vote on the proxy card?

A: If Hypercom stockholders do not include instructions on how to vote their properly signed and dated proxy card, their shares will be voted FOR the approval and adoption of the merger agreement and approval of the merger. They would also be voted FOR the adjournment of the special meeting, if necessary, for any purpose, including to solicit additional proxies in favor of the approval and adoption of the merger agreement and approval of the proxy holders, on any other business that may properly come before the special meeting.

Q: What happens if I do not vote?

A: Approval of the proposal to be presented at the special meeting requires the affirmative vote of holders of a majority of the outstanding shares of Hypercom common stock entitled to vote at the special meeting. Stockholders who represent a majority of the shares of Hypercom entitled to vote must be present in person or represented by proxy in order to constitute a quorum to conduct business at a meeting of stockholders. Abstentions and broker non-votes count as present for establishing a quorum.

If a quorum is present at the special meeting, your failure to return your proxy card or vote in person at the meeting or your abstention from voting will have the same effect as a vote AGAINST the approval and adoption of the merger agreement and approval of the merger.

Q: If my shares are held in street name by my broker, will my broker vote my shares for me?

A: If your shares are held in a stock brokerage account or by a bank or other nominee, you are considered the beneficial holder of the shares held for you in what is known as street name. If this is the case, this proxy statement/prospectus has been forwarded to you by your brokerage firm, bank or other nominee, or their agent. As the beneficial holder, you have the right to direct your broker, bank or other nominee as to how to vote your shares. If you do not provide your broker, bank or other nominee with instructions on how to vote your street name shares, your broker, bank or other nominee will not be permitted to vote them on the proposal to approve and adopt the merger agreement and approve the merger. You should therefore be sure to provide your broker, bank or other nominee with instructions on how to vote your shares.

Q: Have any Hypercom equityholders agreed to vote for the merger agreement and the merger?

A: In connection with the merger, VeriFone entered into a support agreement with FP Hypercom Holdco, LLC and Francisco Partners II, L.P., pursuant to which FP Hypercom Holdco, LLC and Francisco Partners II, L.P. agreed to vote any shares of Hypercom common stock held by them in favor of the merger. As of December 17, 2010, FP Hypercom Holdco, LLC holds a warrant to purchase 10,544,000 shares of Hypercom common stock but based on information received from Francisco Partners II, L.P. neither Francisco Partners II, L.P. nor FP Hypercom Holdco, LLC holds any shares of Hypercom common stock. Accordingly, assuming FP Hypercom Holdco, LLC does not exercise its warrant prior to the record date and neither Francisco Partners II, L.P. nor FP Hypercom Holdco, LLC acquires any shares of Hypercom common stock, FP Hypercom Holdco, LLC and Francisco Partners II, L.P. will not be entitled to vote at the special meeting. The support agreement is attached as Annex B to this proxy statement/prospectus.

Q: Can I change my vote after I have signed and returned my proxy card or voting instruction card?

A: Yes. You can change your vote at any time before your proxy is voted at the special meeting. You can do this in one of three ways:

you can send a written notice stating that you would like to revoke your proxy, provided that the notice is received at least 24 hours prior to the time set for the special meeting or is presented at the special meeting to the chairman of the meeting;

you can complete and submit a new proxy card dated later than the first proxy card, provided that the new proxy card is received at least 24 hours prior to the time set for the special meeting or is presented at the special meeting to the chairman of the meeting; or

you can attend the special meeting, and file a written or make an oral notice of revocation of your proxy with the chairman of the meeting and then vote in person.

Your attendance alone will not revoke your proxy. If you have instructed a broker to vote your shares, you must follow your broker s directions to change those instructions.

Q: If a stockholder purchased Hypercom common stock after the record date, may the stockholder vote these shares at the special meeting?

A: A stockholder is not entitled to vote shares of Hypercom common stock purchased after the record date.

Q: Should Hypercom stockholders send in their Hypercom share certificates now?

A: No, please do not send Hypercom share certificates together with the proxy card.

Q: What should Hypercom stockholders do if they have questions?

A: If Hypercom stockholders have any questions about the special meeting, the merger or this proxy statement/prospectus, or if they need additional copies of this proxy statement/prospectus or the enclosed proxy card, they should contact Hypercom s information and proxy solicitation agent, Innisfree M&A Incorporated at:

Innisfree M&A Incorporated

501 Madison Avenue, 20th Floor

New York, New York 10022

(888) 750-5834 (toll-free)

Banks & Brokers Call Collect:

(212) 750-5833

In addition, if Hypercom stockholders have any questions about the merger or if they need additional copies of this proxy statement/prospectus, they may contact:

Hypercom Corporation

8888 East Raintree Drive, Suite 300

Attention: Investor Relations

Scottsdale, Arizona, 85260

stsujita@hypercom.com

Telephone: (480) 642-5000

If their broker holds their shares, they may also call their broker for additional information.

VeriFone will also provide you with copies of the information relating to VeriFone, without charge, upon written or oral request to:

VeriFone Systems, Inc.

Attention: Investor Relations

2099 Gateway Place, Suite 600

San Jose, California 95110

ir@verifone.com

(408) 232-7979

Summary

This summary highlights selected information from this proxy statement/prospectus and may not contain all of the information that is important to you. To understand the merger fully, you should read carefully this entire proxy statement/prospectus and the documents we refer to. See Where You Can Find More Information on page 117. The merger agreement is attached as Annex A to this proxy statement/prospectus and is incorporated herein by reference. We encourage you to read it, as it is the most important legal document that governs the merger. We have included page references in parentheses to direct you to a more complete description contained elsewhere in this proxy statement/prospectus of the topics presented in this summary. In this proxy statement/prospectus, unless stated to the contrary or the context requires otherwise, the terms we, our, ours, and us refer to VeriFone and Hypercom.

The Companies

(Page 58)

VeriFone Systems, Inc.

2099 Gateway Place, Suite 600

San Jose, California 95110

(408) 232-7800

VeriFone Systems, Inc. is a global leader in secure electronic payment solutions. VeriFone provides expertise, solutions, and services that add value to the point of sale with merchant-operated, consumer-facing, and self-service payment systems for the financial, retail, hospitality, petroleum, transportation, government, and healthcare vertical markets. Since 1981, VeriFone has designed and marketed system solutions that facilitate the long-term shift toward electronic payment transactions and away from cash and checks. VeriFone s net revenues were \$1,001.5 million in the year ended October 31, 2010.

Hypercom Corporation

8888 East Raintree Drive, Suite 300

Scottsdale, Arizona 85260

(480) 642-5000

Hypercom Corporation is one of the largest global providers of complete electronic payment and transaction solutions and value-added services at the point of transaction. Hypercom s customers include domestic and international financial institutions, electronic payment processors, transaction network operators, retailers, system integrators, independent sales organizations and distributors. Hypercom also sells its products to companies in the hospitality, transportation, healthcare, prepaid card and restaurant industries. Hypercom products enable its customers to accept a wide range of payments and other transactions, including credit cards, signature and PIN (Personal Identification Number) based debit cards, contactless identification and near field communications, stored-value cards, and electronic benefits transfer. Hypercom delivers convenience and value to businesses that depend on reliable, secure, high-speed and high-volume electronic transactions.

Honey Acquisition Co.

2099 Gateway Place, Suite 600

San Jose, California 95110

(408) 232-7800

Honey Acquisition Co. is a newly-formed, wholly owned subsidiary of VeriFone. VeriFone formed this subsidiary as a Delaware corporation solely to effect the merger, and this subsidiary has not conducted and will not conduct any business during any period of its existence.

Special Meeting

(Page 54)

The special meeting will be held on , , 2011, at 8888 East Raintree Drive, Suite 300, Scottsdale, Arizona 85260, commencing at 9:00 a.m., Arizona time.

Stockholders Entitled to Vote at the Special Meeting; Required Vote

(Pages 54 and 55)

The close of business on , 2010 has been set as the record date for the special meeting. Only Hypercom stockholders on the record date are entitled to notice of and to vote at the special meeting. Each share of Hypercom common stock will be entitled to one vote on each matter to be acted upon at the special meeting.

The affirmative vote of holders of a majority of the outstanding shares of Hypercom common stock entitled to vote at the special meeting is required to approve and adopt the merger agreement and approve the merger.

The Merger

(Page 59)

In the merger, Honey Acquisition Co. will be merged with and into Hypercom. Hypercom will be the surviving corporation and will become a wholly owned subsidiary of VeriFone.

Consideration

(Page 59)

If the merger agreement is adopted and the merger is completed, each holder of shares of common stock of Hypercom will receive for each such share 0.23 shares of VeriFone common stock.

Fractional Shares

(Page 59)

Hypercom stockholders will not receive fractional shares of VeriFone common stock. Instead, they will receive the cash value, without interest, of any fractional share of VeriFone common stock that they might otherwise have been entitled to receive.

Treatment of Stock Options, Restricted Stock and Warrants

(Page 90)

Under the terms of the merger agreement, prior to the closing of the merger, Hypercom will take actions to fully vest all Hypercom stock options and restricted stock awards that were outstanding as of November 17, 2010. In addition, certain senior officers are entitled to vesting of any equity awards held upon termination of employment without cause or resignation for good cause occurring prior to the merger or within 12 months thereafter. See Interests of Certain Persons in the Merger Change of Control Plan. Following the merger, each outstanding stock option under Hypercom s stock option plans will be converted into an option to purchase a number of shares of VeriFone common stock equal to the product of (i) the number of shares of Hypercom common stock subject to such option immediately prior to the effective time of the merger and (ii) 0.23, subject to certain adjustments that may be required to comply with U.S. tax law. The exercise price of each option will equal the exercise price prior to the merger, divided by 0.23. Holders of restricted stock will receive the same merger consideration as other holders of Hypercom common stock: 0.23 shares of VeriFone common stock for each share of Hypercom common stock, with any unvested stock issued after November 17, 2010 and not subject to acceleration by the terms of the applicable award or under the Hypercom Change of Control Plan continuing to be subject to vesting restrictions.

At the effective time of the merger, each outstanding warrant to purchase shares of Hypercom common stock will be converted into a warrant to acquire a number of shares of VeriFone common stock equal to the

product of (i) the number of shares of Hypercom common stock subject to such warrant and (ii) 0.23. The exercise price of each warrant will equal the exercise price prior to the merger, divided by 0.23.

The Merger Agreement

(Page 83)

The merger agreement is attached as Annex A to this proxy statement/prospectus. You should read the merger agreement in its entirety. It is the most important legal document governing the merger.

Hypercom s Reasons for the Merger; Recommendation of the Hypercom Board of Directors

(Page 70)

After careful consideration, the Hypercom board of directors approved the merger agreement. The Hypercom board of directors recommends that Hypercom stockholders vote FOR the approval and adoption of the merger agreement and the approval of the merger.

In reaching its decision to approve the merger agreement and to recommend that Hypercom stockholders vote to approve and adopt the merger agreement and approve the merger, the Hypercom board of directors consulted with Hypercom s management and Hypercom s financial and legal advisors and considered a number of strategic, financial and other considerations referred to under The Merger Hypercom s Reasons for the Merger; Recommendation of the Hypercom Board of Directors beginning on page 70.

Opinion of Hypercom s Financial Advisor

(Page 72)

UBS Securities LLC delivered its opinion to the Hypercom board of directors that, as of November 16, 2010 and based upon and subject to various assumptions made, matters considered and limitations described in its opinion, the exchange ratio provided for in the merger was fair, from a financial point of view, to the holders of Hypercom s common stock (other than FP Hypercom Holdco, LLC and its affiliates). UBS Securities LLC provided its opinion for the benefit of Hypercom s board of directors in connection with, and for the purpose of, its evaluation of the exchange ratio from a financial point of view and did not address any other aspect of the merger. The UBS Securities LLC opinion is not a recommendation as to how any holder of Hypercom common stock should vote or act with respect to the merger. FP Hypercom Holdco, LLC maintains two designees on the board of directors of Hypercom.

The full text of the written opinion of UBS Securities LLC, dated November 16, 2010, which sets forth the assumptions made, procedures followed, matters considered and limitations on the review undertaken by UBS Securities LLC, is attached as Annex C and is incorporated into this proxy statement/prospectus by reference. UBS Securities LLC is entitled to receive a transaction fee from Hypercom payable upon completion of the merger. We encourage you to read the opinion of UBS Securities LLC in its entirety.

Interests of Certain Persons/Share Ownership by Directors and Executive Officers of Hypercom

(Page 80)

When considering the recommendation by Hypercom s board of directors to vote FOR the proposal to approve and adopt the merger agreement and approve the merger, you should be aware that some officers of Hypercom have interests in the merger that may be different from your interests. No executive officers of Hypercom have executed employment terms sheets or letters with VeriFone.

As of December 17, 2010, the directors and executive officers of Hypercom and their affiliates, as a group, held approximately 1.47% of the outstanding shares of Hypercom common stock, including restricted stock and

excluding stock options. The vote of holders of a majority of the outstanding shares of Hypercom common stock entitled to vote at the special meeting is required to approve and adopt the merger agreement and approve the merger.

Ownership of VeriFone Following the Merger

Hypercom stockholders collectively will receive an estimated 12,878,363 shares of VeriFone common stock in the merger based on the number of shares of Hypercom common stock outstanding, excluding shares issuable upon exercise of options and warrants outstanding, on December 17, 2010.

Based on the number of shares of VeriFone and Hypercom common stock, including restricted stock, outstanding as of December 17, 2010 existing Hypercom stockholders will own 12.8% of the shares of VeriFone common stock outstanding immediately after the merger.

Stock Exchange Listings

(Page 79)

If the merger is completed, Hypercom stockholders will be able to trade the shares of VeriFone common stock they receive in the merger on the New York Stock Exchange.

Conditions to Completion of the Merger

(Page 91)

Mutual Conditions. The obligations of VeriFone and Hypercom to complete the merger are subject to the satisfaction or waiver, at or prior to the effective time of the merger, of each of the following conditions, among others:

the adoption of the merger agreement by Hypercom s stockholders;

unless VeriFone and Hypercom have agreed after consultation with their respective outside counsel that no filing under the HSR Act is required, the waiting period applicable to the completion of the merger under the Hart-Scott-Rodino Antitrust Improvement Act of 1976, or HSR Act, having expired or been terminated, and all governmental consents and any other notification, waiting period, or approval requirements under the comparable antitrust or competition laws of other applicable foreign jurisdictions in which either party has operations or from which either party derives revenues (which in either case are not de minimis) and which are legally required to be made or obtained at or prior to the effective time of the merger having been made or obtained without the imposition of any term, condition or consequence, subject to certain exceptions;

if one or more asset purchase agreements have been entered into for asset dispositions in connection with regulatory matters, then (i) all of the conditions to the closing under such asset purchase agreements having been satisfied or waived, and (ii) Hypercom and the applicable transferee having effectuated the applicable disposition;

the approval for listing on the NYSE of the VeriFone common stock to be issued pursuant to the merger;

the absence of any statute, law, ordinance, rule, regulation, judgment, decree, injunction or other order that restrains, enjoins or otherwise prohibits completion of the merger or the other transactions contemplated by the merger agreement;

the effectiveness of the registration statement of which this proxy statement/prospectus forms a part;

receipt by each of Hypercom and VeriFone of written opinions from their respective counsel to the effect that, for U.S. federal income tax purposes, the merger will constitute a reorganization within the meaning of Section 368 of the Internal Revenue Code;

the accuracy of the representations and warranties made in the merger agreement by the other party, subject to certain materiality thresholds;

performance in all material respects by the other party of the obligations required to be performed by it at or prior to the completion of the merger; and

the absence of a material adverse effect (as defined in the merger agreement) with respect to the other party. *Additional VeriFone Conditions.* VeriFone s obligation to complete the merger is also subject to the satisfaction or waiver by VeriFone at or prior to the effective time of the merger of the following condition, among others:

no court or governmental entity of competent jurisdiction having instituted any suit, action or proceeding seeking to (i) prohibit, limit, restrain or impair VeriFone s ability to own or operate or to retain or change all or a material portion of the assets, licenses, operations, rights, product lines, businesses or interest of Hypercom or of VeriFone, other than with respect to the disposition of businesses or assets that, in the aggregate, would not exceed the threshold of \$124 million of aggregate gross revenues during the 2009 fiscal year as set forth in the merger agreement, or (ii) prohibit or limit VeriFone s ability to vote, transfer, receive dividends or otherwise exercise full ownership rights with respect to Hypercom.

Regulatory Approvals and Reasonable Best Efforts

(Page 94)

VeriFone and Hypercom cannot complete the merger unless they receive certain regulatory approvals. Under the merger agreement and subject to the terms and conditions of the merger agreement, VeriFone and Hypercom have each agreed to cooperate with each other and use their respective reasonable best efforts to take or cause to be taken all actions, and do or cause to be done all things, reasonably necessary, proper or advisable on its part under the merger agreement and applicable laws to complete the merger.

VeriFone has agreed that reasonable best efforts includes a requirement by VeriFone to agree to sell or otherwise dispose of, and to sell or otherwise dispose of, businesses or assets that, subject to certain exceptions, in the aggregate, produced aggregate gross revenues in an amount not in excess of \$124 million for VeriFone or Hypercom during the 2009 fiscal year, if such action should be reasonably necessary for certain closing conditions in the merger agreement to be met.

Termination

(Page 96)

VeriFone and Hypercom can agree to terminate the merger agreement at any time prior to the effective time of the merger, whether before or after the adoption of the merger agreement by Hypercom s stockholders. In addition, either VeriFone or Hypercom can terminate the merger agreement, if any of the following occurs:

the merger is not completed by August 31, 2011 (subject to extension to November 30, 2011 by written notice of VeriFone if, as of that date, all conditions to the merger agreement have been satisfied or waived, other than certain conditions relating to regulatory consents or the absence of certain litigation proceedings);

Hypercom s stockholders do not approve and adopt the merger agreement and approve the merger at the special meeting or at any adjournment or postponement thereof at which a vote upon the merger agreement and merger was taken; or

any order permanently restraining, enjoining or otherwise prohibiting completion of the merger becomes final and non-appealable.

In addition, Hypercom may terminate the merger agreement at any time prior to the effective time of the merger, whether before or after the adoption of the merger agreement by Hypercom stockholders, if

at any time prior to the time of the adoption of the merger agreement by Hypercom stockholders, (i) Hypercom s board of directors authorizes Hypercom, subject to complying with the terms of the merger agreement, to enter into an alternative acquisition agreement with respect to a superior proposal not obtained in violation of the merger agreement, and (ii) Hypercom prior to such termination pays to VeriFone the \$12.2 million termination fee; or

there has been a breach of any representation, warranty, covenant or agreement made by VeriFone in the merger agreement, or any such representation and warranty has become untrue after the date of the merger agreement, in each case such that certain conditions to Hypercom s obligation to complete the merger agreement would not be satisfied and such breach or failure to be true is not curable or, if curable, is not cured within 30 days after written notice thereof is given by Hypercom to VeriFone.

In addition, VeriFone may terminate the merger agreement at any time prior to the effective time of the merger, if:

Hypercom s board of directors has withheld, withdrawn or modified its recommendation to Hypercom s stockholders to approve and adopt the merger agreement or has, following a public announcement of opposition to the merger or the other transactions contemplated by the merger agreement made by a person holding more than five percent of the Hypercom common stock, failed to reconfirm its recommendation to Hypercom s stockholders to approve and adopt the merger agreement within the period of time beginning upon Hypercom s receipt of a written request by VeriFone to do so and ending on the date that is ten business days after Hypercom s receipt of such request (unless VeriFone has already made two such requests and Hypercom has complied with such requests);

either (i) Hypercom has failed to convene and hold the meeting of stockholders prior to the later of the date that is 45 days after the date as of which the Securities and Exchange Commission, or SEC, staff has confirmed that it has no additional comments on the proxy statement (subject to adjournment from time to time) or (ii) the minimum number of shares of Hypercom common stock necessary to constitute a quorum for the transaction of business has been represented in person or by proxy at the Hypercom stockholders meeting and Hypercom has failed to take a vote of stockholders on the merger agreement at such meeting;

a tender offer or exchange offer for outstanding shares of Hypercom common stock has been publicly disclosed (other than by VeriFone) and Hypercom s board of directors recommends that Hypercom s stockholders tender their shares in such tender or exchange offer or, within ten business days after the commencement of such tender or exchange offer, Hypercom s board of directors fails to recommend that Hypercom s stockholders not tender any shares into such tender or exchange offer;

Hypercom s board of directors has recommended to Hypercom s stockholders any alternative acquisition proposal;

Hypercom has entered into an acquisition agreement with respect to an alternative acquisition proposal; or

there has been a breach of any representation, warranty, covenant or agreement made by Hypercom in the merger agreement, or any representations and warranties become untrue after the date of the merger agreement, in each case such that certain conditions to closing of VeriFone would not be satisfied and such breach or condition is not curable or, if curable, is not cured within 30 days after written notice thereof is given by VeriFone to Hypercom.

Termination Fee

(Page 97)

Hypercom Termination Fee. Hypercom has agreed to pay to VeriFone a termination fee equal to \$12.2 million if (i) prior to the Hypercom stockholder vote, the Hypercom board of directors approves a superior proposal and authorizes Hypercom to enter into an alternative acquisition agreement or (ii) VeriFone terminates the merger agreement after:

Hypercom s board of directors has withdrawn, adversely modified or failed to appropriately reconfirm its approval or recommendation of the merger agreement as described above;

Hypercom has failed to convene and hold the Hypercom stockholders meeting prior to the later of the date that is 45 days after the date as of which the SEC has confirmed that it has no additional comments on the proxy statement;

Hypercom has failed to take a vote of Hypercom stockholders on the merger agreement at a stockholders meeting where a quorum for the transaction of business was present;

Hypercom s board of directors has recommended that stockholders tender their shares into a tender or exchange offer of another person or failed to recommend that Hypercom stockholders not tender any shares into such a tender or exchange offer;

Hypercom s board of directors has recommended to its stockholders any alternative acquisition proposal; or

Hypercom has entered into an alternative acquisition agreement with respect to a superior proposal. In addition, if there has been a bona fide acquisition proposal publicly disclosed, announced, commenced, submitted or made by a third-party, and the merger agreement is then terminated by either VeriFone or Hypercom under the provision allowing for termination following August 31, 2011 or November 30, 2011, as applicable, if the transaction has not yet been consummated, or under the provision allowing for termination if Hypercom s stockholders fail to approve the merger agreement in a stockholder meeting at which a vote upon the merger agreement was taken, and within 12 months of such termination, Hypercom enters into a definitive agreement with respect to any other acquisition proposal that is ultimately consummated, then Hypercom must pay VeriFone the \$12.2 million termination fee.

VeriFone Termination Fee. So long as Hypercom has not breached in any material respect its covenants under the merger agreement (to the extent such covenants relate to antitrust or competition law matters), if the merger agreement is terminated:

based on the entry of any order relating to antitrust or competition law, or

based on the failure to consummate the merger by the termination date, and at the time of such termination all the conditions to the closing (other than the conditions relating to regulatory consents, the absence of any order restraining completion of the merger in connection with antitrust or competition law or certain related litigation proceedings, the accuracy of VeriFone s representations and warranties, and VeriFone s performance of its obligations under the merger agreement) have been satisfied or waived,

then VeriFone must pay to Hypercom a fee equal to \$28.4 million. However, if VeriFone elects to extend the termination date and such termination occurs after August 31, 2011, then VeriFone must pay to Hypercom a termination fee equal to \$30.4 million. This termination fee will not be payable by VeriFone if (1) VeriFone has waived in writing any condition related to antitrust or competition matters and is prepared to consummate the merger notwithstanding the failure of such condition, and (2) the completion of the merger notwithstanding the failure of such condition would not reasonably be expected to subject any officer or director of Hypercom to personal liability.

Additional Covenants

(Page 94)

VeriFone and Hypercom have agreed that through the earlier of the effective time of the merger (if the closing occurs) and the six-month period following termination of the merger agreement (if the closing does not occur), neither will solicit for hire certain specified officers or key employees of the other party, or, except where this restriction is prohibited by law, hire any such employee. However, nothing in the merger agreement prohibits solicitation and hiring by means of a general advertisement not directed at any particular individual or the employees of the other party generally, or responding to and hiring an employee of the other party who initiates contact with the applicable party provided that such contact is not solicited by the applicable party.

In addition, VeriFone and Hypercom have agreed that they will not seek to effect, announce or communicate, any potential reduction or elimination of the other party s products or services or change in the pricing thereof.

In the case of the first breach, if any, of either of the two covenants described above, the breaching party will have three days in which to cure the breach. If the initial breach is not cured within this three-day period, or if there is any subsequent breach, the breaching party must pay the other party, within three business days of notice of the breach, \$1 million. However, these amounts must only be paid in the event of a breach caused by or at the instructions of or with the consent of or prior knowledge of (if the senior officer does not use reasonable efforts to prevent such breach and does not direct or re-direct) the applicable person to refrain from such activity after obtaining such knowledge) one or more senior officers of the breaching party.

In addition, VeriFone has agreed with Hypercom that following the closing of the merger, it would fulfill any Hypercom customer agreements or purchase orders entered into in the ordinary course of business consistent with Hypercom s current business practices, unless the customer agrees otherwise, as well as provide certain hardware and software support for specified periods as described under The Merger Agreement Additional Covenants .

Litigation Related to the Merger

(Page 79)

In connection with the announced merger, several purported class action lawsuits have been filed in Arizona and Delaware state courts alleging variously, among other things, that the board of directors of Hypercom breached its fiduciary duties in not securing a higher price in the merger and that VeriFone, Hypercom, FP Hypercom Holdco, LLC and Francisco Partners II, L.P. aided and abetted that alleged breach. The actions seek injunctive relief and unspecified damages.

No Dissenters or Appraisal Rights

(Page 79)

Under Delaware law, no dissenters or appraisal rights are available to Hypercom stockholders.

Material U.S. Federal Income Tax Consequences

(Page 76)

The merger is intended to constitute a reorganization for United States federal income tax purposes. If so treated, the exchange of your shares of Hypercom common stock for shares of VeriFone common stock generally will not cause you to recognize gain or loss for U.S. federal income tax purposes. However, you will recognize income or gain with respect to cash received in lieu of any fractional shares of VeriFone common stock. It is a

condition to the merger that VeriFone and Hypercom receive legal opinions as to the United States federal income tax treatment of the merger with respect to our companies and our stockholders. These opinions will not bind the Internal Revenue Service, which could take a different view.

The consequences of the merger to any particular stockholder will depend on that stockholder s particular facts and circumstances. Accordingly, you are urged to consult your own tax advisor to determine your own tax consequences from the merger. See the section of this proxy statement/prospectus titled The Merger Material U.S. Federal Income Tax Consequences for a summary discussion of material U.S. federal income tax consequences of the merger to U.S. holders.

You should consult your tax advisor about the particular tax consequences of the merger to you.

Risk Factors and Cautionary Statement Concerning Forward-Looking Statements

(Pages 26 and 53)

Both companies have made forward-looking statements in this proxy statement/prospectus and have made forward-looking statements in the documents that are incorporated by reference. Forward-looking statements are subject to risks and uncertainties. Forward-looking statements include information concerning possible or assumed future results of operations of VeriFone, Hypercom or the combined company. When words such as expects, anticipates, intends, plans and similar expressions are used, we are making forward-looking statements.

If you are a Hypercom stockholder, you should note that an investment in VeriFone common stock involves risks and uncertainties. You should consider these risk factors in evaluating how to vote your shares of Hypercom common stock at the special meeting.

Comparison of Rights of Hypercom Stockholders and VeriFone Stockholders

(Page 110)

Hypercom s charter and bylaws and Delaware corporate law govern the rights of Hypercom stockholders. VeriFone s charter and bylaws and Delaware corporate law will govern your rights as a stockholder of VeriFone following the merger. Your rights under VeriFone s charter and bylaws will differ in some respects from your rights under Hypercom s charter and bylaws.

Support Agreement

(Page 82)

In connection with the merger, VeriFone entered into a support agreement with FP Hypercom Holdco, LLC and Francisco Partners II, L.P., pursuant to which FP Hypercom Holdco, LLC and Francisco Partners II, L.P. agreed to vote any shares of Hypercom common stock held by them in favor of the merger. As of December 17, 2010, FP Hypercom Holdco, LLC holds a warrant to purchase 10,544,000 shares of Hypercom common stock but based on information received from Francisco Partners II, L.P. neither Francisco Partners II, L.P. nor FP Hypercom Holdco, LLC holds any shares of Hypercom common stock. Accordingly, assuming FP Hypercom Holdco, LLC does not exercise its warrant prior to the record date and neither Francisco Partners II, L.P. nor FP Hypercom Holdco, LLC acquires any shares of Hypercom common stock, FP Hypercom Holdco, LLC and Francisco Partners II, L.P. will not be entitled to vote at the special meeting. The support agreement is attached as Annex B to this proxy statement/prospectus.

Summary of VeriFone Selected Financial Data

The following selected financial information of VeriFone is provided to aid your analysis of the financial aspects of the merger. When you read this summary historical financial data, it is important that you also read VeriFone s historical consolidated financial statements and related notes incorporated by reference into this proxy statement/prospectus, as well as the sections entitled Management s Discussion and Analysis of Financial Condition and Results of Operations in VeriFone s annual and quarterly reports incorporated by reference into this proxy statement/prospectus. See Where You Can Find More Information on page 117.

The consolidated statements of operations data set forth below for the years ended October 31, 2010, 2009 and 2008 and the consolidated balance sheet data as of October 31, 2010 and 2009 are derived from VeriFone s audited consolidated financial statements that are incorporated by reference in this proxy statement/prospectus. The consolidated statements of operations data for the years ended October 31, 2007 and 2006 and the consolidated balance sheet data as of October 31, 2008, 2007 and 2006 are derived from VeriFone s audited consolidated financial statements are incorporated by reference in this proxy statement/prospectus.

	2006	2007(2)(3)	l Year Ended Oct 2008(2)(4) sands, except per	2009(2)(5)	2010(6)(7)
Consolidated Statements of Operations Data:					
Net revenues	\$ 581,070	\$ 902,892	\$ 921,931	\$ 844,714	\$ 1,001,537
Cost of net revenues	319,525	603,660	628,900	562,585	631,225
Gross profit	261,545	299,232	293,031	282,129	370,312
Operating expenses:					
Research and development	47,353	65,430	75,622	65,148	74,227
Sales and marketing	58,607	96,295	91,457	73,544	94,666
General and administrative	42,573	80,704	126,625	76,468	84,371
Impairment of goodwill and intangible assets			289,119	175,512	
Amortization of purchased intangible assets	4,703	21,571	26,033	20,423	14,624
In-process research and development	,	6,752	,	,	
Total operating expenses	153,236	270,752	608,856	411,095	267,888
Operating income (loss)	108,309	28,480	(315,825)	(128,966)	102,424
Interest expense	(13,617)	(41,310)	(42,209)	(26,476)	(28,344)
Interest income	3,372	6,702	5,981	1,517	1,278
Other income (expense), net	(6,394)	(7,882)	(13,181)	5,716	2,887
Income (loss) before income taxes	91,670	(14,010)	(365,234)	(148,209)	78,245
Provision (benefit) for income taxes	32,159	22,915	45,838	9,246	(20,582)
Net income (loss)	\$ 59,511	\$ (36,925)	\$ (411,072)	\$ (157,455)	\$ 98,827
Net income (loss) per common share:					
Basic	\$ 0.90	\$ (0.45)	\$ (4.88)	\$ (1.86)	\$ 1.16
Diluted	\$ 0.86	\$ (0.45)	\$ (4.88)	\$ (1.86)	\$ 1.13
Weighted-average shares used in computing net income (loss) per common share:					
Basic	66,217	82,194	84,220	84,473	85,203
Diluted	68,894	82,194	84,220	84,473	87,785

	2006	2007(1)(2)(3)	As of October 31, 2008(2)(4) (in thousands)	2009(2)(5)	2010
Consolidated Balance Sheet Data:					
Cash and cash equivalents	\$ 86,564	\$ 215,001	\$ 157,160	\$ 324,996	\$ 445,137
Total assets	452,945	1,517,207	1,077,641	917,290	1,075,326
Long-term debt and capital leases, including current					
portion	192,889	477,772	487,200	468,864	473,511

- (1) In November 2006, VeriFone increased its outstanding balance on its Term B Loan to \$500.0 million. In June 2007, VeriFone sold \$316.2 million of 1.375% Senior Convertible Notes due 2012. As a result of the adoption of Financial Accounting Standards Board (FASB) Accounting Standards Codification (ASC) 470-20 (formerly APB 14-1, Accounting for Convertible Debt Instruments That May Be Settled in Cash upon Conversion (Including Partial Cash Settlement)), which VeriFone adopted in the first quarter of fiscal year 2010, VeriFone allocated \$236.0 million to the liability component and \$80.2 million to the equity component as of the date of issuance. VeriFone repaid \$263.8 million of VeriFone s outstanding Term B Loan with the proceeds from the sale of the 1.375% Senior Convertible Notes.
- (2) Amounts for fiscal years 2009, 2008 and 2007 have been adjusted from the historical consolidated financial data incorporated by reference into this proxy statement/prospectus to reflect the retroactive application of ASC 470-20.
- (3) VeriFone acquired Lipman Electronic Engineering Ltd. on November 1, 2006 and its results of operations are included from the date of acquisition. VeriFone also recognized an in-process research and development expense of \$6.8 million in connection with VeriFone s Lipman acquisition.
- (4) VeriFone s fiscal year 2008 results of operations include \$41.8 million of general and administrative costs related to the restatement of interim financial information for the first three quarters of the fiscal year ended October 31, 2007 which VeriFone completed during fiscal year 2008. VeriFone recorded a \$262.5 million impairment of VeriFone s goodwill in VeriFone s Europe, Middle East and Africa (EMEA) reporting unit and a \$26.6 million impairment of developed and core technology intangible assets due to lower revenue expectations in light of current operating performance and future operating expectations. VeriFone also recognized a \$62.3 million income tax expense for recording a full valuation allowance against all beginning of the year balances for U.S. deferred tax assets.
- (5) VeriFone recorded \$175.5 million of goodwill impairment charges in fiscal year 2009 after VeriFone concluded that the carrying amount of the North America and Asia reporting units exceeded their implied fair values. VeriFone repurchased and extinguished \$33.5 million par value of VeriFone s outstanding 1.375% Convertible Notes for \$19.8 million, excluding accrued interest paid. As a result, VeriFone realized a \$13.1 million gain on these transactions, net of a \$0.6 million write-off of related deferred debt issuance costs. In accordance with the provisions of ASC 470-20, VeriFone recorded a revised gain of approximately \$7.5 million. VeriFone reduced VeriFone s accrual for Brazilian customs penalties by \$14.9 million (28.2 million Brazilian reais) as a result of the dismissal of one of the assessments. Of this amount, \$9.8 million was reversed against general and administrative expenses and \$5.1 million was reversed against interest expense. This reduction was partially offset by the accrual for legal fees related to a contingency legal representation agreement entered into between one of VeriFone s Brazilian subsidiaries (prior to VeriFone s acquisition of that subsidiary from Lipman) and Brazilian counsel pursuant to which VeriFone s Brazilian subsidiary had agreed to pay to Brazilian counsel legal fees in the amount of \$2.8 million (5.0 million Brazilian reais) for achieving the successful dismissal of the assessment. These legal fees are included in general and administrative expenses in the third quarter of fiscal year 2009.
- (6) VeriFone reduced its accrual related to claims against its Brazilian subsidiary (which was acquired as part of the Lipman acquisition in 2006) by \$7.7 million (13.1 million Brazilian reais) as a result of a filed settlement of disputed payables and expiration of the statute of limitation on certain tax contingencies. Of this amount, \$6.6 million was reversed against other income (expense), net, and \$1.1 million (which consisted of accrued interest related to the claims) was reversed against interest expense.
- (7) The income tax benefit for the fiscal year ended October 31, 2010 includes a tax benefit of \$54.0 million attributable to a worthless stock deduction. VeriFone determined that one of its U.K. subsidiaries was insolvent on a fair market value basis and that it became worthless during the year resulting in a worthless stock deduction of \$154.0 million for tax purposes.

Summary of Hypercom Selected Financial Data

The following selected financial information of Hypercom is provided to aid your analysis of the financial aspects of the merger. When you read this summary historical financial data, it is important that you also read Hypercom s historical consolidated financial statements and related notes incorporated by reference into this proxy statement/prospectus, as well as the sections entitled Management s Discussion and Analysis of Financial Condition and Results of Operations in Hypercom s annual and quarterly reports incorporated by reference into this proxy statement/prospectus. See Where You Can Find More Information on page 117.

The table below presents summary selected historical consolidated financial data of Hypercom. These financial statements have been prepared in accordance with U.S. generally accepted accounting principles. The selected consolidated balance sheet data as of September 30, 2010 and the selected consolidated statement of operations data for the nine months ended September 30, 2010 and 2009 have been derived from unaudited financial statements incorporated by reference in this proxy statement/prospectus. The unaudited financial statements include all adjustments, consisting of normal recurring accruals, which Hypercom considers necessary for a fair presentation of the financial position and the results of operations for these periods. Operating results for the nine months ended September 30, 2010 are not necessarily indicative of the results that may be expected for the entire year ending December 31, 2010.

The consolidated statements of operations data set forth below for the years ended December 31, 2009, 2008 and 2007 and the consolidated balance sheet data as of December 31, 2009 and 2008 are derived from Hypercom s audited consolidated financial statements that are incorporated by reference in this proxy statement/prospectus. The consolidated statements of operations data for the years ended December 31, 2006 and 2005 and the consolidated balance sheet data as of December 31, 2007, 2006 and 2005 are derived from Hypercom s audited consolidated financial statements not incorporated by reference in this proxy statement/prospectus.

	2005(1)(2)	Years 2006(3)(4)	Ended Decemb 2007(5)	per 31, 2008(6)	2009(7)	Nine Mon Septem 2009(7) (unau	lber 30, 2010
			(in thousan	ds, except per	share data)		
Consolidated Statements of Operations Data:							
Net revenue	\$ 245,223	\$ 248,565	\$ 289,527	\$ 434,242	\$ 406,903	\$ 289,530	\$ 327,778
Costs of revenue	172,091	159,540	214,848	310,433	278,469	197,857	220,801
Gross profit	73,132	89,025	74,679	123,809	128,434	91,673	106,977
Operating expenses:							
Research and development	27,555	27,706	28,753	45,901	44,486	32,139	33,902
Selling, general and administrative	72,649	60,105	59,964	81,144	74,681	54,820	57,068
Amortization of purchased intangible assets			502	5,678	6,147	4,509	4,092
Impairment of goodwill				67,798			
In-process research and development		1,000					
Gain on sale of real property		(2,953)	(3,796)				(1,515)
Total operating expenses	100,204	85,858	85,423	200,521	125,314	91,468	93,547
Operating income (loss)	(27,072)	3,167	(10,744)	(76,712)	3,120	205	13,430
Interest income	2,679	4,758	3,954	1,466	296	190	287
Interest expense	(787)	(1,072)	(122)	(6,822)	(10,990)	(7,619)	(8,962)
Foreign currency income (loss)	(1,297)	(699)	(1,685)	(1,821)	411	(198)	(704)

	2005(1)(2)	Years E 2006(3)(4)	Ended Decemb 2007(5)	ber 31, 2008(6)	2009(7)	Nine Mont Septem 2009(7) (unau	ber 30, 2010
			(in thousand	ls, except per s	hare data)	(unau	incu)
Other income (expense)	(240)	(71)	(25)	191	390	356	78
Culer meene (enpence)	(2.0)	(71)	(20)	.,	070	000	10
Income (loss) before income taxes and discontinued operations	(26,717)	6,083	(8,622)	(83,698)	(6,773)	(7,066)	4,129
•					() /		,
Benefit (provision) for income taxes	(3,526)	(1,346)	(478)	(1,211)	829	(24)	(462)
Income (loss) before discontinued operations	(30,243)	4,737	(9,100)	(84,909)	(5,944)	(7,090)	3,667
Income (loss) from discontinued operations	(3,123)	2,233	1,623	(494)	(924)	(402)	(73)
	(=,===)	_,	-,	(.,.)	(> = -)	()	()
Net income (loss)	\$ (33,366)	\$ 6,970	\$ (7,477)	\$ (85,403)	\$ (6,868)	\$ (7,492)	\$ 3,594
Net income (loss) per common share:							
Basic	\$ (0.64)	\$ 0.13	\$ (0.14)	\$ (1.60)	\$ (0.13)	\$ (0.14)	\$ 0.07
	φ (0.01)	ф 0.110	¢ (011.)	¢ (1.00)	\$ (0.12)	¢ (011.)	ф 0.0 <i>7</i>
Diluted	\$ (0.64)	\$ 0.13	\$ (0.14)	\$ (1.60)	\$ (0.13)	\$ (0.14)	\$ 0.07
	+ (0.0.)	+ •••••	+ (01-1)	+ (1.0.0)	+ (0122)	+ (012.)	+
Weighted-average shares used in computing net							
income (loss) per common share:	50.005	52.240	52.025	52.221	52 526	52 400	52.045
Basic	52,395	53,248	52,927	53,321	53,526	53,498	53,845
Diluted	52,395	53,966	52,927	53,321	53,526	53,498	54,919

		А	s of December 3	1,			As of tember 30, 2010
	2005	2006	2007	2008	2009	(ur	naudited)
			(in th	ousands)			
Consolidated Balance Sheet Data:							
Cash and cash equivalents	\$ 35,940	\$ 34,190	\$ 76,925	\$ 35,582	\$ 55,041	\$	38,521
Total assets	238,914	236,716	250,244	306,300	306,888		309,019
Short and long-term debt, net of discount	8,377			48,769	56,076		61,691

- (1) In the fourth quarter of 2005, Hypercom sold its U.K. lease portfolio (the U.K. Lease Business) as this leasing business did not align with its core business. As of December 31, 2005, the U.K. Lease Business operating results have been classified as discontinued operations in the statements of operations for all periods presented. Hypercom recognized a \$6.6 million loss on the write-down of the U.K. Lease Business to its estimated fair value based on purchase offers received from potential buyers.
- (2) In 2005, Hypercom performed a comprehensive review of its financial performance; market strategy; products; product development and life cycle management; and employee, customer and vendor relationships. The objective of the business review was to create a plan to improve profitability and customer relations, transition more rapidly to 32-bit technology products, rationalize operations and reorganize Hypercom s corporate structure. As a result of this business review, Hypercom recorded charges of \$16.3 million to costs of revenue and \$2.7 million to operating expenses during 2005.
- (3) In December 2006, Hypercom acquired TPI Software, LLC and incurred a \$1.0 million charge for the acquired in-process research and development.
- (4) In September 2006, Hypercom completed the sale of its real property in Hong Kong for \$5.2 million. Hypercom recorded a gain of approximately \$3.0 million on the sale in the third quarter of 2006.
- (5) In June 2007, Hypercom completed the sale of its former headquarters facilities, located in Phoenix, Arizona, for a sale price of \$16.3 million. Hypercom recorded a gain of \$3.8 million on the sale in the second quarter of 2007. During the second quarter of 2007, Hypercom initiated a reconfiguration of its

Table of Contents

global sales and marketing organizations and incurred charges of \$2.9 million during the year. During the second quarter of 2007, Hypercom increased inventory reserves by \$6.1 million primarily related to non-PCI (payment card industry) compliant products and components.

- (6) Hypercom acquired Thales e-Transactions on April, 1, 2008 and its results of operations are included from the date of acquisition. Hypercom determined that certain of the assets acquired in the Thales e-Transactions transaction were impaired and in December 2008 Hypercom recognized asset impairment charges totaling \$67.8 million for 2008, primarily related to the write-off of goodwill and intangible assets.
- (7) For the nine months ended September 30, 2009 and for the year ended December 31, 2009, Hypercom recorded a net loss of \$7.5 million (\$0.14) per share and a net loss \$6.9 million (\$0.13) per share, respectively. The net loss was principally due to its results during the first quarter in which Hypercom reported a net loss of \$9.9 million. The first quarter was negatively impacted by the global economic downturn and the corresponding decrease in available credit to both businesses and consumers.

Selected Unaudited Pro Forma Condensed Combined Financial Data

The following selected unaudited pro forma condensed combined financial information is based on the historical financial statements of VeriFone and Hypercom after giving effect to the proposed merger between VeriFone and Hypercom and the assumptions and adjustments described in the accompanying notes to the unaudited pro forma condensed combined financial information included elsewhere in this proxy statement/prospectus, including general assumptions as to the scope of Hypercom businesses that may be included in asset dispositions to address potential competitive concerns as described in further detail in the footnotes to the unaudited pro forma condensed combined financial information.

VeriFone s fiscal year ends on October 31 while Hypercom s fiscal year ends on December 31. The selected unaudited pro forma condensed combined balance sheet data as of October 31, 2010 is based on the historical balance sheet of VeriFone as of October 31, 2010 and of Hypercom as of September 30, 2010 and has been prepared to reflect the merger as if it had occurred on October 31, 2010. The selected unaudited pro forma condensed combined statement of operations data for the fiscal year ended October 31, 2010 combines the results of operations of VeriFone for the fiscal year ended October 31, 2010 and of Hypercom for the twelve months ended September 30, 2010, as though the merger had occurred on November 1, 2009, the first day of VeriFone s 2010 fiscal year.

The unaudited pro forma condensed combined financial information assumes that, at closing, (i) each outstanding share of Hypercom common stock will be converted into 0.23 shares of VeriFone common stock, (ii) each restricted stock award (RSA) share will be vested and converted into 0.23 shares of VeriFone common stock, (iii) each outstanding Hypercom warrant will be converted into a warrant to acquire a number of shares of VeriFone common stock equal to the product of (x) the number of shares of Hypercom common stock subject to such warrant and (y) the exchange ratio of 0.23, at an exercise price per share (rounded up to the nearest whole cent) equal to the exercise price per share of Hypercom common stock for such warrant immediately prior to the effective time of the merger divided by 0.23, and (iv) each Hypercom stock option will be converted into an option to purchase a number of shares of VeriFone common stock equal to the product of (i) the number of shares of Hypercom common stock subject to such option immediately prior to the effective time of the merger and (ii) 0.23, at an exercise price per share of shares of Hypercom common stock subject to such option immediately prior to the effective time of the merger and (ii) 0.23, at an exercise price per share of shares of Hypercom common stock subject to such option immediately prior to the effective time of the merger and (ii) 0.23, at an exercise price per share of shares of Hypercom common stock subject to such option immediately prior to the effective time of the merger and (ii) 0.23, at an exercise price per share of shares of Hypercom common stock subject to such option prior to the merger divided by 0.23.

The selected unaudited pro forma condensed combined financial data is based on estimates and assumptions that are preliminary, presented for illustrative purposes only and is not necessarily indicative of what the combined financial position or the results of operations would actually have been if the merger had occurred on those dates. Since VeriFone and Hypercom were not under common control or management for any period presented, the selected unaudited pro forma condensed combined financial data may not be comparable to, or indicative of, future performance. See Risk Factors and Unaudited Pro Forma Condensed Combined Financial Information.

Year Ended October 31, 2010 n thousands, except per share data)		
\$ 1,385,172		
902,023		
483,149		
394,134		
89,015		
(27,197)		
2,904		
64,722		
23,583		
\$ 88,305		
\$ 0.90		
\$ 0.88		
As of October 31, 2010 (in thousands)		
\$ 413,142		
1,769,156		
473,511		
(in thousand \$ \$ \$ \$		

Total stockholders equity

21

678,586

Comparative Historical and Pro Forma Per Share Data

The following tables set forth certain historical per share data of VeriFone and Hypercom and combined per share data on an unaudited pro forma basis after giving effect to the merger using the acquisition method of accounting assuming that all Hypercom stockholders will receive the consideration of 0.23 shares of VeriFone common stock for each share of common stock of Hypercom outstanding. The following data should be read in conjunction with the separate historical consolidated financial statements of VeriFone and the historical consolidated financial statements of Hypercom incorporated by reference into this proxy statement/prospectus. The unaudited pro forma combined per share data do not necessarily indicate the operating results that would have been achieved had the merger been completed as of the beginning of the earliest period presented and should not be taken as representative of future operations. Since becoming a public company, neither VeriFone nor Hypercom has declared or paid any cash dividends on its common stock.

VeriFone		Hypercom	
Net income from continuing operations per share (basic):		Net income (loss) from continuing operations per share (basic):	
Year ended October 31, 2010	\$ 1.16	Year ended December 31, 2009	\$ (0.11)
		Nine months ended September 30, 2010	\$ 0.07
Net income from continuing operations per share		Net income (loss) from continuing operations per	
(diluted):		share (diluted):	
Year ended October 31, 2010	\$ 1.13	Year ended December 31, 2009	\$(0.11)
		Nine months ended September 30, 2010	\$ 0.07
Book value per share(1):		Book value per share(1):	
October 31, 2010	\$ 2.37	December 31, 2009	\$ 1.66
		September 30, 2010	\$ 1.64

	VeriFone Pro Forma Combined Data(2)		Hypercom Equivalent Forma Combined Data(3)	
Net income from continuing operations per share (basic):				
Year ended October 31, 2010	\$	0.90	\$	0.21
Net income from continuing operations per share (diluted):				
Year ended October 31, 2010	\$	0.88	\$	0.20
Book value per share:				
October 31, 2010	\$	6.82	\$	1.57

(1) Historical book value per share is computed by dividing stockholders equity by the number of shares of VeriFone common stock or Hypercom common stock outstanding at the end of each period.

(2) Because of the different fiscal period ends, financial information for VeriFone as of or for the year ended October 31, 2010 has been combined with financial information relating to Hypercom as of or for the twelve months ended September 30, 2010. Pro forma book value per share is computed by dividing pro forma stockholders equity by the pro forma number of shares of VeriFone common stock outstanding at the end of each period.

(3) The Hypercom equivalent pro forma combined per share amounts are calculated by multiplying VeriFone pro forma per share amounts by the exchange ratio in the merger agreement of 0.23 shares of VeriFone common stock for each share of Hypercom common stock.

Comparative Per Share Market Price Data

Hypercom common stock has traded on the New York Stock Exchange under the symbol HYC since November 14, 1997, the date of its initial public offering. VeriFone common stock has traded on the New York Stock Exchange under the symbol PAY since April 29, 2005, the date of its initial public offering.

The following tables show, for the periods indicated, the high and low sale prices per share, adjusted for stock splits and the like, of Hypercom common stock and VeriFone common stock as reported on the New York Stock Exchange.

VeriFone s fiscal year ends on October 31 and Hypercom s fiscal year ends on December 31.

		VeriFone Common Stock New York Stock Exchange		
Fiscal Quarters	High		Low	
2009:				
First Quarter	\$ 12.23	\$	2.31	
Second Quarter	\$ 8.36	\$	3.65	
Third Quarter	\$ 9.28	\$	5.92	
Fourth Quarter	\$ 16.15	\$	8.75	
2010:				
First Quarter	\$ 19.42	\$	12.70	
Second Quarter	\$ 23.75	\$	16.63	
Third Quarter	\$ 22.92	\$	15.62	
Fourth Quarter	\$ 34.06	\$	20.97	
2011:				
First Quarter (through December 20, 2010)	\$ 41.47	\$	31.39	

	• •	Hypercom Common Stock New York Stock Exchange		
Fiscal Quarters	High	High Low		
2009:				
First Quarter	\$ 1.77	\$	0.81	
Second Quarter	\$ 1.75	\$	0.94	
Third Quarter	\$ 3.40	\$	1.32	
Fourth Quarter	\$ 3.53	\$	2.77	
2010:				
First Quarter	\$ 4.07	\$	3.12	
Second Quarter	\$ 5.19	\$	3.75	
Third Quarter	\$ 6.51	\$	3.05	
Fourth Quarter (through December 20, 2010)	\$ 9.00	\$	5.84	

The following table shows the closing sale prices per share of VeriFone common stock and Hypercom common stock as reported on the New York Stock Exchange on:

September 30, 2010, the trading day of the New York Stock Exchange next preceding the announcement of VeriFone s proposal to Hypercom s board of directors to acquire all of the outstanding shares of Hypercom Corporation;

November 16, 2010, the trading day of the New York Stock Exchange next preceding public announcement that VeriFone and Hypercom had entered into the merger agreement; and

, 2010, the last full trading day for which closing prices were available prior to printing this proxy statement/prospectus.

	Veri	VeriFone		Hypercom	
	Commo New Yor Exch	rk Stock	Common Stock New York Stock Exchange		
	High	Low	High	Low	
September 30, 2010	\$ 31.39	\$ 28.50	\$ 6.51	\$ 6.11	
November 16, 2010	32.53	31.39	6.32	6.08	
, 2010					

The above table only shows historical comparisons. These comparisons may not provide meaningful information in determining whether to approve the merger agreement and the merger. VeriFone and Hypercom urge you to obtain current market quotations for VeriFone common stock and Hypercom common stock and to review carefully the information contained in this proxy statement/prospectus or incorporated by reference into this proxy statement/prospectus in considering whether to approve the merger agreement and the merger. You should also review carefully the other information contained in or incorporated by reference into this proxy statement/prospectus.

As of December 17, 2010, there were approximately 111 holders of record of Hypercom common stock.

Dividend Data

Since becoming a public company, VeriFone has not declared or paid any cash dividends on its common stock. Since becoming a public company, Hypercom has not declared or paid any cash dividends on its common stock. VeriFone and Hypercom currently intend to retain any future earnings to fund the growth and development of their businesses and do not anticipate paying any cash dividends in the foreseeable future although their boards of directors may do so subject to applicable legal requirements and contractual restrictions, if any.

Risk Factors

By voting in favor of the proposal to approve and adopt the merger agreement and approve the merger, Hypercom stockholders will be choosing to invest in VeriFone common stock. An investment in VeriFone common stock involves a high degree of risk. In addition to the other information contained in or incorporated by reference into this proxy statement/prospectus, you should carefully consider the following risk factors in deciding whether to vote in favor of the proposal to approve and adopt the merger agreement and approve the merger. This summary of risks is not exhaustive. New risks may emerge from time to time and it is not possible to predict all risk factors, nor can Hypercom or VeriFone assess the impact of all factors on the merger and the combined company following the merger or the extent to which any factor or combination of factors may cause actual results to differ materially from those contained in or implied by any forward-looking statements. For further discussion of risk factors relating to the business of VeriFone and Hypercom, please see the Risk Factors in VeriFone s and Hypercom s current filings with the SEC. See Where You Can Find More Information beginning on page 117.

Risks Related to the Merger

Because the exchange ratio is fixed and the market price of VeriFone common stock may fluctuate, you cannot be sure of the value of the merger consideration you will receive.

Upon completion of the merger, each share of common stock of Hypercom will be exchanged for 0.23 of a share of VeriFone common stock. The merger consideration will not change even if the market price of either or both the Hypercom common stock and VeriFone common stock fluctuates. Neither Hypercom nor VeriFone may withdraw from the merger, and Hypercom may not resolicit the vote of its stockholders, solely because of changes in the market price of Hypercom shares of common stock or VeriFone common stock or because of changes in the operating and financial performance of either company that do not amount to a material adverse effect, as defined in the merger agreement. The specific dollar value of VeriFone common stock you will receive upon completion of the merger will depend on the market value of VeriFone common stock at the time of completion of the merger, which may be different from the closing price of VeriFone common stock on the last full trading day preceding public announcement of the merger agreement, the last full trading day prior to the date of this proxy statement/prospectus or the date of the special meeting. The merger is not expected to be completed immediately following the special meeting, and will not be completed until certain regulatory approvals have been obtained and other conditions have been satisfied or waived. You should obtain current market quotations for shares of VeriFone common stock and for shares of Hypercom common stock. VeriFone cannot assure you that the value of the VeriFone common stock that you will receive in the merger will not decline prior to or after the merger.

VeriFone s stock price has been volatile in the past. VeriFone expects that the price of its common stock may continue to fluctuate substantially in the future and you could lose the value of your investment in VeriFone common stock.

VeriFone s stock price has fluctuated substantially since its initial public offering in 2005 including due to the announcement of its restatement in December 2007 and during the recent turmoil in the worldwide financial markets. The market price of VeriFone common stock may fluctuate significantly and you could lose the value of your investment in VeriFone common stock. In addition to fluctuations related to VeriFone-specific factors, broad market and industry factors may adversely affect the market price of VeriFone common stock, regardless of VeriFone s actual operating performance. Factors that could cause fluctuations in VeriFone s stock price may include, among other things:

actual or anticipated variations in quarterly operating results;

changes in financial estimates by VeriFone or by any securities analysts who might cover its stock, or VeriFone s failure to meet the estimates made by securities analysts;

uncertainty about current global economic conditions;

changes in the market valuations of other companies operating in VeriFone s industry;

announcements by VeriFone or its competitors of significant acquisitions, strategic partnerships or divestitures;

additions or departures of key personnel;

sales or purchases of VeriFone common stock, including sales or purchases of its common stock by its directors and officers or by its principal stockholders; and

any issuance of VeriFone common stock in connection with any conversion of its 1.375% Senior Convertible Notes due 2012. Additionally, other future sales of substantial amounts of VeriFone common stock in the public market, or the availability of such shares for sale, including shares issued upon the exercise of outstanding options, could adversely affect the market price of VeriFone common stock. The stock market in general, and the market prices of stocks of other technology companies in particular, experience price volatility, which may adversely affect the market price of VeriFone s common stock for reasons unrelated to VeriFone s business or operating results.

The merger agreement restricts Hypercom s ability to pursue alternatives to the merger.

The merger agreement contains no shop provisions that, subject to limited exceptions, restrict Hypercom s ability to initiate, solicit, knowingly facilitate, encourage or commit to competing third-party proposals to acquire all or a significant part of Hypercom. Further, there are only limited exceptions to Hypercom s agreement that the Hypercom board of directors will not withhold, withdraw, qualify or modify in a manner adverse to VeriFone its recommendation of the merger agreement, and VeriFone generally has a right to match any competing acquisition proposals that may be made. Although the Hypercom board of directors is permitted to take actions in response to a superior proposal if it determines that doing so is necessary to comply with its fiduciary duties, doing so in specified situations could entitle VeriFone to terminate the merger agreement and to be paid a termination fee of \$12.2 million. Also, if the merger agreement is terminated because the merger is not consummated by August 31, 2011 (or if extended by VeriFone, November 30, 2011) or is not approved at the special meeting and prior to such termination there was publicly announced an acquisition proposal and Hypercom enters into a definitive agreement with respect to any acquisition proposal within 12 months following the termination, then Hypercom would be required to pay VeriFone a termination fee of \$12.2 million.

These provisions could discourage a potential competing acquiror that might have an interest in acquiring all or a significant part of Hypercom from considering or proposing that acquisition, even if it were prepared to pay consideration with a higher per share cash or market value than that proposed to be paid in the merger, or might result in a potential competing acquiror proposing to pay a lower per share price to acquire Hypercom than it might otherwise have proposed to pay because of the added expense of the termination fee that may become payable to Hypercom in certain circumstances.

Regulatory agencies must approve the merger and could impose conditions on, delay or refuse to approve the merger.

VeriFone and Hypercom are subject to the securities and antitrust laws of the United States and the antitrust laws of any other jurisdiction in which the merger is subject to review. VeriFone and Hypercom may be unable to obtain in the anticipated timeframe, or at all, the regulatory approvals required to complete the merger. Governmental entities with which filings are made may seek regulatory concessions, including compliance with material restrictions or conditions, including the divestiture of portions of VeriFone s or Hypercom s business, as conditions for granting approval of the merger. These actions may have a material adverse effect on VeriFone or

Hypercom or a material adverse effect on the combined business. Under the merger agreement, VeriFone and Hypercom have both agreed to use reasonable best efforts to complete the merger, including to gain clearance from antitrust and competition authorities and to obtain other required regulatory approvals. Under the merger agreement, VeriFone s reasonable best efforts includes a requirement by VeriFone to dispose of businesses or assets that produced up to \$124 million in revenues for VeriFone, Hypercom and their respective subsidiaries during the 2009 fiscal year if such disposition is reasonably necessary to gain clearance from antitrust and competition authorities. VeriFone and Hypercom anticipate that Hypercom s U.S. business will be divested in connection with the merger. These actions or any other actions that may be required to obtain regulatory approval for the transaction may have a material adverse effect on VeriFone or Hypercom or a material adverse effect on the combined businesses.

Any delay in completing the merger may reduce or eliminate the benefits expected.

In addition to the required regulatory approvals, the merger is subject to a number of other conditions beyond either company s control that may prevent, delay or otherwise materially adversely affect its completion. Neither VeriFone nor Hypercom can predict whether and when these other conditions will be satisfied. Further, the requirements for obtaining the required clearances and approvals could delay the completion of the merger for a significant period of time or prevent it from occurring. Any delay in completing the merger could cause the combined company not to realize some or all of the synergies that are expected to be achieved if the merger is successfully completed within its expected timeframe.

Although VeriFone and Hypercom expect that the merger will result in benefits to the combined company, those benefits may not occur because of integration and other challenges.

VeriFone and Hypercom have operated and, until the merger is completed, will continue to operate, independently. Achieving the expected benefits of the merger will depend in part on the integration of VeriFone s and Hypercom s technology, operations and personnel in a timely and efficient manner. VeriFone and Hypercom cannot assure you, however, that the integration will be completed as quickly as expected or that the merger will achieve the expected benefits. The challenges involved in this integration include:

incorporating Hypercom s technology and products into VeriFone s next generation of products;

integrating Hypercom s products into VeriFone s business because VeriFone does not currently sell Hypercom products and VeriFone s sales personnel have no experience selling Hypercom s products;

coordinating research and development activities to enhance introduction of new products, services and technologies;

inability to migrate both companies to a common enterprise resource planning information system to integrate all operations, sales and administrative activities for the combined company in a timely and cost effective way;

integrating Hypercom s international operations with those of VeriFone;

coordinating the efforts of the Hypercom sales organization with VeriFone s sales organization;

demonstrating to Hypercom customers that the merger will not result in adverse changes in client service standards or product support;

integrating Hypercom s information technology systems and resources with those of VeriFone;

Edgar Filing: VERIFONE SYSTEMS, INC. - Form S-4

integrating the supply chain of both companies while ensuring that products continue to be manufactured and delivered on a timely basis and with superior quality to customers;

persuading the employees in various jurisdictions that the business cultures of VeriFone and Hypercom are compatible, maintaining employee morale and retaining key employees;

any unforeseen expenses or delays associated with the transaction; and

timely release of products to market.

The integration of VeriFone and Hypercom will be international in scope, complex, time-consuming and expensive, and may disrupt VeriFone s and Hypercom s businesses or result in the loss of customers or key employees or the diversion of the attention of management. Some of Hypercom s suppliers, distributors, customers and licensors are VeriFone s competitors or work with VeriFone s competitors and may terminate their business relationships with Hypercom as a result of the merger. In addition, the integration process may strain the combined company s financial and managerial controls and reporting systems and procedures. This may result in the diversion of management and financial resources from the combined company s core business objectives. There can be no assurance that VeriFone and Hypercom will successfully integrate their businesses or that the combined company will realize any of the anticipated benefits of the merger.

If VeriFone is unable to integrate Hypercom personnel successfully or retain key Hypercom or VeriFone personnel after the merger is completed, the combined company business may suffer.

VeriFone s ability to obtain and successfully execute its business depends upon the personal reputation, judgment, business generation capabilities and project execution skills of VeriFone s and Hypercom s senior professionals. Any management disruption or difficulties in integrating VeriFone s and Hypercom s professionals could result in a loss of clients and customers or revenues from clients and customers and could significantly affect the combined company s business and results of operations.

The success of the merger will depend in part on VeriFone s ability to retain the talents and dedication of the professionals currently employed by VeriFone and Hypercom. It is possible that these employees might decide not to remain with VeriFone or Hypercom while VeriFone and Hypercom work to complete the merger or with the combined company after the merger is completed. If key employees terminate their employment, or insufficient numbers of employees are retained to maintain effective operations, the combined company s business activities might be adversely affected, management s attention might be diverted from successfully integrating Hypercom s operations to hiring suitable replacements, and the combined company s business might suffer. In addition, VeriFone and Hypercom might not be able to locate suitable replacements for any key employees that leave either company or offer employment to potential replacements on reasonable terms.

The combination of VeriFone and Hypercom may result in a loss of common customers.

VeriFone and Hypercom operate in a highly competitive market, and the future performance will be affected by the combined company s ability to retain the customers of both VeriFone and Hypercom before the merger. VeriFone and Hypercom currently sell to several of the same large customers to which each company provides products and services. The combined company s ability to maintain the current level of sales by VeriFone and Hypercom to these common customers may be limited by the desire of these customers to minimize their dependence on a single supplier. If, following the merger, common customers seek alternative suppliers for at least a portion of the products and services currently provided by both VeriFone and Hypercom, the business of the combined company may be materially and adversely impacted.

Third parties may terminate or alter existing contracts or relationships with Hypercom or VeriFone.

Hypercom has contracts with some of its suppliers, distributors, customers, licensors and other business partners. Some of these contracts require Hypercom to obtain consent from these other parties in connection with the merger. If these consents cannot be obtained, Hypercom may suffer a loss of potential future revenue and may lose rights that are material to Hypercom s business and the business of the combined company. In addition, third parties with whom Hypercom or VeriFone currently have relationships may terminate or otherwise reduce the scope of their relationship with Hypercom or VeriFone in anticipation of the merger.

General uncertainty related to the merger could harm the combined company.

VeriFone s or Hypercom s customers may, in response to the announcement of the proposed merger, delay or defer purchasing decisions. If VeriFone s or Hypercom s customers delay or defer purchasing decisions, the

combined company s revenues following the merger could materially decline or any increases in revenue could be lower than expected. Similarly, VeriFone and Hypercom employees may experience or perceive uncertainty about their future roles with the combined company following the merger. This may harm VeriFone s and Hypercom s ability to attract and retain key management, marketing, sales and technical personnel. Also, speculation regarding the likelihood of the completion of the merger could increase the volatility of VeriFone s and Hypercom s share prices prior to the closing.

The pro forma financial statements are presented for illustrative purposes only and may not be an indication of the combined company s financial condition or results of operations following the merger.

The pro forma financial statements contained in this proxy statement/prospectus are presented for illustrative purposes only and may not be an indication of the combined company s financial condition or results of operations following the merger for several reasons. For example, the pro forma financial statements have been derived from the historical financial statements of VeriFone and Hypercom and certain adjustments and assumptions have been made regarding the combined company after giving effect to the merger. The information upon which these adjustments and assumptions have been made is preliminary, and these kinds of adjustments and assumptions are difficult to make with complete accuracy. Moreover, the pro forma financial statements do not reflect all costs that are expected to be incurred by the combined company in connection with the transaction. For example, the impact of any incremental costs incurred in integrating the two companies is not reflected in the pro forma financial statements. As a result, the actual financial condition and results of operations of the combined company following the transaction may not be consistent with, or evident from, these pro forma financial statements.

In addition, the assumptions used in preparing the pro forma financial information may not prove to be accurate, and other factors may affect the combined company s financial condition or results of operations following the transaction. Any potential decline in the combined company s financial condition or results of operations may cause significant variations in the stock price of the combined company. See the section entitled Unaudited Pro Forma Condensed Combined Financial Information beginning on page 100.

VeriFone and Hypercom expect to incur significant costs associated with the merger.

VeriFone estimates that it will incur significant direct transaction costs associated with the merger. In addition, Hypercom estimates that it will incur significant direct transaction costs, which will be recognized as expenses as incurred. VeriFone and Hypercom believe the combined entity may incur charges to operations, which are not currently reasonably estimable, in the quarter in which the merger is completed or the following quarters, to reflect costs associated with integrating the two companies. There can be no assurance that the combined company will not incur additional material charges in subsequent quarters to reflect additional costs associated with the merger and the integration of the two companies. Although VeriFone expects that the elimination of duplicative costs, as well as the realization of other efficiencies related to the integration of the businesses, should allow VeriFone to offset incremental transaction and merger-related costs over time, this net benefit may not be achieved in the near term, or at all.

Certain officers of Hypercom have a personal interest that could have affected their decision to support the merger.

The personal interest of officers of Hypercom in the merger and their participation in arrangements that are different from, or are in addition to, those of Hypercom stockholders generally could have affected those officers decision to support the merger. These interests include the following:

the conversion of options to acquire Hypercom common stock into options to acquire VeriFone common stock and the acceleration of all options and restricted stock awards that were outstanding as of November 17, 2010; and



in certain instances, retention payments payable at closing and severance benefits if the individual s employment is terminated by VeriFone without cause or by the individual for good reason. See The Merger Interests of Certain Persons in the Merger. As a result of these interests, these officers may be more likely to support the merger than they might be if they did not have these interests.

A delay in closing the planned merger could have an adverse effect on Hypercom s revenues in the near term.

To the extent a prolonged delay in completing the planned merger creates uncertainty among users of Hypercom s products, there could be an adverse effect on Hypercom s results of operations, and quarterly revenues could be substantially below the expectations of market analysts which could cause a reduction in the stock price of Hypercom common stock, which effects are likely to be exacerbated the longer the time period between the signing and any termination of the merger agreement.

If the merger is not completed, Hypercom s future business and operations could be harmed.

If the merger is not completed, Hypercom may be subject to the following material risks, among others:

if the merger is terminated and Hypercom s board of directors determines to seek another merger or business combination, Hypercom may not be able to find a partner willing to pay an equivalent or more attractive price than that which would be paid in the merger;

Hypercom may be required to pay VeriFone a termination fee of \$12.2 million;

under the merger agreement, Hypercom is subject to certain restrictions on the conduct of its business which may affect its ability to execute certain of its business strategies or pursue opportunities that may arise during the pendency of the merger;

Hypercom s costs related to the merger, such as legal fees, accounting fees and expenses of its financial advisor must be paid by Hypercom even if the merger is not completed;

Hypercom may be required to record as expenses in its statement of operations items related to the proposed merger that would otherwise be reflected on its balance sheets;

Hypercom would not realize the benefits it expects by being part of the combined company;

the diversion of management attention from Hypercom s day-to-day business and the unavoidable disruption to its employees and its relationships with users and partners as a result of efforts and uncertainties relating to Hypercom s anticipated merger with VeriFone may detract from Hypercom s ability to grow revenues and minimize costs, which, in turn may lead to a loss of market position that Hypercom could be unable to regain if the merger does not occur;

Hypercom may not be able to continue its present level of operations, may need to scale back its business and may not be able to take advantage of future opportunities or effectively respond to competitive pressures, any of which could have a material adverse effect on its business and results of operations; and

Edgar Filing: VERIFONE SYSTEMS, INC. - Form S-4

the price of Hypercom shares of common stock may decline to the extent that the current market price of Hypercom shares of common stock reflects an assumption that the merger will be completed. *The rights of Hypercom stockholders will change when they become stockholders of VeriFone upon completion of the merger.*

When the merger is complete, Hypercom stockholders will become VeriFone stockholders and their rights as stockholders will be governed by VeriFone s certificate of incorporation and bylaws. There are numerous

differences between the rights and responsibilities of a stockholder of Hypercom and the rights and responsibilities of a stockholder of VeriFone. Please see Comparison of Rights of VeriFone Stockholders and Hypercom Stockholders beginning on page 110 for a discussion of the different rights associated with VeriFone common stock.

Stockholders may sell substantial amounts of Hypercom common stock on the public market, which is likely to depress the price of Hypercom common stock.

Any substantial sales of Hypercom common stock on the public market by current stockholders may cause the market price of Hypercom common stock to decline.

If Hypercom s current stockholders sell Hypercom common stock in the public market prior to the merger, it is likely that arbitrageurs will acquire those shares. These arbitrageurs will likely sell all those shares on the public market immediately following any announcement, or anticipated announcement, that the merger with VeriFone failed, or will likely fail, to close for regulatory or other reasons, which, in turn will likely cause the market price of Hypercom common stock to decline.

In addition to the other negative effects on Hypercom, those sales of Hypercom common stock might make it more difficult for Hypercom to sell equity or equity-related securities in the future if the merger with VeriFone is not completed.

Hypercom stockholders will have a significantly reduced ownership and voting interest after the merger and will exercise less influence over management.

Immediately after the completion of the merger, it is expected that former Hypercom stockholders, who collectively own 100% of Hypercom, will own approximately 12.8% of VeriFone, based on the number of shares of Hypercom and VeriFone common stock, including restricted stock, outstanding as of December 17, 2010. Consequently, Hypercom stockholders will have less influence over the management and policies of VeriFone than they currently have over the management and policies of Hypercom.

Risks Related to the Combined Company after the Merger

The risks and uncertainties described below are not the only ones facing the combined company following the merger. Additional risks and uncertainties not presently known to VeriFone or Hypercom or that VeriFone and Hypercom currently deem immaterial also may impair the business operations of the combined company. If any of the following risks or such other risks actually occurs, the business of the combined company could be harmed. When we use the term the combined company in this section or in the section entitled Risks Related to the Industry of this proxy statement/prospectus, we mean the combination of VeriFone and Hypercom after giving effect to the completion of the merger.

Both VeriFone and Hypercom have experienced rapid growth in their operations, and if VeriFone cannot adequately manage the combined company s growth, its results of operations will suffer.

Both VeriFone and Hypercom have experienced rapid growth in their operations in certain periods, both internally and from acquisitions. VeriFone cannot be sure that the combined company will make adequate allowances for the costs and risks associated with its expansion, or that its systems, procedures, and managerial controls will be adequate to support the rapid expansion in its operations. Any delay in implementing, or transitioning to, new or enhanced systems, procedures, or controls to accommodate the requirements of its business and operations and to effectively integrate acquired operations may adversely affect its ability to meet customer requirements, manage its product inventory, and record and report financial and management information on a timely and accurate basis. If the combined company is unable to successfully manage expansion, its results of operations will be adversely affected.

VeriFone s internal processes and control over financial reporting were, until recently, deemed inadequate.

As described under Item 9A. Controls and Procedures in VeriFone s Annual Report on Form 10-K for the fiscal year ended October 31, 2010, as of the fiscal year ended October 31, 2010, VeriFone has remedied the material weakness in control activities related to income taxes. This material weakness and other material weaknesses in VeriFone s internal control over financial reporting contributed to its need to restate previously reported interim financial information for each of the first three quarters of its fiscal year ended October 31, 2007, and to the delays in the filing of its Annual Report on Form 10-K for fiscal year 2007. VeriFone also was unable to file its quarterly reports on Form 10-Q for its fiscal quarters ended January 31, 2008 and April 30, 2008 on a timely basis. VeriFone has implemented a number of additional and enhanced processes and controls to improve its internal control over financial reporting.

Although VeriFone has implemented improved controls that have remedied these material weaknesses, these controls may not be sufficient to detect or prevent errors in financial reporting in future periods. VeriFone has devoted additional resources to its financial control and reporting requirements, including hiring additional qualified employees in these areas. VeriFone may hire additional employees and may also engage additional consultants in these areas. Competition for qualified financial control and accounting professionals in the geographic areas in which VeriFone operates is keen and there can be no assurance that VeriFone will be able to hire and retain these individuals.

A majority of VeriFone s net revenues is generated outside of the United States and VeriFone intends to continue to expand its operations internationally including through acquisitions and strategic partnerships, such as the Hypercom acquisition. The combined company s results of operations could suffer if it is unable to manage its international expansion and operations effectively.

During the fiscal years ended October 31, 2010 and 2009, approximately 60.0% and 63.6%, respectively, of VeriFone's net revenues were generated outside of the United States. During the fiscal years ended December 31, 2009 and 2008, approximately 69.6% and 78.7%, respectively, of Hypercom's net revenues were generated outside of the United States. VeriFone expects its percentage of net revenues generated outside of the United States. to increase over time. In particular VeriFone expects that its recently announced strategic partnership with Gemalto and planned acquisition of Hypercom will significantly increase its business in Africa, Europe and Asia. Part of VeriFone's strategy is to expand its penetration in existing foreign markets and to enter new foreign markets and in particular to enter new emerging markets where VeriFone expects to see growth in electronic payments. VeriFone's ability to penetrate some international markets may be limited due to different technical standards, protocols or product requirements. Expansion of the combined company's international business will require significant management attention and financial resources. The combined company's international net revenues will depend on its success in the following areas:

securing commercial relationships to help establish or increase its presence in new and existing international markets;

hiring and training personnel capable of marketing, installing and integrating its solutions, supporting customers, and effectively managing operations in foreign countries;

localizing its solutions to meet local requirements and to target the specific needs and preferences of foreign customers, which may differ from its traditional customer base in the markets VeriFone and Hypercom currently serve;

building the combined company s brand name and awareness of its services among foreign customers in new and existing international markets; and

implementing new systems, procedures, and controls to monitor its operations in new international markets.

In addition, the combined company is subject to risks and costs associated with operating in foreign countries, including:

multiple, changing, and often inconsistent enforcement of laws and regulations;

satisfying local regulatory or industry-imposed requirements, including security or other certification requirements;

competition from existing market participants, including strong local competitors, that may have a longer history in and greater familiarity with the international markets it enters;

tariffs and trade barriers;

laws and business practices that may favor local competitors;

restrictions on the repatriation of funds, foreign currency exchange restrictions, and currency exchange rate fluctuations;

extended payment terms and the ability to collect accounts receivable;

different and/or more stringent labor laws and practices such as the use of workers councils and labor unions;

different and more stringent data protection, privacy and other laws;

economic and political instability in certain foreign countries;

imposition of limitations on conversion of foreign currencies into U.S. dollars or remittance of dividends and other payments by foreign subsidiaries;

changes in a specific country s or region s political or economic conditions; and

greater difficulty in safeguarding intellectual property in areas such as China, Russia, and Latin America. Many of these factors typically become more prevalent during periods of economic stress; therefore, the persistence or occurrence of weakened global economic conditions in one or more regions where the combined company does business may exacerbate certain of these risks. For example, the combined company will be subject to foreign currency risk and economic and political instability which can lead to significant and unpredictable volatility in currency rates, including significant currency devaluations, which may negatively impact its revenues, gross margins, results of operations and financial position. Although VeriFone engages in some hedging of its foreign currency exposures, it does not hedge all such exposures and its hedging arrangements may not always be effective. In addition, compliance with foreign and U.S. laws and regulations that are applicable to the combined company s international operations is complex and may increase the combined company s cost of doing business in international jurisdictions. The combined company s international operations could expose it to fines and penalties if it fails to comply with these regulations. These laws and regulations include import and export requirements, exchange control regulations, U.S. laws such

Edgar Filing: VERIFONE SYSTEMS, INC. - Form S-4

as the Foreign Corrupt Practices Act, and local laws prohibiting corrupt payments to governmental officials. Although VeriFone has implemented policies and procedures designed to ensure compliance with these laws, there can be no assurance that its employees, contractors, and agents will not take actions in violation of its policies, particularly as the combined company expands its operations through organic growth and acquisitions. For example, two of VeriFone s Brazilian subsidiaries that were acquired as a part of the acquisition of Lipman Electronic Engineering Ltd., or Lipman, have been notified of a number of assessments regarding Brazilian customs penalties that relate to alleged infractions in the importation of goods as well as certain tax assessments and penalties. Hypercom also has certain pending tax assessments and labor matters in Brazil. Defending such assessments can be costly and divert management time. Any such violations could subject the combined company to civil or criminal penalties, including the imposition of substantial fines and interest or prohibitions on its ability to offer its products and services to one or more countries, and could also materially damage its

reputation, brand, international expansion efforts and business, and negatively impact its operating results. In addition, if the combined company fails to address the challenges and risks associated with its international expansion and acquisition strategy, it may encounter difficulties implementing its strategy, which could impede its growth or harm its operating results.

Macroeconomic conditions and economic volatility could materially adversely affect the combined company s business and results of operations.

The combined company s operations and performance will depend significantly on worldwide economic conditions. The significant slowdown and volatility in the U.S. and international economy and financial markets which began in the latter half of 2008 resulted in reduced demand for VeriFone s products, which in turn adversely impacted its revenues, business, financial condition and results of operations. In particular, the slowdown and volatility in the global markets resulted in softer demand in the financial and retail sectors, pricing pressures and more conservative purchasing decisions by customers, including a tendency toward lower-priced products and lower volume of purchases. In some countries where VeriFone does business, the weakened economy resulted in economic instability which had negative impacts, among them a decrease in purchasing power due to currency devaluations.

While a number of markets have shown signs of improvement, certain markets such as parts of southern Europe continue to experience weakened or uncertain economic conditions and some customers, suppliers and partners may continue to be negatively impacted by the global slowdown. VeriFone cannot predict whether such improvements or growth will continue and any future decline or volatility in global conditions could negatively impact its business, operating results and financial condition. Continued volatility in market conditions makes it difficult to forecast earnings, and if the combined company fails to meet its financial guidance or the expectations of investment analysts or investors in any period, the market price of VeriFone s common stock could decline.

Both VeriFone and Hypercom depend on a limited number of customers, including distributors and resellers, for a large percentage of their sales. If the combined company does not effectively manage its relationships with them, its net revenues and operating results will suffer.

A significant percentage of VeriFone s and Hypercom s net revenues are attributable to a limited number of customers, including distributors and independent sales organizations, or ISOs. For the fiscal year ended October 31, 2010, VeriFone s ten largest customers accounted for approximately 27% of its net revenues. Although no individual customer accounted for more than 10% of net revenues in the fiscal year ended October 31, 2010, three customers accounted for approximately 15% of VeriFone s net revenues in that period. For the fiscal year ended December 31, 2009, Hypercom s largest customer accounted for 5.2% of its net revenues, and its five largest customers accounted for 19.0% of its net revenues. If any of the combined company s large customers at reduced prices or on other less favorable terms, the combined company s revenues and income could be materially adversely affected.

VeriFone and Hypercom sell a significant portion of their solutions through third-party resellers such as independent distributors, ISOs, value-added resellers and payment processors. The combined company will depend on their active marketing and sales efforts. These resellers also provide after-sales support and related services to end user customers. When the combined company introduces new applications and solutions, these resellers will provide critical support for developing and supporting the custom software applications to run on its various electronic payment systems and, internationally, in obtaining requisite certifications in the markets in which they are active. Accordingly, the pace at which the combined company is able to introduce new solutions in markets in which these resellers are active depends on the resources such resellers dedicate to these tasks. Moreover, VeriFone s and Hypercom s arrangements with these resellers typically do not prevent them from selling products of other companies, including those of competitors, and they may elect to market competitors products and services in preference to the combined company system solutions. If one or more of the combined

company s major resellers terminates or otherwise adversely changes its relationship with the combined company, the combined company may be unsuccessful in replacing such relationship. The loss of a major reseller could impair the combined company s ability to sell its solutions and result in lower revenues and income. It could also be time-consuming and expensive to replicate, either directly or through other resellers, the certifications and the custom applications owned by these resellers.

In addition, orders from distributors and resellers depend on their sales volumes and inventory management decisions. In response to the global economic downturn a number of distributors and resellers experienced weakened demand and slower sales, which in turn resulted in declines in order volume and deferrals of orders for VeriFone s and Hypercom s products. Declines or deferral of orders could materially adversely affect the combined company s revenues, operating results and cash flows.

The combined company s solutions may have defects or experience field failures that could delay sales, harm the combined company s brand, increase costs and result in product recalls and additional warranty expenses.

The portfolio of products offered by the combined company will include complex solutions that are susceptible to undetected hardware and software errors or failures. Solutions may experience failures when first introduced, as new versions are released, or at any time during their lifecycle. Field failure may result from usage with third-party issued payment cards, for example, if such usage generates excess electrostatic discharge. Defects may also arise from third-party components that the combined company incorporates into its products, such as hardware modules, chipsets or battery cells. Any product recalls as a result of errors or failures could result in the loss of or delays in market acceptance of the combined company s solutions, adversely affect the combined company s business and reputation and increase its product costs, any of which could negatively impact its margins, profitability and results of operations. Any significant returns or warranty claims for any of the combined company s products could result in significant additional costs to the combined company and could adversely affect its results of operations. Customers may also run third-party software applications on the combined company s electronic payment systems. Errors in third-party applications could adversely affect the performance of the combined company s solutions.

The existence of defects and delays in correcting them could result in negative consequences, including the following:

harm to the combined company s brand;

delays in shipping system solutions;

loss of market acceptance for the combined company s system solutions;

additional warranty expenses;

diversion of resources from product development; and

loss of credibility with distributors, customers and partners.

Identifying and correcting defects can be time-consuming, costly and in some circumstances extremely difficult. Software errors may take several months to correct, and hardware defects may take even longer to correct.

Security will be vital to the combined company s customers and end users and therefore breaches in the security of the combined company s solutions could adversely affect the combined company s reputation and results of operations.

Protection against fraud will be of key importance to the purchasers and end users of the combined company s solutions. The combined company will incorporate security features, such as encryption software and

secure hardware, into its solutions to protect against fraud in electronic payment transactions and to ensure the privacy and integrity of consumer data. The combined company s solutions may be vulnerable to breaches in security due to defects in the security mechanisms, the operating system and applications, or the hardware platform. Security vulnerabilities could jeopardize the security of information transmitted or stored using the combined company s solutions. The combined company will also provide its customers with repair, encryption key loading and helpdesk services, and may experience security breaches or fraudulent activities related to unauthorized access to sensitive customer information. If the security of the combined company s solutions is compromised, the combined company s reputation and marketplace acceptance of its solutions will be adversely affected, which would cause its business to suffer, and would subject it to damages claims.

VeriFone intends to pursue additional acquisitions and strategic investments, which will involve numerous risks. VeriFone may not be able to address these risks without substantial expense, delay or other operational or financial problems for the combined company.

VeriFone expects to seek to acquire or make investments in related businesses, technologies, or products in the future. Acquisitions or investments, including the acquisition of Hypercom, involve various risks, such as:

the difficulty of integrating the technologies, operations, and personnel of the acquired business, technology or product;

the potential disruption of VeriFone s ongoing business, including the diversion of management attention to issues related to integration and administration, particularly given the number, size and varying scope of VeriFone s recent acquisitions;

in the case of foreign acquisitions, the need to integrate operations across different cultures and languages and to address the particular economic, currency, political, and regulatory risks associated with specific countries;

the possible inability to obtain the desired financial and strategic benefits from the acquisition or investment;

the loss of all or part of VeriFone s investment;

loss of customers;

the need to integrate each company s accounting, management, information, human resource and other administrative systems to permit effective management, and the lack of control if such integration is delayed or not implemented;

the need to implement controls, procedures and policies appropriate for a larger public company at companies that prior to acquisition had lacked such controls, procedures and policies;

the risk that increasing complexity inherent in operating a larger business may impact the effectiveness of VeriFone s internal controls and adversely affect its financial reporting processes;

assumption of unanticipated liabilities and the incurrence of unforeseen expenditures;

the loss of key employees of an acquired business; and

Edgar Filing: VERIFONE SYSTEMS, INC. - Form S-4

the possibility of VeriFone entering markets in which it has limited prior experience.

Future acquisitions and investments could also result in substantial cash expenditures, potentially dilutive issuances of VeriFone s equity securities and incurrence of additional debt, contingent liabilities and amortization expenses related to other intangible assets, which could adversely affect its business, operating results, and financial condition. Moreover, VeriFone may not realize the anticipated benefits of any or all of its acquisitions, or may not realize them in the timeframe expected. VeriFone will depend on the retention and performance of existing management and employees of acquired businesses for the day-to-day management and future operating results of these businesses.

Both VeriFone and Hypercom are parties to a number of lawsuits and the combined company may be named in additional litigation. Such litigation is likely to require significant management time and attention and expenses and may result in unfavorable outcomes that could have a material adverse effect on the combined company s business, financial condition and results of operations.

VeriFone and Hypercom are currently parties in several litigation proceedings. For example, in connection with the restatement of VeriFone s historical interim financial statements for fiscal year 2007, a number of securities class action complaints were filed against VeriFone and certain of its officers, and purported derivative actions have also been filed against certain of its current and former directors and officers.

The amount of time and resources required to resolve these lawsuits is unpredictable, and defending the combined company is likely to divert management s attention from the day-to-day operations of its business, which could adversely affect the combined company s business, financial condition, and results of operations. In addition, an unfavorable outcome in such litigation or a decision by the combined company to settle such lawsuits to avoid the distraction and expense of continued litigation (even if the combined company deems the claims to be without merit) could have a material adverse effect on the combined company s business, financial condition, and results of operations.

The combined company s insurance may not be sufficient to cover the costs of defending these actions or paying any damages in the event of an unfavorable outcome. In addition, VeriFone may be obligated to indemnify (and advance legal expenses to) both current and former officers, employees and directors in connection with the securities class action and derivative action matters. VeriFone currently holds insurance policies for the benefit of its directors and officers, although its insurance coverage may not be sufficient in some or all of these matters. Furthermore, the combined company s insurance carriers may seek to deny coverage in some or all of these matters, in which case the combined company may have to fund the indemnification amounts owed to such directors and officers itself.

VeriFone is subject to the risk of additional litigation and regulatory proceedings or actions in connection with the restatement of its financial statements. VeriFone has responded to inquiries and provided information and documents related to the restatement to the SEC, the U.S. Department of Justice, the New York Stock Exchange, and the Chicago Board Options Exchange. VeriFone was the subject of a Wells Notice from the SEC stating that the staff of the SEC s Division of Enforcement (the Staff) intends to recommend that the SEC bring a civil injunctive action against it, alleging violations of the federal securities laws arising from the restatement. Although VeriFone settled this matter with the SEC in November 2009, additional regulatory inquiries may also be commenced by other U.S. federal, state or foreign regulatory agencies. In addition, VeriFone may in the future be subject to additional litigation or other proceedings or actions arising in relation to the restatement of its historical interim financial statements. Litigation and any potential regulatory proceeding or action may be time-consuming, expensive and distracting from the conduct of the combined company s business. The adverse resolution of any specific lawsuit or any potential regulatory proceeding or action could have a material adverse effect on the combined company s business, financial condition, and results of operations.

These litigation proceedings could result in substantial additional costs and expenses and adversely affect the combined company s cash flows, and may adversely affect its business, financial condition, and results of operations. VeriFone has already incurred substantial expenses for legal, accounting, tax and other professional services in connection with the investigation by the audit committee of VeriFone s board of directors, VeriFone s internal review of its historical financial statements, the preparation of the restated financial statements, inquiries from government agencies, the related litigation, the amendments to its credit agreement as a result of its failure to timely file its Exchange Act reports with the SEC and other current litigation matters. VeriFone expects to continue to incur significant expenses in connection with these matters. Many members of VeriFone s senior management team and its board of directors have been and will be required to devote a significant amount of time to the litigation related to the restatement. In addition, certain of these individuals are named defendants in the litigation related to the restatement. If the combined company s senior management is unable to devote

sufficient time in the future to developing and pursuing its strategic business initiatives and running ongoing business operations, there may be a material adverse effect on the combined company s business, financial condition and results of operations.

In connection with the announced merger, several purported class action lawsuits have been filed in Arizona and Delaware state courts alleging variously, among other things, that the board of directors of Hypercom breached its fiduciary duties in not securing a higher price in the merger and that VeriFone, Hypercom, FP Hypercom Holdco, LLC and Francisco Partners II, L.P. aided and abetted that alleged breach. The actions seek injunctive relief and unspecified damages.

VeriFone s North American and international operations are not equally profitable, which may promote volatility in the combined company s earnings and adversely impact future growth in its earnings.

VeriFone s International sales of System Solutions have tended to carry lower average selling prices and therefore have lower gross margins than its sales in North America. As a result, if the combined company successfully executes plans to expand its international sales, any improvement in the combined company s results of operations will likely not be as favorable as an expansion of similar magnitude in the United States and Canada. In addition, the combined company will be unable to predict for any future period its proportion of revenues that will result from International sales versus sales in North America. Variations in this proportion from period to period may lead to volatility in its results of operations which, in turn, may depress the trading price of VeriFone s common stock.

Both VeriFone and Hypercom have experienced some fiscal quarters in which a significant percentage of the business for the quarter is executed towards the end of the fiscal quarter. This could negatively impact the combined company s business and results of operations.

Revenues recognized in VeriFone s and Hypercom s fiscal quarters can be back-end loaded, meaning that during a particular fiscal quarter, the timing of orders could be such that a substantial portion of sales orders are received, product is shipped, and revenue is recognized towards the end of the fiscal quarter. The occurrence of such back-end loading could adversely affect the combined company s business and results of operations due to a number of factors, including the following:

the manufacturing processes at third-party contract manufacturers could become concentrated in a shorter time period. This concentration of manufacturing could increase manufacturing costs, such as costs associated with the expediting of orders, and negatively impact gross margins. The risk of higher levels of obsolete or excess inventory write-offs would also increase if the combined company were to hold higher inventory levels to counteract this effect;

the higher concentration of orders may make it difficult to accurately forecast component requirements and, as a result, the combined company could experience a shortage of the components needed for production, possibly delaying shipments and causing lost orders;

if the combined company is unable to fill orders at the end of a quarter, shipments may be delayed. This could cause the combined company to fail to meet its revenue and operating profit expectations for a particular quarter and could increase the fluctuation of quarterly results if shipments are delayed from one fiscal quarter to the next or orders are cancelled by customers; and

in order to fulfill orders at the end of a quarter, the combined company may be forced to deliver its products using air freight which results in increased distribution costs.

The combined company will be exposed to credit risk with some of its customers and to credit exposures and currency controls in certain markets, which could result in material losses.

A significant portion of the combined company s sales is expected to be on an open credit basis, with typical payment terms of up to 60 days in the United States and, because of local customs or conditions, longer in some international markets. In the past, there have been bankruptcies among VeriFone s and Hypercom s customer

base. Although credit losses have not been material to date, future losses, if incurred, could harm the combined company s business and have a materially adverse effect on its operating results and financial condition. Also, certain customers who are invoiced in U.S. dollars, such as those based in Venezuela, have experienced and may continue to experience difficulties in obtaining U.S. dollar currency due to local currency controls, and therefore may not be able to remit timely payment to the combined company. Additionally, to the degree that the recent turmoil in the credit markets continues to make it more difficult for some customers to obtain financing or access U.S. dollar currency, the combined company s customers ability to pay could be adversely impacted, which in turn could have a material adverse impact on the combined company s business, cash flows, operating results and financial condition.

Fluctuations in currency exchange rates may adversely affect the combined company s results of operations.

A substantial portion of the combined company s business is expected to consist of sales made to customers outside the United States. A portion of the net revenues the combined company receives from such sales will be denominated in currencies other than the U.S. dollar. Additionally, portions of the combined company s cost of net revenues and its other operating expenses will be incurred by international operations and denominated in local currencies. Fluctuations in the value of these net revenues, costs and expenses as measured in U.S. dollars have historically affected VeriFone s and Hypercom s results of operations, and adverse currency exchange rate fluctuations may have a material impact in the future. Further, changes in exchange rates that strengthen the U.S. dollar could increase the price of the combined company s products in the local currencies of the foreign markets it serves. This would result in making the combined company s products relatively more expensive than products that are denominated in local currencies, leading to a reduction in sales and profitability in those foreign markets. In addition, the combined company s balance sheet will include monetary assets and liabilities denominated in currencies other than the U.S. dollar, such as cash, intercompany balances, trade receivables and payables, and fluctuations in the exchange rates for these currencies could adversely affect the combined company s results of operations. VeriFone has entered into foreign exchange forward contracts intended to hedge its balance sheet exposure to adverse fluctuations in exchange rates. VeriFone has also effectively priced its System Solutions in U.S. dollars in certain countries. Nevertheless, these hedging arrangements may not always be effective, particularly in the event of imprecise forecasts of non-U.S. denominated assets and liabilities. Additionally, VeriFone s efforts to effectively price products in U.S. dollars may have disadvantages, as they may affect demand for the combined company s products if the local currency strengthens relative to the U.S. dollar. The combined company could also be adversely affected when the U.S. dollar strengthens relative to the local currency between the time of a sale and the time the combined company receives payment, which would be collected in the devalued local currency. Accordingly, if there is an adverse movement in one or more exchange rates, the combined company might suffer significant losses and its results of operations may otherwise be adversely affected. Uncertainty in the global market conditions has resulted in and may continue to cause significant volatility in foreign currency exchange rates which could increase these risks. As the combined company s international operations expand, its exposure to these risks also increases. Additionally, hedging programs will expose the combined company to risks that could adversely affect its operating results, including the following:

the combined company may be unable to hedge currency risk for some transactions because of a high level of uncertainty or the inability to reasonably estimate its foreign exchange exposures; and

the combined company may be unable to acquire foreign exchange hedging instruments in some of the geographic areas where it does business, or, where these derivatives are available, the combined company may choose not to hedge because of the high cost of the derivatives.

The combined company will depend upon third parties to manufacture its systems and to supply the components necessary to manufacture its products.

VeriFone and Hypercom utilize a limited number of third parties to manufacture their hardware products pursuant to specifications and rely upon these contract manufacturers to produce and deliver products to their

customers on a timely basis and at an acceptable cost. In the case of VeriFone, a majority of these third-party manufacturing activities are concentrated in China. Disruptions to the business or operations of these contract manufacturers, or to their ability to produce the products the combined company requires in accordance with its and its customers requirements, and particularly disruptions to the manufacturing operations in China, could significantly affect the combined company s ability to fulfill customer demand on a timely basis which could materially harm the combined company s revenues and results of operations. During the fiscal quarter ended January 31, 2010, VeriFone shifted additional manufacturing activity to its contract manufacturer s facility in Israel, which will increase the combined company s dependency on its third-party contract manufacturers and could exacerbate these risks.

Components such as application specific integrated circuits, or ASICs, microprocessors, wireless modules, modems and printer mechanisms that are necessary to manufacture and assemble the combined company systems will be sourced either directly by the combined company or on its behalf by its contract manufacturers from a variety of component suppliers selected by it. Certain of the components will be specifically customized for use in the combined company s products and are expected to be obtained from sole source suppliers on a purchase order basis. In recent periods, VeriFone and Hypercom each have experienced a tightening in availability of certain semiconductor commodities that are necessary for the manufacture of their products. If the combined company s suppliers are unable or unwilling to deliver the quantities that the combined company requires, the combined company would be faced with a shortage of critical components. VeriFone and Hypercom also experience from time to time an increase in the lead time for delivery of some of their key components. The combined company may not be able to find alternative sources in a timely manner if suppliers of its key components become unwilling or unable to provide it with adequate supplies of these key components when the combined company needs them or if they increase their prices. If the combined company is unable to obtain sufficient key required components, or to develop alternative sources if and as required in the future, or to replace its component and factory tooling for its products in a timely manner if they are damaged or destroyed, the combined company could experience delays or reductions in product shipments. This could harm the combined company is relationships with its customers and cause its revenues to decline. Even if the combined company is able to secure alternative sources or replace its tooling in a timely manner, its costs could increase. Any of these events could adversely affect the combined company is results of operations.

The government tax benefits that VeriFone s subsidiaries have previously received or currently receive require them to meet several conditions and may be terminated or reduced in the future, which could require the combined company to pay increased taxes or refund tax benefits received in the past.

Israel

VeriFone s principal subsidiary in Israel (formerly Lipman) previously received tax benefits under Israeli law for capital investments that are designated as Approved Enterprises. VeriFone received such tax benefits of approximately \$0.4 million in 2009. Due to VeriFone s restructuring and contract manufacturing arrangements entered into in fiscal 2010, VeriFone no longer meets the requirements necessary to maintain the tax benefit status in Israel. Effective as of November 1, 2009, VeriFone has been taxed at the full statutory rate in Israel and no future tax benefit has been recorded. VeriFone s principal subsidiary in Israel has undistributed earnings of approximately \$181 million, the vast majority of which are attributable to Lipman s historic Approved Enterprise programs. As such, these earnings were not subject to Israeli statutory corporate tax at the time they were generated. Distribution or use of these funds outside Israel will subject VeriFone to payment of corporate and withholding taxes. For example, to the extent that these earnings are distributed to the United States in the future, VeriFone s Israeli subsidiary will be required to pay corporate tax at the rate ordinarily applicable to such earnings, currently between 12.5% and 36.25%, which includes the withholding tax between the United States and Israel. VeriFone has accrued approximately \$50 million for taxes associated with potential future distributions of VeriFone s Israeli subsidiary s approximately \$181 million in earnings.

Singapore

VeriFone s principal subsidiary in Singapore has received tax benefits under the Singapore Pioneer Tax Holiday provision. VeriFone received tax benefits of approximately \$8.3 million in 2010 and \$4.1 million in 2009. To maintain its eligibility for these benefits, VeriFone must meet certain agreed conditions, including maintaining agreed levels of Singapore employees and incurring and documenting total local business spend levels as agreed with the Singapore Economic Development Board. VeriFone recently renegotiated the terms and conditions of the Tax Holiday, including revision of certain of these conditions. Although VeriFone expects to be able to meet the terms and conditions to maintain its Tax Holiday, if VeriFone is not able to achieve or maintain the required conditions, VeriFone may lose its eligibility for such benefits which could result in its income in Singapore being taxed at the statutory rate of 17% instead of the agreed Pioneer Tax Holiday rate of 0%. A loss of all or part of these tax benefits would adversely affect the combined company s results of operations and cash flows.

Changes in VeriFone s effective tax rate could adversely affect results of operations.

VeriFone s effective tax rate could be adversely affected by a number of factors, including shifts in the mix of pretax profits and losses by tax jurisdiction, VeriFone s ability to use tax credits, changes in tax laws or related interpretations in the jurisdictions in which VeriFone operates, and tax assessments and related interest and penalties resulting from income tax audits. VeriFone is subject to ongoing tax audits in various jurisdictions. Although VeriFone regularly assesses the likely outcomes of such audits in order to determine the appropriateness of its tax provision, there can be no assurance that VeriFone will accurately predict the outcomes of these audits, and the actual outcomes of these audits could have a material impact on VeriFone s net income or financial condition. VeriFone has not provided for U.S. federal and state income taxes or foreign withholding taxes that may result from future remittances of undistributed earnings of foreign subsidiaries. Recently, there have been proposals to reform U.S. tax rules including proposals which may result in a reduction or elimination of the deferral of U.S. income tax on VeriFone s foreign earnings, in which case VeriFone s effective tax rate could be adversely affected. Any of these changes could have an adverse effect on the combined company s results of operations.

The combined company s quarterly operating results may fluctuate significantly as a result of factors outside of its control, which could cause the market price of VeriFone s common stock to decline.

VeriFone expects the combined company s revenues and operating results to vary from quarter to quarter. As a consequence, its operating results in any single quarter may not meet the expectations of securities analysts and investors, which could cause the price of VeriFone s common stock to decline. Factors that may affect the combined company s operating results include:

the type, timing and size of orders and shipments;

demand for and acceptance of the combined company s new product offerings;

customers willingness to maintain inventories and/or increased overall channel inventories held by customers in a particular quarter;

delays in the implementation and delivery of its products and services, which may impact the timing of the combined company s recognition of revenues;

variations in product mix and cost during any period;

development of new relationships, penetration of new markets and maintenance and enhancement of existing relationships with customers and strategic partners;

component supply, manufacturing, or distribution difficulties;

Edgar Filing: VERIFONE SYSTEMS, INC. - Form S-4

deferral of customer contracts in anticipation of product or service enhancements;

timing of commencement, implementation, or completion of major implementation projects;

timing of governmental, statutory and industry association requirements;

the relative geographic mix of net revenues;

fluctuations in currency exchange rates;

the fixed nature of many of the combined company s expenses; and

industry and economic conditions, including competitive pressures and inventory obsolescence. In particular, differences in relative growth rates between the combined company s businesses in North America and internationally may have a significant effect on its operating results, particularly its reported gross profit percentage, in any individual quarter, with international sales carrying lower margins.

In addition, the combined company may experience periodic variations in sales to its key vertical and international markets. These periodic variations would occur throughout the year and may lead to fluctuations in the combined company s quarterly operating results depending on the impact of any given market during that quarter and could lead to volatility in VeriFone s stock price.

The combined company may accumulate excess or obsolete inventory that could result in unanticipated price reductions and write-downs and adversely affect the combined company s financial condition.

The combined company expects to focus its efforts on providing customers with solutions that have high levels of functionality, which will require the combined company to develop and incorporate cutting edge and evolving technologies. This approach tends to increase the risk of obsolescence for products and components the combined company holds in inventory and may compound the difficulties posed by other factors that affect the combined company s inventory levels, including the following:

the need to maintain significant inventory of components that are in limited supply;

buying components in bulk for the best pricing;

responding to the unpredictable demand for products;

cancellation of customer orders;

responding to customer requests for quick delivery schedules; and

timing of end-of-life decisions regarding products.

Edgar Filing: VERIFONE SYSTEMS, INC. - Form S-4

The accumulation of excess or obsolete inventory has in the past resulted in, and may in future periods result in, price reductions and inventory write-downs and scrap, which could adversely affect the combined company s business, results of operations and financial condition. For example, for the fiscal year ended October 31, 2009, VeriFone incurred costs for obsolete inventory, scrap, and purchase commitments for excess components at contract manufacturers of \$23.0 million due to changing demand VeriFone experienced in fiscal year 2009 as a result of the severe deterioration in the macroeconomic environment.

If the combined company does not accurately forecast customer demand and effectively manage its product mix and inventory levels, the combined company may lose sales from having too few or the wrong mix of products or incur costs associated with excess inventory.

If the combined company improperly forecasts demand for its products it could end up with too many products and be unable to sell the excess inventory in a timely manner, if at all, or, alternatively it could end up with too few products and not be able to satisfy demand. This problem is exacerbated because the combined company s attempt to closely match inventory levels with product demand will leave limited margin for error,

and VeriFone and Hypercom generally receive a significant volume of customer orders towards the end of each fiscal quarter which will leave the combined company little room to adjust inventory mix to match demand. Also, during the transition from an existing product to a new replacement product, the combined company must accurately predict the demand for the existing and the new product. The combined company s inability to properly manage its inventory levels could cause it to incur increased expenses associated with writing off excessive or obsolete inventory, or lose sales, or have to ship products by air freight to meet immediate demand incurring incremental freight costs above sea freight costs, a preferred method, and suffering a corresponding decline in gross margins. If the combined company does not accurately predict demand, it could also incur increased expenses associated with binding commitments to certain third-party contract manufacturers which would negatively impact the combined company s gross margins and operating results. During times of economic uncertainty, such as that of the current global economic environment, it becomes more difficult to accurately forecast demand and manage the combined company s inventory levels. Deteriorating market conditions have in the past and can in future periods cause the combined company to incur additional costs associated with excess and obsolete inventory, scrap and excess inventory held by the combined company s contract manufacturers.

The combined company may be subject to additional impairment charges due to potential declines in the fair value of its assets.

As a result of VeriFone s acquisitions, particularly that of Lipman in November 2006, VeriFone has recorded significant goodwill and intangible assets on its balance sheet, and following the acquisition of Hypercom, VeriFone expects that these amounts will increase. VeriFone tests goodwill and intangible assets for impairment on a periodic basis as required, and whenever events or changes in circumstances indicate that the carrying value may not be recoverable. The events or changes that could require VeriFone to test its goodwill and intangible assets for impairment include a reduction in VeriFone s stock price and market capitalization and changes in VeriFone s estimated future cash flows, as well as changes in rates of growth in VeriFone s industry or in any of its reporting units. In the fourth fiscal quarter of 2008, VeriFone recorded an impairment charge of \$289.1 million for goodwill and developed technology intangible assets due to lower revenue expectations in light of current operating performance and future operating expectations. During the first fiscal quarter of 2009, VeriFone concluded that the carrying amount of the North America and Asia reporting units exceeded their implied fair values and recorded an estimated impairment charge of \$178.2 million. VeriFone finalized the goodwill evaluation process and recorded a \$2.7 million reduction of impairment charge during the second quarter of fiscal year 2009. The final goodwill impairment charge was \$175.5 million as of April 30, 2009. VeriFone has not recorded any further impairment charges since the fiscal quarter ended April 30, 2009.

VeriFone will continue to evaluate the carrying value of its remaining goodwill and intangible assets and if it determines in the future that there is a potential further impairment in any of its reporting units, it may be required to record additional charges to earnings which could materially adversely affect the combined company s financial results and could also materially adversely affect its business. The process of evaluating the potential impairment of goodwill and intangible assets is subjective and requires significant judgment at many points during the analysis and includes estimates of future cash flows attributable to a reporting unit or asset over its estimated remaining useful life. Any changes in VeriFone s estimates, such as its estimates of the future cash flows attributable to a reporting unit or asset, or a longer or more significant decline in its market capitalization or the macroeconomic environment, could require the combined company to record additional impairment charges which could materially adversely affect its financial results.

The combined company s proprietary technology will be difficult to protect and unauthorized use of its proprietary technology by third parties may impair its ability to compete effectively.

The combined company may not be able to protect its proprietary technology, which could enable competitors to develop services that compete with its own. The combined company will rely on patent, copyright,

trademark and trade secret laws, as well as confidentiality, licensing and other contractual arrangements, to establish and protect the proprietary aspects of its solutions. Institution of legal proceedings to enforce its intellectual property rights could be costly and divert the efforts and attention of its management and technical personnel from other business operations. In addition, there can be no assurance that such proceedings would be determined in its favor. The combined company is not expected to have patent protection for certain important aspects of its solutions. The laws of some countries in which the combined company will sell its solutions and services may not protect software and intellectual property rights to the same extent as the laws in the United States. If the combined company is unable to prevent misappropriation of its technology, competitors may be able to use and adapt its technology. The combined company s failure to protect its technology could diminish its competitive advantage and cause it to lose customers to competitors.

The combined company s business may suffer if it is sued for infringing the intellectual property rights of third parties, or if it is unable to obtain rights to third-party intellectual property on which it depends.

Third parties have in the past asserted and may in the future assert claims that VeriFone s and Hypercom s system solutions infringe their proprietary rights. Such infringement claims, even if meritless, may cause the combined company to incur significant costs in defending those claims. The combined company may be required to discontinue using and selling any infringing technology and services, to expend resources to develop non-infringing technology or to purchase licenses or pay royalties for other technology. Similarly, the combined company will depend on its ability to license intellectual property from third parties. These third parties may become unwilling to license to the combined company on acceptable terms intellectual property that is necessary to its business. The combined company may be unable to acquire licenses for other technology on reasonable commercial terms or at all. As a result, the combined company may find that it is unable to continue to offer the solutions and services upon which its business depends.

Both VeriFone and Hypercom have received, and have currently pending, third-party infringement claims and may receive additional notices of claims of infringement in the future. Infringement claims may cause the combined company to incur significant costs in defending such claims or to settle claims to avoid costly or protracted litigation, even if the combined company deems the claims to be without merit. For example, in March 2008, Cardsoft, Inc. and Cardsoft (Assignment for the Benefit of Creditors), LLC commenced an infringement claims are expensive and time-consuming to defend, regardless of the merits or ultimate outcome. Similar claims may result in additional protracted and costly litigation. There can be no assurance that the combined company will prevail in any such actions or that any license required under any such patent or other intellectual property would be made available on commercially acceptable terms, if at all.

The combined company will face litigation and tax assessment risks that could force it to incur substantial defense costs and could result in damages awards against it that would negatively impact its business.

As described in VeriFone s and Hypercom s quarterly and annual reports filed with the SEC, there are a number of pending litigation and tax assessment matters faced by the companies, each of which may be time-consuming to resolve, expensive to defend, and disruptive to normal business operations. The outcome of litigation and tax assessments is inherently difficult to predict. An unfavorable resolution of any specific lawsuit or tax assessment could have a material adverse effect on the combined company s business, results of operations and financial condition.

VeriFone has significant operations in Israel and therefore the combined company s results of operations may be adversely affected by political or economic instability or military operations in or around Israel.

VeriFone has offices and personnel in Israel. In addition, a number of VeriFone s products are manufactured by VeriFone s contract manufacturer in facilities located in Israel and many of VeriFone s suppliers are located in Israel. Therefore, political, economic, and military conditions in Israel directly affect the combined company s

operations. The future of peace efforts between Israel and its Arab neighbors remains uncertain. Any armed conflicts or further political instability in the region is likely to negatively affect business conditions and materially harm the combined company s results of operations. Furthermore, several countries continue to restrict or ban business with Israel and Israeli companies. These restrictive laws and policies may seriously limit the combined company s ability to make sales in those countries.

In addition, many employees in Israel are obligated to perform at least 30 days and up to 40 days, depending on rank and position, of military reserve duty annually and are subject to being called for active duty under emergency circumstances. If a military conflict or war arises, these individuals could be required to serve in the military for extended periods of time. VeriFone s operations in Israel could be disrupted by the absence for a significant period of one or more key employees or a significant number of other employees due to military service. Any disruption in VeriFone s operations in Israel could materially adversely affect the combined company s business.

In early 2010, VeriFone completed the transfer of the manufacture of its NURIT products from an in-house facility to a contract manufacturer with operations in Israel, which manufactures products and warehouses components and finished goods inventories. Disruption of the manufacturing process of VeriFone s contract manufacturer or damage to its facility, whether as a result of fire, natural disaster, act of war, terrorist attack, or otherwise, could materially affect the combined company s ability to deliver products on a timely basis and could materially adversely affect the combined company s results of operations.

The combined company may not be able to attract, integrate, manage and retain qualified personnel.

The combined company s success will depend to a significant degree upon the continued contributions of its key senior management, engineering, sales and marketing, and manufacturing personnel, many of whom would be difficult to replace. In addition, the combined company s success will also depend on its ability to attract, integrate, manage, and retain highly skilled employees throughout its business. Competition for some of these personnel is intense, and in the past, VeriFone has had difficulty hiring employees in its desired timeframe, particularly qualified finance and accounting professionals. The combined company may be unsuccessful in attracting and retaining personnel. The loss of the services of any of the combined company s key personnel, the inability to attract or retain qualified personnel in the future, or delays in hiring required personnel, particularly engineers and sales personnel, could make it difficult for the combined company to manage its business and meet key objectives, such as timely product introductions.

During the last several fiscal years, both VeriFone and Hypercom have implemented work force reduction plans reducing the number of employees and contractors in certain areas due to redundancies and shifting business needs. These reductions have also required both companies to reassign certain employee duties. Workforce reductions and job reassignments could negatively affect employee morale, and make it difficult to motivate and retain the combined company s remaining employees and contractors, which would affect the combined company s ability to deliver its products in a timely fashion and otherwise negatively affect its business.

Shipments of electronic payment systems may be delayed by factors outside of the combined company s control, which could harm its reputation and its relationships with its customers.

The shipment of payment systems will require the combined company or its manufacturers, distributors, or other agents to obtain customs or other government certifications and approvals, and, on occasion, to submit to physical inspection of systems in transit. Failure to satisfy these requirements, and the very process of trying to satisfy them, can lead to lengthy delays in the delivery of the combined company solutions to its direct or indirect customers. Because the combined company will depend upon third-party carriers for the timely delivery of its products, the combined company may face delays in delivery due to reasons outside its control. Delays and unreliable delivery by the combined company may harm its reputation in the industry and its relationships with its customers.

Force majeure events, such as terrorist attacks, other acts of violence or war and political instability may adversely affect the combined company.

Terrorist attacks, war and international political instability may disrupt the combined company s ability to generate revenues. Such events may negatively affect the combined company s ability to maintain sales revenues and to develop new business relationships. Because it is expected that a substantial and growing part of the combined company s revenues will be derived from sales and services to customers outside of the United States and VeriFone and Hypercom have their electronic payment systems manufactured outside the United States, terrorist attacks, war and international political instability anywhere may decrease international demand for the combined company s products and inhibit customer development opportunities abroad, disrupt the combined company s supply chain and impair its ability to deliver its electronic payment systems, which could materially adversely affect the combined company s net revenues or results of operations. Any of these events may also disrupt global financial markets and precipitate a decline in the price of VeriFone s common stock.

Natural or man-made disasters, business interruptions and health epidemics could delay the combined company s ability to receive or ship its products, or otherwise disrupt its business.

The combined company s worldwide operations could be subject to earthquakes, power shortages, telecommunications failures, water shortages, tsunamis, floods, hurricanes, typhoons, fires, extreme weather conditions, health epidemics and other natural or man-made disasters or business interruptions. The occurrence of any of these business disruptions could seriously harm the combined company s revenue and financial condition and increase its costs and expenses. If the combined company s manufacturers or warehousing facilities are damaged or destroyed, the combined company would be unable to distribute its products on a timely basis, which could harm its business. VeriFone s corporate headquarters, and a portion of its research and development activities, are located in California, and other critical business operations and some of its suppliers are located in California and Asia, near major earthquake faults. Certain key servers and information systems are located in Florida, which has in the past experienced major hurricanes and similar extreme weather. Although VeriFone s office facilities have not been impacted by the flooding, power outages or other effects of such storm systems, any disruption of VeriFone s Florida operations could materially affect the combined company s operations and harm its business. Moreover, if the combined company s computer information systems or communication systems, or those of its vendors or customers, are subject to hacker attacks or other disruptions, the combined company s business could suffer. VeriFone has not established a comprehensive disaster recovery plan. The combined company s back-up operations may be inadequate and its business interruption insurance may not be enough to compensate it for any losses that may occur. A significant business interruption could result in losses or damages and harm the combined company s business since much of VeriFone s order fulfillment process is automated and the order information is stored on VeriFone s servers. If its computer systems and servers go down even for a short period at the end of a fiscal quarter, VeriFone s ability to recognize revenue would be delayed until it is again able to process and ship its orders, which could harm the combined company s revenues for that quarter and cause VeriFone s stock price to decline significantly.

VeriFone and Hypercom are exposed to potential risks associated with environmental laws and regulations.

VeriFone and Hypercom are subject to other legal and regulatory requirements, including a European Union directive that places restrictions on the use of hazardous substances (RoHS) in electronic equipment, a European Union directive on Waste Electrical and Electronic Equipment (WEEE), and the environmental regulations promulgated by China s Ministry of Information Industry (China RoHS). RoHS sets a framework for producers obligations in relation to manufacturing (including the amounts of named hazardous substances contained in products sold) and WEEE sets a framework for treatment, labeling, recovery, and recycling of electronic products in the European Union, which may require the combined company to alter the manufacturing of the physical devices that include its solutions and/or require active steps to promote recycling of materials and components. In addition, similar legislation could be enacted in other jurisdictions, including in the United States. Many states in the United States have enacted state-level programs and requirements for recycling

of certain electronic goods. In addition, climate change legislation in the United States is a significant topic of discussion and may generate federal or other regulatory responses in the near future. If the combined company does not comply with environmental law and regulations, it may suffer a loss of revenue, be unable to sell in certain markets or countries, be subject to penalties and enforced fees, and/or suffer a competitive disadvantage. Furthermore, the costs to comply with RoHS, WEEE and China RoHS, or with current and future environmental and worker health and safety laws, may have a material adverse effect on the combined company s business, results of operations and financial condition.

Risks Related to the Combined Company s Industry

The combined company s markets are highly competitive and subject to price erosion.

The markets for the combined company s system solutions and services will be highly competitive, and VeriFone and Hypercom have been subject to price pressures. Competition from manufacturers, distributors, or providers of products similar to or competitive with the combined company s system solutions or services could result in price reductions, reduced margins, and a loss of market share or could render its solutions obsolete. For example, First Data Corporation, a leading provider of payments processing services, and VeriFone s largest customer, has developed and continues to develop a series of proprietary electronic payment systems for the U.S. market.

The combined company expects to continue to experience significant and increasing levels of competition in the future. The combined company will compete with suppliers of cash registers that provide built-in electronic payment capabilities and producers of software that facilitates electronic payment over the internet, as well as other manufacturers or distributors of electronic payment systems. The combined company will also compete with smaller companies that have been able to develop strong local or regional customer bases. In certain foreign countries, some competitors are more established, benefit from greater name recognition and have greater resources within those countries than the combined company will. Further, in certain international markets, such as Brazil, the combined company may face competition from refurbished units which could result in reduced demand and pricing pressures.

The combined company must adhere to industry and government regulations and standards and therefore sales will suffer if the combined company cannot comply with them.

The combined company s system solutions must meet industry standards imposed by payment systems standards setting organizations such as EMVCo LLC, credit card associations such as Visa, MasterCard, and other credit card associations and standard-setting organizations such as PCI SSC, Intermec and the UK Cards Association. New standards are continually being adopted or proposed as a result of worldwide anti-fraud initiatives, the increasing need for system compatibility and technology developments such as wireless and wireline IP communication. The combined company s solutions will also have to comply with government regulations, including those imposed by telecommunications authorities and independent standards groups worldwide, regarding emissions, radiation, and connections with telecommunications and radio networks. VeriFone cannot be sure that the combined company will be able to design its solutions to comply with future standards or regulations on a timely basis, if at all. Compliance with these standards could increase the cost of developing or producing the combined company s solutions. New products designed to meet any new standards need to be introduced to the market and ordinarily need to be certified by the credit card associations and the combined company s customers before being purchased. The certification process is costly and time-consuming and increases the amount of time it will take to sell the combined company s products. Moreover, certain uses of the combined company s products may subject it to additional regulations and licensing requirements. For example, use of VeriFone s products in taxis requires additional licensing and may subject VeriFone to certain taxi business regulations.

The combined company s business, revenues and financial condition could be adversely affected if it cannot comply with new or existing industry standards, or obtain or retain necessary regulatory approval or certifications

in a timely fashion, or if compliance results in increasing the cost of the combined company s products. Selling products that are non-compliant may result in fines against the combined company or its customers, which the combined company may be liable to pay. In addition, even if the combined company s products are designed to be compliant, compliance with certain security standards is determined based on the merchant s or service provider s network environment in which the combined company s systems are installed and, therefore, is dependent upon a number of additional factors, such as proper installation of the components of the environment including the combined company s systems, compliance of software and system components provided by other vendors, implementation of compliant security processes and business practices and adherence to such processes and practices. The combined company s business and financial condition could be adversely affected if the combined company does not comply with new or existing industry standards, or obtain or retain necessary regulatory approval or certifications in a timely fashion, or if compliance results in increasing the cost of the combined company s products.

If the combined company does not continually enhance its existing solutions and develop and market new solutions and enhancements, its net revenues and income will be adversely affected.

The market for electronic payment systems is characterized by:

rapid technological change;

frequent product introductions and enhancements;

evolving industry and government performance and security standards; and

changes in customer and end-user preferences or requirements.

Because of these factors, the combined company must continually enhance its solutions and develop and market new solutions. These efforts will require significant investment in research and development as well as increased costs of manufacturing and distributing the combined company s system solutions, and the combined company may not necessarily be able to increase or maintain prices to account for these costs.

VeriFone cannot be sure that the combined company will successfully complete the development and introduction of new solutions or enhancements or that its new solutions will be accepted in the marketplace. The combined company may also fail to develop and deploy new solutions and enhancements on a timely basis. In either case, the combined company may lose market share to its competitors, and its net revenues and results of operations could suffer.

Risks Related to VeriFone s Capital Structure

VeriFone, Inc. s secured credit facility contains restrictive and financial covenants and, if VeriFone, Inc. is unable to comply with these covenants, it will be in default. A default could result in the acceleration of VeriFone s outstanding indebtedness, which would have an adverse effect on the combined company s business and stock price.

On October 31, 2006, VeriFone Systems, Inc. s principal subsidiary, VeriFone, Inc., entered into a secured credit agreement consisting of a Term B Loan facility of \$500.0 million and a revolving credit facility permitting borrowings of up to \$40.0 million (the Credit Facility). The proceeds from the Term B loan were used to repay all outstanding amounts relating to an existing senior secured credit agreement, pay certain transaction costs, and partially fund the cash consideration in connection with the acquisition of Lipman Electronic Engineering Ltd. on November 1, 2006. Through October 31, 2010, VeriFone, Inc. had repaid an aggregate of \$278.7 million, leaving a Term B Loan balance of \$221.3 million at October 31, 2010.

The Credit Facility contains customary covenants that require VeriFone, Inc. s subsidiaries to maintain certain specified financial ratios and restrict their ability to make certain distributions with respect to their capital

stock, prepay other debt, encumber their assets, incur additional indebtedness, make capital expenditures above specified levels, engage in certain business combinations, or undertake various other corporate activities. Therefore, as a practical matter, these covenants restrict VeriFone s ability to engage in or benefit from such activities. In addition, VeriFone, Inc. has, in order to secure repayment of its Credit Facility, pledged substantially all of its assets and properties. This pledge may reduce VeriFone s operating flexibility because it restricts VeriFone s ability to dispose of these assets or engage in other transactions that may be beneficial to VeriFone.

If VeriFone, Inc. is unable to comply with the covenants in the Credit Facility, it will be in default, which could result in the acceleration of VeriFone s outstanding indebtedness. In addition, if VeriFone, Inc. s leverage exceeds a certain level set out in its Credit Facility, a portion of its excess cash flows must be used to pay down its outstanding debt. If acceleration occurs, VeriFone may not be able to repay its debt and it may not be able to borrow sufficient additional funds to refinance its debt. The U.S. credit markets have contracted significantly and as a result, VeriFone, Inc. may not be able to obtain additional financing on acceptable terms, or at all. If VeriFone, Inc. were to default in performance under the Credit Facility, it may pursue an amendment or waiver of the Credit Facility with its existing lenders, but there can be no assurance that the lenders would grant such an amendment or waiver and, in light of current credit market conditions, any such amendment or waiver requested is likely to be on terms, including additional fees, as well as increased interest rates and other more stringent terms and conditions, that would be materially disadvantageous to VeriFone, Inc. For example, as a result of the delay in VeriFone s financial reports for the 2007 fiscal year and the first two fiscal quarters of 2008, VeriFone, Inc. was required to obtain amendments to the Credit Facility that resulted in an increase in the interest rate payable on its term loan and revolving commitments, as well as increases in the commitment fee for unused revolving commitments and letter of credit fees. VeriFone, Inc. also paid the consenting lenders amendment fees in connection with the amendments. VeriFone expects that in light of recent market conditions any lender s fees or other terms and conditions for a covenant waiver or amendment would be substantially more costly to VeriFone, Inc. today than the cost it incurred for credit agreement amendments in 2008.

The conditions of the U.S. and international capital markets may adversely affect VeriFone, Inc. s ability to draw on the Credit Facility as well as have an adverse effect on other financial transactions.

Lehman Commercial Paper, Inc. (Lehman CP) was a lender under the Credit Facility with a commitment of \$15 million out of the \$40 million facility. As a result of Lehman CP s filing of a voluntary Chapter 11 bankruptcy petition in October 2008, VeriFone, Inc. reduced the revolving credit facility by Lehman CP s commitment.

In addition, the filing by Lehman Brothers of a voluntary Chapter 11 bankruptcy petition in September 2008 constituted an event of default under the note hedge transaction with Lehman Derivatives (which has also filed a voluntary Chapter 11 bankruptcy petition) and the related guaranty by Lehman Brothers, giving VeriFone the immediate right to terminate the note hedge transaction and entitling it to claim reimbursement for the loss incurred in terminating and closing out the transaction. To date, VeriFone has not recovered any amounts or reimbursement related to the note hedge transactions with Lehman Derivatives and there can be no assurance VeriFone will ever receive any amounts, reimbursement or recovery related to the obligations of Lehman Derivatives and Lehman Brothers with respect to this note hedge transaction. Lehman Derivatives and Lehman Brothers ability to perform their obligations under the note hedge transactions is unclear. Because VeriFone has not replaced this hedge transaction, VeriFone is subject to potential dilution on the portion of its convertible notes that is intended to be covered by the hedge transaction with Lehman Derivatives upon conversion if, on the date of conversion, the per share market price of VeriFone s common stock exceeds the conversion price of approximately \$44.02 per share. Further, if VeriFone decided to replace the hedge it could incur significant costs to do so.

If other financial institutions that have extended credit commitments to VeriFone or have entered into hedge, insurance or similar transactions with VeriFone are adversely affected by the conditions of the U.S. and

international capital markets, they may become unable to fund borrowings under their credit commitments to VeriFone or otherwise fulfill their obligations under the relevant transactions, which could have a material and adverse impact on the combined company s financial condition and its ability to borrow additional funds, if needed, for working capital, capital expenditures, acquisitions and other corporate purposes.

VeriFone, Inc. s indebtedness and debt service obligations under the Credit Facility may adversely affect the combined company s cash flow, cash position, and stock price.

Principal payments on VeriFone, Inc. s Term B Loan are due in quarterly installments of \$1.25 million over the term of the loan with the remaining principal balance due October 31, 2013. VeriFone, Inc. intends to fulfill its debt service obligations under the Credit Facility from existing cash, investments and operations. A substantial portion of VeriFone, Inc s cash balances and cash generated from operations are held by its foreign subsidiaries. If VeriFone, Inc. decides to distribute or use such cash and cash equivalents outside those foreign jurisdictions, including a distribution to the United States, it may be subject to additional taxes or costs. In the future, if VeriFone, Inc. is unable to generate or raise additional cash sufficient to meet its debt service obligations and needs to use more of its existing cash than planned or to liquidate investments in order to fund these obligations, VeriFone may have to delay or curtail the development and/or the sales and marketing of new payment systems. The Term B Loan bears interest at 2.75% over the applicable LIBOR rate. As a result of the recent global recession and downturn in financial markets, LIBOR has declined significantly, which has resulted in a lower effective interest rate on the Term B Loan. LIBOR fluctuates based on economic and market factors that are beyond VeriFone s control. Any significant increase in LIBOR would result in a significant increase in interest expense on the Term B Loan, which could negatively impact the combined company s net income and cash flows.

VeriFone, Inc. s indebtedness could have significant additional negative consequences, including, without limitation:

requiring the dedication of a significant portion of the combined company s expected cash flow to service the indebtedness, thereby reducing the amount of expected cash flow available for other purposes, including capital expenditures;

increasing the combined company s vulnerability to general adverse economic conditions;

limiting the combined company s ability to obtain additional financing; and

placing the combined company at a possible competitive disadvantage to less leveraged competitors and competitors that have better access to capital resources.

Additionally, if the combined company is required to refinance or raise additional cash to settle its existing indebtedness on or prior to its maturity, its ability to successfully achieve such objective will be dependent on a number of factors, including but not limited to its business outlook, projected financial performance, general availability of corporate credit and market demand for its securities offerings. VeriFone believes that in the event additional funding is required, the combined company will be able to access the capital markets on terms and in amounts adequate to meet its objectives. However, given the possibility of changes in market conditions or other occurrences, VeriFone cannot assure that such funding will be available on terms favorable to the combined company or at all.

Some provisions of VeriFone s certificate of incorporation and bylaws may delay or prevent transactions that many stockholders may favor.

Some provisions of VeriFone s certificate of incorporation and bylaws may have the effect of delaying, discouraging or preventing a merger or acquisition that VeriFone s stockholders may consider favorable, including transactions in which stockholders might receive a premium for their shares. These provisions include:

authorization of the issuance of blank check preferred stock without the need for action by stockholders;

the amendment of its organizational documents only by the affirmative vote of the holders of two-thirds of the shares of its capital stock entitled to vote;

provision that any vacancy on the board of directors, however occurring, including a vacancy resulting from an enlargement of the board, may only be filled by vote of the directors then in office;

inability of stockholders to call special meetings of stockholders, although stockholders are permitted to act by written consent; and

advance notice requirements for board nominations and proposing matters to be acted on by stockholders at stockholder meetings.

Cautionary Statement Concerning Forward-Looking Statements

This proxy statement/prospectus and the documents incorporated by reference into this proxy statement/prospectus contain forward-looking statements within the meaning of the federal securities laws. These statements relate to future events or the future financial performance of VeriFone, Hypercom or the combined company. In some cases, you can identify forward-looking statements by terminology such as may, should, expect, plan, intend, anticipate, believe, estimate, predict, potential, or continue, the negative of such terms, or compa

Actual events or results may differ materially. In evaluating these statements, you should specifically consider various factors, including the risks discussed under the caption entitled Risk Factors on page 26 and the other cautionary statements made in this proxy statement/prospectus. These factors may cause VeriFone s, Hypercom s and the combined company s actual results to differ materially from any forward-looking statement.

Although VeriFone and Hypercom believe that the expectations reflected in the forward-looking statements are reasonable, VeriFone and Hypercom cannot guarantee future results, events, levels of activity, performance, or achievements. Moreover, none of VeriFone, Hypercom or any other person assumes responsibility for the accuracy and completeness of the forward-looking statements.

These statements relate to future events or the future financial performance of VeriFone, Hypercom or the combined company, and involve known and unknown risks, uncertainties, and other factors that may cause actual results, levels of activity, performance, or achievements to be materially different from any future results, levels of activity, performance, or achievements expressed or implied by these forward-looking statements. These risks and other factors include those listed under the caption entitled Risk Factors on page 26, and elsewhere in this proxy statement/prospectus. VeriFone and Hypercom are under no duty to update any of the forward-looking statements after the date of this proxy statement/prospectus to conform such statements to actual results or to changes in expectations.

Information about the Special Meeting

General; Date; Time and Place

This proxy statement/prospectus is furnished in connection with the solicitation of proxies by Hypercom s board of directors for use at the special meeting. The meeting will be held at 9:00, a.m. Arizona time on , , 2011, at 8888 East Raintree Drive, Suite 300, Scottsdale, Arizona 85260, unless it is postponed or adjourned.

Purpose of the Special Meeting

At the special meeting, Hypercom stockholders of record will be asked to approve and adopt the merger agreement and approve the merger.

Holders of record of Hypercom common stock also will consider any other business that may properly come before the special meeting of stockholders or any adjournment or postponement of the special meeting of stockholders. Hypercom currently does not contemplate that any other matters will be considered at the special meeting.

Record Date; Voting Power

The proxy statement/prospectus and the accompanying proxy card are being mailed beginning on or about , 2010, to holders of record of Hypercom s common stock at the close of business on , 2010. Each share of Hypercom common stock entitles the registered holder thereof to one vote. As of December 17, 2010, there were 55,992,883 shares of Hypercom common stock outstanding, including restricted stock. As of December 17, 2010, approximately 1.47% of the outstanding shares of Hypercom common stock, including restricted stock and excluding stock options, were held by directors and executive officers of Hypercom and their respective affiliates.

You are entitled to vote if you are a holder of record of Hypercom common stock as of the close of business on , 2010. Your shares may be voted at the meeting only if you are present or represented by a valid proxy.

Recommendation of the Board of Directors of Hypercom Concerning the Merger

After careful consideration, the board of directors of Hypercom determined that the merger is fair to and in the best interests of Hypercom and its stockholders and approved the merger agreement, the merger and all of the transactions contemplated by the merger agreement. Accordingly, the board of directors of Hypercom recommends that you vote FOR the proposal to approve and adopt the merger agreement and approve the merger. For a discussion of the merger and the merger agreement, see The Merger beginning on page 59 and The Merger Agreement beginning on page 83.

Method of Voting; Record Date; Stock Entitled to Vote; Quorum

Hypercom stockholders are being asked to vote both shares held directly in their name as stockholders of record and any shares they hold in street name as beneficial owners. Shares held in street name are shares held in a stock brokerage account or shares held by a bank or other nominee.

The method of voting differs for shares held as a record holder and shares held in street name. Record holders will receive proxy cards. Holders of shares in street name will receive voting instruction cards in order to instruct their brokers or nominees how to vote.

Proxy cards and voting instruction cards are being solicited on behalf of the board of directors of Hypercom from Hypercom stockholders in favor of the proposal to approve and adopt the merger agreement and approve

the merger. In addition, Hypercom has retained Innisfree M&A Incorporated of 501 Madison Avenue, 20th Floor, New York, New York 10022, to assist in the solicitation of proxies for a fee of approximately \$30,000, including disbursements based on out-of-pocket expenses, telecommunicators, directory assistance and related telephone expenses.

Stockholders may receive more than one set of voting materials, including multiple copies of this proxy statement/prospectus and multiple proxy cards or voting instruction cards. For example, stockholders who hold shares in more than one brokerage account will receive a separate voting instruction card for each brokerage account in which shares are held. Stockholders of record whose shares are registered in more than one name will receive more than one proxy card. Hypercom stockholders should complete, sign, date and return each proxy card and voting instruction card they receive.

Only stockholders of record of Hypercom at the close of business on , 2010, the record date for the special meeting of stockholders, are entitled to receive notice of, and have the right to vote at, the special meeting of stockholders. Stockholders of record of Hypercom on the record date are entitled to one vote per share of Hypercom common stock on the proposal to approve and adopt the merger agreement and approve the merger.

In accordance with Hypercom s corporate bylaws, stockholders who represent a majority of the shares of Hypercom entitled to vote must be present in person or represented by proxy in order to constitute a quorum to conduct business at a meeting of stockholders.

Abstentions and broker non-votes count as present for establishing a quorum. A broker non-vote occurs on an item when a broker is not permitted to vote on that item without instructions from the beneficial owner of the shares and no instructions are given. Shares held by Hypercom in its treasury do not count toward establishing a quorum.

Adjournment and Postponement

Hypercom stockholders may be asked to vote upon a proposal to adjourn or postpone the special meeting.

Required Vote

Adoption of the merger agreement and approval of the merger will require the affirmative vote of the holders of a majority of the outstanding shares of Hypercom common stock entitled to vote at the special meeting. A proxy card of a record stockholder that is signed and returned that does not indicate a vote for or against a proposal will be counted as a vote FOR that proposal.

A broker or nominee who holds shares for customers, who are the beneficial owners of those shares, are prohibited from giving a proxy to vote those customers shares with respect to the proposals to be voted on at the special meeting of stockholders without instructions from the customer. Shares held by a broker or nominee that are not voted because the customer has not provided instructions to the broker or nominee will have the same effect as a vote against the approval and adoption of the merger agreement and approval of the merger.

Share Ownership of Hypercom Directors and Executive Officers

As of December 17, 2010, directors and executive officers of Hypercom and their affiliates held and were entitled to vote 1.47% of Hypercom common stock, including restricted stock and excluding stock options, outstanding on that date.

How to Vote

Submitting Proxies or Voting Instructions

Hypercom stockholders of record may vote their shares by attending the special meeting of stockholders and voting their shares in person at the special meeting of stockholders or by completing their proxy cards and

signing, dating and mailing them in the enclosed self-addressed envelopes. If a proxy card is signed by a stockholder of record of Hypercom and returned without voting instructions, the shares represented by the proxy will be voted FOR the proposal to approve and adopt the merger agreement and approve the merger and FOR the adjournment of the special meeting, if necessary, for any purpose, including to solicit additional proxies in favor of the approval and adoption of the merger agreement and approval of the merger and in the discretion of the proxy holders, on any other business that may properly come before the special meeting of stockholders or any adjournment of the special meeting.

Hypercom stockholders whose shares are held in the name of a broker or nominee must either direct the record holder of their shares as to how to vote their shares of Hypercom common stock or obtain a proxy from the record holder to vote at the special meeting of stockholders. Beneficial holders of Hypercom common stock should check the voting instruction cards used by their brokers or nominees to see if they may vote by using the telephone or the Internet.

Revoking Proxies or Voting Instructions

Hypercom stockholders of record may revoke their proxies at any time prior to the time their proxies are voted at the special meeting of stockholders. Proxies may be revoked by written notice to the corporate secretary of Hypercom, by a later-dated proxy signed and returned by mail, or by attending the special meeting of stockholders and voting in person.

Hypercom stockholders whose shares are held in the name of a broker or nominee may change their votes by submitting new voting instructions to their brokers or nominees. Those Hypercom stockholders may not vote their shares in person at the special meeting of stockholders unless they obtain a signed proxy from the record holder giving them the right to vote those shares.

Solicitation of Proxies

In addition to solicitation by mail, directors, officers and employees of Hypercom may solicit proxies for the special meeting from Hypercom stockholders personally or by telephone, facsimile and other electronic means without compensation other than reimbursement for their actual expenses. Also, Hypercom has retained Innisfree M&A Incorporated of 501 Madison Avenue, 20th Floor, New York, New York 10022, to assist in the solicitation of proxies for a fee of approximately \$30,000, including disbursements based on out-of-pocket expenses, telecommunicators, directory assistance and related telephone expenses.

Arrangements will also be made with brokerage firms and other custodians, nominees and fiduciaries for the forwarding of solicitation material to the beneficial owners of Hypercom common stock held of record by those persons, and Hypercom will, if requested, reimburse the record holders for their reasonable out-of-pocket expenses in so doing.

Please do not send in any Hypercom share certificates with your proxy cards or voting instruction cards.

To Attend the Special Meeting

Only Hypercom stockholders, including joint holders, who hold shares of record as of the close of business on , 2010 and other persons holding valid proxies for the special meeting of stockholders are entitled to attend the special meeting of stockholders. All stockholders and their proxies should be prepared to present photo identification. In addition, if you are a record holder, your name is subject to verification against the list of record holders on the record date prior to being admitted to the meeting. Hypercom stockholders who are not record holders but hold shares through a broker or nominee (i.e., in street name) should be prepared to provide

proof of beneficial ownership on the record date, such as a recent account statement prior to , 2010, or similar evidence of ownership. If you do not provide photo identification or comply with the other procedures outlined above upon request, you will not be admitted to the special meeting of stockholders.

Contact for Questions and Assistance in Voting

Any Hypercom stockholder who has a question about the merger or how to vote or revoke a proxy should contact:

Hypercom Corporation

8888 East Raintree Drive, Suite 300

Attention: Investor Relations

Scottsdale, Arizona, 85260

stsujita@hypercom.com

Telephone: (480) 642-5000

Any Hypercom stockholder who needs additional copies of this proxy statement/prospectus or voting materials should contact Investor Relations as described above.

Any Hypercom stockholder who has a question about the merger or how to vote or revoke a proxy or who needs additional copies of this proxy statement/prospectus or voting should contact Hypercom s information agent and proxy solicitor for the merger, at the following address and telephone numbers:

Innisfree M&A Incorporated 501 Madison Avenue, 20th Floor New York, New York 10022 (888) 750-5834 (toll-free) Banks & Brokers Call Collect:

(212) 750-5833

Other Matters

Hypercom is not aware of any other business to be acted upon at the special meeting. If, however, other matters are properly brought before the special meeting or any adjournment or postponement of the special meeting, the persons named as proxy holders will each have discretion to act on those matters, or to adjourn or postpone the special meeting.

The Companies

VeriFone

VeriFone is a global leader in secure electronic payment solutions. VeriFone provides expertise, solutions, and services that add value to the point of sale with merchant-operated, consumer-facing, and self-service payment systems for the financial, retail, hospitality, petroleum, transportation, government, and healthcare vertical markets. Since 1981, VeriFone has designed and marketed system solutions that facilitate the long-term shift toward electronic payment transactions and away from cash and checks. VeriFone s net revenues were \$1,001.5 million in the year ended October 31, 2010. VeriFone s principal executive offices are located at 2099 Gateway Place, Suite 600, San Jose, California 95110, and its telephone number is (408) 232-7800.

Additional information concerning VeriFone is included in VeriFone s reports filed under the Exchange Act that are incorporated by reference into this proxy statement/prospectus. See Where You Can Find More Information on page 117.

Hypercom

Hypercom Corporation is one of the largest global providers of complete electronic payment and transaction solutions and value-added services at the point of transaction. Hypercom s customers include domestic and international financial institutions, electronic payment processors, transaction network operators, retailers, system integrators, independent sales organizations and distributors. Hypercom also sells its products to companies in the hospitality, transportation, healthcare, prepaid card and restaurant industries. Hypercom products enable its customers to accept a wide range of payments and other transactions, including credit cards, signature and PIN (Personal Identification Number) based debit cards, contactless identification and near field communications, stored-value cards, and electronic benefits transfer. Hypercom delivers convenience and value to businesses that depend on reliable, secure, high-speed and high-volume electronic transactions. Hypercom s principal executive offices are located at 8888 East Raintree Drive, Suite 300, Scottsdale, Arizona 85260 and its telephone number is (480) 642-5000.

Additional information concerning Hypercom is included in Hypercom s reports filed under the Exchange Act that are incorporated by reference into this proxy statement/prospectus. See Where You Can Find More Information on page 117.

The Merger

General

If the merger agreement is adopted by the holders of a majority of all the issued and outstanding shares of Hypercom common stock at the special meeting, and the other conditions to closing are satisfied or waived, Honey Acquisition Co., a direct, wholly owned subsidiary of VeriFone, will be merged with and into Hypercom. Hypercom will be the surviving corporation and will become a direct, wholly owned subsidiary of VeriFone.

Merger Consideration

At the effective time and as a result of the merger, Hypercom stockholders will receive, for each share of common stock, par value \$0.001 per share, of Hypercom issued and outstanding immediately prior to the effective time of the merger 0.23 shares of VeriFone common stock.

Based on the number of shares outstanding as of the record date, VeriFone currently estimates that the total number of shares of VeriFone common stock to be issued at the closing of the merger will be outstanding as of the record date, VeriFone would issue an additional shares of VeriFone common stock in the merger.

Hypercom stockholders will not receive fractional shares of VeriFone common stock. Instead, they will receive the cash value, without interest, of any fractional share of VeriFone common stock that they might otherwise have been entitled to receive, determined by the per share closing price of VeriFone common stock as reported on the New York Stock Exchange composite transactions reporting system on the last trading day prior to the closing date of the merger.

Schedule of Important Dates

The following schedule shows important dates and events in connection with the special meeting.

Dates	Events
, 2010	Record date for special meeting
, 2011	Special meeting
The merger is expected to close in the second half of 2011. We cannot predict the exact timing because the merger remains subject to	
governmental and regulatory approvals and other conditions.	

Background of the Merger

VeriFone and Hypercom regularly consider a variety of strategic transactions as part of their ongoing evaluation of their respective businesses. VeriFone and Hypercom have been familiar with each other s businesses for several years and from time to time have discussed potential strategic alliances. Over the years, these contacts have been initiated by both parties.

During the second quarter of 2009, Hypercom s board of directors became concerned with Hypercom s cash balance and global economic and market conditions and their potential impact on Hypercom s operations. The board requested an evaluation of Hypercom s strategic alternatives and, based on this evaluation, elected to solicit interest from three companies that participate in the same or a similar sector, including VeriFone, to discuss whether a combination or other strategic alliance might be appropriate.

On June 10, 2009, VeriFone sent Hypercom a letter setting forth proposed terms for combining the two companies. The proposal contemplated that Hypercom stockholders would receive a 40% premium to the average closing price of Hypercom s stock price for each of the 30 consecutive trading days ending five trading days prior to the signing of a definitive transaction agreement, which at the time corresponded to a value of \$1.73 per share. The letter indicated that VeriFone was open to allowing the Hypercom stockholders to elect to receive the merger consideration in the form of stock, cash or a combination of both.

On June 12, 2009, Messrs. Daniel Diethelm, Johann Dreyer, Ian Marsh and Phillip Riese, the independent directors of the Hypercom board, held a meeting to evaluate VeriFone s proposal and compare it to the expected stockholder value that Hypercom could deliver as a stand-alone company, or as a company in combination with other participants in the market. The independent directors discussions focused on Hypercom s improving financial and operational results since initial cash concerns were raised and the resulting expectations for Hypercom s performance as a stand-alone company, the potential availability and value of combinations with other partners, and the potential value of the proposal by VeriFone. The independent directors instructed Hypercom s management to advise VeriFone that it would respond to its indication of interest in due course following further evaluation.

The full board of directors of Hypercom held a meeting on June 18, 2009. Also present at the meeting were UBS, which was engaged as Hypercom s financial advisor as of June 15, 2009, and Hypercom s legal advisor, DLA. The board discussed the proposal received from VeriFone, as well as non-binding proposals submitted by the two other industry participants, UBS s preliminary analysis of such proposals, and (with the assistance of counsel) their fiduciary responsibilities in light of the proposals. Taking into account UBS s preliminary analysis and a financial update by management, the board determined that VeriFone s proposal and the other proposals were inadequate and directed UBS to conduct further analysis, including regarding the synergies that could be achieved in a combination.

During this period, Hypercom and VeriFone negotiated the terms of a confidentiality agreement and, on June 25, 2009, executed an agreement which provided for the exchange of confidential information and the use of such information only for the purpose of evaluating a negotiated transaction. The confidentiality agreement also included limits on each party s ability to solicit certain key employees of the other party, and a standstill provision which prohibited VeriFone, without the consent of Hypercom s board of directors, from acquiring common stock of Hypercom; effecting a tender or exchange offer, merger or other business combination; or soliciting proxies or consents to vote any securities of Hypercom. The standstill and employee solicitation provisions expired June 25, 2010 and the remaining terms generally remain in effect until June 25, 2011 (with the merger agreement providing for a further extension of those remaining terms until June 25, 2012). Hypercom also entered into confidentiality agreements with the other two parties around this time period.

On June 30, 2009, the Hypercom board of directors held another meeting to review Hypercom s business plan for 2009 through 2011 and further discuss VeriFone s June 10, 2009 proposal with DLA and UBS, which had completed its analysis of synergies that could be achieved in a combination. Given the prospects for value creation through the successful execution of Hypercom s business plan, the board requested that UBS solicit significantly improved proposals from VeriFone and the other companies that had submitted proposals, for review at the board s next scheduled meeting on July 29, 2009.

No revised proposals were received prior to July 29, 2009. At its July 29, 2009 meeting, the Hypercom board of directors received and extensively discussed a presentation by UBS regarding the proposal received from VeriFone on June 10, 2009. The board also discussed how to proceed in discussions with VeriFone and unanimously determined that VeriFone s proposal so significantly undervalued Hypercom that Hypercom would not deliver a written response. Hypercom orally confirmed to VeriFone, and UBS orally confirmed to J.P. Morgan Securities LLC (J.P. Morgan), that the June 10, 2009 proposal was inadequate.

On August 2, 2009, August 7, 2009 and August 13, 2009, representatives of J.P. Morgan, financial advisor to VeriFone, and representatives of UBS, financial advisor to Hypercom, conducted discussions regarding VeriFone s June 10, 2009 proposal.

On August 17, 2009, VeriFone sent Hypercom a revised proposal reflecting a price of \$3.00 per share in a stock-for-stock transaction, with up to 50% of the purchase price paid in cash at the stockholders election. VeriFone offered to divest Hypercom s U.S. operations if necessary for antitrust approvals. VeriFone also requested a 40-day exclusivity period. The other companies did not submit revised proposals.

Following receipt of the August 17, 2009 letter and without Hypercom accepting any of the terms set forth in the August 17, 2009 letter, Hypercom and VeriFone undertook limited due diligence, primarily with respect to regulatory and antitrust matters and pending litigation. At various times throughout August and September 2009, representatives of J.P. Morgan and UBS held discussions in which UBS conveyed Hypercom s concern regarding the adequacy of the price set forth in VeriFone s August 17 letter. During this period, on August 25 and August 31, 2009, the Hypercom board of directors held meetings at which it discussed VeriFone s August 17, 2009 proposal and UBS s analysis of the proposal, valuation considerations and perspectives, and a pro forma analysis of the proposed combined company.

On September 23, 2009, VeriFone sent Hypercom a revised indication of interest pursuant to which it proposed to acquire the outstanding shares of Hypercom stock based on an all-stock transaction with an exchange ratio of 0.24 shares of VeriFone common stock per share of Hypercom common stock, which at that date corresponded to a value of \$3.76 per share of Hypercom common stock, representing a 39% premium to the 30-day average trading price of Hypercom common stock.

On September 30, 2009, representatives of J.P. Morgan and UBS discussed VeriFone s September 23, 2009 proposal.

At meetings held on October 6 and October 20, 2009, the Hypercom board of directors met with its financial and legal advisors to review the September 23, 2009 proposal from VeriFone. DLA discussed the board s fiduciary duties. Representatives of UBS reviewed the terms of the offer and financial analyses relating to the offer, which included a detailed review of VeriFone s historical and expected future performance. The board also discussed the possibility of accepting VeriFone s stock as currency, and the need to conduct diligence on VeriFone s future financial prospects. Following extensive discussion, the board directed Messrs. Diethelm and Tartavull to work with UBS to prepare a draft response to the September 23, 2009 proposal, which would include a proposal having both cash and stock components.

On October 21, 2009, Hypercom sent its response rejecting VeriFone s September 23, 2009 proposal and indicating that it would be willing to discuss an all-stock merger at an exchange ratio of 0.30, with the right of stockholders to elect to receive part of the consideration in cash, which at that date corresponded to a value of \$4.57 per share of Hypercom common stock, representing a 50% premium to the 30-day average trading price of Hypercom common stock.

On October 28, 2009, VeriFone sent Hypercom a further revised indication of interest pursuant to which it proposed a revised exchange ratio of 0.28 shares of VeriFone common stock per share of Hypercom common stock, provided that Francisco Partners II, L.P. reduce by 25%, or 2,636,000 shares, the shares subject to its warrant. In addition, VeriFone reserved the right to offer a portion of the consideration in cash to the extent necessary in order that VeriFone would not be required to seek stockholder approval of the transaction. At the time, the proposal corresponded to a value of \$3.91 per share of Hypercom common stock, representing a 27% premium to the 30-day average trading price of Hypercom common stock.

The Hypercom board of directors held meetings on October 28 and November 5, 2009 at which it discussed the revised proposal from VeriFone. Representatives of UBS presented an analysis of the revised proposal. The board considered the key economic terms of VeriFone s proposal, Hypercom s improving business and prospects, including an updated financial forecast, recent relative movements in the prices of Hypercom s stock and VeriFone s proposal to address antitrust concerns, VeriFone s proposal for a combination of stock and cash consideration, certain ongoing regulatory and litigation proceedings to which VeriFone was a party, and alternatives to the proposed transaction, including continuing as a stand-alone entity. The board did not consider other strategic buyers, as the two industry participants that had submitted proposals in June 2009, other than VeriFone, had declined to submit improved proposals. Following such discussions, the Hypercom board of directors concluded that VeriFone s revised proposal was unacceptable, Hypercom s October 21 counterproposal should be withdrawn, and that for the foreseeable future Hypercom should pursue its business strategy and discontinue discussions about a potential business combination with VeriFone. These conclusions were conveyed to VeriFone in a letter dated November 6, 2009.

On November 20, 2009, the Hypercom board of directors held a meeting at which it discussed a proposed acquisition of POS assets of another industry participant, the risks and challenges of the proposed transaction, due diligence matters, the potential acquisition price, and projected financial results. The board authorized management to submit a non-binding term sheet with respect to the transaction. There were several calls between Hypercom and the potential seller in February 2010, followed by meetings on March 19 and 20, 2010. The term sheet was rejected by the seller and a counterproposal was made by the seller at a significantly greater price than offered by Hypercom. On April 10, 2010, the parties determined not to pursue the transaction.

In May 2010, VeriFone re-initiated discussions with Hypercom and, from May 6 to May 12, 2010, Philippe Tartavull, Chief Executive Officer and President of Hypercom, and Douglas Bergeron, Chief Executive Officer of VeriFone, exchanged emails and had telephone communications in which the possibility of a business combination was discussed.

On May 13, 2010, Mr. Bergeron called Keith Geeslin, a member of the board of directors of Hypercom, to discuss a potential business combination between the two companies. VeriFone requested a meeting with the Hypercom board to continue discussions.

Following a meeting of the Hypercom board on May 17, 2010, in response to VeriFone s overture, Mr. Tartavull submitted a letter to Mr. Bergeron which noted that since the prior discussions ended in November 2009, Hypercom s operations and financial results had continued to improve and that, while Hypercom was not soliciting acquisition offers, it believed that a potential business combination between the two companies could provide value to Hypercom s stockholders, provided that VeriFone presented a credible proposal commensurate with Hypercom s improved results. Mr. Tartavull s letter indicated that the Hypercom board would evaluate any credible proposal from VeriFone, consistent with its board s fiduciary duties.

On June 14, 2010, Mr. Bergeron met with Messrs. Tartavull, Diethelm and Norman Stout to discuss a potential business combination between VeriFone and Hypercom.

On June 23, 2010, VeriFone sent Hypercom a proposal to acquire Hypercom in a merger with consideration fixed at \$6.00 per share in cash, with up to 50% of the purchase price paid in stock at Hypercom s stockholders election, subject to restrictions to eliminate the contingency of the approval of VeriFone s stockholders. The other provisions were substantially similar to proposals made in 2009.

On June 30, 2010, the Hypercom board of directors held a meeting at which it discussed VeriFone s revised proposal and recent meetings and telephone conversations between Hypercom directors and Mr. Bergeron. The board also discussed a presentation of UBS s preliminary analysis of the revised proposal. Following such discussion, the board authorized management to meet with VeriFone s management to discuss a potential business combination.

On July 1, 2010, Hypercom submitted a letter to VeriFone accepting VeriFone s offer to hold a meeting of senior executives of the companies to further discuss the potential merits of a business combination. Hypercom noted that it was disappointed with VeriFone s latest revised offer, however, and conveyed the board s belief that the value of Hypercom should be captured in a fixed exchange ratio, rather than a per share price. Hypercom suggested an exchange ratio of 0.32 to 0.34 (representing a value of from \$6.64 to \$7.05 per share of Hypercom common stock based on VeriFone s closing stock price on June 25, 2010), either all in VeriFone common stock or the maximum amount of VeriFone common stock that could be issued without VeriFone stockholder approval, with the remainder in cash. Hypercom also suggested an extension of the standstill provision in the confidentiality agreement (which expired on June 25, 2010) through November 9, 2010. The request for an extended standstill was rejected by VeriFone on July 15, 2010.

On July 22, 2010, Mr. Bergeron, Robert Dykes, Senior Vice President and Chief Financial Officer of VeriFone, and Albert Liu, Senior Vice President and General Counsel of VeriFone, met with Mr. Tartavull, Thomas Sabol, Chief Financial Officer of Hypercom, and Douglas Reich, Senior Vice President and General

Counsel of Hypercom, in Los Angeles to discuss further the merits of a business combination. Hypercom presented its forecast for calendar 2010 through 2012, and VeriFone presented its forecast for fiscal 2010 and 2011.

On July 28, 2010, the Hypercom board of directors held a meeting at which the board discussed communications with VeriFone officers since the June 30 board meeting. The board discussed the history of the exchanges with VeriFone over the previous year and noted that they had not resulted in an acceptable offer.

On August 3, 2010, Hypercom sent VeriFone a letter reiterating its previously suggested exchange ratio of 0.32 to 0.34 and its preference for an all-stock merger and indicating the desire of Hypercom s board of directors to bring negotiations to a prompt and successful conclusion or, alternatively, if a deal could not be reached, to disengage so the parties could focus on their respective businesses. In addition, Mr. Tartavull expressed concern over leaks to Hypercom employees and business partners of the ongoing discussions and the potential adverse impact on Hypercom s business.

On August 26, 2010, Mr. Bergeron met with Mr. Tartavull to further discuss the parameters of a potential business combination.

On August 26, 2010, VeriFone sent Hypercom a letter in which it proposed an all stock merger at a fixed exchange ratio equal to 0.19 of a VeriFone share per Hypercom share, representing a 43% premium to the trading price of Hypercom common stock at the time. The letter included a preliminary financial analysis of the combined companies. VeriFone also informed Hypercom that if the parties were not able to agree on terms, VeriFone intended to present an all-cash offer directly to Hypercom s stockholders in the next couple of weeks.

From August 30, 2010 to September 3, 2010, Messrs. Geeslin, Tartavull and Sabol held meetings and telephone conversations with Messrs. Bergeron and Dykes, and representatives of J.P. Morgan discussed VeriFone s August 26, 2010 proposal with representatives of UBS.

On September 1, 2010, the Hypercom board held a meeting at which it discussed the August 26, 2010 proposal from VeriFone. UBS and management discussed the financial analysis provided by VeriFone, management provided updated financial information, and UBS presented its financial analysis of the offer. The board also discussed the possible adoption of a stockholder rights plan and the legal standards regarding its response to VeriFone s proposal and adoption of a rights plan. Further, the board discussed the terms and conditions upon which it would consider engaging in diligence and further negotiations with VeriFone regarding a potential combination.

On September 7, 2010, Messrs. Bergeron, Dykes and Liu and Douglas Reed, Vice President and Treasurer of VeriFone, and representatives of J.P. Morgan met with Messrs. Tartavull, Sabol, Reich, Geeslin and Diethelm and representatives of UBS in San Francisco to have further discussions regarding a potential combination. At the meeting, Hypercom representatives expressed the view that the August 26, 2010 proposal was inadequate.

On September 10, 2010, Mr. Bergeron initiated a call with Mr. Geeslin to discuss VeriFone s August 26, 2010 proposal. Also on September 10, 2010, Hypercom submitted a letter to VeriFone reaffirming Hypercom s view that the revised proposal failed to provide reasonable value to Hypercom s stockholders. On the same day, VeriFone sent a response letter to Hypercom discussing VeriFone s justifications for its August 26, 2010 proposal.

On September 11, 2010, representatives of J.P. Morgan and UBS discussed VeriFone s August 26, 2010 proposal and Hypercom s September 10, 2010 response letter. On that same date, Mr. Geeslin received a call from Mr. Bergeron to discuss VeriFone s August 26 proposal.

Also on September 11, 2010, Mr. Dykes talked telephonically with Mr. Sabol regarding VeriFone s August 26, 2010 proposal.

On September 12, 2010, the board of Hypercom held a meeting to discuss the status of discussions among the various participants, including the discussions Mr. Geeslin and Mr. Sabol had with Mr. Bergeron and Mr. Dykes, and those among representatives of UBS and J.P. Morgan. Messrs. Tartavull and Diethelm reported on recent contacts by Hypercom stockholders (confirming no confidential information was provided). The board requested that Messrs. Stout and Riese contact certain directors of VeriFone to discuss the August 26 proposal and Hypercom s position with respect to it.

On September 14, 2010, Messrs. Dykes and Liu had a further call with Messrs. Sabol and Reich regarding the August 26, 2010 proposal, during which VeriFone again indicated its intention to go public with an all-cash offer to Hypercom s stockholders.

On September 17, 2010, Mr. Stout called Charles Rinehart, Chairman of the Board of VeriFone, to discuss VeriFone s proposal, the involvement of the respective boards of directors with respect to a potential transaction between the two companies, the merits of a potential combination from a business standpoint, and the positive and negative factors regarding a friendly rather than a hostile transaction, and to establish another line of communication between the two companies.

Also on September 17, 2010, the Hypercom board of directors held a meeting, at which representatives of UBS, DLA and Hypercom s public relations advisors were present, and at which the directors discussed the communications with VeriFone since the last meeting. The board further discussed VeriFone s August 26, 2010 proposal in light of Hypercom s expected third quarter 2010 results, updated financial forecast, and continued improved performance and the recent relative movement in the prices of Hypercom s and VeriFone s stock. The board unanimously concluded that the economic terms of the August 26, 2010 proposal were opportunistic and unacceptable under the circumstances. Following the meeting, with the authorization of the board, Mr. Riese called Alex (Pete) Hart, a member of VeriFone s board of directors, to discuss VeriFone s proposal and open an additional line of communication between the boards.

On September 20, 2010, VeriFone sent Hypercom a letter proposing to acquire all of the outstanding Hypercom shares for \$5.70 per share, either in cash or shares of VeriFone common stock. VeriFone informed Hypercom that if the proposal was not accepted by September 22, 2010, VeriFone s board of directors would approve an all-cash tender offer directly to Hypercom s stockholders on September 27, 2010.

On September 22, 2010, Mr. Dykes had a telephone call with Mr. Tartavull regarding VeriFone s latest proposal.

Also on September 22, 2010, the Hypercom board of directors held a meeting at which it discussed recent communications between Hypercom and VeriFone regarding VeriFone s September 20, 2010 offer and intent to make a public offer if the Hypercom board did not accept its terms. Management provided an overview of these discussions as well as a financial update. The board discussed the provisions of the confidentiality agreement and its fiduciary duties with Hypercom s legal advisors. UBS presented a financial analysis of the latest VeriFone proposal. Following such discussions and presentation, the board determined that the proposal failed to provide adequate consideration to Hypercom s stockholders and was unacceptable.

On September 23, 2010, Hypercom sent VeriFone a letter rejecting VeriFone s September 20, 2010 offer as inadequate. Hypercom stated that it would be willing to consider a bona fide offer representing an exchange ratio of 0.24.

Also on September 23, 2010, Hypercom filed a complaint under seal against VeriFone in the Court of Chancery of the State of Delaware along with a motion for a temporary restraining order and motion for a preliminary injunction, seeking to enjoin VeriFone from making public disclosures regarding its proposed offer in light of the parties confidentiality agreement.

On September 24, 2010, the Delaware Chancery Court held a teleconference with counsel for the parties during which the Court scheduled a hearing for October 14, 2010. VeriFone voluntarily agreed to refrain from making public disclosures until the hearing; however, the Court left open the possibility that VeriFone could make an application to determine whether it could publicly announce a new offer to the Hypercom board of directors without violating the restrictions in the confidentiality agreement.

On September 24, 2010, VeriFone sent Hypercom a letter proposing a best and final offer of an exchange ratio of 0.21, in stock or the equivalent value in cash, which based on the closing price as of September 23, 2010 corresponded to a value of \$5.82 per share of Hypercom common stock, a 52% premium over Hypercom s closing share price on September 23, 2010 and a 69% premium over Hypercom s average closing share price for the past 30 trading days prior to the proposal. VeriFone informed Hypercom that if the offer were not accepted, VeriFone would move the Delaware Chancery Court to make public a different proposal on September 27, 2010, which proposal would be an all-cash offer at a materially reduced per share price to appropriately reflect the additional costs associated with a hostile bid.

On September 26, 2010, Mr. Tartavull polled each Hypercom director, and then confirmed with VeriFone that the views set forth in Hypercom s September 23, 2010 letter remained unchanged. Hypercom followed up with a letter to VeriFone rejecting its revised offer as inadequate. In addition, Hypercom affirmed its belief that a friendly transaction was in the best interests of all stakeholders.

On September 27, 2010, VeriFone sent a letter to the Hypercom board of directors proposing an all-cash offer to Hypercom of \$5.25, and also indicated that because VeriFone had not received a positive response to its September 24 proposal, it intended to make public the new offer letter.

On September 28, 2010, the Delaware Chancery Court granted VeriFone s request to make the September 27, 2010 letter public. Also on September 28, 2010, Mr. Dykes and Mr. Tartavull discussed telephonically VeriFone s most recent proposal.

On September 28, 2010, the Hypercom board approved the appointment of Thomas Ludwig as a Class I Director, to serve until the 2012 annual meeting of Hypercom s stockholders. The appointment was made pursuant the contractual right of Francisco Partners II, L.P. to maintain two designees on Hypercom s board of directors. Mr. Ludwig was added to the class of directors up for election in 2012.

On September 29, 2010, the Hypercom board held a meeting at which it discussed VeriFone s recent proposals. In addition, the board revisited the adoption of a stockholders rights plan, in conjunction with reports of legal counsel and UBS. Following such discussion, the board approved a stockholder rights plan in response to VeriFone s public offer.

On September 29, 2010, VeriFone publicized its September 27, 2010 letter to Hypercom.

On September 30, 2010, Hypercom issued a press release announcing that its board had received and unanimously rejected VeriFone s non-binding proposal to acquire Hypercom for \$5.25 per share in an all-cash deal, because the proposal significantly undervalued Hypercom and its future prospects and was not in the best interests of stockholders. Hypercom also announced that it expected to significantly exceed Wall Street securities analysts consensus estimate of \$112 million in net revenues for the third quarter of 2010, that the board had adopted the stockholder rights plan, and that Mr. Ludwig, a representative of Francisco Partners II, L.P., had been appointed as a director.

On October 1, 2010, a purported holder of common stock of Hypercom filed a putative class action and derivative lawsuit in the Delaware Court of Chancery alleging, among other things, that the Hypercom directors had breached their fiduciary duties in failing to fairly consider the offer from VeriFone.

On October 2, 2010, representatives from J.P. Morgan contacted representatives from UBS to determine whether in light of the recent announcements by both parties, there might be an opportunity to reach a mutually agreeable transaction.

On October 4, 2010, in light of the recently filed stockholder lawsuit as well as other statements and inquiries following VeriFone s public announcement of its September 27 letter to the Hypercom board, representatives from Morris, Nichols, Arsht, & Tunnell LLP, Delaware counsel for Hypercom, contacted representatives from Richards, Layton & Finger, Delaware counsel for VeriFone, indicating that Hypercom was willing to agree that public disclosures could be made for purposes of VeriFone s publicly announced offer and possible public tender offer under the parties confidentiality agreement without either party admitting or denying that such restrictions would have been applicable to any proposal made by VeriFone. Thereafter, pursuant to a waiver agreement dated October 7, 2010, the parties agreed to waive certain confidentiality and use restrictions contained in the confidentiality agreement, to the extent, if at all, they would apply, for purposes of any proposal by VeriFone to purchase common stock of Hypercom. On October 7, 2010, Hypercom filed a voluntary notice of dismissal dismissing its Delaware lawsuit against VeriFone.

Between October 8, 2010 and October 18, 2010, representatives of J.P. Morgan and UBS conducted further discussions regarding a potential transaction.

On October 12, 2010, the Hypercom board held a meeting at which it discussed an updated report regarding Hypercom s expected third quarter results and fourth quarter forecast. The board also discussed a presentation by UBS of its recent communications with J.P. Morgan regarding a possible revised proposal from VeriFone.

On October 14, 2010, the Compensation Committee of Hypercom s board of directors adopted a Change of Control Plan to incentivize employees to remain with Hypercom despite the publicly announced desire of VeriFone to acquire Hypercom, the willingness of VeriFone to sell off Hypercom s U.S. operations if needed to address potential competitive concerns, and concerns that in light of the ongoing events, key employees may be solicited by competitors.

On October 18, 2010, Mr. Bergeron called Mr. Diethelm to express his disappointment with the inability of the two companies to come to a friendly agreement in a timely manner. Mr. Bergeron informed Mr. Diethelm that VeriFone was prepared to move forward with a tender offer at \$6.00 if the parties could not come to agreement quickly. Mr. Bergeron communicated the same message to Mr. Tartavull. On October 18, 2010 and October 19, 2010, Mr. Bergeron and Mr. Tartavull corresponded regarding the work that would need to be done to complete an agreement, including due diligence and discussions regarding required antitrust approvals.

Representatives of J.P. Morgan and UBS continued to have conversations from October 19 through October 22, 2010. On October 22, 2010, Mr. Dykes and Mr. Sabol discussed diligence matters. Also on October 22, 2010, representatives of Sullivan & Cromwell LLP, counsel to VeriFone, sent a draft merger agreement to a representative of DLA, counsel to Hypercom.

Between October 23, 2010 and October 29, 2010, representatives of J.P. Morgan and UBS held further discussions to determine whether the parties could reach agreement on the economic terms of a transaction. Representatives of UBS discussed whether a transaction based on an exchange ratio of 0.24 with a cash component would be acceptable to VeriFone. VeriFone offered an exchange ratio of .207 plus a cash component equal to 0.033 per share of VeriFone stock, but VeriFone later withdrew this proposal.

On October 27, 2010, the Hypercom board of directors held a meeting at which it discussed an update of Hypercom s expected third quarter results and fourth quarter forecast. The board also discussed a presentation by UBS regarding its recent communications with J.P. Morgan and possible strategies for negotiating improved economic terms of the proposed transaction with VeriFone. Mr. Tartavull reported on a contact by a large private equity firm and Mr. Riese reported on a contact by the purported representative of another industry participant, regarding their potential interest in a transaction with Hypercom. The board authorized Mr. Tartavull to conduct discussions with these parties. The board also discussed the draft merger agreement received from Sullivan & Cromwell LLP.

Between October 29, 2010 and November 1, 2010, representatives of VeriFone, Hypercom, Sullivan & Cromwell LLP and DLA exchanged drafts of the merger agreement, began negotiation of the terms of the merger agreement, and conducted financial, legal and business due diligence on Hypercom and VeriFone.

On October 30, 2010, representatives of Hypercom and UBS met with representatives of VeriFone and J.P. Morgan at VeriFone s offices in San Jose to conduct financial diligence on Hypercom and VeriFone.

On October 30 and October 31, 2010, Mr. Stout had discussions with Mr. Bergeron regarding VeriFone s proposal, the valuation of Hypercom, the merits of a potential combination from a business standpoint, and the positive and negative factors regarding a friendly rather than a hostile transaction and the timing of a potential transaction.

On November 1, 2010, Messrs. Bergeron, Dykes, Liu, Tartavull and Stout conducted a telephone conversation regarding the potential economic terms of the transaction. Hypercom and VeriFone discussed a range of valuation between \$7.00 and \$7.50 per Hypercom share in stock or stock and cash. Representatives of Hypercom expressed a desire for deal certainty in regard to antitrust issues. VeriFone proposed an antitrust termination fee of 3%, and also agreed to undertake certain divestitures in the event required to obtain regulatory approvals. On November 4, 2010, Messrs. Bergeron, Dykes and Liu had further discussions with Messrs. Tartavull, Sabol, Stout and Reich in which Hypercom proposed an antitrust termination fee equal to \$50 million.

Between November 1, 2010 and November 11, 2010, representatives of VeriFone, Hypercom, Sullivan & Cromwell LLP and DLA exchanged additional drafts of the merger agreement and held telephone calls to negotiate the terms of the merger agreement. Key issues included the ability of Hypercom to accept a superior proposal upon payment of a termination fee, the size of which was the subject of ongoing discussion, with VeriFone requesting 4% and Hypercom proposing 3%; antitrust matters, including Hypercom s request of a significant termination fee if the parties were unable to close due to regulatory barriers, as well as its request that VeriFone be required to dispose of certain business operations (of either party) in order to obtain regulatory approvals; provisions regarding certain communications by the parties between signing and closing; and provisions relating to VeriFone s required post-closing support of Hypercom products and services. The parties also conducted additional financial, legal and business due diligence on one another.

On November 3, 2010, the Hypercom board of directors held a telephonic meeting during which it discussed an updated report regarding Hypercom s fourth quarter forecast. The board also discussed the terms and conditions of the draft merger agreement with VeriFone, including the proposed consideration, provisions dealing with anticipated antitrust matters, and the parties operation of their respective businesses pending closing. Mr. Tartavull updated the board on his discussions with the representative of another industry participant.

On November 7, 2010, the Hypercom board of directors held a telephonic meeting during which it discussed the terms and conditions of the draft merger agreement with VeriFone, including proposed provisions dealing with the resolution of anticipated antitrust issues and the allocation of the attendant risks.

On November 10, 2010, the Hypercom board of directors held a telephonic meeting during which it discussed a report by UBS regarding its recent communications with J.P. Morgan about the proposed terms of the proposed merger transaction with VeriFone. The board also discussed the antitrust issues involved in a potential merger with VeriFone, as well as methods to allocate the antitrust risk between the parties. The board also discussed the consideration proposed to be received by Hypercom stockholders in the merger transaction.

On November 11, 2010, representatives of VeriFone, Hypercom, Sullivan & Cromwell LLP and DLA had further discussions regarding the size of the VeriFone antitrust-related reverse termination fee. During those discussions, VeriFone agreed to a 7% termination fee, which would increase to 7.5% if VeriFone exercised its right to extend the drop-dead date in the merger agreement from August 31, 2011 to November 30, 2011, and the

obligation of VeriFone to dispose of certain operations if necessary to secure regulatory approvals. Between November 11, 2010 and November 14, 2010, representatives of VeriFone, Hypercom, Sullivan & Cromwell LLP and DLA continued to negotiate the terms of the merger agreement and conduct financial, legal and business due diligence.

During the first two weeks of November 2010, Hypercom met separately with representatives of the potential strategic partner and large private equity firm that had approached Messrs. Tartavull and Riese in October 2010. Neither of these discussions led to a proposal for a transaction. Except for these inquiries, from VeriFone s publication on September 29, 2010 of its offer of \$5.25 in cash, until the execution of the merger agreement on November 17, 2010, no other potential buyer, including industry participants that had previously expressed interest in 2009, contacted Hypercom or UBS regarding a potential bid.

On November 13, 2010, Messrs. Bergeron, Dykes and Liu had further discussions with Messrs. Tartavull, Sabol, Stout and Reich regarding the consideration for the transaction. Mr. Tartavull indicated that Hypercom would be willing to accept an exchange ratio equal to 0.24, and Mr. Bergeron responded that VeriFone would not agree to such a ratio. On November 14, 2010, Messrs. Bergeron, Dykes and Liu had further discussions with Messrs. Tartavull, Sabol, Stout and Reich, with Mr. Bergeron proposing that the parties discuss an exchange ratio between 0.225 and 0.23, with a portion of the consideration to be paid in cash.

On November 15, 2010, Messrs. Bergeron, Dykes and Liu had further discussions with Messrs. Tartavull, Stout, Sabol and Reich regarding the terms of a potential transaction, including the purchase price. Representatives of VeriFone, Hypercom, Sullivan & Cromwell LLP and DLA continued to negotiate the terms of the merger agreement and undertake financial, legal and business due diligence.

On November 15, 2010, the Hypercom board of directors held a special telephonic meeting to discuss the status of the negotiations with VeriFone and the open issues in the draft merger agreement. DLA provided an updated summary of the merger agreement, and discussed open issues and the board s fiduciary duties, UBS presented analysis of the relative merits of the proposed economic terms of the merger, and the board received reports from its public relations and proxy advisors. The board discussed in extensive detail the factors weighing in favor of and against the proposed merger. No specific resolutions were passed with respect to the merger at that time.

Also on such date, the Compensation Committee of the Hypercom board met to discuss Hypercom s Change of Control Plan and other severance/change in control arrangements, and determined to make modifications to the Change of Control Plan designed to assure its effectiveness in the VeriFone transaction and to incentivize key employees to stay with Hypercom during any transition period or beyond.

On November 16, 2010, Messrs. Bergeron and Tartavull had further discussions regarding a proposed purchase price and exchange ratio for purposes of the merger agreement. After further discussions, including whether a portion of the merger consideration would be paid in cash as proposed by Mr. Bergeron, Messrs. Bergeron and Tartavull agreed on an exchange ratio of 0.23, payable in all stock, which represented a valuation of \$7.32 per Hypercom share based on VeriFone s November 16, 2010 closing price of \$31.84 per share. Also on November 16, 2010, Messrs. Bergeron, Tartavull and Stout further discussed the size of Hypercom s termination fee. In a discussion the same day between Messrs. Dykes and Tartavull, VeriFone ultimately agreed to a termination fee of 3%, which, in the view of Hypercom s board, was consistent with fees payable in comparable transactions and not likely to preclude another party from making a competing superior proposal. On November 16, 2010, representatives of VeriFone, Hypercom, Sullivan & Cromwell LLP and DLA continued to negotiate the terms of the merger agreement, as well as conclude each party s financial, legal and business due diligence.

On November 16, 2010, the VeriFone board of directors met to formally consider the transaction, at which meeting members of VeriFone s management and its legal and financial advisors made various presentations about, and the VeriFone board of directors discussed, the potential merger. Representatives from Sullivan &

Cromwell LLP reviewed the board s fiduciary duties in connection with the consideration of the proposed transaction and presented summaries of the terms of the merger agreement. J.P. Morgan reviewed the financial terms of the proposed merger and delivered advice relating to financial matters involving the transaction. Following deliberation, the board of directors of VeriFone unanimously approved the merger agreement.

On November 16, 2010, the Hypercom board of directors held a meeting to formally consider the merger with VeriFone and the terms and conditions of the merger agreement. Management presented an overview of the transaction and expressed its support for it. Representatives from DLA reviewed the board s fiduciary duties in connection with the proposed transaction, updated the status of open issues, and, with Ian Marsh, the Chair of the Compensation Committee, discussed revisions to Hypercom s severance/Change of Control Plan to ensure effectiveness in connection with the proposed transaction. Representatives from UBS made a presentation and delivered its oral opinion that, as of November 16, 2010 and based upon and subject to various assumptions made, matters considered and limitations described in its opinion, the exchange ratio provided for in the merger was fair, from a financial point of view, to the holders of Hypercom s common stock (other than FP Hypercom Holdco, LLC and its affiliates, as to which no opinion was rendered). The board of directors asked clarifying questions of representatives of DLA and UBS and the representatives responded to such questions. The board of directors considered the terms and conditions of the merger agreement, the merger and the other transactions contemplated by the merger agreement. After discussing the terms of the proposed transactions, including the personal interests of certain of Hypercom s officers and directors in the proposed transactions, the directors in attendance unanimously approved the merger agreement (with Mr. Tartavull abstaining from compensation matters), the merger and the other transactions contemplated by the merger agreement to Hypercom s stockholders for approval at the special meeting. Johan Dreyer, who was unable to attend the meeting, later ratified all actions taken and resolutions adopted.

On November 17, 2010, representatives of VeriFone and Hypercom executed the merger agreement on the same terms as presented to each party s board of directors, and the relevant parties executed the support agreement contemplated by the merger agreement. VeriFone and Hypercom issued a joint press release announcing that they had entered into the merger agreement.

VeriFone s Reasons for the Merger

VeriFone believes that there are several potential benefits of the merger that will contribute to the success of the combined company, including:

enhancing VeriFone s ability to expand its business in continental Europe, an area where it historically has had lower market penetration as compared to many other parts of the world;

enabling VeriFone to utilize its distribution channels, international presence, customer base, and brand recognition to accelerate Hypercom s market penetration and growth;

the potential to realize operating synergies from overlapping sales regions and efficiencies in administrative and engineering costs;

the potential to enhance the combined companies extensive research and development efforts and expertise in payment systems to offer more innovative products and services to customers and distributors;

the potential to deliver better quality and more widely available services to customers and distributors as a result of expanded reach; and

the potential to realize supply chain efficiencies as a result of increases in joint product volumes.

Hypercom s Reasons for the Merger; Recommendation of the Hypercom Board of Directors

Hypercom s Reasons for the Merger

In the course of reaching its decision to approve the merger and the merger agreement and the transactions contemplated thereby, the Hypercom board of directors consulted Hypercom s legal and financial advisors and considered a number of factors that it believed supports its decision, including the following:

the opinion of UBS that the exchange ratio provided for in the merger was, as of the date of the opinion and based upon and subject to factors and assumptions set forth herein, fair from a financial point of view to Hypercom s common stockholders (other than FP Hypercom Holdco, LLC and its affiliates);

the premium to Hypercom s average closing prices of 105.0%, 87.5% and 75.5%, respectively, for the 30, 60 and 90 trading days prior to VeriFone s first public offer of \$5.25 made on September 29, 2010;

the 39% premium that the merger consideration payable to the Hypercom common stockholders represents over VeriFone s first public offer of \$5.25;

the fact that the merger consideration will be paid in stock, giving common stockholders an opportunity to participate in any future earnings or growth of the combined company and future appreciation in the value of the VeriFone shares following the merger;

the benefits to the combined company that could result from the merger, including an enhanced competitive and financial position, increased diversity in its product line, additional scale in specific geographic areas, and potential cost and other synergies;

the likelihood that the merger will be consummated, in light of VeriFone s commitment to sell or otherwise dispose of businesses or assets that, subject to certain exceptions, in the aggregate, produced aggregate gross revenues in an amount not in excess of \$124 million for VeriFone or Hypercom during the 2009 fiscal year, if such action would be reasonably necessary for the closing conditions in the merger agreement to be met;

the terms and conditions of the merger agreement, including:

the scope of the representations, warranties and covenants being made by VeriFone;

the conditions to the consummation of the merger, including the requirement that the merger agreement be approved by Hypercom s stockholders;

the ability of the Hypercom board of directors, if the failure to do so would be inconsistent with its fiduciary duty, to provide information to and engage in negotiations with another party in connection with an unsolicited, bona fide written proposal that the Hypercom board determines in good faith and after consultation with outside legal counsel is reasonably likely to lead to a superior proposal and, subject to paying a \$12.2 million termination fee to VeriFone, accept a superior proposal;

Edgar Filing: VERIFONE SYSTEMS, INC. - Form S-4

the Board s belief that the termination fee payable to VeriFone is consistent with fees payable in comparable transactions and would not be likely to preclude another party from making a competing proposal;

the fact that the merger is intended to qualify as a reorganization within the meaning of Section 368(a) of the Internal Revenue Code for U.S. federal income tax purposes;

the fact that the merger agreement provides for specific enforcement, and that Hypercom may pursue damages reflecting the economic benefits of the transaction to its stockholders if VeriFone fails to close in breach of the merger agreement;

the fact that VeriFone will be obligated to pay Hypercom a termination fee of \$28.4 million (or \$30.4 million if VeriFone elects to extend the termination date to November 30, 2011 and such termination occurs after August 31, 2011), if, subject to certain exceptions, the merger agreement is terminated based on (i) the entry of any order relating to antitrust or competition law, or (ii) the

failure to consummate the merger by the termination date, and at the time of such termination all the closing conditions (other than regulatory conditions and the accuracy of VeriFone s representations and warranties, and VeriFone s performance of its obligations under the merger agreement) have been satisfied or waived; and

the fact that the merger agreement provides that VeriFone must (i) through closing or for six months following termination of the merger agreement, not solicit certain of Hypercom s employees, (ii) through closing or termination of the merger agreement, not announce any discontinuance of any product line of Hypercom (or pricing changes), and (iii) after closing, continue to support Hypercom s hardware and software for fixed periods, with the breach of (i) and (ii) giving to Hypercom the right to \$1 million in liquidated damages per occurrence, in addition to other remedies Hypercom may have under the merger agreement; and

the business, financial, market and execution risks associated with remaining independent and the challenges of successfully executing Hypercom s current business strategies, with the resulting uncertainty as to the future value of Hypercom s common stock. The Hypercom board of directors also considered a number of potentially countervailing factors in its deliberations concerning the merger, including the following:

the fact that, because the merger consideration is a fixed exchange ratio of shares of VeriFone common stock to Hypercom common stock, Hypercom s common stockholders could be adversely affected by a decrease in the trading price of VeriFone common stock during the pendency of the merger (that is, there is no price-based termination right or other similar protection);

the risk that the potential benefits and synergies sought in the merger will not be realized or will not be realized within the expected time period, and other risks generally associated with the integration by VeriFone of Hypercom;

the possibility that the merger might not be completed as a result of, among other matters, the failure of Hypercom s stockholders to adopt the merger agreement or failure to secure regulatory approvals or consents, and the effect the termination of the transaction may have on the trading price of Hypercom s common stock, its business, operating results and prospects, which effect is likely to be exacerbated the longer the time period between the signing and any termination of the merger agreement;

the possible effects of the pendency (or termination) of the merger agreement on Hypercom s business, operating results, prospects, employees, customers, distributors and suppliers, which effects are likely to be exacerbated the longer the time period between the signing and any termination of the merger agreement;

that, under the terms of the merger agreement, Hypercom cannot solicit other acquisition proposals and must pay VeriFone a termination fee of \$12.2 million in cash if the merger agreement is terminated under certain circumstances, including if the Hypercom board exercises its right to terminate the merger agreement and enter into a superior transaction, which may deter others from proposing an alternative transaction that may be more advantageous to Hypercom s stockholders;

the fact that upon the closing of the merger, stockholders will be required to surrender their shares involuntarily in exchange for a price determined by the Hypercom board and approved by a majority of the stockholders;

the fact that if VeriFone fails to complete the merger as a result of a breach of the merger agreement, it may be expensive and difficult to pursue rights and remedies, the success of any action would be uncertain, and, depending upon the reason for not closing, Hypercom s remedies may be limited to the termination fee payable by VeriFone, which is likely to be inadequate to compensate Hypercom for the damage caused;

Edgar Filing: VERIFONE SYSTEMS, INC. - Form S-4

the fact that Hypercom is subject to various remedies should it fail to complete the merger or breach the merger agreement; and

that the restrictions imposed by the merger agreement on the conduct of Hypercom s business prior to completion of the merger, requiring Hypercom to conduct its business only in the ordinary course and imposing additional specific restrictions, may delay, limit or prevent Hypercom from undertaking business opportunities that may arise during that period. *Recommendation of Hypercom s Board of Directors*

The Hypercom board of directors concluded that, on balance, the potential benefits to Hypercom and its stockholders of the transactions contemplated by the merger agreement outweighed the potential disadvantages and risks associated with those transactions. The foregoing discussion of the information and factors considered by the Hypercom board is not intended to be exhaustive. In view of the variety of factors considered in connection with its evaluations, the Hypercom board did not find it practicable to, and did not quantify or otherwise assign relative weight to, the specific factors considered in reaching its determination. Instead, the Hypercom board conducted an overall analysis of the factors described above, including summaries of discussions of Hypercom s management with Hypercom s legal, financial, accounting, tax, and other advisors and made its determination based on the totality of the information provided. In considering the factors described above, individual directors may have given different weights to different factors. Hypercom s board of directors believes that the merger is advisable, and is fair and in the best interests of Hypercom and its stockholders and unanimously recommends that Hypercom s stockholders vote FOR the proposal to approve and adopt the merger agreement and approve the merger.

Opinion of Hypercom s Financial Advisor

On November 16, 2010, at a meeting of Hypercom s board of directors held to evaluate the proposed merger, UBS delivered to Hypercom s board of directors an oral opinion, which opinion was confirmed by delivery of a written opinion, dated November 16, 2010 to the effect that, as of that date and based on and subject to various assumptions, matters considered and limitations described in its opinion, the exchange ratio provided for in the merger was fair, from a financial point of view, to holders of Hypercom common stock (other than FP Hypercom Holdco LLC and its affiliates, as to which UBS expressed no opinion). The opinion addressed the exchange ratio on a per share basis, without taking into consideration any individual holder s percentage ownership of Hypercom common stock.

The following summary of UBS s opinion is qualified in its entirety by reference to the full text of UBS s opinion. The full text of UBS s opinion describes the assumptions made, procedures followed, matters considered and limitations on the review undertaken by UBS. This opinion is attached as Annex C and is incorporated into this proxy statement/prospectus by reference. Holders of Hypercom common stock are encouraged to read UBS s opinion carefully in its entirety. UBS s opinion was provided for the benefit of Hypercom s board of directors in connection with, and for the purpose of, its evaluation of the exchange ratio provided for in the merger and did not address any other aspect of the merger. The opinion does not address the relative merits of the merger as compared to other business strategies or transactions that might be available with respect to Hypercom or Hypercom s underlying business decision to effect the merger. The opinion does not constitute a recommendation to any stockholder as to how to vote or act with respect to the merger.

In arriving at its opinion, UBS, among other things:

reviewed certain publicly available business and financial information relating to Hypercom and VeriFone;

reviewed certain internal financial information and other data relating to Hypercom s business and financial prospects that were not publicly available, including financial forecasts and estimates prepared by Hypercom s management that Hypercom s board of directors directed UBS to utilize for purposes of its analysis;

reviewed certain internal financial information and other data relating to VeriFone s business and financial prospects that were not publicly available, including financial forecasts and estimates prepared by VeriFone s management, as modified by Hypercom s management, that Hypercom s board of directors directed UBS to utilize for purposes of its analysis;

reviewed certain estimates of synergies prepared by Hypercom s management that were not publicly available that Hypercom s board of directors directed UBS to utilize for purposes of its analysis;

conducted discussions with members of the senior managements of Hypercom and VeriFone concerning the business and financial prospects of Hypercom and VeriFone;

reviewed publicly available financial and stock market data with respect to certain other companies UBS believed to be generally relevant;

reviewed current and historical market prices of Hypercom common stock and VeriFone common stock;

reviewed a draft, dated November 13, 2010, of the merger agreement; and

conducted such other financial studies, analyses and investigations, and considered such other information, as UBS deemed necessary or appropriate.

In connection with its review, with the consent of Hypercom s board of directors, UBS assumed and relied upon, without independent verification, the accuracy and completeness in all material respects of the information provided to or reviewed by UBS for the purpose of its opinion. In addition, with the consent of Hypercom s board of directors, UBS did not make any independent evaluation or appraisal of any of the assets or liabilities (contingent or otherwise) of Hypercom or VeriFone, and was not furnished with any such evaluation or appraisal. With respect to the financial forecasts, estimates and synergies referred to above, UBS assumed, at the direction of Hypercom s board of directors, that such forecasts, estimates and synergies had been reasonably prepared on a basis reflecting the best currently available estimates and judgments of Hypercom s management as to the future financial performance of Hypercom, VeriFone and such synergies, would be achieved at the times and in the amounts projected. UBS also assumed, with the consent of Hypercom s board of directors, that the merger would qualify for U.S. federal income tax purposes as a reorganization within the meaning of Section 368(a) of the Internal Revenue Code of 1986, as amended. UBS s opinion was necessarily based on economic, monetary, market and other conditions as in effect on, and the information available to UBS as of, the date of its opinion.

In addition, at the direction of Hypercom s board of directors, UBS was not asked to, and it did not, offer any opinion as to the terms, other than the exchange ratio to the extent expressly specified in UBS s opinion, of the merger agreement or the form of the merger. In addition, UBS expressed no opinion as to the fairness of the amount or nature of any compensation to be received by any officers, directors or employees of any parties to the merger, or any class of such persons, relative to the exchange ratio. UBS expressed no opinion as to what the value of VeriFone common stock would be when issued pursuant to the merger or the price at which VeriFone common stock or Hypercom common stock would trade at any time. In rendering its opinion, UBS assumed, with the consent of Hypercom s board of directors, that (i) the final executed form of the merger agreement would not differ in any material respect from the draft that UBS reviewed, (ii) the parties to the merger agreement without any adverse waiver or amendment of any material term or condition of the merger agreement. UBS also assumed that all governmental, regulatory or other consents and approvals necessary for the consummation of the merger would be obtained without any material adverse effect on Hypercom, VeriFone or the merger. Except as described above, Hypercom imposed no other instructions or limitations on UBS with respect to the investigations made or the procedures followed by UBS in rendering its opinion. The issuance of UBS s opinion was approved by an authorized committee of UBS.

In connection with rendering its opinion to Hypercom s board of directors, UBS performed a variety of financial and comparative analyses which are summarized below. The following summary is not a complete description of all analyses performed and factors considered by UBS in connection with its opinion. The preparation of a financial opinion is a complex process involving subjective judgments and is not necessarily susceptible to partial analysis or summary description. With respect to the selected companies analysis, no company used as a comparison was identical to Hypercom. These analyses necessarily involve complex considerations and judgments concerning financial and operating characteristics and other factors that could affect the public trading values of the companies concerned.

UBS believes that its analyses and the summary below must be considered as a whole and that selecting portions of its analyses and factors or focusing on information presented in tabular format, without considering all analyses and factors or the narrative description of the analyses, could create a misleading or incomplete view of the processes underlying UBS s analyses and opinion. UBS did not draw, in isolation, conclusions from or with regard to any one factor or method of analysis for purposes of its opinion, but rather arrived at its ultimate opinion based on the results of all analyses undertaken by it and assessed as a whole.

The estimates of the future performance of Hypercom and VeriFone provided by Hypercom and VeriFone in or underlying UBS s analyses are not necessarily indicative of actual future results or values, which may be significantly more or less favorable than those estimates. In performing its analyses, UBS considered industry performance, general business and economic conditions and other matters, many of which were beyond the control of Hypercom and VeriFone. Estimates of the financial value of companies do not purport to be appraisals or necessarily reflect the prices at which businesses or securities actually may be sold or acquired.

The exchange ratio was determined through negotiation between Hypercom and VeriFone and the decision by Hypercom to enter into the merger was solely that of Hypercom s board of directors. UBS s opinion and financial analyses were only one of many factors considered by Hypercom s board of directors in its evaluation of the merger and should not be viewed as determinative of the views of Hypercom s board of directors or management with respect to the merger or the exchange ratio.

The following is a brief summary of the material financial analyses performed by UBS and reviewed with Hypercom s board of directors on November 16, 2010, in connection with its opinion relating to the proposed merger. The financial analyses summarized below include information presented in tabular format. In order for UBS s financial analyses to be fully understood, the tables must be read together with the text of each summary. The tables alone do not constitute a complete description of the financial analyses. Considering the data below without considering the full narrative description of the financial analyses, including the methodologies and assumptions underlying the analyses, could create a misleading or incomplete view of UBS s financial analyses. For purposes of the financial analyses described below, the term implied per share value of the merger consideration refers to the \$7.32 implied per share value of the merger consideration based on the merger exchange ratio of 0.23 and the closing price of VeriFone common stock on November 16, 2010 of \$31.84.

Selected Public Companies Analysis

UBS compared selected financial and stock market data of Hypercom with corresponding data of the following two publicly traded point-of-sale terminal companies:

Ingenico

VeriFone Systems, Inc.

UBS reviewed, among other things, the enterprise values of the selected companies, calculated as diluted equity market value based on closing share prices on November 16, 2010, plus face value of total debt and book value of minority interests, less cash and cash equivalents, as multiples of estimated revenue and estimated

earnings before interest, taxes, depreciation and amortization (EBITDA) for calendar years 2010 and 2011. UBS also reviewed closing share prices of the selected companies on November 16, 2010 as a multiple of calendar years 2010 and 2011 estimated earnings per share. UBS then compared these multiples derived for the selected companies with corresponding multiples implied for Hypercom based on the closing price of Hypercom common stock on the New York Stock Exchange on September 29, 2010, the last trading day prior to the initial publicly disclosed VeriFone proposal, and on November 16, 2010, and based on the implied per share value of the merger consideration. Estimated financial data for Hypercom were based on estimates provided by the Institutional Brokerage Estimate System (IBES) and referred to as Street in the table below, and on estimates provided by the management of Hypercom and referred to as Management in the table below. Estimated financial data for the selected public companies were based on publicly available information and estimates provided by IBES and, in the case of VeriFone, additionally based on estimates provided by the management of VeriFone as modified by Hypercom s management.

This analysis indicated the following implied revenue, EBITDA and price to earnings multiples for the selected point-of-sale terminal companies as compared to the corresponding implied multiples for Hypercom based on the closing share prices of Hypercom common stock on September 29, 2010 and November 16, 2010, and based on the implied per share value of the merger consideration:

		Enterprise Value /				
	Rev	enue	EBITDA		P/E	
	2010E	2011E	2010E	2011E	2010E	2011E
Point-of-Sale Terminal Companies						
Ingenico (Street)	1.5x	1.4x	9.4x	8.4x	17.6x	15.5x
VeriFone (Street)	3.0x	2.7x	15.8x	14.1x	24.0x	19.3x
VeriFone (Management)	2.9x	2.4x	16.5x	11.5x	22.0x	14.9x
Hypercom (Street)						
As of 9/29/10	0.6x	0.5x	6.0x	4.9x	10.2x	8.2x
As of 11/16/2010	0.8x	0.8x	9.0x	7.2x	14.7x	11.8x
At \$7.32 Offer	1.0x	0.9x	11.0x	8.8x	17.6x	14.1x
Hypercom (Management)						
As of 9/29/10	0.6x	0.5x	5.9x	4.4x	9.2x	6.3x
As of 11/16/2010	0.8x	0.8x	8.8x	6.6x	13.3x	9.2x
At \$7.32 Offer	1.0x	0.9x	10.8x	8.1x	15.9x	11.0x

Hypercom Stand-alone Discounted Cash Flow Analysis

UBS performed a discounted cash flow analysis of Hypercom stand-alone using financial forecasts and estimates relating to Hypercom prepared by Hypercom s management. UBS calculated a range of implied present values (as of December 31, 2010) of the stand-alone unlevered, after-tax free cash flows that Hypercom was forecasted to generate from January 1, 2011 through calendar year 2015 using discount rates ranging from 14% to 16%. Implied terminal values were derived by applying to Hypercom s calendar year 2015 estimated EBITDA a range of EBITDA terminal value multiples of 6x to 8x. Present values of cash flows and terminal values were calculated using discount rates ranging from 14% to 16%. The discounted cash flow analysis resulted in a range of implied present values of equity of approximately \$8.70 to \$11.00 per share of Hypercom common stock.

VeriFone Stand-alone Discounted Cash Flow Analysis

UBS performed a discounted cash flow analysis of VeriFone stand-alone using financial forecasts and estimates relating to VeriFone prepared by VeriFone s management, as modified by Hypercom s management. UBS calculated a range of implied present values (as of January 31, 2011) of the stand-alone unlevered, after-tax free cash flows that VeriFone was forecasted to generate from February 1, 2011 through fiscal year 2015 and of

terminal values for VeriFone based on VeriFone s fiscal year 2015 estimated EBITDA. Implied terminal values were derived by applying to VeriFone s fiscal year 2015 estimated EBITDA a range of EBITDA terminal value multiples of 13x to 15x. Present values of cash flows and terminal values were calculated using discount rates ranging from 12.5% to 14.5%. The discounted cash flow analysis resulted in a range of implied present values of approximately \$48.25 to \$58.50 per share of VeriFone common stock.

Pro Forma Discounted Cash Flow Analysis

UBS performed a discounted cash flow analysis of VeriFone pro forma for the merger with Hypercom, using financial forecasts and estimates relating to VeriFone prepared by VeriFone s management, as adjusted by Hypercom s management, and financial forecasts and estimates relating to Hypercom and estimated synergies prepared by Hypercom s management. UBS calculated a range of implied present values (as of January 31, 2011) of the stand-alone unlevered, after-tax free cash flows that VeriFone pro forma for the merger with Hypercom was forecasted to generate from February 1, 2011 through fiscal year 2015 and of terminal values for VeriFone pro forma for the merger with Hypercom based on the fiscal year 2015 estimated EBITDA for VeriFone pro forma for the merger with Hypercom. Implied terminal values were derived by applying to fiscal year 2015 estimated EBITDA for VeriFone pro forma for the merger with Hypercom a range of EBITDA terminal value multiples of 11.75x to 13.75x. Present values of cash flows and terminal values were calculated using discount rates ranging from 12.5% to 14.5%. The pro forma discounted cash flow analysis resulted in a range of implied present values of approximately \$50.50 to \$61.75 per share of common stock of VeriFone pro forma for the merger arange of implied present values of equity of approximately \$11.60 to \$14.20 per share of Hypercom common stock, as compared to the range of implied present values of equity of approximately \$8.70 to \$11.00 per share of Hypercom common stock on a stand-alone basis.

Miscellaneous

Under the terms of UBS s engagement, Hypercom agreed to pay UBS for its financial advisory services in connection with the merger an aggregate fee currently estimated to be approximately \$6,500,000, a portion of which was payable in connection with UBS s opinion and a significant portion of which is contingent upon consummation of the merger. In addition, Hypercom agreed to reimburse UBS for its reasonable expenses, including fees, disbursements and other charges of counsel, and to indemnify UBS and related parties against liabilities, including liabilities under federal securities laws, relating to, or arising out of, its engagement. In the past, UBS and its affiliates have provided investment banking services to Hypercom unrelated to the proposed merger, for which UBS and its affiliates have received compensation, including having acted as financial advisor to Hypercom in 2007 and 2008 in its acquisition of the e-Transaction business line of Thales SA. In the ordinary course of business, UBS and its affiliates may hold or trade, for their own accounts and the accounts of their customers, securities of Hypercom and VeriFone, and, accordingly, may at any time hold a long or short position in such securities. Hypercom selected UBS as its financial advisor in connection with the merger because UBS is an internationally recognized investment banking firm with substantial experience in similar transactions and because of UBS s familiarity with Hypercom. UBS is regularly engaged in the valuation of businesses and their securities in connection with mergers and acquisitions, leveraged buyouts, negotiated underwritings, competitive bids, secondary distributions of listed and unlisted securities and private placements.

Material U.S. Federal Income Tax Consequences

In the opinion of Sullivan & Cromwell LLP and in the opinion of DLA Piper LLP (US), the following section describes the anticipated material United States federal income tax consequences of the merger generally applicable to Hypercom stockholders. These opinions and the following discussion is based on the Internal Revenue Code of 1986, as amended, which is referred to as the Code, Treasury regulations, administrative rulings and court decisions in effect as of the date of this proxy statement/ prospectus, all of which are subject to change at any time, possibly with retroactive effect.

For purposes of this discussion, the term U.S. holder means:

a citizen or resident of the United States;

a corporation, or other entity taxable as a corporation for United States federal income tax purposes, created or organized under the laws of the United States or any of its political subdivisions;

a trust if it (i) is subject to the primary supervision of a court within the United States and one or more United States persons have the authority to control all substantial decisions of the trust, or (ii) has a valid election in effect under applicable United States Treasury regulations to be treated as a United States person; or

an estate that is subject to United States federal income tax on its income regardless of its source. If a partnership holds Hypercom common stock, the tax treatment of a partner in the partnership generally will depend on the status of the partners and the activities of the partnership. If a U.S. holder is a partner in a partnership holding Hypercom common stock, such holder should consult its tax advisor.

This discussion only addresses United States federal income tax consequences of the merger to U.S. holders of Hypercom common stock that hold their Hypercom common stock as a capital asset within the meaning of Section 1221 of the Code. Further, this summary does not address all aspects of United States federal income taxation that may be relevant to a U.S. holder of Hypercom common stock in light of such holder s particular circumstances or that may be applicable to holders subject to special treatment under United States federal income tax law (including, for example, non-United States persons, financial institutions, dealers in securities, insurance companies, tax-exempt entities, holders who acquired Hypercom common stock pursuant to the exercise of stock options or otherwise as compensation or through exercise of warrants, holders subject to the alternative minimum tax provisions of the Code and holders who hold Hypercom common stock as part of a hedge, straddle, constructive sale or conversion transaction). In addition, no information is provided herein with respect to the tax consequences of the merger under applicable state, local or foreign laws.

HOLDERS OF HYPERCOM COMMON STOCK ARE URGED TO CONSULT WITH THEIR TAX ADVISORS REGARDING THE TAX CONSEQUENCES OF THE MERGER TO THEM, INCLUDING THE EFFECTS OF UNITED STATES FEDERAL, STATE AND LOCAL, FOREIGN AND OTHER TAX LAWS.

The Merger. The merger has been structured to qualify as a reorganization under Section 368(a) of the Code for United States federal income tax purposes. It is a condition to the closing of the merger that each of VeriFone and Hypercom receive opinions from its respective counsel that the merger will constitute a reorganization within the meaning of Section 368 of the Code. These opinions will be based on assumptions, representations, warranties and covenants, including those contained in the merger agreement and in tax representation letters to be provided by VeriFone and Hypercom. Although the merger agreement allows each of VeriFone and Hypercom to waive its tax opinion closing condition, neither VeriFone nor Hypercom currently anticipates it will waive this closing condition. If either VeriFone or Hypercom waives this condition, Hypercom will inform you of the decision to waive this condition and will ask you to vote on the merger taking such waiver into consideration.

Assuming that the merger will be treated for federal income tax purposes as a reorganization within the meaning of Section 368(a) of the Code, the material United States federal income tax consequences of the merger to U.S. holders of Hypercom common stock are as follows:

a U.S. holder will not recognize any gain or loss upon receipt of VeriFone common stock solely in exchange for Hypercom common stock in the merger, except with respect to cash received in lieu of a fractional share of VeriFone common stock (as discussed below);

a U.S. holder s aggregate tax basis in the VeriFone common stock received in the merger (including any fractional shares deemed received and redeemed as described below) will be equal to the U.S. holder s aggregate tax basis in the Hypercom common stock surrendered; and

a U.S. holder s holding period of the VeriFone common stock received in the merger (including any fractional shares deemed received and redeemed as described below) will include the U.S. holder s holding period of the Hypercom common stock surrendered.

Cash in Lieu of Fractional Shares. A U.S. holder of Hypercom common stock who receives cash in lieu of a fractional VeriFone common share in the merger generally will be treated as having received such fractional share in the merger and then as having received cash in redemption of such fractional share. Gain or loss generally will be recognized based on the difference between the amount of cash received in lieu of the fractional share and the portion of the U.S. holder s aggregate tax basis in the Hypercom common stock surrendered which is allocable to the fractional share. This gain or loss generally will be long-term capital gain or loss if the holding period for the Hypercom common stock is more than one year at the effective time of the merger. The deductibility of capital losses is subject to limitations.

Ruling. No ruling has been or will be sought from the Internal Revenue Service as to the United States federal income tax consequences of the merger, and the opinions of counsel described above are not binding upon the Internal Revenue Service or any court. Accordingly, there can be no assurances that the Internal Revenue Service will not disagree with or challenge any of the conclusions described herein.

Backup Withholding and Information Reporting. Payments of cash made to a U.S. holder of Hypercom common stock in lieu of fractional shares of VeriFone common stock may be subject to information reporting and backup withholding, unless the U.S. holder of Hypercom common stock: (i) provides a correct taxpayer identification number and any other required information to the exchange agent; or (ii) is a corporation or comes within certain exempt categories and otherwise complies with applicable requirements of the backup withholding rules.

All non-corporate U.S. holders of Hypercom common stock should complete and sign the Substitute Form W-9 that will be included as part of the letter of transmittal to be delivered following completion of the merger. Backup withholding does not constitute an additional tax, but is merely an advance payment of tax, which may be refunded to the extent it results in an overpayment of tax if the required information is supplied to the Internal Revenue Service.

Accounting Treatment of the Merger

In accordance with U.S. generally accepted accounting principles, VeriFone will account for the merger using the acquisition method of accounting.

Regulatory Matters

U.S. Antitrust Approvals

The Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, or the HSR Act, and the rules and regulations thereunder, provide that certain merger transactions may not be consummated until required information and materials have been furnished to the U.S. Department of Justice (DOJ) and/or the Federal Trade Commission (FTC), and certain waiting periods have expired or been terminated.

The conditions to the merger agreement include the expiration or termination of the waiting period under the HSR Act, unless VeriFone and Hypercom have agreed after consultation with their respective outside counsel that no filing under the HSR Act is required, including as a result of divestitures by Hypercom prior to the closing of the merger. VeriFone and Hypercom anticipate that Hypercom s U.S. business will be divested in connection with the merger.

At any time before or after the merger, the DOJ or FTC could take such action under the antitrust laws as it deems necessary or desirable in the public interest, including seeking to enjoin the merger or seeking changes or restrictions in the operations of any of the assets or businesses of VeriFone, Hypercom or their affiliates. Private

parties and state attorneys general may also bring an action under federal or state antitrust laws under certain circumstances. There can be no assurance that a challenge to the merger on antitrust grounds will not be made or, if such a challenge is made, that it would not be successful.

Non-U.S. Antitrust Approvals

VeriFone and Hypercom own property or conduct business in various foreign countries and jurisdictions. In connection with the merger, the laws and regulations of certain of those foreign countries and jurisdictions likely will require the filing of information with, or the obtaining of the approval of, governmental authorities in such countries and jurisdictions. The regulatory authorities in such countries and jurisdictions might attempt to impose additional conditions on VeriFone s or Hypercom s operations conducted in such countries and jurisdictions as a result of the merger.

As of December 20, 2010, the parties anticipate submitting international merger filings in Portugal and Spain and are continuing to evaluate the applicability of other jurisdictions. The competition regimes of Portugal and Spain include processes analogous to those applicable in the United States, requiring an initial filing with the applicable authorities of information about the merger, Hypercom, Honey Acquisition Co. and VeriFone, mandatory review/waiting periods and prohibitions on proceeding without the affirmative approval from the appropriate authority or the lapsing of the applicable mandatory waiting period.

VeriFone and Hypercom have filed a notification with respect to the merger to the Brazilian competition authorities, which include the Secretariat for Economic Monitoring of the Ministry of Finance (SEAE), the Secretariat of Economic Law of the Ministry of Justice (SDE) and the Administrative Council of Economic Defense (CADE), under the applicable Brazilian antitrust merger control law. The review of the merger by the Brazilian competition authorities usually does not prevent the parties from completing the merger. In complex merger cases the Brazilian antitrust authorities may prevent the parties from closing a transaction before CADE s approval, by way of an injunction or an APRO (Agreement designed to Preserve the Reversibility of the Transaction).

Federal Securities Laws Consequences

Shares of VeriFone common stock issued in the merger will not be subject to any restrictions on transfer arising under the Securities Act or the Exchange Act, except for shares of VeriFone common stock issued to any Hypercom stockholder who may be deemed to be an affiliate of VeriFone after the completion of the merger. This proxy statement/prospectus does not cover resales of VeriFone common stock received by any person upon the completion of the merger, and no person is authorized to make any use of this proxy statement/prospectus in connection with any resale.

New York Stock Exchange Listing

The shares of VeriFone common stock to be issued in the merger will be listed for trading on the New York Stock Exchange. If the merger is complete, Hypercom common stock will no longer be registered under the Exchange Act or listed for trading on the New York Stock Exchange.

Litigation Related to the Merger

In connection with the announced merger, several purported class action lawsuits have been filed in Arizona and Delaware state courts alleging variously, among other things, that the board of directors of Hypercom breached its fiduciary duties in not securing a higher price in the merger and that VeriFone, Hypercom, FP Hypercom Holdco, LLC and Francisco Partners II, L.P. aided and abetted that alleged breach. The actions seek injunctive relief and unspecified damages.

No Dissenters or Appraisal Rights

No dissenters or appraisal rights are available to Hypercom stockholders in connection with the merger.

Executives; Executive Compensation; Stock Ownership of Directors, Executive Officers and Five Percent Stockholders

Information concerning current directors and officers of VeriFone, executive compensation and ownership of VeriFone common stock by VeriFone s management and principal stockholders is set forth in VeriFone s Annual Report on Form 10-K for the fiscal year ended October 31, 2010, filed with the SEC on December 21, 2010, and is incorporated herein by reference. See Where You Can Find More Information on page 117.

Information concerning current directors and executive officers of Hypercom, aggregate executive compensation, certain relationships and related transactions and ownership of Hypercom common stock by Hypercom s management and principal is set forth in Hypercom s Annual Report on Form 10-K and Form 10-K/A for the fiscal year ended December 31, 2009, filed with the SEC on March 12, 2010 and December 1, 2010, respectively, and the proxy statement for Hypercom s 2010 annual meeting of stockholders filed with the SEC on April 26, 2010, and is incorporated herein by reference. See Where You Can Find More Information on page 117.

Interests of Certain Persons in the Merger

In considering the recommendations of Hypercom s board of directors with respect to the merger, Hypercom stockholders should be aware that certain officers of Hypercom have agreements or arrangements that provide them interests in the merger that may be different from, or in addition to, the interests of the other stockholders of Hypercom. Hypercom s board of directors was aware of these agreements during its deliberations of the merger and in determining to recommend to Hypercom stockholders that they vote for the proposal to approve and adopt the merger agreement and approve the merger and the agreements.

Common Stock

As of December 17, 2010, the executive officers and directors of Hypercom and their respective affiliates held an aggregate of approximately 1.47% of outstanding Hypercom common stock, including restricted stock and excluding stock options, which will be treated in the merger in the same manner as shares of Hypercom common stock held by other stockholders of Hypercom.

Hypercom Stock Options, Restricted Stock and Hypercom Warrants

As of December 17, 2010, the total number of shares of Hypercom common stock issuable upon the exercise of stock options held by the executive officers and directors of Hypercom as a group (12 persons) was 2,633,922. These options have exercise prices ranging from \$0.96 to \$13.38 per share and a weighted average exercise price of \$4.46 per share. These options and all other Hypercom stock options outstanding at the time of the merger will be assumed by VeriFone and solely exercisable for VeriFone common stock following the merger. Prior to the completion of the merger, all Hypercom stock options and restricted stock that were outstanding on November 17, 2010, the date that the merger agreement was entered into, will be fully vested. Holders of restricted stock will receive the same merger consideration as other holders of Hypercom common stock: 0.23 shares of VeriFone common stock for each share of Hypercom common stock, with any unvested stock issued after November 17, 2010 and not subject to acceleration by the terms of the applicable award or under the Hypercom Change of Control Plan continuing to be subject to vesting restrictions.

As of December 17, 2010, there is an outstanding warrant to purchase 10,544,000 shares of Hypercom common stock held by FP Hypercom Holdco, LLC, whose affiliate, Francisco Partners II, L.P., has the right to nominate two candidates for election to Hypercom s board of directors if certain conditions are met. Francisco Partners II, L.P. has used its right to appoint two directors, Keith Geeslin and Thomas Ludwig.

Change of Control Plan

Under the Hypercom Change of Control Plan, as amended November 15, 2010 by the compensation committee of Hypercom s board of directors and ratified by the board, approximately 92 employees are entitled to severance benefits if they are terminated following a change of control (as defined). The Change of Control Plan was initially implemented to incentivize key employees to remain with Hypercom despite the publicly announced desire of VeriFone to acquire Hypercom, the willingness of Verifone to sell off Hypercom s U.S. operations if needed to address potential competitive concerns and concerns that in light of the ongoing events, key employees may be solicited by competitors. The Change of Control Plan was amended on November 15, 2010 to assure its effectiveness in the merger transaction and to incentivize key employees to remain with Hypercom through any transition period or beyond.

The Change of Control Plan also covers certain senior officers and generally codifies benefits that they had in pre-existing arrangements. With respect to Philippe Tartavull, Thomas Sabol, Scott Tsujita, Douglas Reich and Shawn Rathje, the Change of Control Plan provides for payment of a retention bonus to each of these individuals upon the closing of the merger or their earlier termination of employment without cause or resignation for good reason. Further, each individual will be entitled to company-paid continuing healthcare coverage under the Consolidated Omnibus Budget Reconciliation Act of 1985 (COBRA) for a specified period and accelerated vesting of any equity awards held upon any such termination occurring prior to the merger or within 12 months thereafter. The estimated cash value of benefits available to such named executive officers under the amended Change of Control Plan are summarized in the table below.

Name(1)	Title	Retention Bonus	COBRA Premiums	Relocation Expenses	Equity Awards(4)	Total
Philippe Tartavull	Chief Executive Officer and President	\$ 675,000(2)	\$ 32,400	\$ 75,000(3)	\$ 4,274,550	\$ 5,056,950
Thomas Sabol	Chief Financial Officer	\$450,000	\$ 32,400		\$ 1,871,650	\$ 2,354,050
Scott Tsujita	Senior Vice President, Finance, Treasury and Investor Relations	\$ 192,077	\$ 7,200		\$ 208,200	\$ 407,477
Douglas Reich	Senior Vice President, General Counsel, Chief Compliance Officer and Corporate Secretary	\$ 223,000	\$ 18,000		\$ 375,400	\$ 616,400
Shawn Rathje	Chief Accounting Officer and Corporate Controller	\$ 182,000	\$ 21,600		\$ 421,100	\$ 624,700

- (1) Henry Gaillard resigned on December 13, 2010 and is not entitled to benefits under the Change of Control Plan. See Hypercom s Form 8-K filed on December 16, 2010 for more information.
- (2) Upon completion of the merger Mr. Tartavull will be entitled to a cash retention bonus equal to his base salary for the greater of 18 months or the period remaining until December 30, 2012, the termination date of his employment agreement with Hypercom. Because the amount set forth in the table above represents 18 months of Mr. Tartavull s base salary, this amount will increase if the merger is completed before June 30, 2011. The pre-existing change of control arrangements for Messrs. Sabol and Rathje were set to expire by their terms in April 2011 and February 2011, respectively, but such terms have been replaced and superseded by the Change of Control Plan.
- (3) In addition to the benefits provided by the Change of Control Plan, Mr. Tartavull will be entitled under his employment agreement with Hypercom to reimbursement of reasonable moving expenses to relocate his primary residence during a period of six months following his termination of employment without cause or resignation for good reason within 12 months following the merger.
- (4) The amounts listed in this column represent the intrinsic value of all outstanding equity awards for each executive officer, including vested and unvested awards or portions thereof, as of the date of signing of the merger agreement and as if the closing of the merger had occurred as of such date (based on the closing market price of \$31.84 for VeriFone common stock on November 16, 2010). The board of directors of

Hypercom has approved accelerated vesting in full of all Hypercom equity awards outstanding on November 17, 2010, including those held by the executive officers, effective ten days prior to the expected closing date of the merger (without regard to whether a qualifying termination has occurred).

Employee Benefits

Please refer to The Merger Agreement Employee Benefits on page 90 for a discussion of the employee benefits to be provided to Hypercom employees, including any executive officers, who remain at the combined company following the merger.

Support Agreement

In connection with the merger, VeriFone has entered into a support agreement with FP Hypercom Holdco, LLC and Francisco Partners II, L.P. pursuant to which, among other things, FP Hypercom Holdco, LLC and Francisco Partners II, L.P. agreed to vote any shares of Hypercom common stock owned by them (including any shares of Hypercom common stock that may be issued to FP Hypercom Holdco, LLC upon exercise of the Hypercom warrant held by it) in favor of the merger and against competing acquisition proposals, in each case subject to and on the conditions set forth in the support agreement. As of December 17, 2010, FP Hypercom Holdco, LLC holds a warrant to purchase 10,544,000 shares of Hypercom common stock but based on information received from Francisco Partners II, L.P. neither Francisco Partners II, L.P. nor FP Hypercom Holdco, LLC holds any shares of Hypercom common stock. Accordingly, assuming FP Hypercom Holdco, LLC does not exercise its warrant prior to the record date and neither Francisco Partners II, L.P. nor FP Hypercom Holdco, LLC adquires any shares of Hypercom common stock, FP Hypercom Holdco, LLC and Francisco Partners II, L.P. will not be entitled to vote at the special meeting. For a discussion of this support agreement, see Other Material Agreements Relating to the Merger Support Agreement on page 99.

Indemnification and Insurance

Pursuant to the merger agreement, VeriFone has agreed that, after the effective time of the merger, it will provide certain indemnification and liability insurance benefits to present and former directors and officers of Hypercom. See The Merger Agreement Indemnification and Insurance on page 98.

The Merger Agreement

This section of the proxy statement/prospectus describes the material provisions of the merger agreement. Because the description of the merger agreement contained in this proxy statement/prospectus is a summary, it does not contain all of the information that may be important to you. You should carefully read the entire copy of the merger agreement attached as Annex A to this proxy statement/prospectus, which is hereby incorporated into this proxy statement/prospectus by reference, before you decide how to vote.

The representations, warranties and covenants contained in the merger agreement and described in this proxy statement/prospectus:

were made only for purposes of the merger agreement and as of specific dates and may be subject to more recent developments;

were made solely for the benefit of the parties to the merger agreement and may be subject to limitations agreed upon by the contracting parties, including being qualified by reference to confidential disclosures, for the purposes of allocating risk between parties to the merger agreement instead of establishing these matters as facts; and

may apply standards of materiality in a way that is different from what may be viewed as material by you or by other investors. Accordingly, these representations and warranties alone may not describe the actual state of affairs as of the date they were made or at any other time. The representations and warranties contained in the merger agreement do not survive the effective time of the merger. Investors should not rely on the representations, warranties and covenants or any description thereof as characterizations of the actual state of facts or condition of VeriFone, Hypercom, Honey Acquisition Co. or any of their respective subsidiaries or affiliates. Moreover, information concerning the subject matter of the representations, warranties and covenants may change after the date of the merger agreement, which subsequent information may or may not be fully reflected in public disclosures by VeriFone and Hypercom.

Structure of the Merger

The merger agreement provides for the merger of Honey Acquisition Co., a newly formed and direct, wholly owned subsidiary of VeriFone, into Hypercom. After the merger, Hypercom will become a direct, wholly owned subsidiary of VeriFone.

Completion and Effectiveness of the Merger

We will complete the merger as promptly as practicable, but in no event later than the third business day after the last of the conditions to completion are satisfied or waived, or at such other time or date as the parties to the merger agreement may mutually agree.

The merger will become effective at the time when the certificate of merger has been duly filed with the Secretary of State of the State of Delaware.

Conversion of Hypercom Common Stock in the Merger

Each share of Hypercom common stock issued and outstanding at the effective time of the merger, other than shares owned by VeriFone, Honey Acquisition Co. or any other direct or indirect subsidiary of VeriFone or shares that are owned by Hypercom or its direct or indirect subsidiaries that in each case are not held on behalf of third parties, will convert into the right to receive 0.23 shares of common stock, par value \$0.01 per share, of VeriFone.

VeriFone will not issue any fractional shares. Instead, Hypercom stockholders will receive a cash payment in accordance with the terms of the merger agreement, as described below in Fractional Shares.

Exchange Procedures

You should not send stock certificates with your proxy card and should not surrender stock certificates prior to the adoption of the merger agreement and approval of the merger by Hypercom stockholders and the receipt of a transmittal form.

Fractional Shares

Each Hypercom stockholder who would otherwise have been entitled to receive a fraction of a share of VeriFone common stock will receive cash, without interest and rounded to the nearest whole cent, an amount determined by multiplying such fraction by the per share closing price of VeriFone common stock on the New York Stock Exchange on the last trading day immediately prior to the closing date.

Dividends and Distributions

All shares of VeriFone common stock to be issued pursuant to the merger shall be deemed issued and outstanding as of the effective time of the merger. Whenever a dividend or other distribution is declared by VeriFone in respect of VeriFone common stock, and the record date for the dividend or distribution is on or after the date of the effective time of the merger, the declaration will include those dividends or other distributions in respect of all VeriFone common stock issued or issuable pursuant to the merger agreement. No dividends or other distributions with respect to VeriFone common stock will be paid unless a Hypercom share certificate is surrendered.

Transfer of Ownership and Lost Stock Certificates

If a Hypercom share certificate has been lost, stolen or destroyed, the Hypercom stockholder may need to deliver an affidavit or bond prior to receiving a VeriFone stock certificate.

Hypercom s Representations and Warranties

Hypercom made a number of representations and warranties to VeriFone in the merger agreement. These representations and warranties include representations as to:

corporate organization, good standing and qualification to do business of Hypercom and its subsidiaries;

the governing documents of Hypercom and its subsidiaries;

Hypercom s capital structure;

authorization of the merger agreement by Hypercom;

enforceability of the merger agreement;

the fairness opinion received by Hypercom from UBS;

governmental consents, filings and regulatory approvals necessary to complete the merger;

the merger not conflicting with the organizational documents or material contracts of Hypercom or its subsidiaries or applicable laws;

contracts that limit the type of business in which Hypercom or its subsidiaries may engage or the manner or locations in which any of them may so engage in any business;

Hypercom s filings with the SEC, including financial statements, and the accuracy of the information contained therein;

absence of certain changes, events or material adverse effects involving Hypercom or its subsidiaries since the most recent audited balance sheet;

litigation and liabilities involving Hypercom or its subsidiaries;

employee benefit plans of Hypercom and its subsidiaries;

compliance with laws and regulatory matters by Hypercom and its subsidiaries;

the inapplicability of takeover statutes and any anti-takeover provisions in Hypercom s certificate of incorporation or bylaws;

environmental matters relating to Hypercom and its subsidiaries;

taxes and tax matters relating to Hypercom and its subsidiaries;

Hypercom s and its subsidiaries labor relations;

intellectual property matters involving Hypercom and its subsidiaries;

transactions between Hypercom and its subsidiaries, on the one hand, and Hypercom s affiliates, on the other hand;

Hypercom s and its subsidiaries insurance;

Hypercom s material contracts and governmental contracts;

Hypercom s good and marketable title to property;

lease agreements of Hypercom and its subsidiaries;

Hypercom s warranties and product liability;

product certifications given or granted;

Hypercom s inventories;

possession of and compliance with all licenses necessary to conduct Hypercom s business;

absence of any violation to the U.S. Foreign Corrupt Practice Act of 1977;

absence of any sanctions by the Office of Foreign Assets Control;

brokers and finders;

neither Hypercom nor its subsidiaries is an investment company as defined in the Investment Company Act of 1940;

the inapplicability of Hypercom s rights agreement; and

reorganization treatment of the merger. VeriFone s and Honey Acquisition Co. s Representations and Warranties

VeriFone and Honey Acquisition Co. made a number of representations and warranties to Hypercom in the merger agreement. These representations and warranties include representations as to:

corporate organization, good standing and qualification to do business of VeriFone, Honey Acquisition Co. and VeriFone s subsidiaries;

the certificate of incorporation and bylaws of VeriFone and its subsidiaries;

capitalization of Honey Acquisition Co.;

VeriFone s capital structure;

authorization of the merger agreement by VeriFone;

enforceability of the merger agreement;

VeriFone common stock to be issued pursuant to the merger;

governmental consents, filings and regulatory approvals necessary to complete the merger;

the merger not conflicting with the organizational documents or material contracts of VeriFone or its significant subsidiaries or applicable laws;

VeriFone s filings with the SEC, including financial statements, and the accuracy of the information contained therein;

absence of certain changes, events or material adverse effects involving VeriFone or its subsidiaries since the most recent audited balance sheet;

litigation and liabilities involving VeriFone or its subsidiaries;

brokers and finders;

none of VeriFone, Honey Acquisition Co., or any of their affiliates being an interested stockholder of Hypercom for purposes of Section 203 of the Delaware General Corporation Law;

compliance with laws and regulatory matters by VeriFone and VeriFone s significant subsidiaries; and

reorganization treatment of the merger.

The representations and warranties contained in the merger agreement are complicated and not easily summarized. You are urged to read carefully Sections 5.1 and 5.2 of the merger agreement entitled Representations and Warranties of the Company and Representations and Warranties of Parent and Merger Sub. The representations and warranties of Hypercom and VeriFone are qualified by disclosure schedules and as to materiality and material adverse effect.

Hypercom s Conduct of Business Before Completion of the Merger

Pursuant to the terms of the merger agreement, Hypercom has agreed that until the effective time of the merger, or unless VeriFone approves in writing, Hypercom and its subsidiaries business will be conducted in the ordinary and usual course and Hypercom and its subsidiaries will use reasonable best efforts to:

preserve their business organizations intact; and

maintain existing relations and goodwill with governmental entities, customers, suppliers, distributors, creditors, lessors, employees and business associates and keep available the services of its and its subsidiaries present employees and agents.

In addition, Hypercom has also agreed that until the effective time of the merger, subject to specified exceptions, unless VeriFone approves in writing, it will not, and none of its subsidiaries will:

amend its certificate of incorporation, bylaws or other applicable governing instruments;

split, combine, subdivide or reclassify its outstanding shares of capital stock;

declare, set aside or pay any dividend or distribution payable in cash, stock or property in respect of any capital stock;

repurchase, redeem or otherwise acquire any shares of its capital stock or any securities convertible into or exchangeable or exercisable for any shares of its capital stock;

merge or consolidate itself or any of its subsidiaries with any other person or restructure, reorganize or completely or partially liquidate or otherwise enter into any agreements or arrangements imposing material changes or restrictions on its assets, operations or businesses;

spend in excess of \$6,500,000 individually or \$10,000,000 in the aggregate to acquire any business, subject to certain conditions, including that any such acquisition not prevent, delay or impair Hypercom s ability to consummate the transaction;

incur any indebtedness, except for (A) drawings under and refinancings or replacements of Hypercom s or any of its subsidiaries current credit agreements in an amount not to exceed \$25,000,000 in the aggregate at any one time outstanding, (B) other indebtedness for borrowed money incurred in the ordinary course of business consistent with past practices not to exceed \$10,000,000 in the aggregate, or (C) interest rate or currency swaps on customary commercial terms consistent with past practice and in compliance with its risk management policies in effect on the date of the merger agreement and not to exceed \$5,000,000 of notional debt in the aggregate;

make or commit to any capital expenditures other than in the ordinary course of business and in any event not in excess of the aggregate amount reflected in Hypercom s capital expenditure budget for the year in which such capital expenditures are made;

transfer, sell, lease, license, mortgage, pledge, surrender, encumber, divest, cancel, abandon or allow to lapse or expire or otherwise dispose of any of Hypercom s assets, product lines or businesses or of its subsidiaries, except for dispositions of assets with a fair market value not in excess of \$10,000,000 in the aggregate, other than pursuant to contracts in effect as of the date of the merger agreement;

issue, sell, pledge, dispose of, grant, transfer, encumber, or authorize the issuance, sale, pledge, disposition, grant, transfer, lease, license, guarantee or encumbrance of any shares of, or securities convertible or exchangeable into or exercisable for, or any options, warrants or other rights to acquire shares of, its capital stock or of any its subsidiaries, subject to certain exceptions, including the issuance of shares pursuant to existing options, warrants and awards;

make any change with respect to accounting policies or procedures, except as required by changes in GAAP or by law;

except as required by law:

make any material tax election or take any position on any tax return filed on or after the date of the merger agreement or adopt any method that is inconsistent with elections made, positions taken or methods used in preparing or filing similar tax returns in prior periods;

settle or resolve any material tax controversy, claim or assessment;

enter into any material closing agreement;

waive or extend any statute of limitations with respect to taxes; or

surrender any right to claim a refund for taxes; waive or extend any statute of limitations with respect to taxes or surrender any right to claim a refund for taxes;

make any loans, advances or capital contributions to or investments in any person in excess of \$1,000,000 individually or \$2,000,000 in the aggregate;

enter into any non-competition contract or any other contract that (i) purports to limit in any material respect either the type of business in which Hypercom or its subsidiaries (or, after the effective time of the merger, VeriFone or its affiliates) may engage or the manner or locations in which any of them may so engage in any business or (ii) could require the disposition of any material assets or line of business of Hypercom or its subsidiaries or, after the effective time of the merger, VeriFone or its affiliates;

grant or provide any severance or termination payments or benefits, increase the compensation, bonus or pension, welfare, severance or other benefits of, pay any bonus to, or make any new equity awards to any of its directors, officers or employees, except in the ordinary course of business consistent with past practices;

establish, adopt, amend or terminate any of its benefit plans or amend the terms of any outstanding equity-based awards;

subject to certain exceptions, take any action to accelerate the vesting or payment, or fund or in any other way secure the payment, of compensation or benefits under any of its benefit plans, to the extent not already provided in those benefit plans;

change any actuarial or other assumptions used to calculate funding obligations with respect to any benefit plan or change the manner in which contributions to those plans are made or the basis on which contributions are determined, except as may be required by GAAP;

forgive any loans to any of its or of any of its subsidiaries directors, officers or employees;

amend, modify or terminate any material contract or cancel, modify or waive any debts or claims held by it or waive any rights other than in the ordinary course of business having a value in excess of \$1,000,000;

settle any litigation if the settlements would exceed \$1,000,000 in the aggregate; and

take any action that is reasonably likely to result in any of the conditions to its obligations to consummate the merger not to be satisfied.

In addition, VeriFone has also agreed that, until the effective time of the merger, it will not and will not permit its subsidiaries to take any action that is reasonably likely to result in any of the conditions to its obligations to consummate the merger not to be satisfied.

No Solicitation of Acquisition Proposals by Hypercom

The merger agreement provides that neither Hypercom nor any of its subsidiaries nor any of the officers and directors of Hypercom or its subsidiaries will, and that Hypercom will instruct and cause its and its subsidiaries directors, officers, employees, investment bankers, attorneys, accountants and other advisors or representatives not to, directly or indirectly:

initiate, solicit, or knowingly facilitate or encourage any inquiries or the making of any proposal or offer that constitutes, or would reasonably be expected to lead to, an acquisition proposal;

engage in, continue or otherwise participate in any discussions or negotiations regarding, or provide any non-public information or data to any person relating to, any acquisition proposal; or

otherwise knowingly facilitate any effort or attempt to make an acquisition proposal. Under the merger agreement, an acquisition proposal is:

any proposal or offer with respect to a merger, joint venture, consolidation, dissolution, liquidation, tender offer, recapitalization, reorganization, share exchange, business combination or similar transaction involving Hypercom; or

any acquisition by any person resulting in (or any proposal or offer to acquire in any manner which if consummated would result in) any person becoming the beneficial owner of, directly or indirectly, in one or a series of related transactions 15% or more of the total voting power or of any class of equity securities of Hypercom or any of its subsidiaries, or 15% or more of the consolidated total assets of Hypercom and its subsidiaries taken as a whole, in each case other than the merger with VeriFone.

The merger agreement does not prevent Hypercom from doing the following prior to, but not after, the receipt of the vote of a majority of the holders of the outstanding shares of Hypercom common stock entitled to vote at the special meeting:

providing information in response to a request by a person who has made an unsolicited bona fide written acquisition proposal if the board of directors receives from the person requesting such information an executed confidentiality agreement on terms not less restrictive to such person than those contained in the confidentiality agreement between Hypercom and VeriFone;

engaging or participating in any discussions or negotiations with any person who has made an unsolicited bona fide written acquisition proposal; or

approving, recommending, or otherwise declaring advisable such an acquisition proposal to Hypercom stockholders;

if and only to the extent that, in each case, Hypercom s board of directors determines in good faith after consultation with outside legal counsel and financial advisors that the failure to take such action would be inconsistent with the directors fiduciary duties under applicable law, and Hypercom s board of directors has determined in good faith based on the information then available and after consultation with its outside legal counsel and financial advisor that such acquisition proposal constitutes a superior proposal (or in the case of the first two bullet points above, is reasonably likely to result in a superior proposal). In addition, during the two-business-day period prior to taking any action described in the third bullet point above, Hypercom must negotiate in good faith with VeriFone, if requested by VeriFone, any revisions to the terms of the merger agreement.

Under the merger agreement, a superior proposal is an unsolicited bona fide written acquisition proposal involving 80% or more of the assets (on a consolidated basis) or the total voting power of the equity securities of Hypercom that Hypercom s board of directors has determined in its good faith judgment is reasonably capable of being consummated in accordance with its terms, and that, taking into account all legal, financial and regulatory aspects of the proposal and the person making the proposal, if consummated, would result in a transaction more favorable to Hypercom s stockholders than the transaction contemplated by the merger agreement.

In addition, Hypercom has agreed, pursuant to the merger agreement, that it will promptly disclose to VeriFone any non-public information provided to any third-party pursuant to a request by a third-party who has made an unsolicited bona fide written acquisition proposal and executed a confidentiality agreement on terms not less restrictive than those contained in the confidentiality agreement between Hypercom and VeriFone.

Changes in Hypercom s Board of Directors Recommendation

Under the merger agreement, Hypercom has also agreed that subject to certain exceptions, the board of directors of Hypercom may not:

withdraw, qualify or modify (or publicly propose or resolve to withhold, withdraw, qualify or modify), in a manner adverse to VeriFone, its recommendation to Hypercom stockholders with respect to the merger; or

cause or permit Hypercom to enter into any alternative acquisition agreement relating to any acquisition proposal. The Hypercom board of directors may make an adverse recommendation change or recommend a superior proposal if the board of directors determines in good faith, after consultation with its outside counsel and financial advisors, that failure to take such action would be inconsistent with the directors fiduciary obligations under applicable law, either in connection with:

a superior proposal; or

as a result of a fact, event, circumstance, development or a change in circumstances or facts that was not known (or if known, the magnitude and material consequences of which were not known) by the board of directors as of the date of the merger agreement. Hypercom must provide VeriFone with two-business-days notice prior to taking any such action, and during the two-business-day period, it must negotiate in good faith with VeriFone revisions to the merger agreement, if requested by VeriFone. Following the two-business-day period, Hypercom may take action with respect to a superior proposal only if the proposal continues to be a superior proposal in light of any revisions to the merger agreement proposed by VeriFone.

Solicitation of Votes

The merger agreement provides that Hypercom will take, in accordance with applicable law and its certificate of incorporation and bylaws, all action necessary to convene a meeting of holders of Hypercom common stock as promptly as practicable after the registration statement of which this proxy statement/

prospectus forms a part is declared effective to consider and vote upon the adoption of the merger agreement and approval of the merger and to cause such vote to be taken. Except on the determination of the occurrence of a superior proposal and during such time as there remains a superior proposal or as Hypercom s board of directors may determine in good faith, after consultation with its outside legal counsel, in order to comply with its fiduciary duties under applicable law, Hypercom s board of directors will recommend such adoption and approval and will take all lawful action to solicit such adoption. Hypercom s obligation to call, give notice of, convene and hold the stockholders meeting will not be limited to or otherwise affected by the commencement, disclosure, announcement or submission of any acquisition proposal to Hypercom.

Effect on Hypercom Stock Options and Hypercom Restricted Stock Awards

Prior to the effective time of the merger, Hypercom will take actions to fully vest all Hypercom stock options and restricted stock awards that were outstanding on November 17, 2010, the date that VeriFone and Hypercom entered into the merger agreement. In addition, certain senior officers are entitled to vesting of any equity awards held upon termination of employment without cause or resignation for good cause occurring prior to the merger or within 12 months thereafter. See Interests of Certain Persons in the Merger Change of Control Plan. At the effective time of the merger, each outstanding option to purchase Hypercom common stock, whether vested or unvested, will be converted into an option to acquire a number of shares of VeriFone common stock equal to the product of (x) the number of shares of Hypercom common stock subject to an option and (y) the exchange ratio of 0.23, at an exercise price per share (rounded up to the nearest whole cent) equal to (A) the exercise price per share of Hypercom common stock of such outstanding option immediately prior to the effective time divided by (B) 0.23. Holders of restricted stock will receive the same merger consideration as other holders of Hypercom common stock: 0.23 shares of VeriFone common stock for each share of Hypercom common stock, with any unvested stock issued after November 17, 2010 and not subject to acceleration by the terms of the applicable award or under the Hypercom Change of Control Plan continuing to be subject to vesting restrictions.

As soon as reasonably practicable (but not more than five days) after the effective time of the merger, VeriFone will file a registration statement on Form S-8 and use its reasonable best efforts to maintain the effectiveness of such registration statement for so long as the relevant Hypercom stock plans or other benefit plans remain in effect and such registration continues to be required.

Effect on Hypercom Warrants

At the effective time of the merger, each outstanding warrant to purchase shares of Hypercom common stock will be converted into a warrant to acquire a number of shares of VeriFone common stock equal to the product of (x) the number of shares of Hypercom common stock subject to such warrant and (y) the exchange ratio of 0.23, at an exercise price per share (rounded up to the nearest whole cent) equal to (A) the exercise price per share of Hypercom common stock for such warrant immediately prior to the effective time divided by (B) 0.23. Following the effective time of the merger, each Hypercom warrant will continue to be governed by the same terms and conditions as were applicable under such warrant immediately prior to the effective time of the merger.

Employee Benefits

For a period ending not earlier than 12 months following the closing of the merger, employees of Hypercom and its subsidiaries who continue their employment after the effective time of the merger will be provided with employee benefits under either:

the employee benefit programs applicable to similarly situated employees of VeriFone; or

the employee benefit programs applicable to affected employees immediately prior to the effective time, as determined in the sole discretion of VeriFone.

VeriFone will, and will cause Hypercom to, honor and pay from and after the effective time any bonus payment obligations of Hypercom and each subsidiary of Hypercom properly accrued on Hypercom s financial statements in accordance with GAAP for Hypercom s fiscal year (or any portion thereof) ending with or before

the effective time under bonus plans of Hypercom and its subsidiaries in accordance with their terms as in effect immediately prior to the effective time. With respect to any benefit plan, program or arrangement, VeriFone will recognize the service of employees of Hypercom and its subsidiaries (and any of their predecessors) prior to the effective time of the merger for purposes of eligibility, vesting, level of benefits and benefit accruals, except with respect to any defined benefit pension plan and to the extent such recognition would result in a duplication of benefits.

With respect to any welfare plan in which employees of Hypercom and its subsidiaries and their covered dependents are eligible to participate after the effective time, VeriFone will, and will cause Hypercom to use reasonable best efforts to, waive all limitations as to preexisting conditions, exclusions and waiting periods with respect to participation and coverage requirements applicable to such employees and covered dependents to the extent such conditions were satisfied, inapplicable or waived under the welfare plans of Hypercom and its subsidiaries prior to the effective time, and (ii) provide each such employee and covered dependent with credit for any co-payments, deductibles and maximum out-of pocket requirements paid or incurred prior to the effective time in satisfying any analogous deductible or maximum out-of-pocket requirements to the extent applicable under any such plan.

VeriFone will also cause Hypercom to honor all of its obligations under its compensation and benefit plans, and compensation and severance arrangements and agreements, in accordance with their terms as in effect immediately prior to the effective time; VeriFone and Hypercom may, however, amend or terminate any particular compensation and benefit plan to the extent permitted by its terms or applicable laws. VeriFone will cause Hypercom to provide affected employees whose employment terminates during the 12-month period following the effective time of the merger with severance benefits at levels no less than and pursuant to the terms of either Hypercom s severance policy as in effect on the date of the merger agreement or under VeriFone s severance policy, as determined in the sole discretion of VeriFone. VeriFone will not be obligated to provide these benefits to individuals who are parties to individual agreements providing severance or termination benefits or entitled to severance or termination benefits under Hypercom s Change of Control Plan.

Conditions to Completion of the Merger

Mutual Conditions. The obligations of VeriFone and Hypercom to complete the merger are subject to the satisfaction or waiver, at or prior to the effective time of the merger, of each of the following conditions, among others:

the adoption of the merger agreement by Hypercom s stockholders;

unless VeriFone and Hypercom have agreed after consultation with their respective outside counsel that no filing under the HSR Act is required, the expiration or the termination of the waiting period applicable to the completion of the merger under the HSR Act and all governmental consents and any other notification, waiting period, or approval requirements under the comparable antitrust or competition laws of other applicable foreign jurisdictions in which either party has operations or from which either party derives revenues (which in either case are not de minimis) and which are legally required to be made or obtained at or prior to the effective time of the merger must have been made or obtained without the imposition of any term, condition or consequence the acceptance of which would be reasonably likely to have a material adverse effect on Hypercom or VeriFone or the requirement to divest, hold separate or license assets or businesses (other than divestitures of businesses or assets that produced aggregate gross revenues in an amount not in excess of \$124 million (or a higher amount, if agreed to by VeriFone) during the 2009 fiscal year);

the approval for listing on the NYSE of the VeriFone common stock issuable to Hypercom stockholders;

no court or governmental entity of competent jurisdiction having enacted, issued, promulgated, enforced or entered any statute, law, ordinance, rule, regulation, judgment, decree, injunction or other order that is in effect and restrains, enjoins or otherwise prohibits completion of the merger or the other transactions contemplated by the merger agreement;

the registration statement of which this proxy statement/prospectus is a part having become effective under the Securities Act, no stop order suspending the effectiveness of the registration statement having been issued, and no proceeding for that purpose having been initiated or threatened, by the SEC;

VeriFone having received all state securities and blue sky permits and approvals necessary to consummate the transactions contemplated in the merger agreement;

if one or more asset purchase agreements have been entered into for asset dispositions in connection with regulatory matters, then (i) all of the conditions to the closing under such asset purchase agreements having been satisfied or waived, and (ii) Hypercom and the applicable transferee having effectuated the applicable disposition and unless VeriFone otherwise consents (such consent not to be unreasonably withheld), without deviation from the terms set forth in each such asset purchase agreement;

each of Hypercom and VeriFone having received a written opinion from its respective counsel to the effect that the merger will constitute a reorganization within the meaning of Section 368 of the Internal Revenue Code for U.S. federal income tax purposes;

the representations and warranties of the other party set forth in the merger agreement (i) which are qualified by a material adverse effect qualification, must be true and correct in all respects as so qualified and (ii) which are not so qualified, must be true and correct, in each case at and as of the closing date as if made on the closing date, except that this condition will be deemed to have been satisfied even if the representations or warranties are not so true and correct unless the failure of the representations or warranties to be so true and correct, individually or in the aggregate, has had, or is reasonably likely to have, a material adverse effect on the other party;

the other party having performed in all material respects all obligations required to be performed by it under the merger agreement at or prior to the closing date; and

there not having occurred any event, occurrence, discovery or development that, individually or in the aggregate, has resulted, or would reasonably be likely to result, in a material adverse effect on the other party. *Additional VeriFone Conditions.* VeriFone s obligation to complete the merger is also subject to the satisfaction or waiver by VeriFone at or prior to the effective time of the merger of the following conditions, among others:

the representations and warranties of Hypercom relating to the fully-diluted capitalization of Hypercom must be true and correct on the date of the merger agreement and at the effective time as though made on and as of the effective time, other than:

changes in capitalization due to any permitted capitalization change; and

other deviations in Hypercom s actual fully-diluted capitalization by an amount that does not exceed one percent of such fully-diluted capitalization;

no court or governmental entity of competent jurisdiction having instituted any suit, action or proceeding seeking to:

prohibit completion of the merger or prohibit, limit, restrain or impair VeriFone s ability to own or operate or to retain or change all or a material portion of the assets, licenses, operations, rights, product lines, businesses or interest therein of Hypercom or of VeriFone, other than with respect to the disposition of businesses or assets that, in the aggregate, exceed the threshold of \$124 million of aggregate gross revenues during the 2009 fiscal year as set forth in the merger agreement; or

prohibit or limit VeriFone s ability to vote, transfer, receive dividends or otherwise exercise full ownership rights with respect to Hypercom;

the receipt of payoff letters under Hypercom s credit facilities; and

the resignation of each director of Hypercom effective as of the closing. **Determination of Material Adverse Effect**

Under the terms of the merger agreement, a VeriFone material adverse effect is defined to mean a material adverse effect on the financial condition, assets, liabilities, business or results of operations of VeriFone and its subsidiaries taken as a whole, or a material adverse effect on the ability of VeriFone or Honey Acquisition Co. to consummate the transactions contemplated by the merger agreement, except that the following are excluded from the definition of VeriFone material adverse effect and from the determination of whether such a VeriFone material adverse effect has occurred:

changes in general political, economic or market (including currency) conditions, or in the industry in which VeriFone and its subsidiaries operate, except to the extent such changes or conditions disproportionately affect in any material respect VeriFone and its subsidiaries, taken as a whole, as compared to other persons engaged in similar businesses;

acts of terrorism or war (whether or not declared) or natural disasters occurring after the date of the merger agreement, except to the extent such changes or conditions disproportionately affect in any material respect VeriFone and its subsidiaries, taken as a whole, as compared to other persons engaged in similar businesses;

the merger agreement, the merger and other transactions contemplated thereby, or the announcement or performance thereof, including any negative impact on or disruption in relationships with customers, suppliers, distributors, employees or similar relationships;

the taking of any action required by the merger agreement or expressly approved or permitted in writing by Hypercom or reasonably required to obtain the approval of any governmental entity or the failure to take any action prohibited by the merger agreement;

changes in the price or trading volume of VeriFone s stock (provided that the occurrence which caused or contributed to such change in market price or trading volume shall not be excluded);

any failure by VeriFone to meet public or internal revenue, earnings or other projections, in and of itself; and

changes in law or any applicable accounting regulations or principles or the interpretations thereof. Under the terms of the merger agreement, a Hypercom material adverse effect is defined to mean a material adverse effect on the financial condition, assets, liabilities, business or results of operations of Hypercom and its subsidiaries taken as a whole, or a material adverse effect on the ability of Hypercom to consummate the transactions contemplated by the merger agreement, except that the following are excluded from the definition of Hypercom material adverse effect and from the determination of whether such a Hypercom material adverse effect has occurred:

changes in general political, economic or market (including currency) conditions, or in the industry in which Hypercom and its subsidiaries operate, except to the extent such changes or conditions disproportionately affect in any material respect Hypercom and its subsidiaries, taken as a whole, as compared to other persons engaged in similar businesses;

acts of terrorism or war (whether or not declared) or natural disasters occurring after the date of the merger agreement, except to the extent such changes or conditions disproportionately affect in any material respect Hypercom and its subsidiaries, taken as a whole, as compared to other persons engaged in similar businesses;

the merger agreement, the merger and other transactions contemplated thereby, or the announcement or performance thereof, including any negative impact on or disruption in relationships with customers, suppliers, distributors, employees or similar relationships;

changes in the price or trading volume of Hypercom s stock (provided that the occurrence which caused or contributed to such change in market price or trading volume shall not be excluded);

any failure by Hypercom to meet public or internal revenue, earnings or other projections, in and of itself;

changes in law or any applicable accounting regulations or principles or the interpretations thereof;

the taking of any action required by the merger agreement or expressly approved or permitted in writing by VeriFone or reasonably required to obtain the approval of any governmental entity or the failure to take any action prohibited by the merger agreement; and

any legal proceedings commenced or threatened by any of the current or former stockholders of Hypercom (on their own behalf or on behalf of Hypercom) against Hypercom that assert allegations of a breach of fiduciary duty or violations of securities laws relating to the merger agreement or the transactions contemplated by the merger agreement.

The conditions to each of the parties obligations to consummate the merger are for the sole benefit of such party and may be waived by such party in whole or in part to the extent permitted by applicable law.

Reasonable Best Efforts Covenant

Under the merger agreement and subject to the terms and conditions of the merger agreement, VeriFone and Hypercom have each agreed to cooperate with each other and use their respective reasonable best efforts to take or cause to be taken all actions, and do or cause to be done all things, reasonably necessary, proper or advisable on its part under the merger agreement and applicable laws to complete the merger.

VeriFone has agreed that reasonable best efforts includes a requirement by VeriFone to agree to sell or otherwise dispose of, and to sell or otherwise dispose of, businesses or assets that, subject to certain exceptions, in the aggregate, produced aggregate gross revenues in an amount not in excess of \$124 million for VeriFone or Hypercom during the 2009 fiscal year, if such action should be reasonably necessary to cause the closing conditions to be met in the merger agreement. Except with respect to this covenant, the merger agreement provides that VeriFone will not be required to proffer to, or agree to, sell, divest, lease, license, transfer, dispose of or otherwise encumber or hold separate and agree to sell, divest, lease, license, transfer, dispose of or otherwise encumber of the merger, any assets, licenses, operations, rights, product lines, businesses or interest therein of VeriFone or Hypercom (or to consent to any sale, divestiture, lease, license, transfer, disposition or other encumbrance by Hypercom of any of its assets, licenses, operations, rights, product lines, businesses or interest therein, loss of ability to set off for tax purposes in the future any and all losses accumulated by Hypercom or to any agreement by Hypercom to take any of the foregoing actions) or to agree to any material changes (including through a licensing arrangement) or restriction on, or other impairment of VeriFone s ability to own or operate, of any such assets, licenses, product lines, businesses or interest therein of veriFone s ability to vote, transfer, receive dividends or otherwise exercise full ownership rights with respect to the stock of Hypercom.

In addition, so long as VeriFone requests, and subject to certain limitations, Hypercom has agreed to use reasonable best efforts as promptly as practicable to effectuate the disposition of portions of its business identified by VeriFone as reasonably necessary to satisfy the conditions to closing in the merger agreement.

Additional Covenants

Non-Solicitation of Employees. VeriFone and Hypercom have agreed that through the earlier of the effective time of the merger (if the closing occurs) and the six-month period following termination of the merger agreement (if the closing does not occur), neither will solicit for hire certain specified officers or key employees of the other party, or, except where this restriction is prohibited by law, hire any such employee. However,

nothing in the merger agreement prohibits solicitation and hiring by means of a general advertisement not directed at any particular individual or the employees of the other party generally, or responding to and hiring an employee of the other party who initiates contact with the applicable party, provided that such contact is not solicited by the applicable party.

Certain Communications. In addition, VeriFone and Hypercom have agreed that they will not seek to effect, announce or communicate any potential reduction or elimination of the other party s products or services or change in the pricing thereof.

Damages. In the case of the first breach, if any, of either of the two covenants described above, the breaching party will have three days in which to cure the breach. If the initial breach is not cured within this three-day period, or if there is any subsequent breach, the breaching party must pay the other party, within three business days of notice of the breach, \$1 million. However, these amounts must only be paid in the event of a breach caused by or at the instructions of or with the consent of or prior knowledge of (if the senior officer does not use reasonable efforts to prevent such breach and does not direct or re-direct the applicable person to refrain from such activity after obtaining such knowledge) one or more senior officers of the breaching party.

Customer Commitments. In addition, if the closing occurs, VeriFone has agreed with Hypercom to do the following (and not to announce, communicate or take any action to the contrary during the period through the closing):

fulfill any Hypercom customer agreements or purchase orders entered into in the ordinary course of business consistent with Hypercom s current business practices, unless the customer agrees otherwise;

to the extent requested by the applicable customer, provide warranty service on Hypercom products required by contracts entered into in the ordinary course of business consistent with Hypercom s current business practices, as well as repair and maintenance for the longer of three years from closing or as required by law, at service levels consistent with Hypercom s current business practices; and

to the extent requested by the applicable customer, continue to support software applications, including the provision of bug fixes and updates, for the longer of 12 months from closing or as required by law, at service levels consistent with Hypercom s current business practices.

Rights Agreement. Prior to the effective time of the merger, the Hypercom board of directors will take any action necessary to prevent the merger and the merger agreement from causing the rights granted under Hypercom s stockholder rights agreement from becoming exercisable.

Reorganization. VeriFone and Hypercom have agreed to use commercially reasonable efforts to cause the merger to be treated as a reorganization within the meaning of Section 368(a) of the Internal Revenue Code and to not knowingly take any action, or fail to take any action, that would prevent, or reasonably be expected to prevent, the merger from qualifying as a reorganization under Section 368(a).

Waiver and Amendment of the Merger Agreement

Subject to the provisions of applicable law, at any time prior to the effective time of the merger, the parties to the merger agreement may modify or amend the merger agreement, by written agreement executed and delivered by duly authorized officers of the respective parties. After adoption of the merger agreement by Hypercom stockholders, the parties may not amend or waive any provision of the merger agreement if such amendment or waiver would require further approval of Hypercom stockholders under Delaware law, unless such approval has first been obtained.

Any provision of the merger agreement may be waived prior to the effective time of the merger if, and only if, such waiver is in writing and signed by the party against whom the waiver is to be effective.

Termination of the Merger Agreement

The merger agreement may be terminated at any time prior to the effective time of the merger, whether before or after adoption of the merger agreement by Hypercom s stockholders, by mutual written consent of Hypercom and VeriFone.

The merger agreement may be terminated at any time prior to the effective time of the merger, whether before or after approval by stockholders of Hypercom by action of either VeriFone or Hypercom if:

the merger has not occurred by August 31, 2011, provided that if all other conditions have been satisfied or waived except for the expiration or termination of the waiting period under the HSR Act or the obtaining of required governmental consents or certain other conditions relating to regulatory matters, then VeriFone may, upon written notice to Hypercom, extend the termination date to November 30, 2011;

the adoption of the merger agreement by Hypercom s stockholders required by the merger agreement has not been obtained at a special meeting, or at any adjournment or postponement thereof, at which a vote upon the merger agreement was taken; or

any order permanently restraining, enjoining or otherwise prohibiting completion of the merger becomes final and non-appealable. However, the right to terminate in the above circumstances of VeriFone or Hypercom will not be available to any party that breached in any material respect its obligations under the merger agreement in any manner that shall have been a principal cause of or resulted in the occurrence of the failure of the merger to be consummated on or before the termination date.

The merger agreement may be terminated by VeriFone or Hypercom at any time prior to the effective time of the merger, whether before or after approval by Hypercom s stockholders, if there has been a breach of any representation, warranty, covenant or agreement made by the other party in the merger agreement, or any such representation and warranty becomes untrue after the date of the merger agreement, so that certain conditions to closing of the terminating party would not be satisfied and such breach or condition is not curable or, if curable, is not cured within 30 days after written notice thereof is given.

The merger agreement may be terminated at any time prior to the approval of the merger agreement by Hypercom s stockholders if Hypercom s board of directors authorizes Hypercom, subject to compliance with the terms of the merger agreement, to enter into an alternative acquisition agreement with respect to a superior proposal that has been obtained without violation of Hypercom s covenant not to solicit or negotiate other acquisition proposals, provided that Hypercom first pays to VeriFone the \$12.2 million termination fee.

The merger agreement may be terminated at any time prior to the effective time of the merger, by action of the board of directors of VeriFone, if:

at any time prior to the adoption of the merger agreement by Hypercom s stockholders, the board of directors of Hypercom has withdrawn, adversely modified, or has, following a public announcement of opposition to the merger or the other transactions contemplated by the merger agreement made by a person holding more than five percent of the Hypercom common stock, failed to reconfirm within ten business days from the receipt of a written request from VeriFone to do so, its approval or recommendation of the merger agreement (unless VeriFone has previously made two such requests and Hypercom has complied with those requests);

either (i) Hypercom failed to convene and hold the Hypercom stockholders meeting prior to the later of the date that is 45 days after the date as of which the SEC has confirmed that it has no additional comments on the proxy statement (subject to adjournment from time to time as permitted under the

merger agreement), or (ii) the minimum number of shares of Hypercom common stock necessary to constitute a quorum for the transaction of business has been represented in person or by proxy at the Hypercom stockholders meeting and Hypercom failed to take a vote of the stockholders on the merger and the merger agreement at such meeting;

a tender offer or exchange offer for outstanding Hypercom common stock has been publicly disclosed, other than by Hypercom or one of its affiliates, and the board of directors of Hypercom recommends that the stockholders of Hypercom tender their shares in such tender or exchange offer, or within ten business days after the commencement of such tender or exchange offer, Hypercom s board of directors fails to recommend that the stockholders of Hypercom not tender any shares into such tender or exchange offer;

the board of directors of Hypercom recommends to the stockholders of Hypercom an acquisition proposal other than the proposed merger with VeriFone; or

Hypercom enters into a letter of intent, agreement in principle, merger agreement, share purchase agreement, asset purchase agreement, share exchange agreement, option agreement or other similar contract relating to an acquisition proposal other than the proposed merger with VeriFone.

Termination Fee Payable by Hypercom

Hypercom has agreed to pay to VeriFone a termination fee equal to \$12.2 million if (i) prior to the Hypercom stockholder vote, the Hypercom board of directors approves a superior proposal and authorizes Hypercom to enter into an alternative acquisition agreement or (ii) VeriFone terminates the merger agreement after:

Hypercom s board of directors has withdrawn, adversely modified or failed to appropriately reconfirm its approval or recommendation of the merger agreement;

Hypercom has failed to convene and hold the Hypercom stockholders meeting prior to the later of the date that is 45 days after the date as of which the SEC has confirmed that it has no additional comments on the proxy statement;

Hypercom s board of directors has either recommended that stockholders tender their shares into a tender or exchange offer of another person or failed to recommend that Hypercom stockholders not tender any shares into such a tender or exchange offer;

Hypercom s board of directors has recommended to its stockholders another acquisition proposal;

Hypercom has failed to take a vote of Hypercom stockholders on the merger agreement at a stockholders meeting where a quorum for the transaction of business was present; or

Hypercom has entered into an alternative acquisition agreement with respect to a superior proposal.

In addition, if prior to termination there has been a bona fide acquisition proposal publicly disclosed, announced, commenced, submitted or made by a third-party, and the merger agreement is then terminated by either VeriFone or Hypercom under the provision allowing for termination following August 31, 2011 (or November 30, 2011, if extended) if the transaction has not yet been consummated, or under the provision allowing for termination if Hypercom stockholders fail to approve the merger agreement in a stockholder meeting at which a vote upon the merger agreement was taken, and within 12 months of such termination, Hypercom shall enter into a definitive agreement with respect to any other acquisition proposal that is ultimately consummated, then Hypercom shall pay VeriFone the termination fee.

Termination Fee Payable by VeriFone

So long as Hypercom has not breached in any material respect its obligations under the covenants of the merger agreement (to the extent such obligations or covenants relate to antitrust or competition law matters), if the merger agreement is terminated:

based on the entry of any order relating to antitrust or competition law, or

based on the failure to consummate the merger by the termination date, and at the time of such termination all the conditions to the closing (other than the conditions relating to regulatory consents, the absence of any order restraining completion of the merger in connection with antitrust or competition law, the accuracy of VeriFone s representations and warranties, and VeriFone s performance of its obligations under the merger agreement) have been satisfied or waived,

then VeriFone must pay to Hypercom a fee equal to \$28.4 million. However, if VeriFone elects to extend the termination date and such termination occurs after August 31, 2011, then VeriFone must pay to Hypercom a termination fee equal to \$30.4 million. This termination fee will not be payable by VeriFone if:

VeriFone has waived in writing any condition related to antitrust or competition matters and is prepared to consummate the merger notwithstanding the failure of such condition; and

the completion of the merger notwithstanding the failure of such condition would not reasonably be expected to subject any officer or director of Hypercom to personal liability.

Indemnification and Insurance

VeriFone has agreed that from and after the effective time of the merger, it will indemnify and hold harmless each present and former director and officer of Hypercom, when acting in such capacity, determined as of the effective time of the merger, against any costs or expenses, including reasonable attorneys fees, judgments, fines, losses, claims, damages or liabilities incurred in connection with any claim, action, suit, proceeding or investigation, whether civil, criminal, administrative or investigative, arising out of or pertaining to matters existing or occurring at or prior to the effective time of the merger.

However, any person to whom expenses are advanced must provide an undertaking to repay such advance if it is ultimately determined that such person is not entitled to indemnification.

VeriFone has agreed to maintain in effect Hypercom s current officers and directors insurance for six years after the effective time of the merger, subject to certain limitations.

Third-Party Beneficiaries

The merger agreement generally is not intended to confer upon any person other than VeriFone, Honey Acquisition Co. and Hypercom benefits or remedies, other than the rights to indemnification and insurance described above.

Other Material Agreements Relating to the Merger

Support Agreement

Contemporaneously with the execution and delivery of the merger agreement, VeriFone has entered into a support agreement with FP Hypercom Holdco, LLC and Francisco Partners II, L.P., pursuant to which FP Hypercom Holdco, LLC agreed, to the extent that FP Hypercom Holdco, LLC exercises its warrant, to deliver, promptly upon VeriFone s request, an irrevocable proxy to VeriFone to vote any shares of Hypercom common stock owned by it:

in favor of adoption of the merger agreement and approval of the merger and the transactions contemplated by the merger agreement;

against any action or agreement that would compete with, materially impede or interfere with the merger or inhibit its timely completion.

In addition, FP Hypercom Holdco, LLC and Francisco Partners II, L.P. agreed that any shares of Hypercom common stock would be subject to the support agreement and that prior to stockholder approval of the merger, they would not transfer any Hypercom shares of common stock held by them unless the transferee becomes subject to the support agreement.

Pursuant to the support agreement, Francisco Partners II, L.P. agreed to deliver to VeriFone at the closing of the merger a letter stating that upon the satisfaction of certain conditions, all of Hypercom s obligations under the credit agreement between Hypercom and Francisco Partners II, L.P. will be released and satisfied in full.

Unaudited Pro Forma Condensed Combined Financial Information

The following unaudited pro forma condensed combined financial information is based on the historical financial statements of VeriFone and Hypercom after giving effect to the proposed merger between VeriFone and Hypercom and the assumptions and adjustments described in the accompanying notes to the unaudited pro forma condensed combined financial information. The pro forma condensed combined financial information also reflects the divestiture of Hypercom s U.S. business, which is expected to be disposed of in connection with the merger.

VeriFone s fiscal year ends on October 31 while Hypercom s fiscal year ends on December 31. The unaudited pro forma condensed combined balance sheet data as of October 31, 2010 is based on the historical balance sheet of VeriFone as of October 31, 2010 and of Hypercom as of September 30, 2010 and has been prepared to reflect the merger as if it had occurred on October 31, 2010. The unaudited pro forma condensed combined statement of operations data for the year ended October 31, 2010 combines the results of operations of VeriFone for the year ended October 31, 2010 and of Hypercom for the twelve months ended September 30, 2010, as though the merger had occurred on November 1, 2009, the first day of VeriFone s 2010 fiscal year.

The unaudited pro forma condensed combined financial information assumes that, at closing, (i) each outstanding share of Hypercom common stock will be converted into 0.23 shares of VeriFone common stock, (ii) each restricted stock award (RSA) will be vested and converted into 0.23 shares of VeriFone common stock, (iii) each outstanding Hypercom warrant will be converted into a warrant to acquire a number of shares of VeriFone common stock equal to the product of (x) the number of shares of Hypercom common stock subject to such warrant and (y) the exchange ratio of 0.23, at an exercise price per share (rounded up to the nearest whole cent) equal to the exercise price per share of Hypercom common stock for such warrant immediately prior to the effective time of the merger divided by 0.23, and (iv) each Hypercom stock option will be converted into an option to purchase a number of shares of VeriFone common stock equal to the product of (x) the number of shares of Hypercom common stock subject to such option immediately prior to the effective time of the merger and (y) 0.23, at an exercise price equal to the merger divided by 0.23.

The estimated number of shares of VeriFone common stock to be issued in the proposed merger is approximately 13 million shares based on the number of outstanding shares of Hypercom common stock as of September 30, 2010, assuming that none of the 5,049,457 Hypercom stock options and 10,544,000 Hypercom warrants outstanding as of September 30, 2010 will be exercised prior to closing. Any stock option or warrant exercised prior to closing will increase the number of VeriFone stock issued by the product of 0.23 times the number of shares issued as a result of option and warrant exercises prior to the closing.

The preliminary allocation of the purchase price used in the unaudited pro forma condensed combined financial information is based on preliminary estimates and currently available information. These assumptions and estimates, which cannot be finalized until the consummation of the merger, will be revised as additional information becomes available, upon completion of the merger and finalization of the valuation of Hypercom s assets and liabilities.

Certain reclassification adjustments to the pro forma financial information were made to the Hypercom pro forma financial information to conform to VeriFone s financial statement presentation.

The unaudited pro forma condensed combined financial information should be read in conjunction with (i) VeriFone s historical consolidated financial statements and related notes contained in VeriFone s Annual Report on Form 10-K for the year ended October 31, 2010 and (ii) Hypercom s historical consolidated financial statements and related notes contained in Hypercom s Annual Report on Form 10-K, as amended, for the year ended December 31, 2009, and Hypercom s unaudited consolidated financial statements contained in Hypercom s Quarterly Reports on Form 10-Q for the periods ended March 31, 2010, June 30, 2010 and September 30, 2010, which are incorporated by reference into this proxy statement/prospectus. See Where You Can Find More Information on page 117.

UNAUDITED PRO FORMA CONDENSED COMBINED BALANCE SHEET AS OF OCTOBER 31, 2010

(DOLLARS IN THOUSANDS)

			Adiust	Adjustments	
			Purchase Price	Divestiture of	Pro forma
Arresta	VeriFone	Hypercom	Allocation(1)	US Operations(2)	Combined
Assets					
Current assets:	\$ 445,137	\$ 38,521	\$ (70,516)(G)	\$	\$ 413,142
Cash and cash equivalents Accounts receivable, net of allowances			\$ (70,310)(0)	¢	
	132,988	103,208			236,196
Current portion of net investment in sales-type		4,292			4 202
leases Inventories	111,901	,	9 447(D)	(3,528)	4,292 151,023
Deferred tax assets, net	10,953	34,203 1,230	8,447(D)	(5,528)	
	,				12,183
Prepaid expenses and other current assets Assets held for sale	60,112	11,522		4,074	71,634
Assets held for sale		5,821		4,074	9,895
Total current assets	761,091	198,797	(62,069)	546	898,365
Property, plant, and equipment, net	46,007	23,434	(0_,000)	(546)	68,895
Net investment in sales-type leases	10,007	6,530		(510)	6,530
Purchased intangible assets, net	50,121	44,520	195,480(C)		290,121
Goodwill	169,322	26,638	251,400(B)		447,360
Deferred tax assets, net	9,933	20,050	231,100(B)		9,933
Debt issuance costs, net	5,069				5,069
Other assets	33,783	9,100			42,883
	55,765	9,100			42,005
Total assets	\$ 1,075,326	\$ 309,019	\$ 384,811	\$	\$ 1,769,156
Liabilities and Equity					
Current liabilities:					
Accounts payable	\$ 64.016	\$ 51,865	\$	\$	\$ 115,881
Income taxes payable	651	4,957			5,608
Accrued compensation	27,316	14,135		(1,156)	40,295
Accrued warranty	10,898	5,411			16,309
Deferred revenue, net	55,264	15,518	(4,218)(E)	(1,709)	64,855
Deferred tax liabilities	400	23			423
Other current liabilities	95,981	34,811			130,792
Liabilities held for sale		1,419		6,086	7,505
Current portion of long-term debt	5,280	, -		- ,	5,280
Total current liabilities	259,806	128,139	(4,218)	3,221	386,948
Accrued warranty	1,849		(.,)	-,	1,849
Deferred revenue, net	22,344				22,344
Long-term debt, less current portion	468,231	61,691	(61,691)(F)(G)		468,231
Deferred tax liabilities	62,081	11,900	68,800(H)		142,781
Other long-term liabilities	53,552	16,648	00,000(11)	(3,221)	66,979
		10,010		(3,221)	00,717
Total liabilities	867,863	218,378	2,891		1,089,132
Commitments and contingencies					
Redeemable noncontrolling interest	866				866
Stockholders equity:					
Preferred Stock					
Common Stock	868	59	(59)(A)		868
Additional paid-in-capital	763,212	277,144	195,417(A)		1,235,773
para in express		_ / / ,1			1,200,110

Edgar Filing: VERIFONE SYSTEMS, INC. - Form S-4

Accumulated deficit	(551,460)	(142,519)	142,519(A)		(551,460)
Treasury Stock		(22,911)	22,911(A)		
Accumulated other comprehensive loss	(6,595)	(21,132)	21,132(A)		(6,595)
Total stockholders equity	206,025	90,641	381,920		678,586
Noncontrolling interests	572				572
-					
Total equity	206.597	90.641	381,920		679,158
1 5	,	,	,		,
Total liabilities and equity	\$ 1,075,326	\$ 309.019	\$ 384,811	\$	\$ 1,769,156
rour nuomnes una equity	φ1,07 <i>5</i> ,520	φ <i>50</i> ,017	\$ 50 1,011	Ψ	ψ ¹ ,709,150

UNAUDITED PRO FORMA CONDENSED COMBINED STATEMENT OF OPERATIONS

FOR THE YEAR ENDED OCTOBER 31, 2010

(DOLLARS IN THOUSANDS, EXCEPT PER SHARE DATA)

			Adjus Divestiture of	stments Combined	Pro forma
	VeriFone	Hypercom	US Operations(2)	Adjustments(3)	Combined
Net revenues:	vern one	nypercom	es operations(2)	Augustinents(5)	combined
System Solutions	\$ 828,949	\$ 344,537	\$ (53,996)		\$ 1,119,490
Services	172,588	100,614	(7,520)		265,682
	1,2,000	100,011	(1,5=0)		200,002
Total net revenues	1,001,537	445,151	(61,516)		1,385,172
Cost of net revenues:					
System Solutions	530,821	227,144	(35,322)	8,087(I)	730,730
Services	100,404	74,269	(3,380)		171,293
Total cost of net revenues	631,225	301,413	(38,702)	8,087	902,023
Gross profit	370,312	143,738	(22,814)	(8,087)	483,149
Operating expenses:					
Research and development	74,227	46,249	(6,247)		114,229
Selling, general and administrative	179,037	76,929	(11,670)		244,296
Amortization of purchased intangible assets	14,624	5,730		16,770(I)	37,124
Gain on sale of real property		(1,515)			(1,515)
Total operating expenses	267,888	127,393	(17,917)	16,770	394,134
Operating income	102,424	16,345	(4,897)	(24,857)	89,015
Interest expense	(28,344)	(12,333)		11,809(J)	(28,868)
Interest income	1,278	393			1,671
Other income, net	2,887	17			2,904
	, ,				
Income before income taxes	78,245	4,422	(4,897)	(13,048)	64,722
Benefit from income taxes	20,582	391	(1,0)7)	2,610(M)	23,583
	20,002	0,71		2,010(11)	20,000
Net income from continued operations	\$ 98,827	\$ 4,813	\$ (4,897)	\$ (10,438)	\$ 88,305
Net meome from continued operations	\$ 90,027	\$ 4,015	\$ (4,097)	\$ (10,458)	\$ 88,505
Net income from continued operations per					
share:					
Basic	\$ 1.16	\$ 0.09			\$ 0.90
Diluted	\$ 1.13	\$ 0.09			\$ 0.88
Weighted average number of shares used in	φ 1.15	φ 0.09			φ 0.00
computing net income per share:					
Basic	85,203	53,787		(41,075)(N)	97,915
Diluted	87,785	54,768		(41,075)(N) (42,056)(N)	100,497
Diraco	07,705	54,700		(72,030)(11)	100,797

NOTES TO THE UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL STATEMENTS

Basis of Presentation

On November 17, 2010, VeriFone and Hypercom entered into a definitive merger agreement that will result in Hypercom becoming a wholly owned subsidiary of VeriFone. The transaction is subject to a number of conditions. See the section of this proxy statement/prospectus titled The Merger Agreement Conditions to Completion of the Merger for a summary description of these conditions.

The unaudited pro forma condensed combined financial information reflects the effects of VeriFone acquiring Hypercom for an aggregate preliminary estimated consideration of \$543 million as calculated below (in millions):

Value of common stock to be issued by VeriFone for Hypercom common stock	\$ 405
Estimated fair value of Hypercom s debt that needs to be repaid at closing	71
Value of warrants to be issued by VeriFone for Hypercom warrants	45
Value of stock options to be issued by VeriFone for Hypercom stock options	22
Total	\$ 543

The preliminary estimated consideration was calculated with the following significant assumptions:

- 1) Hypercom will have 55,270,132 shares of common stock outstanding at the closing of the merger consistent with the number of Hypercom shares outstanding as of September 30, 2010.
- 2) The fair value of VeriFone common stock at closing will be \$31.84, which was VeriFone s closing common stock price on November 16, 2010, the day before VeriFone and Hypercom announced the merger transaction on November 17, 2010.
- 3) The merger agreement between VeriFone and Hypercom did not require the repayment of Hypercom s debt balance at closing, however, the debt agreement classifies the merger as a default event and requires the immediate repayment of the debt. Therefore, VeriFone has included the repayment of debt in the purchase price and has assumed that the principal debt balance outstanding at closing is the same as the balance outstanding as of September 30, 2010.
- 4) Hypercom will have 1,206,022 RSAs outstanding at the closing of the merger consistent with the number of Hypercom RSAs outstanding as of September 30, 2010.
- 5) Hypercom has change in control clauses in its stock option plans that allow for the full acceleration of vesting of all Hypercom employee stock awards at closing.
- 6) Hypercom will have 10,544,000 warrants with an exercise price of \$5 outstanding at the closing of the merger consistent with the number of Hypercom warrants outstanding as of September 30, 2010. The warrants will be converted to 2,425,120 VeriFone warrants with an exercise price of \$21.74 and will expire on April 1, 2013. VeriFone used the Black Scholes valuation model to calculate the fair value of these warrants. The fair value of each warrant was estimated at \$18.66. The major assumptions VeriFone used were as follows:

- expected term of 2.37 years;
- volatility of 84.77%, which was based on VeriFone s historic stock volatility over the past 2.37 years;
- strike price of \$21.74; and
- market price of \$31.84.

- 7) Hypercom will have 5,049,457 stock options outstanding at the closing of the merger consistent with the number of Hypercom s stock options outstanding as of September 30, 2010. VeriFone used the weighted average exercise price of \$4.53 and a weighted average term of 7.37 years for all the outstanding stock options consistent with the weighted average exercise price and weighted average term of Hypercom s stock options that were outstanding as of September 30, 2010. The Hypercom stock options will be converted into 1,161,375 VeriFone stock options with an average strike price of \$19.70 and an average term of 7.37 years. VeriFone used the Black Scholes valuation model to calculate the fair value of the options. The fair value of each option was estimated at \$19.42. The major assumptions VeriFone used were as follows:
 - expected term of 2.37 years;
 - volatility of 84.77% which was based on VeriFone s historic stock volatility over the past 2.37 years
 - strike price of \$19.70; and
 - market price of \$31.84.
- 8) VeriFone has assumed that there was no incremental fair value to Hypercom option holders on the exchange given the current terms and all the options will be fully vested at the time of the close of the transaction due to Hypercom s initiated change in control clauses. As a result, the full amount of the fair value of the options will be assigned to the purchase consideration.

The final purchase consideration at closing is expected to change due to expected changes in the market price of VeriFone s common stock. In addition, the exact number of employee stock awards outstanding will change due to exercises and cancellations as a result of Hypercom staff turnover which may occur before the transaction is closed.

The merger will be accounted for as a business combination using the acquisition method of accounting and, accordingly, will result in the recognition of assets acquired and liabilities assumed at fair value. The preliminary allocation of the purchase price used in the unaudited pro forma condensed combined financial information is based on preliminary estimates and currently available information. These assumptions and estimates, some of which cannot be finalized until the consummation of the merger, will be revised as additional information becomes available, upon consummation of the merger and finalization of the valuation of Hypercom s assets and liabilities. The final determination of the allocation of the purchase price will be based on the actual intangible assets, net tangible assets and in-process research and development of Hypercom existing as of the closing date of the merger.

The unaudited pro forma condensed combined financial information included herein does not give effect to any potential cost reductions or other operating efficiencies that could result from the merger, including but not limited to, those associated with potential (i) reductions of corporate overhead, (ii) eliminations of duplicate functions and (iii) increased operational efficiencies through the adoption of best practices and capabilities from each company. The unaudited pro forma condensed combined financial information is not intended to represent what VeriFone s financial position or results of operations would actually have been if the merger had occurred on those dates. Since VeriFone and Hypercom were not under common control or management for any period presented, the unaudited pro forma condensed combined financial information does not include any adjustments for liabilities resulting from integration planning. Management of VeriFone is in the process of assessing the costs associated with integration.

Notes to Unaudited Pro Forma Condensed Combined Financial Information

(1) Basis of Preliminary Purchase Price Allocation

The following preliminary allocation of the Hypercom purchase price is based on VeriFone s preliminary estimates of the fair value of the tangible and intangible assets and liabilities of Hypercom as of September 30, 2010. The final determination of the allocation of the purchase price will be based on the fair value of such assets and liabilities as of the actual consummation date of the acquisition and will be completed after the acquisition is consummated. Such final determination of the purchase price allocation may be different than the preliminary used in these pro forma condensed combined financial statements.

The estimated purchase price of Hypercom (as calculated in the manner described above) is allocated to the assets to be acquired and liabilities to be assumed based on the following preliminary basis (in thousands):

Net assets of Hypercom as of September 30, 2010	\$ 90,641
Add: Debt to be repaid at closing	70,516
Adjusted net assets of Hypercom at closing	161,157
Acquisition method adjustments:	
Less: existing intangible assets, net	(44,520)
Add: estimated developed technology intangible asset	60,000
Add: estimated customer relationships intangible asset	150,000
Add: estimated trademarks and trade names intangible asset	30,000
Less: existing goodwill	(26,638)
Add: estimated goodwill	278,038
Add: mark up inventory to estimated fair value	8,447
Add: reduce deferred revenue to estimated fair value	4,218
Less: estimated deferred tax liability assigned to acquired intangibles	(68,800)
Less: discounts on warrants netted with long-term debt	(8,825)
Total estimated purchase price	\$ 543,077

Other considerations in the preliminary allocations of the estimated purchase price of Hypercom include the following:

- Our preliminary valuation used to allocate the purchase price uses a third-party market participant view and assumes there are no synergies unique to VeriFone. If there were any synergies unique to VeriFone, then a higher portion of the purchase consideration would be allocated to goodwill;
- 2) The preliminary valuation relies on publicly available information and general statistics from prior acquisitions within our industry;
- 3) Accounts receivable book value approximates fair value;
- 4) The effective interest rate of the net investment in sales-type leases was assumed to approximate current market interest rate and therefore no fair value adjustments were required;
- 5) Operating leases for facilities on average are assumed to be at market prices;

Edgar Filing: VERIFONE SYSTEMS, INC. - Form S-4

- 6) The property, plant and equipment s current carrying value approximates the fair value of the assets;
- 7) The location of the intangibles determine the tax rate used to determine the deferred tax liabilities, VeriFone assumed that the developed technology and trademarks and trade name will be located in the United States and the customer relationships intangibles will be in locations where Hypercom generates their revenues; and
- 8) Carrying value of debt is assumed to approximate fair value.

(2) Divestiture of US Operations Anticipated Asset Dispositions to Address Potential Competitive Concerns

To address potential competitive concerns, VeriFone and Hypercom have agreed to sell or otherwise dispose of, businesses or assets that, subject to certain exceptions, in the aggregate produced aggregate gross revenues in an amount not to exceed \$124 million for VeriFone or Hypercom during the 2009 fiscal year.

VeriFone has made general assumptions as to the scope of Hypercom s businesses that will be included in the sale; however, the actual assets to be sold will depend on negotiations with any buyer and regulatory authorities and may be materially different.

The divestiture of US operations is reflected in the unaudited pro forma condensed balance sheet as an adjustment that classifies those assets and liabilities that VeriFone and Hypercom intend to divest as Assets held for sale and Liabilities held for sale. No pro forma adjustments have been made to reflect any anticipated gain or loss from the disposition of these assets, as the method of disposition and sales proceeds are not known, but any such effect could be material with respect to earnings in any given period. However, it is not expected that the dispositions will materially affect the combined company s financial position or negatively affect the combined company s liquidity.

The unaudited pro forma condensed combined statements of income for the fiscal year ended October 31, 2010 include adjustments to eliminate Hypercom s operating results attributable to the divestiture of US operations. Additionally, no pro forma adjustments have been made to reflect any earnings benefits from the reinvestment of any proceeds which might be recovered, or the reduction of debt which may arise as a consequence of the asset dispositions.

(3) Pro Forma Combined Adjustments

The following pro forma combined adjustments have been reflected in the unaudited pro forma condensed combined financial information. These adjustments give effect to pro forma events that are (i) directly attributable to the merger, (ii) factually supportable and (iii) with respect to the statements of operations, expected to have a continuing impact on the combined company.

All adjustments are based on current assumptions and are subject to change upon completion of the final purchase price allocation at the merger closing date.

VeriFone and Hypercom s consolidated financial statements were prepared in conformity with accounting principles generally accepted in the United States and the accounting policies of VeriFone and Hypercom are substantially comparable. Thus, no accounting policy conformance adjustments have been made to the unaudited pro forma condensed combined financial information.

Balance Sheet Adjustments

(A) To reflect the fair value of the common shares, stock options, restricted stock awards and warrants issued to Hypercom stockholders, employees and warrant holders by VeriFone as consideration to acquire Hypercom of \$473 million and the elimination of Hypercom s stockholders equity balances.

(B) To reflect the establishment of goodwill of \$278 million, estimated as a result of the preliminary purchase price allocation described in Note (1). This adjustment also includes the elimination of the carrying value of Hypercom s existing goodwill that has been adjusted to fair value in the purchase price allocation process.

(C) To reflect the preliminary estimates of fair value of the identifiable intangible assets of Hypercom which were estimated based on preliminary fair value review and benchmarked against recent comparable acquisitions

of related industries. These allocations may materially change once a final appraisal is performed. The estimated useful life of the developed technology, customer relationships and trademarks and trade names were assumed to be 6 years, 8 years and 8 years, respectively. This adjustment also includes the elimination of the carrying value of Hypercom s existing intangible assets that has been adjusted to fair value in the purchase price allocation process.

	Increase (Decrease) (in thousands)	
Establish developed technology asset	\$ 60,000	
Establish customer relationships asset	150,000	
Establish trademarks and trade names asset	30,000	
Elimination of intangible assets from prior Hypercom acquisitions	(44,520)	
Net pro forma combined adjustment	\$ 195,480	

(D) To increase inventory to estimated fair value. The estimated fair value is calculated using the estimated selling price less a reasonable margin.

(E) To reflect a decrease in deferred revenue as required by acquisition accounting. The estimated fair value represents the cost to provide services under contractual obligations plus a reasonable margin.

(F) To eliminate the remaining value of \$8.8 million assigned to the 10.5 million warrants that were issued by Hypercom in connection with issuing debt. The fair value of the warrants was recorded as debt issuance costs and resulted in an unamortized discount to the debt. In purchase accounting, the debt discount has been eliminated and the debt is recorded at its estimated fair value. For purposes of the condensed combined pro forma financial information it has been assumed that the carrying value of the Hypercom debt approximates fair value.

(G) To reflect the extinguishment of Hypercom s \$70.5 million outstanding debt. The merger agreement does not require the repayment of debt. However, Hypercom s credit agreement defines a change of control event as an event of default which requires immediate repayment of the debt; therefore VeriFone has assumed the repayment of the debt in full at closing.

(H) Adjustment to reflect the expected deferred tax liability for the intangible assets. VeriFone has assumed the developed technology and trademarks and trade name will use its U.S. tax rate of 38% and for the customer relationship intangible asset a foreign tax rate of 31%.

Income Statement Adjustments

(I) To reflect amortization expense associated with the developed technology, customer relationships and trademarks and trade names in Note (C) above assuming an estimated life of 6 years, 8 years and 8 years, respectively. These expenses are offset by the elimination of the amortization expense associated with Hypercom s acquisitions.

	Ended Oct	al Year ober 31, 2010 ousands)
Amortization of developed technology	\$	10,000
Amortization of customer relationships		18,750
Amortization of trademarks and trade names		3,750
Hypercom acquisitions		(7,643)
	\$	24,857
Amortized to cost of net revenues System Solutions	\$	8,087
Amortized to operating expenses	\$	16,770

(J) To eliminate the amortization of the deferred warrant costs to interest expense associated with the warrants described in Note (F) above and interest expense associated with the loan as described in Note (G). The effect was \$4.9 million and \$6.9 million, respectively, for the fiscal year ended October 31, 2010.

(K) In relation to the inventory and deferred revenue fair value adjustments described above in Note (D) and Note (E), the items represent one-time adjustments directly attributed to the merger that will not have recurring significant ongoing impact in excess of one year. Therefore, the impact on gross margins due to the write-up of inventory and the reduction in revenues due to the fair value adjustment made to deferred revenue are not reflected in the unaudited pro forma condensed combined statement of operations.

(L) Treatment of stock-based compensation expense: Although all of the employee awards of Hypercom will be fully vested, VeriFone did not adjust for the stock-based compensation expense recorded by Hypercom because VeriFone will grant new options to Hypercom s employees in the ordinary course of business.

(M) To reflect the tax effects of adjustments using an estimated effective income tax rate of 20% for the fiscal year ended October 31, 2010.

(N) To reflect (i) the elimination of Hypercom s basic and diluted shares outstanding, net of (ii) the assumed issuance of VeriFone common stock as a result of the transaction calculated by multiplying 55,270,132 of Hypercom common stock outstanding as of September 30, 2010 by the 0.23 exchange ratio and (iii) no dilution from the new VeriFone stock options issued in exchange for Hypercom stock options. For each option that is exercised from September 30, 2010, the average number of shares used in the calculation of basic and diluted earnings per share will increase by 0.23 multiplied by the number of shares subject to that option.

(O) The unaudited pro forma condensed combined financial information has not been adjusted for merger costs expected to be incurred by VeriFone as these costs are directly attributed to the merger and will not have a continuing impact in excess of one year. These merger costs are not readily predictable but are anticipated to be significant.

Description of VeriFone Capital Stock

The following description of material terms of the capital stock of VeriFone does not purport to be complete and is qualified in its entirety by reference to the amended and restated certificate of incorporation and bylaws of VeriFone, which documents are incorporated by reference as exhibits to the registration statement of which this proxy statement/prospectus is a part, and to the applicable provisions of the Delaware General Corporation Law.

The authorized capital stock of VeriFone consists of 200,000,000 shares of common stock, par value \$0.01 per share, and 10,000,000 shares of preferred stock, par value \$0.01 per share. As of December 17, 2010, there were 87,532,653 shares of common stock, including restricted stock, outstanding and no shares of preferred stock outstanding.

Common Stock

VeriFone is authorized to issue one class of common stock. Stockholders are entitled to one vote for each share of common stock held of record on all matters on which stockholders are entitled or permitted to vote. The holders of VeriFone common stock do not have cumulative voting rights in the election of directors. As a result, holders of a majority of the shares of the common stock voting for the election of directors can elect all the directors standing for election. Holders of VeriFone common stock will be entitled to receive dividends, if any, out of legally available funds when and if declared from time to time by the board of directors. In the event of VeriFone s liquidation, dissolution or winding-up, the holders of VeriFone common stock are entitled to share ratably in all assets remaining after payment of liabilities, subject to the rights of any then-outstanding preferred stock. VeriFone common stock has no preemptive, subscription or conversion rights, and there are no redemption or sinking fund provisions applicable to VeriFone common stock. The rights, preferences and privileges of holders of VeriFone common stock are subject to, and may be adversely affected by, the rights of holders of shares of any series of preferred stock that VeriFone may designate and issue in the future. All outstanding shares of VeriFone common stock are fully paid and nonassessable and the shares of common stock offered hereby will be fully paid and nonassessable.

Preferred Stock

The board of directors of VeriFone may authorize the issuance of preferred stock with voting or conversion rights that could adversely affect the voting power or other rights of VeriFone common stockholders. The issuance of preferred stock, while providing flexibility in connection with possible acquisitions and other corporate purposes, could have the effect of delaying or preventing a change in control, and may cause the market price of VeriFone common stock to decline or impair the voting and other rights of the holders of VeriFone common stock. VeriFone has no current plans to issue any shares of preferred stock.

No Stockholder Rights Plan

VeriFone currently does not have a stockholder rights plan.

Transfer Agent and Registrar

The transfer agent and registrar for VeriFone common stock is Computershare Limited.

Comparison of Rights of VeriFone Stockholders and Hypercom Stockholders

VeriFone and Hypercom are incorporated under the laws of the State of Delaware. If the merger is completed, Hypercom stockholders will exchange their respective shares of Hypercom common stock for VeriFone common stock, which trades on the New York Stock Exchange. The following is a summary comparison of material differences between the rights of a VeriFone stockholder and a Hypercom stockholder arising from the differences between the governing documents of the respective companies.

The following summary does not purport to be a complete statement of the rights of holders of VeriFone common stock or Hypercom common stock under the applicable provisions of Delaware law and VeriFone s or Hypercom s certificate of incorporation and bylaws or a complete description of the specific provisions referred to herein. This summary contains a list of the material differences but is not meant to be relied upon as an exhaustive list or a detailed description of the provisions discussed and is qualified in its entirety by reference to the laws of Delaware and the United States, VeriFone s certificate of incorporation and bylaws and Hypercom s certificate of incorporation and bylaws.

Copies of such governing corporate instruments of VeriFone and Hypercom are available, without charge, to any person, including any beneficial owner to whom this proxy statement/prospectus is delivered, by following the instructions listed under Where You Can Find More Information.

Summary of Material Differences Between the Rights of VeriFone Stockholders and the Rights of Hypercom Stockholders

Number of Directors	VeriFone Stockholder Rights The bylaws of VeriFone provide that the number of directors of VeriFone will be determined from time to time by the directors then serving in office.	Hypercom Stockholder Rights The bylaws of Hypercom provide that the number of directors of Hypercom cannot be less than one or more than nine, with the exact number of directors to be determined from time to time by the directors then serving in office.
Term and Classes of Directors	VeriFone s board of directors is not classified into more than one class. Each director serves until the next annual meeting and until his or her successor is elected and qualified.	Hypercom s board of directors is classified into two classes, as nearly equal in number as possible, with each class holding office for two years.
Removal of Directors	The DGCL provides that a director or directors may be removed, with or without cause, by the holders of a majority in voting power of the shares then entitled to vote on the election of directors.	Any director, or the entire board of directors, may be removed, with cause, by the holders of a majority of the outstanding voting stock. No director may be removed without cause.
Vacancies on the Board	The bylaws of VeriFone provide that vacancies on the board of directors created by an increase in the number of directors will be	The bylaws of Hypercom provide that any vacancy or newly created directorship may be filled by a majority of the

	VeriFone Stockholder Rights filled by a majority vote of the directors then remaining in office, and a successor elected to fill the vacancy will serve until the next annual election of the class in which that director served.	Hypercom Stockholder Rights directors then in office, although less than a quorum, or by a sole remaining director.
Stockholder Meetings	The bylaws of VeriFone provide that an annual meeting of the stockholders for the election of directors and for the transaction of such other business will be held at such date, time and place as may be designated by the board of directors from time to time.	The bylaws of Hypercom provide that the annual meeting of stockholders shall be held on such date and at such time as shall be designated from time to time by the board of directors and stated in the notice of meeting.
	The certificate of incorporation and bylaws of VeriFone provide that special meetings of the stockholders may be called at any time, either by the board of directors or by the chairman of the board or the president. This request must specify the place, date, hour and purpose of the meeting.	Hypercom s bylaws provide that special stockholder meetings may be called only by the board of directors pursuant to a resolution approved by a majority of the total number of directors then constituting the entire board, without regard to any vacancies on the board, or by the chairman or a vice-chairman.
Business Combinations; Interested Stockholder Transactions; Anti-Takeover Effects	The DGCL contains a business combination statute (DGCL § 203) that prohibits some transactions once an acquiror has gained a significant holding in the corporation. The DGCL generally prohibits business combinations, including mergers, sales and leases of assets, issuances of securities and similar transactions by a corporation or a subsidiary with an interested stockholder (defined as including the beneficial owner of 15 percent or more of a corporation s voting shares), within three years after the person or entity becomes an interested stockholder, unless:	Because Hypercom has not elected to opt out of DGCL § 203, this provision governs business combinations with interested stockholders.
	the board of directors has approved, before the acquisition date, either the business	

111

combination or the

VeriFone Stockholder Rights transaction that resulted in the person becoming an interested stockholder;

upon completion of the transaction that resulted in the person becoming an interested stockholder, the person owns at least 85 percent of the corporation s voting shares, excluding shares owned by directors who are officers and shares owned by employee stock plans in which participants do not have the right to determine confidentially whether shares will be tendered in a tender or exchange offer; or

after the person or entity becomes an interested stockholder, the business combination is approved by the board of directors and authorized by the vote of at least 66 2/3 percent of the outstanding voting shares not owned by the interested stockholder at an annual or special meeting of stockholders and not by written consent.

The amended and restated certificate of incorporation has elected not to be governed by this statute.

VeriFone currently does not have a stockholder rights plan.

Hypercom Stockholder Rights

Hypercom entered into a stockholder rights plan on September 29, 2010 and amended the plan on November 17, 2010 in connection with the merger agreement. Pursuant to Hypercom s stockholder rights plan, holders of Hypercom common stock received a dividend of one preferred share purchase right per outstanding

112

Stockholder Rights Plan

VeriFone Stockholder Rights

Amendment of Certificate of Incorporation and Bylaws

VeriFone s certificate of incorporation provides that the affirmative vote of the holders of at least two-thirds of the outstanding shares of capital stock of the corporation entitled to vote at an election of directors shall be required to amend, alter, change or repeal any provision contained in the certificate of incorporation.

VeriFone s bylaws provide that they may be amended or repealed or new bylaws adopted only in accordance with VeriFone s certificate of incorporation. The certificate of incorporation provides that the board of directors is authorized to make, alter, amend, change or repeal the bylaws, or the bylaws may be amended by the affirmative vote of the holders of at least two-thirds of the outstanding shares of capital stock entitled to vote at an election of directors.

Hypercom Stockholder Rights share of Hypercom common stock. Each share purchase right entitles the registered holder to purchase from Hypercom one one-thousandth of a share of Series A Junior Participating Preferred Stock, par value \$0.001 per share, of Hypercom at a price of \$25.02 per one one-thousandth of a share of preferred stock, subject to adjustment. The preferred share purchase rights have certain anti-takeover effects and under certain circumstances could cause substantial dilution to a person or group who attempts to acquire Hypercom on terms not approved by its board of directors. The preferred share purchase rights should not interfere with any merger or other business combination approved by Hypercom s board of directors.

Hypercom s certificate of incorporation provides that the certificate of incorporation may be amended in accordance with the DGCL. The DGCL provides that the right to amend, alter, change or repeal the certificate of incorporation requires an affirmative vote of a majority of the voting power of the outstanding shares entitled to vote and a majority of the voting power of the outstanding shares of each class entitled to vote as a class on the amendment.

Hypercom s bylaws provide that they may be adopted, amended or repealed by the board of directors, or the bylaws may be amended or repealed by the affirmative vote of the holders of at least two-thirds of the shares of voting stock then entitled to vote.

Nominations to the Board of Directors

Stockholder Proposals

VeriFone Stockholder Rights

VeriFone s bylaws contain provisions requiring a stockholder to give advance written notice to VeriFone of a director nomination. The written notice, which must include information specified in the bylaws, must be received, in the case of an annual meeting, not less than 90 days prior to the date of the anniversary of the previous year s annual meeting; provided, however, that in the event the annual meeting is scheduled to be held on a date more than 30 days prior to or delayed by more than 60 days after such anniversary date, notice by the stockholder in order to be timely must be so received not later than the later of the close of business 90 days prior to such annual meeting or the tenth day following the day on which such notice of the date of the annual meeting was mailed or such public disclosure of the date of the annual meeting was made.

In the case of a special meeting called for the purpose of electing directors, a stockholder s notice must be received not later than the close of business on the tenth day following the day on which notice of the date of the special meeting was mailed or public disclosure of the date of the special meeting was made, whichever first occurs.

VeriFone s bylaws contain provisions requiring a stockholder to give advance written notice to VeriFone of a proposal in order to have it considered at the stockholder meeting. The written notice, which must include information specified in the bylaws, must be received not less than 90 days prior to the date of the anniversary of the previous year s annual meeting; provided, however, that in the event the

114

Hypercom Stockholder Rights Hypercom s bylaws contain provisions requiring a stockholder to give advance written notice to Hypercom of a director nomination. The written notice, which must include information specified in the bylaws, must be received, in the case of an annual meeting, by the Secretary not earlier than 120 and not later than 90 days prior to the anniversary date of the immediately preceding annual meeting, or, in the case of a special meeting, within seven days after notice of the meeting is mailed to stockholders.

Unless otherwise required by law, if the stockholder (or a qualified representative of the stockholder) does not appear at the annual or special meeting of stockholders of the corporation to present a nomination, such nomination shall be disregarded, notwithstanding that proxies in respect of such vote may have been received by the corporation.

Hypercom s bylaws contain provisions requiring a stockholder to give advance written notice to Hypercom of a proposal in order to have it considered at the stockholder meeting. The written notice, which must include information specified in the bylaws, must be received by the Secretary not earlier than 120 and not later than 90 days prior to the anniversary date of the

VeriFone Stockholder Rights

annual meeting is scheduled to be held on a date more than 30 days prior to or delayed by more than 60 days after such anniversary date, notice by the stockholder in order to be timely must be so received not later than the later of the close of business 90 days prior to such annual meeting or the tenth day following the day on which such notice of the date of the annual meeting was mailed or such public disclosure of the date of the annual meeting was made. Hypercom Stockholder Rights immediately preceding annual meeting, provided, however, that if the annual meeting is called for a date that is not within 30 days before or after such anniversary date, notice by the stockholder must be so received not later than the close of business on the tenth day following the day on which such notice of the date of the annual meeting was mailed or such public disclosure of the date of the annual meeting was made, whichever first occurs.

Unless otherwise required by law, if the stockholder (or a qualified representative of the stockholder) does not appear at the annual meeting of stockholders of the corporation to present proposed business, such proposed business shall not be transacted, notwithstanding that proxies in respect of such vote may have been received by the corporation.

Validity of Common Stock

The validity of the shares of VeriFone common stock offered hereby will be passed upon for VeriFone by Sullivan & Cromwell LLP, Palo Alto, California. Hypercom has been represented by DLA Piper LLP (US).

Experts

The consolidated financial statements of VeriFone Systems, Inc., appearing in VeriFone Systems, Inc. s Annual Report (Form 10-K) for the year ended October 31, 2010, and the effectiveness of VeriFone Systems, Inc. s internal control over financial reporting as of October 31, 2010 have been audited by Ernst & Young LLP, independent registered public accounting firm, as set forth in their reports thereon, included therein, and incorporated herein by reference. Such consolidated financial statements and VeriFone Systems, Inc. management s assessment of the effectiveness of internal control over financial reporting as of October 31, 2010 are incorporated herein by reference in reliance upon such reports given on the authority of such firm as experts in accounting and auditing.

The consolidated financial statements of Hypercom Corporation, appearing in Hypercom Corporation s Annual Report (Form 10-K) for the year ended December 31, 2009, and the effectiveness of Hypercom Corporation s internal control over financial reporting as of December 31, 2009 have been audited by Ernst & Young LLP, independent registered public accounting firm, as set forth in their reports thereon, included therein, and incorporated herein by reference. Such consolidated financial statements and Hypercom Corporation management s assessment of the effectiveness of internal control over financial reporting as of December 31, 2009 are incorporated herein by reference in reliance upon such reports given on the authority of such firm as experts in accounting and auditing.

Future Stockholder Proposals and Director Nominations

Hypercom intends to hold a 2011 annual meeting of stockholders only if the merger agreement is terminated. Any Hypercom stockholder satisfying the SEC requirements and wishing to submit a proposal to be included in Hypercom s proxy statement for Hypercom s 2011 Annual Meeting of Stockholders (in the event this meeting is held) should submit the proposal in writing to Hypercom s Corporate Secretary, 8888 East Raintree Drive, Suite 300, Scottsdale, Arizona 85260. Generally, under SEC rules such notice would have to be received by December 31, 2010 in order to be considered for inclusion in the proxy statement.

Proposals to be presented at Hypercom s 2011 Annual Meeting of Stockholders (in the event this meeting is held) that are not intended for inclusion in the proxy statement must be submitted in accordance with applicable advance notice provisions of Hypercom s bylaws. Generally, Hypercom s bylaws provide that a stockholder may sponsor a proposal to be considered at the annual meeting if written notice of such proposal is timely received by Hypercom s Corporate Secretary. Generally, a notice is timely given if received by Hypercom s Corporate Secretary date of Hypercom s immediately preceding annual meeting. If Hypercom s 2011 Annual Meeting of Stockholders (in the event this meeting is held) is called for a date within 30 days of June 10, 2011, a notice is timely given if received by Hypercom s Corporate Secretary not later than March 12, 2011. If Hypercom s 2011 Annual Meeting of Stockholders (in the event this meeting is held) is not called for a date that is within 30 days before or after June 10, 2011, pursuant to Hypercom s bylaws, a notice is timely given if received by Hypercom s 2011 Annual Meeting of Stockholders was made, whichever first occurs. If Hypercom does not receive notice of a proposal within the time frame specified by Hypercom s applicable advance notice provisions of its bylaws, such proposal will not be presented at Hypercom s 2011 Annual Meeting of Stockholders (in the event first occurs. If Hypercom does not receive notice of a proposal within the time frame specified by Hypercom s applicable advance notice provisions of its bylaws, such proposal will not be presented at Hypercom s 2011 Annual Meeting of Stockholders (in the event this meeting is held).

Subject to advance notice provisions contained in Hypercom s bylaws, a stockholder of record may propose the nomination of someone for election as a director by timely written notice to Hypercom s Corporate Secretary.

Generally, a notice is timely given if received by Hypercom s Corporate Secretary not less than 90 or more than 120 days before the anniversary date of Hypercom s immediately preceding annual meeting. For Hypercom s 2011 Annual Meeting of Stockholders (in the event this meeting is held), a notice is timely given if received by Hypercom s Corporate Secretary not later than March 12, 2011. A nomination or other proposal may be disregarded if it does not comply with the above procedures and any additional requirements set forth in Hypercom s bylaws. Please note these requirements relate only to the matters a Hypercom stockholder wishes to bring before fellow Hypercom stockholders at an annual meeting and are separate from the SEC s requirements to have a Hypercom stockholder s proposal included in Hypercom s proxy statement, as discussed above.

Delivery of Documents to Stockholders Sharing an Address

Only one copy of this proxy statement/prospectus is being delivered to multiple stockholders of Hypercom unless Hypercom has previously received contrary instructions from one or more stockholders. If you are the beneficial owner, but not the record holder, of shares of Hypercom stock, your broker, bank or other nominee may only deliver one copy of this proxy statement/prospectus to multiple stockholders who share an address unless that nominee has received contrary instructions from one or more of the stockholders. This practice is commonly known as

householding. Hypercom will deliver promptly, upon written or oral request, a separate copy of this proxy statement/prospectus to a Hypercom stockholder at a shared address to which a single copy of the documents was delivered. A Hypercom stockholder who wishes to receive a separate copy of this proxy statement/prospectus should submit this request by writing to Hypercom s Corporate Secretary, 8888 East Raintree Drive, Suite 300, Scottsdale, Arizona 85260, by submitting an e-mail to the Corporate Secretary at corporatesecretary@hypercom.com or by phone at (480) 642-5000. Beneficial owners sharing an address who are receiving multiple copies of this proxy statement/prospectus and who wish to receive a single copy of such materials in the future will need to contact their broker, bank or other nominee to request that only a single copy of such document be mailed to all stockholders at the shared address in the future.

Where You Can Find More Information

VeriFone files annual, quarterly and current reports, proxy statements and other information with the SEC. Hypercom files annual and current reports and other information with the SEC. You may read and copy materials that VeriFone and Hypercom have filed with the SEC at the following SEC public reference room:

100 F Street, N.E.

Washington, D.C. 20549

Please call the SEC at 1-800-SEC-0330 for further information on the operation of the public reference room.

VeriFone s and Hypercom s SEC filings are also available for free to the public on the SEC s Internet website at www.sec.gov, which contains reports, proxy and information statements and other information regarding companies that file electronically with the SEC. In addition, VeriFone s SEC filings are also available for free to the public on VeriFone s website, www.verifone.com, and Hypercom s filings with the SEC are also available for free to the public on Hypercom.com. Information contained on VeriFone s website is not incorporated by reference into this proxy statement/prospectus, and you should not consider information contained on those websites as part of this proxy statement/prospectus.

VeriFone incorporates by reference into this proxy statement/prospectus the documents listed below, and any filings VeriFone makes with the SEC under Section 13(a), 13(c), 14 or 15(d) of the Exchange Act after the date of this proxy statement/prospectus until the date of the special meeting shall be deemed to be incorporated by reference into this proxy statement/prospectus and is considered part of this prospectus from the date VeriFone files that document. In addition, all filings filed by VeriFone pursuant to Sections 13(a), 13(c), 14 or

15(d) of the Exchange Act after the date of the initial registration statement and prior to effectiveness of the registration statement shall be deemed to be incorporated by reference into this proxy statement/prospectus and are considered part of this proxy statement/prospectus. Any statement in a document incorporated by reference into this proxy statement/prospectus will be deemed to be modified or superseded for purposes of this proxy statement/prospectus to the extent a statement contained in this or any other subsequently filed document that is incorporated by reference into this proxy statement. Any statement so modified or superseded will be not deemed, except as so modified or superseded, to constitute a part of this proxy statement/prospectus.

VeriFone SEC Filings

Commission file number 1-32465

Annual Report on Form 10-K

Current Report on Form 8-K

The description of VeriFone common stock set forth in a registration statement filed pursuant to Section 12 of the Exchange Act and any amendment or report filed for the purpose of updating those descriptions.

a registration

Filed on November 19, 2010

Period

Year Ended October 31, 2010 (filed on December 21, 2010)

VeriFone has supplied all information contained in or incorporated by reference into this proxy statement/prospectus relating to VeriFone and its affiliates.

You can obtain a copy of any document incorporated by reference into this proxy statement/prospectus except for the exhibits to those documents from VeriFone. You may also obtain these documents from the SEC or through the SEC s website described above. Documents incorporated by reference are available from VeriFone without charge, excluding all exhibits unless specifically incorporated by reference as an exhibit into this proxy statement/prospectus. You may obtain documents incorporated by reference into this proxy statement/prospectus by requesting them in writing or by telephone from VeriFone at the following address and telephone number:

VeriFone Systems, Inc.

2099 Gateway Place, Suite 600

San Jose, California 95110

Attention: Investor Relations

ir@verifone.com

Telephone: (408) 232-7979

If you would like to request documents, please do so by , 2011 to receive them before the special meeting. If you request any of these documents from VeriFone, VeriFone will mail them to you by first-class mail, or similar means.

Hypercom incorporates by reference into this proxy statement/prospectus the documents listed below, and any filings Hypercom makes with the SEC under Section 13(a), 13(c), 14 or 15(d) of the Exchange Act after the date of this proxy statement/prospectus until the date of the special meeting shall be deemed to be incorporated by reference into this proxy statement/prospectus and is considered part of this prospectus from the date Hypercom files that document. In addition, all filings filed by Hypercom pursuant to Sections 13(a), 13(c), 14 or 15(d) of the Exchange Act after the date of the initial registration statement and prior to effectiveness of the registration statement shall be deemed to be incorporated by reference into this proxy statement/prospectus from the date that Hypercom files that document. The information incorporated by reference is an important part of this proxy statement/prospectus. Any statement in a document

incorporated by reference into this proxy statement/prospectus will be deemed to be modified or superseded for purposes of this proxy statement/prospectus to the extent a statement contained in this or any other subsequently filed document that is incorporated by reference into this proxy statement/prospectus modifies or supersedes such statement. Any statement so modified or superseded will be not deemed, except as so modified or superseded, to constitute a part of this proxy statement/prospectus.

Hypercom SEC Filings

Commission file number 0-50544	Period
Quarterly Reports on Form 10-Q	Quarter ended March 31, 2010 (filed on May 10, 2010); Quarter ended June 30, 2010 (filed on August 9, 2010); Quarter ended September 30, 2010 (filed on November 8, 2010)
Annual Report on Form 10-K and Form 10-K/A	Year Ended December 31, 2009 (filed on March 12, 2010 and December 1, 2010)
Current Reports on Form 8-K	Filed on January 5, 2010, June 11, 2010, September 30, 2010, October 8, 2010, November 19, 2010, and December 16, 2010

Hypercom has supplied all information contained in or incorporated by reference into this proxy statement/prospectus relating to Hypercom and its affiliates.

You can obtain a copy of any document incorporated by reference into this proxy statement/prospectus except for the exhibits to those documents from Hypercom. You may also obtain these documents from the SEC or through the SEC s website described above. Documents incorporated by reference are available from Hypercom without charge, excluding all exhibits unless specifically incorporated by reference as an exhibit into this proxy statement/prospectus. You may obtain documents incorporated by reference into this proxy statement/prospectus by requesting them in writing or by telephone from Hypercom at the following address and telephone number:

Hypercom Corporation

Attention: Investor Relations

8888 East Raintree Drive, Suite 300

Scottsdale, Arizona, 85260

stsujita@hypercom.com

(480) 642-5000

If you would like to request documents, please do so by , 2011 to receive them before the special meeting. If you request any of these documents from Hypercom, Hypercom will mail them to you by first-class mail, or similar means.

You also may obtain documents incorporated by reference into this proxy statement/prospectus by requesting them in writing or by telephone from Innisfree M&A Incorporated, Hypercom s information agent and proxy solicitor for the Hypercom special meeting and merger, at the following address and telephone numbers:

Innisfree M&A Incorporated

501 Madison Avenue, 20th Floor

New York, New York 10022

(888) 750-5834 (toll-free)

Edgar Filing: VERIFONE SYSTEMS, INC. - Form S-4

Banks & Brokers Call Collect:

(212) 750-5833

ANNEX A

AGREEMENT AND PLAN OF MERGER

Among

HYPERCOM CORPORATION,

VERIFONE SYSTEMS, INC.

and

HONEY ACQUISITION CO.

Dated as of November 17, 2010

TABLE OF CONTENTS

ARTICLE I THE MERGER; CLOSING; EFFECTIVE TIME	Page A-1
<u>1.1 The Merger</u>	A-1
<u>1.2</u> Closing	A-1
<u>1.3 Effective Time</u>	A-2
ARTICLE II CERTIFICATE OF INCORPORATION AND BY-LAWS OF THE SURVIVING CORPORATION	A-2
2.1 The Certificate of Incorporation	A-2
<u>2.2 The By-Laws</u>	A-2
ARTICLE III OFFICERS AND DIRECTORS	A-2
3.1 Directors of Surviving Corporation	A-2
3.2 Officers of Surviving Corporation	A-2
ARTICLE IV EFFECT OF THE MERGER ON CAPITAL STOCK; EXCHANGE OF CERTIFICATES	A-2
4.1 Effect on Capital Stock	A-2
(a) Merger Consideration	A-2
(b) Cancellation of Shares	A-3
(c) Merger Sub	A-3
4.2 Exchange of Certificates for Shares	A-3
(a) Exchange Agent	A-3
(b) Exchange Procedures	A-3
(c) Distributions with Respect to Unexchanged Shares; Voting	A-4
(d) Transfers	A-5
(e) Fractional Shares	A-5
(f) Termination of Exchange Period; Unclaimed Stock	A-5
(g) Lost, Stolen or Destroyed Certificates	A-5
(h) Withholding Rights	A-5
4.3 Adjustments to Prevent Dilution	A-6
4.4 Company Stock Based Plans	A-6
4.5 Warrants	A-7
4.6 Tax Consequences	A-7
ARTICLE V REPRESENTATIONS AND WARRANTIES	A-8
5.1 Representations and Warranties of the Company	A-8
(a) Organization, Good Standing and Qualification	A-8
(b) Capital Structure	A-9
(c) Corporate Authority; Approval and Fairness	A-10
(d) Governmental Filings; No Violations	A-11
(e) Reports; Financial Statements	A-12
(f) Absence of Certain Changes	A-13
(g) Litigation and Liabilities	A-14
(h) Employee Benefits	A-15
(i) Compliance with Laws	A-16
(i) Takeover Statutes	A-17
(k) Environmental Matters	A-17
(l) Taxes	A-18
(m) Labor Matters	A-20

	Page
(n) Intellectual Property	A-20
(o) Affiliate Transactions	A-22
(p) Insurance	A-22
(q) Material Contracts	A-22
(r) Property	A-24
(s) Warranties/Product Liability	A-24
(t) Product Certifications	A-25
<u>(u) Inventory</u>	A-25
(v) Licenses	A-25
(w) Anti-Bribery Laws	A-25
(x) OFAC	A-26
(y) Investment Company Act of 1940	A-26
(z) Brokers and Finders	A-26
(aa) Rights Agreement	A-26
(bb) Reorganization Treatment	A-26
(cc) No Other Representations and Warranties	A-26
5.2 Representations and Warranties of Parent and Merger Sub	A-26
(a) Organization, Good Standing and Qualification	A-20 A-27
(b) Capitalization of Merger Sub	A-27 A-27
(c) Capital Structure of Parent	A-27 A-27
(d) Corporate Authority	A-27 A-28
(e) Governmental Filings; No Violations; Certain Contracts, Etc.	A-28
	A-28 A-29
	A-29 A-30
	A-30 A-30
 (h) Litigation and Liabilities (i) Compliance with Laws 	A-30 A-30
	A-30 A-30
(k) Section 203 (l) Reorganization Treatment	A-30 A-30
(m) No Other Representations and Warranties	A-30 A-30
(iii) No Other Representations and warranties	A-30
<u>ARTICLE VI COVENANTS</u>	A-30
6.1 Interim Operations	A-30
6.2 Acquisition Proposals	A-33
(a) No Solicitation or Negotiation	A-33
(b) Definitions	A-34
(c) No Change in Recommendation Agreement	A-34
(d) Certain Permitted Disclosure	A-35
(e) Existing Discussions	A-35
(f) Notice	A-35
6.3 Information Supplied	A-35
6.4 Stockholders Meeting	A-36
6.5 Filings; Other Actions; Notification	A-37
6.6 Dispositions	A-39
6.7 Access; Consultation	A-39
6.8 Stock Exchange Listing and Delisting	A-40
6.9 Publicity	A-40
6.10 Expenses	A-40
6.11 Indemnification; Directors and Officers Insurance	A-40
6.12 Takeover Statute	A-42

	Page
6.13 Section 16(b)	A-42
6.14 Rights	A-42
6.15 Stockholder Litigation	A-42
6.16 Employee Matters	A-42
6.17 Tax Treatment	A-43
6.18 Certain Actions	A-43
ARTICLE VII CONDITIONS	A-45
7.1 Conditions to Each Party s Obligation to Effect the Merger	A-45
(a) Stockholder Approval	A-45
(b) NYSE Listing	A-45
(c) Regulatory Consents	A-45
(d) No Injunction or Restraint	A-45
(e) S-4 Registration Statement	A-45
(f) Blue Sky Approvals	A-46
(g) Dispositions	A-46
(h) Opinion	A-46
7.2 Conditions to Obligations of Parent and Merger Sub	A-46
(a) Representations and Warranties	A-46
(b) Performance of Obligations of the Company	A-46
(c) No Restraints	A-46
(d) Material Adverse Effect	A-47
(e) Receipt of Payoff Letter	A-47
(f) Resignations	A-47
7.3 Conditions to Obligation of the Company	A-47
(a) Representations and Warranties	A-47
(b) Performance of Obligations of Parent and Merger Sub	A-48
(c) Parent Material Adverse Effect	A-48
ARTICLE VIII TERMINATION	A-48
8.1 Termination by Mutual Consent	A-48
8.2 Termination by Either Parent or the Company	A-48
8.3 Termination by the Company	A-48
8.4 Termination by Parent	A-49
8.5 Effect of Termination and Abandonment	A-49
ARTICLE IX MISCELLANEOUS AND GENERAL	A-50
<u>9.1 Survival</u>	A-50
9.2 Modification or Amendment	A-50
9.3 Waiver of Conditions	A-50
9.4 Counterparts	A-51
9.5 Governing Law and Venue; Waiver of Jury Trial	A-51
9.6 Notices	A-51
9.7 Entire Agreement	A-52
9.8 No Third-Party Beneficiaries	A-52
9.9 Obligations of Parent and of the Company	A-53
9.10 Severability	A-53
9.11 Definitions	A-53
9.12 Interpretation	A-53
9.13 Captions	A-53

		Page
<u>9.14</u>	Assignment	A-53
<u>9.15</u>	Enforcement	A-53
Attachments:		

Annex A Definitions

Exhibit A Form of Support Agreement

A-iv

AGREEMENT AND PLAN OF MERGER

AGREEMENT AND PLAN OF MERGER (hereinafter called this <u>Agreement</u>), dated as of November 17, 2010 among Hypercom Corporation, a Delaware corporation (the <u>Company</u>), VeriFone Systems, Inc., a Delaware corporation (<u>Parent</u>), and Honey Acquisition Co., a Delaware corporation and a direct wholly-owned subsidiary of Parent (<u>Merger Sub</u>).

RECITALS

WHEREAS, the respective Boards of Directors of each of the Company, Parent and Merger Sub have approved the merger of Merger Sub with and into the Company (the <u>Merger</u>) upon the terms and subject to the conditions set forth in this Agreement and have approved and declared advisable this Agreement;

WHEREAS, the Boards of Directors of each of the Company and Merger Sub have resolved to submit this Agreement to their respective stockholders for adoption;

WHEREAS, Parent, Merger Sub and the Company intend that the Merger will qualify as a reorganization within the meaning of Section 368(a) of the Internal Revenue Code of 1986, as amended (the <u>Code</u>) for U.S. federal income tax purposes and intend, by executing this Agreement, to adopt a plan of reorganization within the meaning of Treasury Regulations Sections 1.368-2(g) and 1.368-3(a);

WHEREAS, concurrently with the execution and delivery of this Agreement, and as a condition and inducement to Parent s willingness to enter into this Agreement, FP Hypercom Holdco, LLC and Francisco Partners II, L.P. have entered into a Support Agreement in the form attached hereto as Exhibit A (the <u>Support Agreement</u>); and

WHEREAS, the Company, Parent and Merger Sub desire to make certain representations, warranties, covenants and agreements in connection with this Agreement.

NOW, THEREFORE, in consideration of the premises, and of the representations, warranties, covenants and agreements contained herein, the parties hereto agree as follows:

ARTICLE I

THE MERGER; CLOSING; EFFECTIVE TIME

1.1 <u>The Merger</u>. Upon the terms and subject to the conditions set forth in this Agreement, at the Effective Time, Merger Sub shall be merged with and into the Company and the separate corporate existence of Merger Sub shall thereupon cease. The Company shall be the surviving corporation in the Merger (sometimes hereinafter referred to as the <u>Surviving Corporation</u>), and the separate corporate existence of the Company with all its rights, privileges, immunities, powers and franchises shall continue unaffected by the Merger, except as set forth in Article III hereof. The Merger shall have the effects specified in the Delaware General Corporation Law, as amended (the <u>DGCL</u>).

1.2 <u>Closing</u>. Unless otherwise mutually agreed in writing between Parent and the Company, the closing for the Merger (the <u>Closing</u>) shall take place, subject to the terms and conditions of this Agreement, at the offices of Sullivan & Cromwell LLP, 1870 Embarcadero Road, Palo Alto, California 94303, at a time and date to be designated by the parties (the <u>Closing Date</u>) which shall be no later than the third business day after the satisfaction or waiver of the conditions set forth in Article VII (other than those conditions that by their nature are to be satisfied at the Closing, but subject to the fulfillment or waiver of those conditions) or at such other time, date and location as the parties hereto shall mutually agree. For purposes of this Agreement, the term <u>business day</u> shall mean any day ending at 11:59 p.m. (Pacific Time) other than a Saturday or Sunday or a day on which banks are required or authorized to close in The City of New York.

1.3 <u>Effective Time</u>. At the Closing, the Company and Merger Sub will cause a Certificate of Merger (the <u>Certificate of Merger</u>) to be completed, executed, acknowledged and filed with the Secretary of State of the State of Delaware as provided in Section 251 of the DGCL. The Merger shall become effective at the time when the Certificate of Merger has been duly filed with the Secretary of State of Delaware or such other time as shall be agreed upon by the parties hereto in writing and set forth in the Certificate of Merger in accordance with the DGCL (the <u>Effective Time</u>).

ARTICLE II

CERTIFICATE OF INCORPORATION AND BY-LAWS

OF THE SURVIVING CORPORATION

2.1 <u>The Certificate of Incorporation</u>. The certificate of incorporation of Merger Sub as in effect immediately prior to the Effective Time shall be the certificate of incorporation of the Surviving Corporation (the <u>Charter</u>), until duly amended as provided therein or by applicable Law.

2.2 <u>The By-Laws</u>. The by-laws of Merger Sub in effect at the Effective Time shall be the by-laws of the Surviving Corporation (the <u>By-Laws</u>), until thereafter amended as provided therein or by applicable Law.

ARTICLE III

OFFICERS AND DIRECTORS

3.1 <u>Directors of Surviving Corporation</u>. The directors of Merger Sub at the Effective Time shall, from and after the Effective Time, be the directors of the Surviving Corporation until their successors have been duly elected or appointed and qualified or until their earlier death, resignation or removal in accordance with the Charter and the By-Laws. The Company shall take all actions necessary to ensure that such directors shall be the directors of the Surviving Corporation at the Effective Time.

3.2 <u>Officers of Surviving Corporation</u>. The parties hereto shall take all actions necessary so that from and after the Effective Time, the individuals designated by Parent shall be the officers of the Surviving Corporation until their successors have been duly elected or appointed and qualified or until their earlier death, resignation or removal in accordance with the Charter and the By-Laws.

ARTICLE IV

EFFECT OF THE MERGER ON CAPITAL STOCK;

EXCHANGE OF CERTIFICATES

4.1 <u>Effect on Capital Stock</u>. At the Effective Time, as a result of the Merger and without any action on the part of the holder of any capital stock of the Company:

(a) Merger Consideration.

(i) Each share of common stock, par value \$0.001 per share, of the Company (each a <u>Company Share</u>, and together, the <u>Company Shares</u>) issued and outstanding at the Effective Time (other than Company Shares that are owned by Parent or by the Company or any direct or indirect wholly-owned Subsidiary of the Company and in each case not held on behalf of third parties (<u>Excluded Company Shares</u>)) shall be converted into and shall become exchangeable for 0.23 (the <u>Exchange Ratio</u>) shares (the <u>Merger Consideration</u>) of common stock, par value \$0.01 per share, of Parent (the <u>Parent Common Stock</u>).

(ii) At the Effective Time, all Company Shares shall no longer be outstanding, shall be cancelled and retired and shall cease to exist, and (A) each certificate (a <u>Certificate</u>) formerly representing any of such Company Shares (other than Excluded Company Shares) and (B) each uncertificated Company Share (a <u>Book-Entry Company Share</u>) registered to a holder on the stock transfer books of the Company (other than Excluded Company Shares), shall thereafter represent only the right to the Merger Consideration and the right, if any, to receive pursuant to Section 4.2(e) cash in lieu of fractional shares into which such Company Shares have been converted pursuant to this Section 4.1(a) and any distribution or dividend pursuant to Section 4.2(c).

(b) <u>Cancellation of Shares</u>. Each Excluded Company Share shall, by virtue of the Merger and without any action on the part of the holder thereof, no longer be outstanding, shall be cancelled and retired without payment of any consideration therefor and shall cease to exist.

(c) <u>Merger Sub</u>. At the Effective Time, each share of common stock, par value \$0.001 per share, of Merger Sub issued and outstanding immediately prior to the Effective Time shall be converted into one share of Common Stock, par value \$0.001 per share, of the Surviving Corporation.

4.2 Exchange of Certificates for Shares.

(a) Exchange Agent. At or prior to the Effective Time, Parent shall deposit, or shall cause to be deposited, with an exchange agent selected by Parent and reasonably satisfactory to the Company (the <u>Exchange Agent</u>), for the benefit of the holders of Company Shares (other than Excluded Company Shares), certificates representing the shares of Parent Common Stock to be exchanged for Company Shares (other than Excluded Company Shares) in respect of the Merger Consideration to be paid in the Merger, cash necessary to pay for cash in lieu of any fractional shares of Parent Common Stock issuable in connection with the Merger pursuant to Section 4.2(e), plus an amount of cash with respect to any dividends or other distributions with respect to the Parent Common Stock to be paid or to be issued pursuant to Section 4.2(c) in exchange for Company Shares (other than Excluded Company Shares) (such certificates for shares of Parent Common Stock, together with the amount of any cash payable pursuant to Section 4.2(e) in lieu of fractional shares and dividends or other distributions payable with respect thereto pursuant to Section 4.2(c), being hereinafter referred to as the <u>Exchange Fund</u>). With respect to the amount of cash to be deposited as of the Effective Time to satisfy its obligations under Section 4.2(e), Parent shall only be required to make a reasonable estimate of the amount of such cash that will be necessary. The Exchange Agent shall invest the Exchange Fund as directed by Parent. Any interest and other income resulting from such investment shall become a part of the Exchange Fund, and any amounts in excess of the amounts payable under Section 4.1(a) shall be promptly returned to Parent. If there are any losses realized as a result of such investments, Parent shall promptly deposit in the Exchange Fund the amount of such losses.

(b) Exchange Procedures.

(i) Promptly after the Effective Time, the Surviving Corporation shall cause the Exchange Agent to mail to each holder of record as of the Effective Time of Company Shares (other than holders of Excluded Company Shares) (A) (x) in the case of holders of Certificates, a letter of transmittal specifying that delivery shall be effected, and that risk of loss and title to the Certificates shall pass, only upon delivery of the Certificates (or affidavits of loss in lieu thereof) to the Exchange Agent (the <u>Certificate Letter of Transmittal</u>) or (y) in the case of holders of Book-Entry Company Shares, a letter of transmittal specifying that the exchange for uncertificated shares of Parent Common Stock registered on the stock transfer books of Parent in the name of each such holder (<u>Book-Entry Parent Shares</u>) or certificates representing shares of Parent Common Stock shall occur only upon delivery of such letter of transmittal to the Exchange Agent (the <u>Registered Letter of Transmittal</u>), each such letter of transmittal to be in such form and have such other provisions as Parent and the Company may reasonably agree, and (B) instructions for exchanging Certificates or Book-Entry Company Shares for the applicable Merger Consideration and any unpaid dividends and other distributions, if applicable, and cash in lieu of fractional shares.

(ii) Following the Effective Time, upon (A) surrender of a Certificate for cancellation to the Exchange Agent together with a Certificate Letter of Transmittal, duly executed, the holder of such Certificate or (B) delivery of a Registered Letter of Transmittal, duly executed, the holder of such Certificate or Book-Entry Company Shares, as the case may be, shall be entitled to receive in exchange therefor (x) Book-Entry Parent Shares or, at the election of such holder, a certificate representing that number of whole shares of Parent Common Stock that such holder is entitled to receive pursuant to this Article IV, if any, and (y) a check in the amount (after giving effect to any required Tax withholdings) of (I) any cash in lieu of fractional shares determined in accordance with Section 4.2(e) hereof plus (II) any cash dividends and any other dividends or other distributions that such holder has the right to receive pursuant to the provisions of this Article IV, and any Certificate so surrendered shall forthwith be cancelled.

(iii) No interest will be paid or accrued on any amount payable upon due surrender of any Certificate or delivery of a duly executed Registered Letter of Transmittal, as the case may be. In the event of a transfer of ownership of Company Shares that is not registered in the transfer records of the Company, Book-Entry Parent Shares or a certificate representing the proper number of shares of Parent Common Stock, as the case may be, together with a check for the cash to be paid upon due surrender of the Certificate or upon the delivery to the Exchange Agent of the duly executed Registered Letter of Transmittal and any other dividends or distributions in respect thereof, may be issued and/or paid to such a transferee if, in the case of holders of Certificates, the Certificate formerly representing such Company Shares is presented to the Exchange Agent, and, in the case of holders of Book-Entry Company Shares, if the Registered Letter of Transmittal is delivered to the Exchange Agent, in either case accompanied by all documents required to evidence and effect such transfer and to evidence that any applicable stock transfer Taxes have been paid.

(iv) If any Book-Entry Parent Shares or any certificate for shares of Parent Common Stock is to be issued in a name other than that in which the Certificate surrendered in exchange therefor or the Book-Entry Company Shares exchanged therefor, as the case may be, is registered, it shall be a condition of such exchange that the Person requesting such exchange shall pay any transfer or other Taxes required by reason of the issuance of Book-Entry Parent Shares or a certificate for shares of Parent Common Stock in a name other than that of the registered holder of the Certificate surrendered or the Book-Entry Company Shares exchanged, as the case may be, or shall establish to the satisfaction of Parent or the Exchange Agent that such Tax has been paid or is not applicable.

For the purposes of this Agreement, the term <u>Person</u> or <u>person</u> shall mean any individual, corporation (including not-for-profit), general or limited partnership, limited liability company, joint venture, estate, trust, association, organization, Governmental Entity or other entity of any kind or nature.

(c) <u>Distributions with Respect to Unexchanged Shares</u>; <u>Voting</u>. All shares of Parent Common Stock to be issued pursuant to the Merger shall be deemed issued and outstanding as of the Effective Time, and whenever a dividend or other distribution is declared by Parent in respect of the Parent Common Stock, the record date for which is at or after the Effective Time, that declaration shall include dividends or other distributions in respect of all shares of Parent Common Stock issuable pursuant to this Agreement. No dividends or other distributions in respect of such Parent Common Stock shall be paid to any holder of any unsurrendered Certificate or Book-Entry Company Shares for which a Registered Letter of Transmittal shall not have been delivered, until such Certificate (or affidavits of loss in lieu thereof) is surrendered for exchange or such Registered Letter of Transmittal is delivered, as the case may be, in accordance with this Article IV. Subject to the effect of applicable Laws, following surrender of any such Certificate (or affidavits of loss in lieu thereof) or delivery of any such Registered Letter of Transmittal, as the case may be, there shall be issued and/or paid to the holder of the Book-Entry Parent Shares or the certificates representing whole shares of Parent Common Stock, as the case may be, issued in exchange therefor, without interest, (A) at the time of such surrender or delivery, as the case may be, the dividends or other distributions with a record date at or after the Effective Time and a payment date, the dividends or other distributions payable with respect to such whole shares of Parent Common Stock with a record date at or after the Effective Time but with a payment date

subsequent to surrender or delivery, as the case may be. Holders of unsurrendered Certificates or Book-Entry Company Shares shall be entitled to vote after the Effective Time at any meeting of Parent stockholders the number of whole shares of Parent Common Stock represented by such Certificates or Book-Entry Company Shares, regardless of whether such holders have exchanged their certificates.

(d) <u>Transfers</u>. From and after the Effective Time, there shall be no transfers on the stock transfer books of the Company of Company Shares that were outstanding immediately prior to the Effective Time.

(e) <u>Fractional Shares</u>. Notwithstanding any other provision of this Agreement, no fractional shares of Parent Common Stock will be issued and any holder of Company Shares entitled to receive a fractional share of Parent Common Stock but for this Section 4.2(e) shall be entitled to receive an amount in cash (without interest) determined by multiplying such fraction (rounded to the nearest one-hundredth of a share) by the Average Closing Price means the per share closing price of Parent Common Stock as reported on the NYSE composite transactions reporting system (as reported in the Eastern Edition of *The Wall Street Journal* or, if not reported thereby, another authoritative source) on the last trading day prior to the Closing Date.

(f) <u>Termination of Exchange Period; Unclaimed Stock</u>. Any portion of the Exchange Fund (including the proceeds of any investments thereof and any shares of Parent Common Stock) that remains unclaimed by the stockholders of the Company one year after the Effective Time shall be delivered at Parent s option, to Parent. Any stockholders of the Company who have not theretofore complied with this Article IV shall thereafter look only to Parent for delivery of any Merger Consideration and any cash, dividends and other distributions in respect thereof payable or deliverable pursuant to Section 4.2(c) and Section 4.2(e) upon due (i) surrender of their Certificates (or affidavits of loss in lieu thereof) or (ii) delivery of duly executed Registered Letters of Transmittal, as the case may be, in each case with respect to both clauses (i) and (ii), without any interest thereon. Notwithstanding the foregoing, none of Parent, the Surviving Corporation, the Exchange Agent or any other Person shall be liable to any former holder of Company Shares for any amount properly delivered to a public official pursuant to applicable abandoned property, escheat or similar Laws.

(g) Lost, Stolen or Destroyed Certificates. In the event any Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the Person claiming such Certificate to be lost, stolen or destroyed and, if reasonably required by Parent, the posting by such Person of a bond in customary amount and upon such terms as may be required by Parent as indemnity against any claim that may be made against it with respect to such Certificate, the Exchange Agent will issue in exchange for such lost, stolen or destroyed Certificate the applicable Merger Consideration and any cash, unpaid dividends or other distributions that would be payable or deliverable in respect thereof pursuant to this Agreement had such lost, stolen or destroyed Certificate been surrendered.

(h) <u>Withholding Rights</u>. Parent shall be entitled to deduct and withhold from the consideration otherwise payable pursuant to this Agreement to any holder of Company Shares such amounts as Parent is required to deduct and withhold with respect to the making of any payment pursuant to this Agreement under the Code and by rules and regulations promulgated thereunder or any other applicable state, local, or foreign Tax Law; <u>it</u> <u>being understood that</u> Parent shall be entitled but not obligated to deduct and withhold from any Merger Consideration payable or otherwise deliverable by withholding a portion or all of the Merger Consideration otherwise deliverable and selling such Merger Consideration to cover its withholding obligations. To the extent that amounts are so withheld by Parent, as the case may be, such withheld amounts (i) shall be remitted by Parent or the Surviving Corporation, as applicable, to the applicable Governmental Entity, and (ii) shall be treated for all purposes of this Agreement as having been paid to the holder of Company Shares in respect of which such deduction and withholding was made by Parent or the Surviving Corporation, as the case may be.

4.3 <u>Adjustments to Prevent Dilution</u>. Notwithstanding anything in this Agreement to the contrary, if, between the date of this Agreement and the Effective Time, the issued and outstanding Company Shares or securities convertible or exchangeable into or exercisable for Company Shares, or the issued and outstanding shares of Parent Common Stock or securities convertible or exchangeable into or exercisable for shares of Parent Common Stock, shall have been changed into a different number of shares or a different class by reason of any reclassification, stock split (including a reverse stock split), stock dividend or distribution, recapitalization or other similar transaction, then the Merger Consideration, the Exchange Ratio and any other dependent items shall be equitably adjusted and as so adjusted shall, from and after the date of such event, be the Merger Consideration, the Exchange Ratio or other dependent item.

4.4 Company Stock Based Plans.

(a) At the Effective Time, each outstanding option to purchase Company Shares (a <u>Company Option</u>), whether vested or unvested, shall be converted into an option to acquire a number of shares of Parent Common Stock equal to the product (rounded down to the nearest whole number) of (x) the number of Company Shares subject to the Company Option immediately prior to the Effective Time and (y) the Exchange Ratio at an exercise price per share (rounded up to the nearest whole cent) equal to (A) the exercise price per Company Share of such Company Option immediately prior to the Effective Time divided by (B) the Exchange Ratio; <u>provided</u>, <u>however</u>, the exercise price and the number of shares of Parent Common Stock purchasable pursuant to Company Option to which Section 422 of the Code applies, the exercise price, the number of shares of Parent Common Stock purchasable pursuant to such option and the terms and conditions of exercise of such option shall be determined in accordance with the foregoing, subject to such adjustments as are necessary in order to satisfy the requirements of Section 424(a) of the Code. Except as specifically provided above, following the Effective Time, each Company Option shall continue to be governed by the same terms and conditions as were applicable under such Company Option immediately prior to the Effective Time.

(b) At the Effective Time, each right of any kind (other than Company Warrants, which are the subject of Section 4.5), contingent or accrued, to acquire or receive Company Shares or benefits measured by the value of Company Shares, and each award of any kind consisting of Company Shares that may be held, awarded, outstanding, payable or reserved for issuance under the Company Stock Plans and any other Company Compensation and Benefits Plans, other than Company Options (the <u>Company Awards</u>), shall be deemed to be converted into the right to acquire or receive, or receive benefits measured by the value of, as the case may be, the number of shares of Parent Common Stock equal to the product (rounded down to the nearest whole number) of (x) the number of Company Shares subject to such Company Award immediately prior to the Effective Time and (y) the Exchange Ratio. Except as specifically provided above, following the Effective Time, each such right shall be subject to the same terms and conditions as were applicable to the rights under the relevant Company Stock Plan or other Company Compensation and Benefit Plan immediately prior to the Effective Time. At or prior to the Effective Time, the Company shall cause to be effectuated the provisions of this Section 4.4(b).

(c) As soon as reasonably practicable (but not more than five (5) days) after the Effective Time, Parent shall, if registration of any interests in the Company Stock Plans or other Company Compensation and Benefit Plans or the shares of Parent Common Stock issuable thereunder is required under the Securities Act of 1933, as amended (the <u>Securities Act</u>), file with the Securities and Exchange Commission (the <u>SEC</u>) a registration statement on Form S-8 (or any successor form), or another appropriate form with respect to such interests and Parent Company Stock Plans or other Company Compensation and Benefit Plans, as applicable, remain in effect and such registration of interests therein or the shares of Parent Common Stock issuable thereunder continues to be required. As soon as practicable after the registration of such interests or shares, as applicable, Parent shall deliver to the holders of Company Options and Company Awards appropriate notices setting forth such holders rights pursuant to the respective Company Stock Plans and agreements evidencing the

grants of such Company Options and Company Awards, and stating that such Company Options and Company Awards and agreements have been assumed by Parent and shall continue in effect on the same terms and conditions (subject to the adjustments required by this Section 4.4 after giving effect to the Merger and the terms of the Company Stock Plans).

(d) At or prior to the Effective Time, the Company, the Board of Directors of the Company and the compensation committee of the Board of Directors of the Company, as applicable, shall adopt any resolutions and take any actions which are necessary to effectuate the provisions of Sections 4.4(a) and 4.4(b). Parent shall take all actions which are necessary for the assumption of the Company Options and Company Awards pursuant to Sections 4.4(a) and 4.4(b) including the reservation, issuance (subject to Section 4.4(c)) and listing of Parent Common Stock as necessary to effect the transactions contemplated by this Section 4.4. The Company shall take all necessary action to ensure that from and after the Effective Time neither Parent nor the Surviving Corporation will be required to deliver Company Shares or other capital stock of the Company to any Person pursuant to or in settlement of Company Options or Company Awards, and all Company Stock Plans or other Company Compensation and Benefits Plans conferring any rights to Company Shares or other capital stock of the Company shall be deemed to be in conformity with this Section 4.4(d).

(e) The Company shall take all actions necessary to cause the Company s Employee Stock Purchase Plan to terminate at or prior to the Effective Time in accordance with the terms of the plan and to cause each Company employee s then current plan account balance to be distributed in cash to such employee at the time of plan termination.

(f) Notwithstanding any other provision of this Agreement, Parent acknowledges and agrees that the Board of Directors of the Company has taken or may take prior to Closing such action as may be required (i) to fully vest as of or prior to the Closing all Company Options and Company Awards outstanding as of the date hereof, and (ii) to provide for net or cashless exercise of Company Options and Company Awards, such that any Tax withholding is satisfied from Company Shares underlying such Company Options and Company Awards. Company Shares purchased upon exercise of Company Options and Company Shares delivered upon vesting of Company Awards shall be outstanding as of the Effective Time and entitled to receive the Merger Consideration as provided in Sections 4.1 and 4.2, and Company Options not exercised will be assumed as provided in Section 4.4(a).

4.5 <u>Warrants</u>. At the Effective Time, each outstanding warrant to purchase Company Shares (a <u>Company Warrant</u>) shall be converted into a warrant to acquire a number of shares of Parent Common Stock (rounded down to the nearest whole number) equal to the product of (a) the number of Company Shares subject to such Company Warrant immediately prior to the Effective Time, multiplied by (b) the Exchange Ratio, at an exercise price per share (rounded up to the nearest whole cent) equal to (x) the exercise price per Company Share of such Company Warrant immediately prior to the Effective Time divided by (b) the Exchange Ratio. Except as specifically provided above, following the Effective Time, each Company Warrant shall continue to be governed by the same terms and conditions as were applicable under such Company Warrant immediately prior to the Effective Time.

4.6 <u>Tax Consequences</u>. For U.S. federal income tax purposes, the Merger is intended to constitute a reorganization within the meaning of Section 368(a) of the Code. The parties to this Agreement hereby adopt this Agreement as a plan of reorganization within the meaning of Treasury Regulations Sections 1.368-2(g) and 1.368-3(a).

ARTICLE V

REPRESENTATIONS AND WARRANTIES

5.1 <u>Representations and Warranties of the Company</u>. The Company hereby represents and warrants to Parent and Merger Sub, as of the date of this Agreement and as of the Closing Date (it being understood that to the extent any such representation and warranty speaks as of a particular date, the Company makes such representation and warranty with respect to such particular date), that, except as otherwise permitted in this Agreement or as set forth (i) in the Company Reports filed with the SEC since January 1, 2010 and prior to the date hereof (excluding, in each case, any disclosures set forth in any risk factor section or in any other section to the extent such statements are cautionary, predictive or forward-looking in nature), (ii) in the corresponding sections or subsections of the disclosure letter delivered to Parent by the Company at the time of entering into this Agreement (the <u>Company Disclosure Letter</u>) or (iii) to the extent that the qualifying nature of a disclosure in the Company Disclosure Letter with respect to another section or subsection is reasonably apparent on its face, in such other disclosure:

(a) <u>Organization. Good Standing and Qualification</u>. Other than as set forth in Section 5.1(a) of the Company Disclosure Letter, each of the Company and its Subsidiaries is a legal entity duly organized, validly existing and in good standing under the Laws of its respective jurisdiction of organization and has all requisite corporate or similar power and authority to own, lease and operate its properties and assets and to carry on its business as presently conducted and is qualified to do business and is in good standing as a foreign corporation or other legal entity in each jurisdiction where the ownership, leasing or operation of its assets or properties or conduct of its business requires such qualification, except where the failure to be so organized, qualified or in good standing, or to have such power or authority, individually or in the aggregate, has not had or would not reasonably be likely to have a Company Material Adverse Effect. Prior to the date of this Agreement, the Company has made available to Parent a complete and correct copy of the Company s and its Subsidiaries certificates of incorporation and by-laws or comparable governing documents, each as amended through the date of this Agreement, and each as so delivered is in full force and effect. Section 5.1(a) of the Company Disclosure Letter contains a correct and complete list of each jurisdiction where the Company and its Subsidiaries are organized and qualified to do business.

As used in this Agreement, (i) the term <u>Subsidiary</u> means, with respect to any Person, any other Person of which at least a majority of the securities or ownership interests having by their terms ordinary voting power to elect a majority of the Board of Directors or other persons performing similar functions is directly or indirectly owned or controlled by such Person and/or one or more of its Subsidiaries; and (ii) the term <u>Company Material Adverse Effect</u> means a material adverse effect on (x) the financial condition, assets, liabilities, business, or results of operations of the Company and its Subsidiaries, taken as a whole, or (y) the ability of the Company to perform its obligations under this Agreement or consummate the Merger and the other transactions contemplated hereby; provided, however, that Company Material Adverse Effect shall not include any event, condition, change, occurrence or development of a state of circumstances resulting from, primarily attributable to, or arising in connection with (A) general political, economic or market (including currency) conditions or general changes or developments in the industry in which the Company and its Subsidiaries operate, (B) acts of terrorism or war (whether or not declared) or natural disasters occurring after the date hereof, (C) this Agreement, the Merger and the other transactions contemplated hereby, or the announcement or performance thereof, including any negative impact on or disruption in relationships with customers, suppliers, distributors, employees or similar relationships, (D) changes in Law or any applicable accounting regulations or principles or the interpretations thereof, (E) changes in the price or trading volume of the Company s stock (provided that any event, condition, change, occurrence or development of a state of circumstances that may have caused or contributed to such change in market price or trading volume shall not be excluded under this proviso), (F) any failure by the Company to meet public or internal revenue, earnings or other projections, in and of itself (provided that any event, condition, change, occurrence or development of a state of circumstances that may have caused or contributed to such failure to meet published revenue, earnings or other projections shall not be excluded under this proviso), (G) the taking of any action required by this Agreement or expressly approved or permitted in

writing by Parent, or reasonably required to obtain the approval of any Governmental Entity or the failure to take any action prohibited by this Agreement or (H) any legal proceedings commenced or threatened by any of the current or former stockholders of the Company (on their own behalf or on behalf of the Company) against the Company that assert allegations of a breach of fiduciary duty or violations of securities laws relating to this Agreement or the transactions contemplated by this Agreement, except in the case of clause (A) or (B), to the extent such changes or conditions disproportionately affect in any material respect the Company and its Subsidiaries, taken as a whole, as compared to other Persons engaged in similar businesses.

(b) Capital Structure. (i) The authorized capital stock of the Company consists of 100,000,000 Company Shares, of which 55,515,330 Company Shares were issued and outstanding as of the close of business on November 16, 2010, and 10,000,000 shares of Preferred Stock, \$0.001 par value per share (the <u>Company Preferred Shares</u>) (including a series of Preferred Stock constituting 100,000 Company Preferred Shares designated as <u>Series A Junior Participating Preferred Stock</u>), none of which were outstanding as of the date of this Agreement. Other than as set forth in Section 5.1(b)(i) of the Company Disclosure Letter, no Company Shares are held in treasury by the Company or its Subsidiaries. All of the outstanding Company Shares have been duly authorized and validly issued and are fully paid and nonassessable. All of the outstanding Company Shares have been issued in material compliance with applicable securities Laws. The Company has no Company Shares or Company Preferred Shares reserved for issuance, except that as of November 16, 2010, there were an aggregate of 5,986,416 Company Shares reserved for issuance pursuant to the Company Compensation and Benefit Plans identified in Section 5.1(h)(i) of the Company Disclosure Letter as being the only Company Compensation and Benefit Plans pursuant to which Company Shares may be issued (the <u>Company Stock Plans</u>), 10,544,000 Company Shares issuable upon exercise of the Company Warrants and 100,000 Company Preferred Shares reserved for issuance pursuant to the Rights Agreement (the <u>Rights Agreement</u>), dated September 29, 2010, between the Company and Computershare Trust Company, N.A., as Rights Agent. Section 5.1(b)(i) of the Company Disclosure Letter contains a correct and complete list as of October 30, 2010 of (x) each outstanding Company Option and Company Award, (y) each other outstanding right, including those issued under the Company Stock Plans or Company Compensation and Benefits Plans, to receive, the value of which is determined by reference to, Company Shares (including restricted stock and restricted stock units) (each a <u>Common Stock Unit</u>), and (z) each Company Warrant, including, in each case, the holder, date of grant, term, number of Company Shares subject thereto and, where applicable, exercise price and vesting schedule, including whether the vesting will be accelerated by the execution of this Agreement or consummation of the Merger or by termination of employment or change of position following consummation of the Merger. Since October 30, 2010, the Company has not issued any Company Options, Company Awards or Common Stock Units. All outstanding grants of Company Shares, Company Awards and Common Stock Units were made under the Company Stock Plans or Company Compensation and Benefits Plans. Upon any issuance of Company Shares pursuant to any Company Options, Company Awards or Common Stock Units, such Company Shares will be duly authorized, validly issued, fully paid and nonassessable and free and clear of any lien, charge, pledge, security interest, claim or other encumbrance (each, a Lien). Except as set forth in this Section 5.1(b)(i) and pursuant to the Rights Agreement, there are no preemptive or other outstanding rights, options, warrants, conversion rights, stock appreciation rights, redemption rights, repurchase rights, agreements, arrangements, calls, commitments or rights of any kind that obligate the Company or any of its Subsidiaries to issue or sell any shares of capital stock or other securities of the Company or any of its Subsidiaries or any securities or obligations convertible or exchangeable into or exercisable for, or giving any Person a right to subscribe for or acquire, any securities of the Company or any of its Subsidiaries, and no securities or obligations evidencing such rights are authorized, issued or outstanding. The Company does not have outstanding any bonds, debentures, notes or other obligations the holders of which have the right to vote (or which are convertible into or exercisable for securities having the right to vote) with the stockholders of the Company on any matter.

(ii) Each of the outstanding shares of capital stock or other securities of each of the Company s Subsidiaries has been duly authorized and validly issued and is fully paid and nonassessable and owned by the Company or by a direct or indirect wholly-owned Subsidiary of the Company, free and clear of any Liens. Section 5.1(b)(ii) of the Company Disclosure Letter sets forth (x) each of the Company s Subsidiaries

and the ownership interest of the Company in each such Subsidiary, as well as the ownership interest of any other Person or Persons in each such Subsidiary and (y) the Company s or its Subsidiaries capital stock, equity interest or other direct or indirect ownership interest in any other Person. The Company does not own, directly or indirectly, any voting interest in any Person the acquisition of which requires a separate filing by the ultimate parent entity of Parent under the Hart-Scott-Rodino Antitrust Improvement Act of 1976, as amended, and the rules and regulations thereunder (the <u>HSR Act</u>) or any other antitrust or other Laws.

(iii) Each Company Option (A) was granted in material compliance with all applicable Laws and all of the terms and conditions of the Company Stock Plan pursuant to which it was issued, (B) has an exercise price per Company Share equal to or greater than the fair market value of a Company Share on the date of such grant, (C) has a grant date identical to (or later than, as specified by the granting body) the date on which the Company s Board of Directors or compensation committee actually awarded such Company Option, and (D) qualifies for the Tax and accounting treatment afforded to such Company Option in the Company s Tax returns and the Company Reports, respectively.

(iv) Other than as set forth in Section 5.1(b)(iv) of the Company Disclosure Letter, to the knowledge of the Company, as of the date of this Agreement, no Person or group beneficially owns 5% or more of the Company s voting securities, with the terms group and beneficially owns having the meanings ascribed to them under Rule 13d-3 and Rule 13d-5 under the Exchange Act. As used in this Agreement, <u>knowledge</u> of (i) the Company means the actual knowledge of the Company s executive officers after reasonable inquiry of their direct reports and (ii) Parent or Merger Sub means the actual knowledge of such party s executive officers after reasonable inquiry of their direct reports.

(c) Corporate Authority; Approval and Fairness.

(i) Assuming the accuracy of the representation set forth in Section 5.2(k) of this Agreement, the Company has all requisite corporate power and authority and has taken all corporate action necessary in order to execute, deliver and perform its obligations under this Agreement to consummate the Merger, subject only to adoption of this Agreement by the holders of a majority of the outstanding Company Shares entitled to vote on such matter at a stockholders meeting duly called and held for such purpose (the <u>Company Requisite Vote</u>). The holders of the Company Warrants have consented to the treatment of the Company Warrants contemplated by this Agreement. This Agreement has been duly executed and delivered by the Company and constitutes a valid and binding agreement of the Company enforceable against the Company in accordance with its terms, subject to bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar Laws of general applicability relating to or affecting creditors rights and to general equity principles (the <u>Bankruptcy and Equity Except</u>ion).

(ii) The Board of Directors of the Company (A) has unanimously determined that the Merger is fair to, and in the best interests of, the Company and its stockholders, approved and declared advisable this Agreement and the Merger and the other transactions contemplated hereby and thereby and resolved to recommend adoption of this Agreement to the holders of the Company Shares (the <u>Company Recommendation</u>), (B) received the opinion of its financial advisor, UBS Securities LLC, dated as of the date of this Agreement, to the effect that the Merger Consideration to be received by the holders of Company Shares in the Merger is fair to such holders from a financial point of view; it being understood that such opinion is for the benefit of the Company Shares for their approval. Assuming the accuracy of the representation set forth in Section 5.2(k) of this Agreement, the Board of Directors of the Company has taken all action so that Parent will not be an interested stockholder or prohibited from entering into or consummating a business combination with the Company (in each case as such term is used in Section 203 of the DGCL) as a result of the execution of this Agreement or the Support Agreement or the consummation of the transactions in the manner contemplated hereby or thereby.

(d) <u>Governmental Filings: No Violations</u>. (i) Other than the necessary notices, reports, filings, consents, registrations, approvals, permits, authorizations, actions or non-actions (A) pursuant to Section 1.3, (B) required under the HSR Act, the Securities Exchange Act of 1934, as amended (the <u>Exchange Act</u>) and the Securities Act, (C) to comply with state securities or blue-sky laws, (D) to be made with the New York Stock Exchange (the <u>NYSE</u>) and (E) other filings and recordations as required by Governmental Entities other than those in the United States as set forth on Section 5.1(d) of the Company Disclosure Letter, no filings, notices and/or reports are required to be made by the Company with, nor are any consents, registrations, approvals, permits or authorizations required to be obtained by the Company from, any domestic or foreign governmental or regulatory authority, court, agency, commission, body or other legislative, executive or judicial governmental entity (each, a <u>Governmental Entity</u>), in connection with the execution, delivery and performance of this Agreement by the Company and the consummation of the Merger and the other transactions contemplated hereby, or in connection with the continuing operation of the business of the Company and its Subsidiaries following the Effective Time, except those that the failure to make or obtain are not, individually or in the aggregate, reasonably likely to have a Company Material Adverse Effect.

(ii) The execution, delivery and performance of this Agreement by the Company does not, and the consummation of the Merger and the other transactions contemplated hereby will not, constitute or result in (A) a breach or violation of, or a default under, the certificate of incorporation or by-laws of the Company or the comparable governing documents of any of its Subsidiaries, (B) with or without notice, lapse of time or both, a breach or violation of, a default or termination (or right of termination) under, the creation or acceleration of any obligations under, or the creation of a Lien or pledge, security interest or other encumbrance on its assets or the assets of any of its Subsidiaries pursuant to, any agreement, lease, license, contract, note, mortgage, indenture, arrangement or other obligation (each, a <u>Contract</u>) binding upon the Company or any of its Subsidiaries is subject or (C) any change in the rights or obligations of any party under any Contract binding upon the Company or any of its Subsidiaries, except, in the case of clause (B) or (C) above, for any such breach, violation, termination, default, acceleration, creation or change that is not, individually or in the aggregate, reasonably likely to have a Company Material Adverse Effect. Section 5.1(d)(ii) of the Company Disclosure Letter sets forth a correct and complete list of material Contracts of the Company and its Subsidiaries pursuant to which consents or waivers are or may be required prior to consummation of the transactions contemplated by this Agreement (whether or not subject to the exception set forth with respect to clauses (B) and (C) above).

(iii) Other than as described in Section 5.1(d)(iii) of the Company Disclosure Letter, except for (A) relationships with the Company or any of its Subsidiaries as an officer, director, or employee thereof (and compensation by the Company or any of its Subsidiaries in consideration of such services) in accordance with the terms of their employment and (B) relationships with the Company as stockholders or option holders therein, to the knowledge of the Company, none of the directors or officers of the Company or Persons holding more than five percent (5%) of the Company Shares (<u>5% Holders</u>), or any known member of any of their families or affiliates, is presently a party to, or was a party to, during the year preceding the date of this Agreement, any material transaction, agreement or arrangement with the Company or any of its Subsidiaries. To the knowledge of the Company, none of the officers or directors of the Company or 5% Holders has any interest in any material property, real or personal, tangible or intangible, including inventions, copyrights, trademarks, or trade names, used in or pertaining to the business, or any supplier, distributor, or customer of the Company, except for the normal rights of a stockholder, and except for rights under the Company Stock Plans and the Company Warrants.

(iv) Neither the Company nor any of its Subsidiaries is a party to or bound by any non-competition Contracts or other Contract that purports to limit in any material respect either the type of business in which the Company or its Subsidiaries (or, after giving effect to the Merger, Parent or its Subsidiaries) may engage or the manner or locations in which any of them may so engage in any business.

(v) The Company and its Subsidiaries are not creditors or claimants with respect to any debtor or debtor-in-possession subject to proceedings under chapter 11 of title 11 of the United States Code with respect to claims that, in the aggregate, constitute more than 25% of the gross assets of the Company and its Subsidiaries (excluding cash and cash equivalents).

(e) Reports: Financial Statements. (i) The Company has made available to Parent each registration statement, report, proxy statement or information statement prepared by it since December 31, 2009 (the <u>Audit Date</u>) and filed with or furnished to the SEC, including (x) its Annual Report on Form 10-K for the year ended December 31, 2009, and (y) its Quarterly Reports on Form 10-Q for the quarterly periods ended March 31, 2010, June 30, 2010 and September 30, 2010, each in the form (including exhibits, annexes and any amendments thereto) filed with or furnished to the SEC. The Company has filed or furnished, as applicable, on a timely basis all forms, statements, reports and documents required to be filed or furnished by it with or to the SEC pursuant to the Exchange Act or the Securities Act since December 31, 2006 (the _Applicable Date) (collectively, the forms, statements, reports and documents filed with or furnished to the SEC since the Applicable Date, and those filed with or furnished to the SEC subsequent to the date of this Agreement, including any amendments thereto, the <u>Company Reports</u>). Each of the Company Reports was prepared in all material respects in accordance with the applicable requirements of the Securities Act, the Exchange Act and the rules and regulations thereunder and complied in all material respects with then applicable accounting standards. Each of the Company Reports, at the time of its filing or being furnished complied, or if not yet filed or furnished, will comply, in all material respects with the applicable requirements of the Securities Act, the Exchange Act and the Sarbanes-Oxley Act of 2002 (<u>Sarbanes-Oxley</u>), and any rules and regulations promulgated thereunder applicable to the Company Reports. As of their respective dates (or, if amended prior to the date of this Agreement, as of the date of such amendment) the Company Reports did not, and any Company Reports filed with or furnished to the SEC subsequent to the date of this Agreement will not, contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements made therein, in light of the circumstances in which they were made, not misleading, except to the extent that the information in such Company Report has been amended or superseded by a later Company Report filed prior to the date of this Agreement.

(ii) The Company is in compliance in all material respects with the applicable listing and corporate governance rules and regulations of the NYSE. Except as permitted by the Exchange Act, including Sections 13(k)(2) and (3) thereof or rules of the SEC, since the enactment of Sarbanes-Oxley, neither the Company nor any of its affiliates (as defined in Rule 405 promulgated under the Securities Act. Affiliates) has made, arranged or modified (in any material way) any extension of credit in the form of a personal loan to any executive officer or director of the Company.

(iii) The Company maintains effective disclosure controls and procedures as required by Rule 13a-15 or 15d-15 under the Exchange Act. The Company maintains effective internal control over financial reporting (as defined in Rule 13a-15 or 15d-15, as applicable, under the Exchange Act). The Company has disclosed to the Company s auditors and the audit committee of the Company s Board of Directors (A) any significant deficiencies in the design or operation of its internal controls over financial reporting and has identified for the Company s auditors and audit committee of the Company s Board of Directors any material weaknesses in internal control over financial reporting and (B) any fraud, whether or not material, that involves management or other employees who have a significant role in the Company s internal control over financial reporting. The Company has made available to Parent (i) a summary of any such disclosure made by management or the Company s auditors and audit committee since the Audit Date and (ii) any material communication since the Audit Date made by management or the Company s auditors to the audit committee required or contemplated by listing standards of the NYSE, the audit committee s charter or professional standards of the Public Company Accounting Oversight Board. Since the Audit Date, no material complaints from any source regarding accounting, internal accounting controls or auditing matters, and no concerns from Company employees regarding questionable accounting or auditing matters, have been received by the Company. The Company has made available to Parent prior to the date of this Agreement a summary of all material complaints or concerns relating to other matters made since the Audit

Date through the Company s whistleblower hot-line or equivalent system for receipt of employee concerns regarding accounting matters or possible violations of Law. No attorney representing the Company or any of its Subsidiaries, whether or not employed by the Company or any of its Subsidiaries, has reported evidence of a violation of securities Laws, breach of fiduciary duty or similar violation by the Company or any of its officers, directors, employees or agents to the Company s chief legal officer, audit committee (or other committee designated for the purpose) of the Board of Directors or the Board of Directors pursuant to the rules adopted pursuant to Section 307 of Sarbanes-Oxley or any Company policy contemplating such reporting, including in instances not required by those rules.

(iv) Each of the consolidated balance sheets included in or incorporated by reference into any Company Reports (including the related notes and schedules) fairly presents, or, in the case of Company Reports filed after the date hereof, will fairly present the consolidated financial position of the Company and its consolidated Subsidiaries as of its date and each of the consolidated statements of income, changes in stockholders equity (deficit) and cash flows included in or incorporated by reference into the Company Reports (including any related notes and schedules) fairly presents, or in the case of the Company Reports filed after the date hereof, will fairly present the results of operations, retained earnings (loss) and changes in financial position, as the case may be, for the periods set forth therein (subject, in the case of unaudited statements, to notes that comply with the requirements of Rule 3-10 of Regulation S-X only and normal year-end audit adjustments that will not be material in amount or effect), in each case in accordance with U.S. generally accepted accounting principles (<u>GAAP</u>) consistently applied during the periods involved, except as may be noted therein.

(v) The Company has previously furnished to Parent a complete and correct copy of any amendments or modifications, which have not yet been filed with the SEC but which are required to be filed, to agreements, reports, schedules, documents or other instruments which previously had been filed by the Company with the SEC pursuant to the Securities Act or the Exchange Act.

(vi) Except as reflected or reserved against in the Company s most recent consolidated balance sheet (or the notes thereto) included in the Company Reports, neither the Company nor any Subsidiary of the Company has any liabilities or obligations of any nature, whether or not accrued, contingent or otherwise, whether known or unknown and whether due or to become due, that would, individually or in the aggregate, have a Company Material Adverse Effect, and, except for this Agreement and the transactions specifically provided for herein, since the Audit Date, the Company and its Subsidiaries have not incurred any material liability other than in the ordinary course of business consistent with past practice.

(vii) The Company s revenue recognition policies and practices are and have been in compliance with all rules, regulations and statements of the SEC with respect thereto, including SAB 101, <u>Revenue Recognition in Financial Statements</u>, and SAB 104. <u>Revenue Recognition</u>, and, to the extent required, the Company recognizes revenue in accordance with Accounting Standards Codification (ASC) 985, <u>Certain Revenue</u> <u>Arrangements That Contain Software Elements</u>, and ASC 605-25, <u>Revenue Recognition Multiple Element Arrangements</u>; and the Company maintains adequate controls over its revenue recognition policies and practices all of which are properly communicated to and applied by its sales organizations.

(f) <u>Absence of Certain Changes</u>. Other than as set forth in Section 5.1(f) of the Company Disclosure Letter, since the Audit Date, the Company and its Subsidiaries have conducted their respective businesses only in, and have not engaged in any material transaction other than as contemplated by this Agreement or in accordance with, the ordinary course of such businesses consistent with past practices, and there has not been:

(i) any event, development or circumstance involving, or any change in the financial condition, assets, liabilities, business or results of their operations or any circumstance, occurrence or development (including any adverse change with respect to any circumstance, occurrence or development existing on or prior to the most recent fiscal year end) which, individually or in the aggregate, has had or would reasonably be expected to have a Company Material Adverse Effect;

(ii) any damage, destruction, or other casualty loss with respect to any asset or property owned, leased or otherwise used by the Company or any of its Subsidiaries whether or not covered by insurance, which, individually or in the aggregate, has had or would reasonably be expected to have a Company Material Adverse Effect;

(iii) any declaration, setting aside or payment of any dividend or other distribution with respect to any shares of capital stock of the Company or any of its Subsidiaries (except for dividends or other distributions by any direct or indirect wholly-owned Subsidiary to the Company or to any wholly-owned Subsidiary of the Company) or any repurchase, redemption or other acquisition by the Company or any of its Subsidiaries of any outstanding shares of capital stock or other securities of the Company or any of its Subsidiaries;

(iv) any incurrence of any material indebtedness for borrowed money by the Company or any of its Subsidiaries and neither the Company nor any of its Subsidiaries has guaranteed such indebtedness of another Person, or issued or sold any debt securities or warrants or other rights to acquire any debt security of the Company or any of its Subsidiaries;

(v) any incurrence or authorization of capital expenditures in excess of \$15,000,000 in the aggregate;

(vi) any change in any method of accounting or accounting practices by the Company or any of its Subsidiaries;

(vii) any making, changing or rescinding of any material election in respect of Taxes by the Company or any of its Subsidiaries or changing of any method of Tax accounting by the Company or any of its Subsidiaries;

(viii) any transfer, lease, license, sale, mortgage, pledge, placement of a Lien upon or other disposition of any of the Company s or its Subsidiaries property or assets (including capital stock of any of the Company s Subsidiaries) with a fair market value in excess of \$10,500,000;

(ix) (A) any increase in the compensation payable or to become payable to its officers or employees (except for increases in the ordinary course of business and consistent with past practice) or (B) any establishment, adoption, entry into or amendment of any collective bargaining, bonus, profit sharing, thrift, compensation, employment, termination, severance or other plan, agreement, trust, fund, policy or arrangement for the benefit of any director, officer or employee, except to the extent required by applicable Laws;

(x) any event, development or circumstance as a result of which the Company incurred a material Tax liability (or material decrease in a material Tax asset or attribute) not in the ordinary course of business; or

(xi) any agreement to do any of the foregoing.

(g) <u>Litigation and Liabilities</u>. Except as set forth in Section 5.1(g) of the Company Disclosure Letter, there are no (i) civil, criminal or administrative actions, suits, claims, hearings, arbitrations, investigations or other proceedings pending or, to the Company s knowledge, threatened against the Company or any of its Subsidiaries, except for those that would not, individually or in the aggregate, reasonably be expected to have a Company Material Adverse Effect or (ii) obligations or liabilities of the Company or any of its Subsidiaries, whether or not accrued, contingent or otherwise, and whether or not required to be disclosed in a Company Report or any other facts or circumstances known to the Company that could reasonably be expected to result in any claims against, or obligations or liabilities of, the Company or any of its Subsidiaries, including those relating to environmental and occupational safety and health matters, except for those that, individually or in the aggregate, have not had and would not reasonably be expected to have, a Company Material Adverse Effect. Neither the Company nor any of its Subsidiaries is a party to or subject to the provisions of any judgment, order, writ, injunction, decree or award of any Governmental Entity which, individually or in the aggregate, have had, or would reasonably be expected to have, a Company Material Adverse Effect.

(h) Employee Benefits.

(i) Section 5.1(h)(i) of the Company Disclosure Letter sets forth a true and complete list of each Company Compensation and Benefit Plan sponsored or maintained with respect to or for the benefit of employees of the Company or its Subsidiaries, other than broad-based employee welfare and similar plans that do not provide for salary, bonus, incentive or equity compensation or severance arrangements. For purposes of this Agreement, the term <u>Company Compensation and Benefit Plan</u> shall mean any employee or director benefit plan, arrangement or agreement, including any such plan that is an employee welfare benefit plan within the meaning of Section 3(1) of the Employee Retirement Income Security Act of 1974, as amended (<u>ERISA</u>), an employee pension benefit plan within the meaning of Section 3(2) of ERISA (whether or not such plan is subject to ERISA) or a non-*de minimis* bonus, incentive, deferred compensation, vacation, stock purchase, stock option, stock-based severance, employment, change of control or fringe benefit plan, program or agreement that is sponsored or maintained by the Company or any of its Subsidiaries to or for the benefit of the current or former employees, independent contractors or directors of the Company and its Subsidiaries other than Company Compensation and Benefit Plans maintained outside of the U.S. primarily for the benefit of employees working outside of the U.S. (such plans <u>Company Non-U.S. Compensation and Benefit Plans</u>). Section 5.1(h)(i)(A) of the Company Disclosure Letter sets forth for each individual party to an agreement providing severance or termination benefits or entitled to severance or termination benefits under the 2007 change in control severance arrangement or the Company s 2010 Change of Control Plan, the current salary of such individual.

(ii) For each Company Compensation and Benefit Plan listed in Section 5.1(h)(i) of the Company Disclosure Letter, the Company has heretofore made available to Parent true and complete copies of each of Company Compensation and Benefit Plans and (A) each writing constituting a part of such Company Compensation and Benefit Plan, including all amendments thereto; (B) the most recent (x) Annual Reports (Form 5500 Series) and accompanying schedules, if any, (y) audited financial statements and (z) actuarial valuation reports; (C) the most recent determination letter from the Internal Revenue Service (<u>IRS</u>) (if applicable) for such Company Compensation and Benefit Plan; and (D) any related trust agreement or funding instrument now in effect or required in the future as a result of the transactions contemplated by this Agreement.

(iii) (A) Each of the Company Compensation and Benefit Plans has been established, operated and administered in all material respects in compliance with its terms and applicable Laws, including, but not limited to, ERISA, the Code and, in each case, the regulations thereunder; (B) each of the Company Compensation and Benefit Plans intended to be qualified within the meaning of Section 401(a) of the Code has received a favorable determination letter from the IRS, and to the knowledge of the Company, there are no existing circumstances or events that have occurred that could reasonably be expected to result in the revocation of such letter; (C) no Company Compensation and Benefit Plan is subject to Title IV of ERISA; (D) no Company Compensation and Benefit Plan provides health, life insurance or disability benefits (whether or not insured), with respect to current or former employees or directors of the Company or its Subsidiaries beyond their retirement or other termination of service, other than (x) coverage mandated by applicable Law or (y) death benefits or retirement benefits under any employee pension plan (as such term is defined in Section 3(2) of ERISA); (E) no material liability under Title IV of ERISA has been incurred by the Company, its Subsidiaries or any ERISA Affiliate of the Company that has not been satisfied in full, and, to the knowledge of the Company, no condition exists that presents a risk to the Company, its Subsidiaries or any ERISA Affiliate of the Company of incurring a material liability thereunder; (F) no Company Compensation and Benefit Plan is a multiemployer pension plan (as such term is defined in Section 3(37) of ERISA) or a plan that has two or more contributing sponsors at least two of whom are not under common control, within the meaning of Section 4063 of ERISA; (G) all contributions or other amounts payable by the Company or its Subsidiaries as of the date hereof with respect to each Company Compensation and Benefit Plan in respect of current or prior plan years have been paid or accrued in accordance with GAAP; (H) neither the Company nor its Subsidiaries has engaged in a transaction in connection with which the Company or its Subsidiaries could reasonably be expected to be

subject to either a material civil penalty assessed pursuant to Section 409 of ERISA or a material Tax imposed pursuant to Section 4975 of the Code; (I) there are no material pending, threatened or anticipated claims (other than routine claims for benefits) by, on behalf of or against any of the Company Compensation and Benefit Plans or any trusts related thereto which could reasonably be expected to result in any liability of the Company or any of its Subsidiaries; and (J) the Company or its Subsidiaries may amend or terminate any Company Compensation and Benefit Plan at any time without incurring any liability thereunder other than in respect of claims incurred prior to such amendment or termination. <u>ERISA Affiliate</u> means, with respect to any entity, trade or business, any other entity, trade or business that is a member of a group described in Section 414(b), (c), (m) or (o) of the Code or Section 4001(b)(1) of ERISA that includes the first entity, trade or business, or that is a member of the same controlled group as the first entity, trade or business pursuant to Section 4001(a)(14) of ERISA.

(iv) Since the Audit Date, there has been no amendment to, announcement by the Company or any of its Subsidiaries relating to, or change in employee participation or coverage under, any of the Company Compensation and Benefit Plans that would increase materially the expense of maintaining such plan above the level of the expense incurred therefor for the most recent fiscal year. Except as set forth in Section 5.1(h)(iv) of the Company Disclosure Letter, none of the execution of this Agreement, stockholder approval of this Agreement, receipt of approval or clearance from any one or more Governmental Entities of the Merger or the other transactions contemplated by this Agreement, or the consummation of the Merger and the other transactions contemplated by this Agreement either alone or together with any other event will, (A) entitle any employees of the Company or any of its Subsidiaries to severance pay or any increase in severance pay upon any termination of employment after the date of this Agreement, (B) accelerate the time of payment or vesting or result in any payment or funding (through a grantor trust or otherwise) of compensation or benefits under, increase the amount payable or result in any other material obligation pursuant to, any of the Company Compensation and Benefit Plans, (C) limit or restrict the right of the Company Compensation and Benefit Plans, (D) entitle any employees of the Company or any of its Subsidiaries to any tax gross-up or other reimbursement of taxes or tax penalties or (E) result in payments under any of the Company Compensation and Benefit Plans which would not be deductible under Section 162(m) or Section 280G of the Code.

(v) All of the Company Non-U.S. Compensation and Benefit Plans sponsored or maintained with respect to or for the benefit of employees of the Company or its Subsidiaries outside of the United States, other than broad-based employee welfare and similar plans that do not provide for salary, bonus, incentive or equity compensation or severance arrangements are listed in Section 5.1(h)(v) of the Company Disclosure Letter and comply with applicable local Law, except as would not reasonably be expected to have a Company Material Adverse Effect. The Company and its Subsidiaries have no unfunded liabilities with respect to any such Company Non-U.S. Compensation and Benefit Plans that are not set forth in the consolidated balance sheets included in or incorporated by reference into Company Reports filed prior to the date of this Agreement, there is no pending or, to the knowledge of the Company, threatened litigation relating to the Company Non-U.S. Compensation and Benefit Plans.

(vi) All Company Compensation and Benefit Plans that are nonqualified deferred compensation plans (within the meaning of Section 409A of the Code) have been maintained and administered in all respects in compliance with the requirements of Section 409A of the Code and any regulations or other guidance issued thereunder.

(i) <u>Compliance with Laws</u>. (i) The businesses of each of the Company and its Subsidiaries have not been, and are not being, conducted in violation of any federal, state, local or foreign law, statute or ordinance, or any rule, regulation, standard, judgment, determination, order, writ, decree, injunction, arbitration award, license, authorization, opinion, agency requirement or permit of any Governmental Entity or common law (collectively, <u>Laws</u>), except for violations that individually or in the aggregate, have not had and would not reasonably be expected to have a Company Material Adverse Effect. To the knowledge of the Company, no investigation or review by any Governmental Entity with respect to the Company or any of its Subsidiaries is pending or

threatened, nor has any Governmental Entity indicated an intention to conduct the same. No material change is required in the Company s or any of its Subsidiaries processes, properties or procedures in connection with any such Laws, and the Company has not received any notice or communication of any material noncompliance with any such Laws that has not been cured as of the date of this Agreement, except for such changes and noncompliance that, individually or in the aggregate, have not had and would not reasonably be expected to have a Company Material Adverse Effect.

(ii) Neither the Company nor any of its Subsidiaries maintains or conducts, and since January 1, 2007 has maintained or conducted, any business, investment, operation or other activity in or with: (i) any country or person targeted by any of the economic sanctions of the United States of America administered by the United States Treasury Department s Office of Foreign Assets Control; (ii) any person appearing on the list of Specially Designated Nationals and Blocked Persons issued by the United States Treasury Department s Office of Foreign Assets Control; or (iii) any country or person designated by the United States Secretary of the Treasury pursuant to the USA PATRIOT Act as being of primary money laundering concern .

(iii) Neither the Company nor any of its Subsidiaries has collected, used, transferred or disclosed any personally identifiable information in violation of any privacy policy, statement, or agreement maintained by the Company or any of its Subsidiaries, except for violations that, individually or in the aggregate, have not had and would not reasonably be expected to have a Company Material Adverse Effect. The Company and each of its Subsidiaries has commercially reasonable security measures and safeguards in place to protect personally identifiable information from illegal or unauthorized access, download or use by its personnel or third parties and, to the knowledge of the Company, no such illegal or unauthorized access, downloading or use has occurred.

(iv) The Company and each of its Subsidiaries is in compliance in all respects with all Laws applicable to the transfer of personally identifiable information, except for such noncompliance that, individually or in the aggregate, have not had and would not reasonably be expected to have a Company Material Adverse Effect.

(j) <u>Takeover Statutes</u>. Assuming the accuracy of the representation and warranty of Parent contained in Section 5.2(k), the Company has taken all actions required to ensure that no fair price, moratorium, control share acquisition or other similar anti-takeover statute or regulation (each, a <u>Takeover Statute</u>) or any anti-takeover provision contained in the Company s certificate of incorporation or by-laws is, or at the Effective Time will be, applicable to the Company, the Company Shares, the Merger or the other transactions contemplated by this Agreement.

(k) Environmental Matters. (i) Except as would not, individually or in the aggregate, reasonably be expected to have a Company Material Adverse Effect, (a) each of the Company and its Subsidiaries has complied at all times with all applicable Environmental Laws; (b) the properties currently owned, leased or operated by the Company or any of its Subsidiaries (including soils, groundwater, surface water, buildings or other structures) are not contaminated with any Hazardous Substances; (c) the properties formerly owned, leased or operated by the Company or any of its Subsidiaries were not contaminated with Hazardous Substances during the period of ownership, tenancy or operation by the Company or any of its Subsidiaries; (d) neither the Company nor any of its Subsidiaries is subject to liability for any Hazardous Substance disposal or contamination on any third-party property; (e) neither the Company nor any of its Subsidiaries has been associated with any release or threat of release of any Hazardous Substance; (f) neither the Company nor any of its Subsidiaries has received any notice, demand, letter, claim or request for information alleging that the Company or any of its Subsidiaries may be in violation of or subject to liability under any Environmental Law (including any claims relating to electromagnetic fields or microwave transmissions); (g) neither the Company nor any of its Subsidiaries is subject to any indemnity or other agreement with any third party relating to liability under any Environmental Law or relating to Hazardous Substances; (h) there are no circumstances or conditions involving the Company or any of its Subsidiaries that could reasonably be expected to result in any claim, liability, investigation, cost or restriction on the ownership, use, or transfer of any of its properties pursuant to any Environmental Law; and (i) the Company

has delivered to Parent copies of all environmental reports, studies, assessments, sampling data and other environmental information in its possession relating to Company or its Subsidiaries or their respective current and former properties or operations.

(ii) Except as would not reasonably be expected to have a Company Material Adverse Effect, each of the Company s products does and has complied with any and all applicable Laws pertaining to the presence (or absence) of specified substances in electrical or electronic or other products; registration or notification of chemical substances in products; labeling of product or product packaging as respects product content or as respects health, safety or environmental effects or attributes or as respects required end-of-life handling or disposition of products or product packaging; and coverage and payment of fees under an approved scheme for end-of-life, return and recycling of products or of product packaging. Without limiting the foregoing, to the Company s knowledge each of the products sold by the Company in the European Union meets, in all material respects, the restrictions of the European Union s Directive 2002/95/EC on the Restriction of the Use of Certain Hazardous Substances in Electronic Equipment (the <u>RoHS Directive</u>) and the European Union s Directive 2002/96/EC on Waste Electrical and Electronic Equipment (as amended by European Directive 2003/108/EC) (<u>WEEE Directive</u>). To the Company s knowledge, the products currently sold by the Company fully comply with the requirements of the RoHS Directive and WEEE Directive, including any amending directives thereto, and are applicably marked.

As used herein, the term <u>Environmental La</u>w means any Law relating to (A) the protection, investigation or restoration of the environment, (B) the handling, use, presence, disposal, release or threatened release of any Hazardous Substance or (C) noise, odor, indoor air, employee exposure, wetlands, pollution, contamination or any injury or threat of injury to persons or property relating to any Hazardous Substance.

As used herein, the term <u>Hazardous Substance</u> means any substance that is (A) listed, classified or regulated pursuant to or in connection with any Environmental Law and (B) any petroleum product or by-product, asbestos-containing material, lead-containing paint or plumbing, polychlorinated biphenyls, mold, radioactive material or radon.

(l) Taxes.

(i) The Company and each of its Subsidiaries have prepared and timely filed (taking into account any valid extension of time within which to file) all material Tax Returns required to be filed by or with respect to any of them and all such Tax Returns are complete and accurate in all material respects;

(ii) the Company and each of its Subsidiaries have timely paid all material Taxes that are required to be paid by any of them (whether or not shown on any Tax Return), except with respect to matters contested in good faith through appropriate proceedings and for which adequate reserves have been established on the financial statements of the Company and its Subsidiaries in accordance with GAAP;

(iii) other than as set forth in Section 5.1(1)(iii) of the Company Disclosure Letter, there is no pending audit, examination, investigation, dispute, proceeding or claim relating to any Tax on the Company or any of its Subsidiaries (collectively, a <u>Proceeding</u>) and neither the Company nor any of its Subsidiaries has received any written notice that a Proceeding is pending or threatened;

(iv) all assessments for material Taxes due with respect to completed and settled examinations or any concluded litigation have been fully paid;

(v) no Tax authority has asserted any adjustment that would result in an additional material Tax on the Company or any of its Subsidiaries which has not been fully paid or which adjustment, if asserted with respect to another Tax period, would result in an additional Tax on the Company or on any of its Subsidiaries or in the loss or reduction of a Tax asset or attribute of the Company or of any of its Subsidiaries;

(vi) there are no Liens for Taxes on any of the assets of the Company or any of its Subsidiaries other than statutory Liens for Taxes not yet due and payable or Liens for Taxes that are being contested in good faith through appropriate proceedings and for which adequate reserves have been established on the financial statements of the Company and its Subsidiaries in accordance with GAAP;

(vii) the Company has made a provision in accordance with GAAP, in the financial statements of the Company, for all Taxes that accrued on or before the end of the most recent period covered in such financial statements;

(viii) none of the Company or any of its Subsidiaries has been a controlled corporation or a distributing corporation in any distribution that was purported or intended to be governed by Section 355 of the Code (or any similar provision of state, local or foreign Law) occurring during the two-year period ending on the date hereof;

(ix) the Company and its Subsidiaries have withheld or collected and paid over to the appropriate Taxing authority or deposited in accordance with applicable Laws in a timely manner all Taxes required to have been withheld or collected and paid or deposited in connection with amounts paid or owing to any employee, independent contractor, creditor, stockholder or other third party;

(x) neither the Company nor any of its Subsidiaries has executed any waiver of any statute of limitations on or extension of the period for the assessment or collection of any Tax, and no such waivers of statutes of limitations or extensions have been requested from the Company or any Subsidiary;

(xi) neither the Company nor any of its Subsidiaries (a) has ever been a member of an affiliated group (within the meaning of Section 1504(a) of the Code) filing a consolidated federal income Tax Return (other than a group the common parent of which was the Company), (b) has ever been a member of an affiliated, combined, consolidated or unitary Tax group for purposes of filing any foreign, state or local Tax Return, (c) owes any amount under any Tax sharing, indemnification or allocation agreement, (d) is or has ever been a party to any Tax sharing or Tax allocation agreement, arrangement or understanding or (e) has any liability for the Taxes of any Person (other than the Company or any of its Subsidiaries) under Treasury Regulations Section 1.1502 6 (or any similar Law), as a transferee or successor, by contract, or otherwise;

(xii) no closing agreements, private letter rulings, technical advance memoranda or similar agreement or rulings have been entered into or issued by any Taxing authority with respect to the Company or any of its Subsidiaries, or the business of the Company or any of its Subsidiaries and neither the Company nor any of its Subsidiaries has outstanding any ruling request, request for consent to change a method of accounting, subpoena or request for information with or from a Taxing authority in connection with any Tax matter; and

(xiii) neither the Company nor any of its Subsidiaries (including current or former subsidiaries) (A) has engaged or entered into any transaction that is the same as, or substantially similar to, a transaction which is a reportable transaction within the meaning of Section 6707A(c) of the Code or Treasury Regulations Section 1.6011-4 or any predecessor thereof, including any transaction which the IRS has determined to be a listed transaction for purposes of Treasury Regulations Section 1.6011-4(b)(2), or would be reportable to a similar extent under any other provision of state, local or foreign Tax law, (B) has engaged in a transaction of which it made disclosure to any taxing authority to avoid penalties, or (C) has participated in any tax amnesty or similar program offered by any Taxing authority to avoid the assessment of penalties or other additions to Tax.

As used in this Agreement, (i) the term <u>Tax</u> (including, with correlative meaning, the ter<u>m</u> <u>Taxes</u>) includes all federal, state, local and foreign income, profits, franchise, gross receipts, environmental, customs duty, capital stock, severance, stamp, payroll, sales, employment, unemployment, disability, use, property, withholding, excise, production, value added, occupancy and other taxes, duties or assessments imposed by a Governmental Entity, together with all interest, penalties and additions imposed with respect to such amounts and any interest in respect of such penalties and additions, and (ii) the term <u>Tax Return</u> includes all returns and reports (including elections, declarations, disclosures, schedules, estimates and information returns) required to be supplied to a Tax authority relating to Taxes.

(m) Labor Matters. Neither the Company nor any of its Subsidiaries is a party to or otherwise bound by any collective bargaining agreement or other Contract with a labor union, labor organization, works council or employer organization, nor is the Company or any of its Subsidiaries the subject of any material proceeding that asserts that the Company or any of its Subsidiaries has committed an unfair labor practice or that seeks to compel it to bargain with any labor union, works council or labor organization nor is there pending or, to the Company s knowledge, threatened, nor has there been for the past five years, any labor strike, dispute, walk-out, work stoppage, slow-down or lockout involving the Company or any of its Subsidiaries. To the Company s knowledge, there are no organizational efforts with respect to the formation of a collective bargaining unit presently being made involving employees of the Company or any of its Subsidiaries. The Company has previously made available to Parent correct and complete copies of all labor, works council and collective bargaining agreements, Contracts or other agreements or understandings with a labor union, works council or labor organization to which the Company or any of its Subsidiaries is party or by which any of them are otherwise bound (collectively, the <u>Company Labor Agreements</u>). The consummation of the Merger and the other transactions contemplated by this Agreement will not entitle any third party (including any labor union, works council or labor organization) to any payments under any of the Company Labor Agreements. The Company and its Subsidiaries have complied in all material respects with the reporting requirements of the Labor Management Reporting and Disclosure Act.

(n) Intellectual Property.

(i) Each of the Company and its Subsidiaries has sufficient rights, through ownership or license, to use all material Intellectual Property used in its business as presently conducted and to be used in its business as proposed to be conducted, all of which rights shall survive unchanged the consummation of the transactions contemplated by this Agreement. Section 5.1(n)(i) of the Company Disclosure Letter sets forth a true and complete list of all (x) Registered and/or material Intellectual Property owned by the Company and its Subsidiaries, indicating for each Registered item the registration or application number and the applicable filing jurisdiction and (y) Intellectual Property Contracts (other than licenses for commercial off-the-shelf or shrink-wrap Software that has not been modified or customized for the Company). Each of the Company and its Subsidiaries has exclusive ownership of all material Intellectual Property owned by it, free and clear of all Liens, non-exclusive licenses (other than those granted in the ordinary course of business) and exclusive licenses. The Intellectual Property owned by the Company and its Subsidiaries is valid, subsisting and enforceable, and is not subject to any outstanding order, judgment, decree or agreement adversely affecting the Company s or any of its Subsidiaries use thereof or its rights thereto. To the knowledge of the Company and its Subsidiaries, neither the Company nor any of its Subsidiaries, nor any of the products or services of the Company or any of its Subsidiaries, infringes, misappropriates or otherwise violates, or has infringed, misappropriated or otherwise violated the Intellectual Property rights of any third party during the five (5) year period immediately preceding the date of this Agreement, and, except as set forth in Section 5.1(n)(i) of the Company Disclosure Letter, neither the Company nor any of its Subsidiaries has received any notice or claim challenging the Company s or any Subsidiary s ownership of any of the Intellectual Property owned by the Company or any Subsidiary or claiming that the Company or any Subsidiary infringes or misappropriates the Intellectual Property of any third party.

(ii) The Company and its Subsidiaries have taken all reasonable measures to protect the confidentiality and value of all Trade Secrets that are owned, used or held by the Company and its Subsidiaries, and to the Company s knowledge, such Trade Secrets have not been used, disclosed to or discovered by any person, except pursuant to valid and appropriate non-disclosure and/or license agreements which have not been breached. To the Company s knowledge, none of the Company s or its Subsidiaries current employees or subcontractors has any patents issued or patent applications pending for any device, process, design or invention of any kind now used in, or required by the Company or its Subsidiaries to be used in the Company s or such Subsidiary s business, other than patents or applications that have been assigned to the Company pursuant to valid intellectual property and confidentiality agreements. All of the Company s and its Subsidiaries current and prior employees have executed valid intellectual property and confidentiality agreements for the benefit of the Company in a form which the Company has prior to the date of this

Agreement provided to Parent for its review. Every contract or agreement under which Intellectual Property was developed, created or otherwise made, for the Company or any of its Subsidiaries assigns or licenses all rights to such Intellectual Property to the Company to the extent allowed by applicable Law. To the knowledge of the Company, the Company s and its Subsidiaries employees performance of their employment activities does not violate any third party s Intellectual Property rights or such employees contractual obligations to any third person.

(iii) Neither the Company nor any of its Subsidiaries has granted any licenses or other rights to third parties to use the Company s or its Subsidiaries Intellectual Property other than non-exclusive licenses granted in the ordinary course of business pursuant to standard terms which have been previously provided to Parent and exclusive licenses set forth in Section 5.1(n)(iii) of the Company Disclosure Letter.

(iv) The IT Assets owned, used or held for use by the Company or any of its Subsidiaries operate and perform in all material respects in accordance with their documentation and functional specifications and otherwise as required by the Company and its Subsidiaries in connection with their business and have not materially malfunctioned or failed within the past three (3) years. To the Company s knowledge, no person has gained unauthorized access to the IT Assets. The Company and its Subsidiaries have implemented reasonable backup and disaster recovery technology consistent with industry practices.

(v) The Company and its Subsidiaries have source code for each version of Software owned by them and used in the past five (5) years. The source code for such Software will compile into object code or otherwise is capable of being installed and operated. Once compiled and/or installed, such Software will have the features, functions and performance described in the documentation pertaining to it and will execute on the computer platforms for which it is designed. All source code and other documentation concerning such Software is sufficiently documented to enable a Software developer of reasonable skill to understand, modify, debug, enhance, compile, support and otherwise utilize all aspects of Software to which it pertains, without reference to other sources of information. None of the Software owned by the Company or any of its Subsidiaries contains any shareware, open source code, or other Software or is subject to the terms or conditions of any open source license (including but not limited to any General Public License) whose use requires or imposes on the Company or any of its Subsidiaries any obligation with respect to any Intellectual Property or Software that is owned by the Company or any of its Subsidiaries, and further, the Company and its Subsidiaries are in compliance with all terms and conditions of any such open source licenses.

(vi) Section 5.1(n)(vi) of the Company Disclosure Letter contains a complete and accurate list, together with applicable Export Control Classification Number (ECCN), of all Software (other than Software embedded in terminal products) that is sold, licensed, leased or otherwise distributed by the Company or its Affiliates or resellers (the <u>Software Products</u>) indicating in each case, the name, owner and most recent version of the Software Product.

(vii) As used herein,

(1) <u>Intellectual Property</u> means all (i) trademarks, service marks, brand names, certification marks, collective marks, d/b/a s, Internet domain names, logos, symbols, trade dress, assumed names, fictitious names, trade names, and other indicia of origin, all applications and registrations for the foregoing, and all goodwill associated therewith and symbolized thereby, including all renewals of same; (ii) inventions and discoveries, whether patentable or not, and all patents, registrations, invention disclosures and applications therefor, including divisions, continuations, continuations-in-part and renewal applications, and including renewals, extensions and reissues; (iii) confidential information, trade secrets and know-how, including processes, schematics, business methods, formulae, drawings, prototypes, models, designs, customer lists and supplier lists (collectively, <u>Trade Secrets</u>); (iv) published and unpublished works of authorship, whether copyrightable or not (including databases and other compilations of information), copyrights therein and thereto, and registrations and applications therefor, and all renewals, extensions, restorations and reversions thereof; and (v) moral rights and all other intellectual property or proprietary rights.

(2) <u>IT Assets</u> means the Company and its Subsidiaries computers, computer software, firmware, middleware, servers, workstations, routers, hubs, switches, data communications lines, and all other information technology equipment, and all associated documentation.

(3) <u>Registered</u> means issued by, registered with, renewed by or the subject of a pending application before any Governmental Entity or Internet domain name registrar.

(4) <u>Software</u> means all software and databases (including source code, object code and all related documentation).

(o) <u>Affiliate Transactions</u>. As of the date of this Agreement, there are no transactions, arrangements or Contracts between the Company and its Subsidiaries, on the one hand, and its Affiliates (other than its wholly-owned Subsidiaries) or other Persons, on the other hand, that would be required to be disclosed under Item 404 of Regulation S-K under the Securities Act.

(p) <u>Insurance</u>. The Company has made available to Parent prior to the date of this Agreement true, correct and complete copies of the Company s director and officer and error and omissions insurance policies and all other material policies of insurance to which the Company or any of its Subsidiaries is a beneficiary or named insured (<u>Insurance Policies</u>). Each Insurance Policy is in full force and effect and all premiums due with respect to all Insurance Policies have been paid.

(q) <u>Material Contracts</u>. (i) As of the date of this Agreement, except as described in the relevant subsection of Section 5.1(q)(i) of the Company Disclosure Letter, neither the Company nor any of its Subsidiaries is a party to or bound by:

(A) any lease of real or personal property providing for annual rentals of \$1,000,000 or more;

(B) other than purchase orders or agreements related to the sale of the Company s products in the ordinary course of business, including end-of-life raw materials and finished goods inventory purchases, (w) any Contract for the purchase of raw materials that is reasonably likely to require payments of \$500,000 or more in any year, (x) any Contract for the acquisition of or investment in capital equipment for an aggregate purchase price or investment value of \$500,000 or more, (y) any Contract authorizing the distribution or resale by any Person of any of the Company s or its Subsidiaries products or services or (z) any Contract for the sale or rental of products or services that is reasonably likely to result in payments to the Company and its Subsidiaries of \$500,000 or more in any year;

(C) any partnership, joint venture or other similar agreement or arrangement relating to the formation, creation, operation, management or control of any partnership or joint venture material to the Company or any of its Subsidiaries or in which the Company owns more than a 5% voting or economic interest, or any interest valued at more than \$500,000 without regard to percentage voting or economic interest;

(D) any Contract (other than among direct or indirect wholly-owned Subsidiaries of the Company) relating to indebtedness for borrowed money or the deferred purchase price of property (in either case, whether incurred, assumed, guaranteed or secured by any asset) in excess of \$1,000,000;

(E) any Contract required to be filed as an exhibit to any Company Report;

(F) any non-competition Contract or other Contract that (I) purports to limit in any material respect either the type of business in which the Company or its Subsidiaries (or, after the Effective Time, Parent or its Subsidiaries) may engage or the manner or locations in which any of them may so engage in any business, (II) could require the disposition of any material assets, line of business or product line of the Company or any of its Subsidiaries or, after the Effective Time, Parent or any of its Subsidiaries, (III) grants most favored nation status including any that, following the Merger, would apply to Parent and its Subsidiaries, including the Company and its Subsidiaries, (IV) prohibits or limits in any material respect the rights of the Company or any of its Subsidiaries to make, sell or distribute any products or services, or use, transfer, license, distribute or enforce any of their respective Intellectual Property rights, or (V) provides for

a material indemnification obligation by the Company or any of its Subsidiaries, other than indemnification arrangements with customers, suppliers or manufacturers entered into in the ordinary course of business, which arrangements are required to be recorded as liabilities on the Company s balance sheets included in the Company Reports;

(G) any Contract to which the Company or any of its Subsidiaries is a party containing a standstill or similar agreement pursuant to which the party has agreed not to acquire assets or securities of the other party or any of its Affiliates;

(H) any Contract between the Company or any of its Subsidiaries and any director or officer of the Company or any Person beneficially owning five (5%) percent or more of the outstanding Company Shares;

(I) any Contract providing for indemnification by the Company or any of its Subsidiaries of any Person, except for any such Contract that is (x) not material to the Company and its Subsidiaries and (y) entered into in the ordinary course of business;

(J) any Contract that contains a put, call or similar right pursuant to which the Company or any of its Subsidiaries could be required to purchase or sell, as applicable, any equity interests of any Person or assets that have a fair market value or purchase price of more than \$500,000; and

(K) any other Contract or group of related Contracts that, if terminated or subject to a default by any party thereto, would, individually or in the aggregate, be reasonably likely to have a Company Material Adverse Effect (the Contracts described in clauses (A) (K), together with all exhibits and schedules to such Contracts, being the <u>Material Contracts</u>).

(ii) Section 5.1(q)(ii) of the Company Disclosure Letter sets forth a list of (x) the Company s top ten (10) suppliers by value for the fiscal year ended December 31, 2009, and (y) the Company s top ten (10) customers by value of sales for the fiscal year ended December 31, 2009. Except as set forth in Section 5.1(q)(ii) of the Company Disclosure Letter, neither the Company nor any of its Subsidiaries has received any written notice from any supplier required to be identified in Section 5.1(q)(ii) of the Company Disclosure Letter stating that such supplier intends to stop supplying, or materially reduce the level of, materials, products or services supplied to the Company or any of its Subsidiaries. Except as set forth in Section 5.1(q)(ii) of the Company Disclosure Letter, neither the Company nor any of its Subsidiaries has received any written notice from any customer required to be identified in Section 5.1(q)(ii) of the Company nor any of its Subsidiaries. Except as set forth in Section 5.1(q)(ii) of the Company Disclosure Letter, neither the Company nor any of its Subsidiaries has received any written notice from any customer required to be identified in Section 5.1(q)(ii) of the Company Disclosure Letter, neither the Company nor any of its Subsidiaries has received any written notice from any customer required to be identified in Section 5.1(q)(ii) of the Company Disclosure Letter, neither the Company Disclosure Letter stating that such customer intends to stop buying, or materially reduce the level of, products of services bought from the Company or any of its Subsidiaries or otherwise materially and adversely change its business relationship or terms of any Material Contract with the Company or any of its Subsidiaries.

(iii) A true and correct copy of each Material Contract has previously been delivered to Parent and each such Contract is a valid and binding obligation of the Company or one of its Subsidiaries, as the case may be, and is in full force and effect, and neither the Company nor any of its Subsidiaries nor, to the Company s knowledge, any other party thereto is in material default under or material breach in any respect under the terms of any such Material Contract. In addition, none of the Company or any of its Subsidiaries has received any written claim of default under any Material Contract, and no event has occurred which would result in a breach or violation of, or a default under, any Material Contract (in each case, with or without notice or lapse of time or both).

(iv) (A) With respect to each Government Contract, except as has not had and would not reasonably be expected to have a Company Material Adverse Effect, (x) all representations and certifications executed, acknowledged or set forth in or pertaining to such Government Contract were complete and correct as of their effective date, and the Company and each of its Subsidiaries have complied in all material respects with all such representations and certifications, (y) neither the United States government nor any prime contractor, subcontractor or other Person has notified the Company or any of its Subsidiaries that the Company or any such Subsidiary has breached or violated any material certification, representation, clause,

provision or requirement, pertaining to such Government Contract, and (z) no termination for convenience, termination for default, cure notice or show cause notice is in effect as of the date hereof pertaining to any Government Contract.

(B) Except as has not had and would not reasonably be expected to have a Company Material Adverse Effect, (x) to the Company s knowledge, neither the Company nor any of its Subsidiaries nor any of their respective personnel is or has been under administrative, civil, or criminal investigation, or indictment or audit by any Governmental Entity with respect to any alleged irregularity, misstatement or omission arising under or relating to any Government Contract, (y) neither the Company nor any of its Subsidiaries has conducted or initiated any internal investigation or made a voluntary disclosure to the United States government with respect to any alleged irregularity, misstatement or omission arising under or relating to a Government Contract, and (z) neither the Company nor any of its Subsidiaries nor, to the Company s knowledge, any of their respective personnel has been suspended or debarred from doing business with the United States government or is, or at any time has been, the subject of a finding of non-responsibility or ineligibility for United States government contracting.

As used herein, <u>Government Contract</u> means any Material Contract to which the Company or any of its Subsidiaries is a party, or by which any of them are bound, the ultimate contracting party of which is a Governmental Entity (including any subcontract with a prime contractor or other subcontractor who is a party to any such contract).

(r) <u>Property</u>. Other than as set forth in Section 5.1(r) of the Company Disclosure Letter, (i) the Company has good and marketable title to, or, in the case of securities and investments, a security entitlement (as defined in the Uniform Commercial Code) in, or in the case of leased property, a valid leasehold interest in, all material property (whether real or personal, tangible or intangible, and including securities and investments) and assets purported to be owned or leased by it or any of its Subsidiaries, and no such material property and assets are subject to any Liens except mechanics , workmen s, repairmen s, warehousemen s, carriers or similar Liens arising in the ordinary course of business consistent with past practice or Tax Liens for current Taxes not yet due and payable.

(ii) The lease agreements entered into by the Company and its Subsidiaries in connection with their premises are in full force and effect and enforceable, and, to the knowledge of the Company, there are no existing material defaults of the Company and its Subsidiaries or any other party to the leases thereunder, and neither the Company nor its Subsidiaries has received or given notice of default or claimed default with respect to such leases, nor is there, to the knowledge of the Company, any event that with notice or lapse of time, or both, would constitute a material default thereunder. Other than the lease agreements referred to above, the Company and its Subsidiaries have no other interests of any type in any real property.

(s) <u>Warranties/Product Liability</u>. Except as set forth on Section 5.1(s) of the Company Disclosure Letter and except as specifically reflected, reserved against (all such reserves being made in accordance with GAAP) or otherwise disclosed on the consolidated balance sheet of the Company included in the Company s Quarterly Report on Form 10-Q for the quarter ended September 30, 2010, or incurred since September 30, 2010 in the ordinary course of business, (i) the Company has not received any notice, demand, claim, action, suit, inquiry, hearing, proceeding, notice of violation or investigation from, by or before any Governmental Entity relating to any product, including the packaging and advertising related thereto, designed, formulated, manufactured, processed, sold or placed in the stream of commerce by the Company or any services provided by the Company (a <u>Product</u>), or claim or lawsuit involving a Product which is pending or, to the Company s knowledge, threatened by any Person, (ii) there has not been, nor is there under consideration by the Company, any Product recall or post-sale warning of a material nature concerning any Product, (iii) there are no material warranties, express or implied, written or oral, with respect to any Product, (iv) there are no pending or, to the Company s knowledge, threatened claims with respect to any such warranty which could reasonably be expected to result in a Company Material Adverse Effect, and (v) there are no material pending or, to the Company s knowledge, threatened product liability claims with respect to any Product and no such claims have been settled or adjudicated. All Products comply in all material respects with applicable governmental authorizations and Laws, and there have not been and there are no material defects or deficiencies in such Products.

(t) Product Certifications.

(i) The product certifications given or granted by processors or manufacturers with respect to Products (<u>Product Certifications</u>) constitute all the Product Certifications necessary for the Company and its Subsidiaries to conduct their respective businesses as currently conducted, and each Product is certified in the countries in which it is sold, other than as set forth in Section 5.1(t) of the Company Disclosure Letter. The Company has not made any material modifications or updates to the Products which would require Product Certifications different from or in addition to those in effect as of the date of this Agreement, and, other than as set forth in Section 5.1(t) of the Company Disclosure Letter, to the Company s knowledge, none of the Product Certifications would be terminated, rescinded or modified as a result of this Agreement or the completion of the Merger.

(ii) No claims have been made or to the Company s knowledge, threatened alleging that any product designed, manufactured, marketed, sold or distributed by the Company or any of its Subsidiaries or bearing any trademark or trade name of the Company or any of its Subsidiaries has been subject to any security breach or other compromise that has or may lead to the disclosure, theft or other misappropriation of confidential, proprietary or private information not to the Company s knowledge, has there been any such breach or compromise involving or related to any such product.

(u) <u>Inventory</u>. All inventories of raw materials, work-in-process, finished goods, products under research and development, demonstration equipment, office and other supplies, parts, packaging materials and other accessories related thereto which are held at, or are in transit from or to, any Company location, or located at customers premises on consignment, in each case, which are used or held for use by the Company, including any of the foregoing purchased subject to any conditional sales or title retention agreement in favor of any other Person, together with all rights of the Company against suppliers of such inventories, but excluding any of the foregoing which are used or held for use by the Company (directly or indirectly through a Subsidiary) on consignment from a third party (the <u>Inventory</u>), consists of a quality and quantity usable and salable in the ordinary course of business consistent with past practice. Except as disclosed in Section 5.1(u) of the Company Disclosure Letter, none of the items included in the Inventory are held by the Company on consignment from others.

(v) <u>Licenses</u>. (i) Each of the Company and its Subsidiaries has obtained and is in substantial compliance with all permits, licenses, certifications, approvals, registrations, consents, authorizations, franchises, variances, exemptions and orders issued or granted by a Governmental Entity (collectively, <u>Licenses</u>) necessary to conduct its business as presently conducted, except for those the absence of which or failure to be in compliance with, individually or in the aggregate, have not had and would not reasonably be expected to have a Company Material Adverse Effect. No License will be violated, revoked, terminated prior to its normal expiration date or not renewed solely as a result of the consummation of the transactions contemplated by this Agreement, except, in all cases, for any violation, revocation, termination or renewal that would not reasonably be expected to have a Company Material Adverse Effect.

(ii) The Company and each of its Subsidiaries is and has been in substantial compliance with all applicable rules and regulations (<u>Card Regulations</u>) of any card association or other financial institution that licenses the issuance of, or directly issues, credit cards, charge cards, and/or debit cards to cardholders, including Visa International, Inc., Visa U.S.A., Inc., MasterCard International Services, Inc., and American Express Travel Related Services, Inc. (collectively, <u>Card Organizations</u>). Each of the Company s products offered for sale meets all required Card Regulations, including, as applicable, operating regulations of the various Card Organizations, the Payment Card Industry Encrypting PIN Pad (EPP) Security Requirements Manual, the POS PIN Entry Device Security Requirements Manual, the Payment Card Industry Data Security Standard, the Account Information Security Program, and Payment Application Best Practices.

(w) <u>Anti-Bribery Laws</u>. Neither the Company nor any of its Subsidiaries nor, to the Company s knowledge, any director, officer, agent or employee of the Company or any of its Subsidiaries is aware of or has taken any action, directly or indirectly, that would result in a violation by any such persons of the U.S. Foreign Corrupt

Practices Act of 1977, as amended, and the rules and regulations thereunder, the U.K. Bribery Act of 2010 and the rules and regulations thereunder or any other anti-bribery/corruption legislation promulgated by any Governmental Entity (collectively, <u>Anti-Bribery Laws</u>). The Company and its Subsidiaries have conducted their businesses in compliance in all material respects with all Anti-Bribery Laws and have instituted and maintain policies and procedures designed to ensure, and which are reasonably expected to continue to ensure, continued compliance therewith and the Company has no knowledge of any violation of any Anti-Bribery Law by any of its Affiliates.

(x) <u>OFAC</u>. To the knowledge of the Company, neither the Company nor any of its Subsidiaries, nor any director, officer, agent or employee of the Company or any of its Subsidiaries is currently targeted by any U.S. sanctions administered by the Office of Foreign Assets Control of the U.S. Treasury Department (<u>OFAC</u>); and the Company has not directly or indirectly lent, contributed or otherwise made available funds to any Subsidiary, joint venture partner or other person or entity, for the purpose of financing the activities of any Person currently targeted by any U.S. sanctions administered by OFAC.

(y) <u>Investment Company Act of 1940</u>. Neither of the Company nor any of its Subsidiaries is, or at the Effective Time will be, an investment company, as defined in the Investment Company Act of 1940, as amended.

(z) <u>Brokers and Finders</u>. Neither the Company nor any of the Company s officers, directors or employees has employed any broker or finder or incurred any liability for any brokerage fees, commissions or finders fees in connection with the Merger or the other transactions contemplated by this Agreement, except that the Company has employed UBS Securities LLC as the Company s financial advisor. The Company has made available to Parent a complete and accurate copy of all agreements pursuant to which UBS Securities LLC is entitled to any fees and expenses in connection with any of the transactions contemplated by this Agreement.

(aa) <u>Rights Agreement</u>. The Board of Directors of the Company has taken all necessary action to render the Rights Agreement inapplicable to the Merger and the other transactions contemplated hereby and to cause the rights distributed to holders of Company Shares (the <u>Rights</u>) to expire at or immediately prior to the Effective Time, and neither the execution and delivery of this Agreement nor the consummation of the Merger and the other transactions contemplated by this Agreement will result in the occurrence of a Distribution Date, as defined in the Rights Agreement, or otherwise cause the Rights to become exercisable by the holders hereof.

(bb) <u>Reorganization Treatment</u>. Neither the Company nor any of its Affiliates has taken or agreed to take (or failed to take or agree to take) any action or knows of any facts or circumstances that would reasonably be expected to prevent the Merger from qualifying as a reorganization within the meaning of Section 368(a) of the Code.

(cc) <u>No Other Representations and Warranties</u>. Except for the representations and warranties of the Company contained in this Section 5.1, the Company is not making and has not made, and no other Person is making or has made on behalf of the Company, any express or implied representation or warranty in connection with this Agreement or the transactions contemplated hereby, and no Person is authorized to make any such representation or warranty on behalf of the Company.

5.2 <u>Representations and Warranties of Parent and Merger Sub</u>. Parent and Merger Sub jointly and severally hereby represent and warrant to the Company, as of the date of this Agreement and as of the Closing Date (it being understood that to the extent any such representation and warranty speaks as of a particular date, the Company makes such representation and warranty with respect to such particular date), that, except as otherwise permitted in this Agreement or as set forth (i) in the Parent Reports filed with the SEC since November 1, 2009 and prior to the date hereof (excluding, in each case, any disclosures set forth in any risk factor section or in any other section to the extent such statements are cautionary, predictive or forward-looking in nature), (ii) in the corresponding sections or subsections of the disclosure letter delivered to the Company by Parent and Merger

Sub at the time of entering into this Agreement (the <u>Parent Disclosure Letter</u>) or (iii) to the extent that the qualifying nature of a disclosure in the Parent Disclosure Letter with respect to another section or subsection is reasonably apparent on its face, in such other disclosure:

(a) <u>Organization</u>, <u>Good Standing and Qualification</u>. Each of Parent and its Subsidiaries is a legal entity duly organized, validly existing and in good standing under the Laws of its respective jurisdiction of organization and has all requisite corporate or similar power and authority to own, lease and operate its properties and assets and to carry on its business as presently conducted and is qualified to do business and is in good standing as a foreign corporation or other legal entity in each jurisdiction where the ownership, leasing or operation of its assets or properties or conduct of its business requires such qualification, except where the failure to be so organized, qualified or in such good standing, or to have such power or authority, individually or in the aggregate, has not had or would not reasonably be likely to have a Parent Material Adverse Effect (as defined below). Parent has made available to the Company complete and correct copies of Parent s and Merger Sub s certificate of incorporation and by-laws, each as amended through the date of this Agreement, and each as so delivered is in full force and effect.

As used in this Agreement, the term <u>Parent Material Adverse Effect</u> means a material adverse effect on (x) the financial condition, assets, liabilities, business, or results of operations of Parent and its Subsidiaries, taken as a whole, or (y) the ability of Parent to perform its obligations under this Agreement or consummate the Merger and the other transactions contemplated hereby; provided, however, that Parent Material Adverse Effect shall not include any event, condition, change, occurrence or development of a state of circumstances resulting from, primarily attributable to, or arising in connection with (A) general political, economic or market (including currency) conditions or general changes or developments in the industry in which Parent and its Subsidiaries operate, (B) acts of terrorism or war (whether or not declared) or natural disasters occurring after the date hereof, (C) this Agreement, the Merger and the other transactions contemplated hereby, or the announcement or performance thereof, including any negative impact on or disruption in relationships with customers, suppliers, distributors, employees or similar relationships, (D) changes in Law or any applicable accounting regulations or principles or the interpretations thereof (E) changes in the price or trading volume of Parent s stock (provided that any event, condition, change, occurrence or development of a state of circumstances that may have caused or contributed to such change in market price or trading volume shall not be excluded under this proviso), (F) any failure by Parent to meet public or internal revenue, earnings or other projections, in and of itself (provided that any event, condition, change, occurrence or development of a state of circumstances that may have caused or contributed to such failure to meet published revenue, earnings or other projections shall not be excluded under this proviso) or (G) the taking of any action required by this Agreement or expressly approved or permitted in writing by the Company or reasonably required to obtain the approval of any Governmental Entity, or the failure to take any action prohibited by this Agreement, except in the case of clause (A) or (B), to the extent such changes or conditions disproportionately affect in any material respect Parent and its Subsidiaries, taken as a whole, as compared to other Persons engaged in similar businesses.

(b) <u>Capitalization of Merger Sub</u>. The authorized capital stock of Merger Sub consists of 1,000 shares of Common Stock, par value \$0.001 per share, all of which are validly issued and outstanding. All of the issued and outstanding capital stock of Merger Sub is, and at the Effective Time will be, owned by Parent, and there are (i) no other shares of capital stock or voting securities of Merger Sub, (ii) no securities of Merger Sub convertible into or exchangeable for shares of capital stock or voting securities or securities convertible into or exchangeable for shares of merger Sub, and no obligations of Merger Sub to issue, any capital stock, voting securities or securities convertible into or exchangeable for capital stock or voting securities or securities or securities of Merger and has no, and prior to the Effective Time will have no, assets, liabilities or obligations of any nature other than those incident to its formation and pursuant to this Agreement and the Merger and the other transactions contemplated by this Agreement.

(c) <u>Capital Structure of Parent</u>. The authorized capital stock of Parent consists of 200,000,000 shares of Parent Common Stock, of which 86,918,790 shares were outstanding as of the close of business on

November 15, 2010, and 10,000,000 shares of Preferred Stock par value \$0.01 per share (the <u>Parent Preferred Shares</u>), of which no shares were outstanding as of the close of business on the date of this Agreement. All of the outstanding shares of Parent Common Stock have been duly authorized and are validly issued, fully paid and nonassessable. As of the date of this Agreement, Parent has no Parent Common Stock or Parent Preferred Shares reserved for issuance, except for shares of Parent Common Stock reserved for issuance pursuant to (A) Parent s New Founders Stock Option Plan, Outside Directors Stock Option Plan and 2005 Employee Equity Incentive Plan, (B) Parent s 1.375% Senior Convertible Notes due 2012 and (C) warrants to purchase Parent Common Stock. Each of the outstanding shares of capital stock of each of Parent s Subsidiaries is duly authorized, validly issued, fully paid and nonassessable and, except for directors qualifying shares, owned by Parent or by a direct or indirect wholly-owned Subsidiary of Parent, free and clear of any Liens. Except as set forth above, there are no preemptive or other outstanding rights, options, warrants, conversion rights, stock appreciation rights, redemption rights, repurchase rights, agreements, arrangements, calls, commitments or rights of any kind that obligate Parent or any of its Subsidiaries to issue or to sell any shares of capital stock or other securities of Parent or any of its Subsidiaries or any securities or obligations convertible or exchangeable into or exercisable for, or giving any Person a right to subscribe for or acquire, any securities of Parent or any of its Subsidiaries, and no securities or obligations evidencing such rights are authorized, issued or outstanding. Parent does not have outstanding any bonds, debentures, notes or other obligations the holders of which have the right to vote (or that are convertible into or exercisable for securities having the right to vote) with the stockholders of Parent on any matter.

(d) <u>Corporate Authority</u>. Assuming the accuracy of the Company s representations and warranties set forth in Section 5.1(b) with respect to capitalization matters and the Company s compliance with Section 6.1(a)(vii), no vote of holders of capital stock of Parent is necessary to approve this Agreement and the Merger and the other transactions contemplated hereby. Each of Parent and Merger Sub has all requisite corporate power and authority and has taken all corporate and stockholder action necessary in order to execute, deliver and perform its obligations under this Agreement and to consummate the Merger. This Agreement has been duly executed and delivered by each of Parent and Merger Sub and is a valid and binding agreement of Parent and Merger Sub, enforceable against each of Parent and Merger Sub in accordance with its terms, subject to the Bankruptcy and Equity Exception. The shares of Parent Common Stock, when issued pursuant to this Agreement, will be validly issued, fully paid and nonassessable, and no stockholder of Parent will have any preemptive right of subscription or purchase in respect thereof. The Board of Directors of Parent has unanimously approved this Agreement and the other transactions contemplated hereby.

(e) <u>Governmental Filings: No Violations; Certain Contracts, Etc</u>. (i) Other than the necessary notices, reports, filings, consents, registrations, approvals, permits, authorizations, actions or non-actions (A) pursuant to Section 1.3, (B) required under the HSR Act, Exchange Act and the Securities Act, (C) to comply with state securities or blue-sky laws, (D) to be made with the NYSE and (E) other filings and recordations as required by Governmental Entities other than those in the United States as set forth on Section 5.2(e) of the Parent Disclosure Letter, no filings, notices and/or reports are required to be made by Parent or Merger Sub with, nor are any consents, registrations, approvals, permits or authorizations required to be obtained by Parent or Merger Sub from, any Governmental Entity in connection with the execution, delivery and performance of this Agreement by Parent and Merger Sub and the consummation of the Merger and the other transactions contemplated hereby or in connection with the continuing operation of the business of Parent and its Subsidiaries following the Effective Time, except those that the failure to make or obtain are not, individually or in the aggregate, reasonably likely to have a Parent Material Adverse Effect.

(ii) The execution, delivery and performance of this Agreement by Parent and Merger Sub do not, and the consummation of the Merger and the other transactions contemplated hereby will not, constitute or result in (A) a breach or violation of, or a default under, the certificate of incorporation or by-laws of Parent or Merger Sub or the comparable governing documents of any of Parent s Subsidiaries, (B) with or without notice, lapse of time or both, a breach or violation of, a default or termination (or right of termination) under, the creation or acceleration of any obligations under, or the creation of a Lien or pledge, security interest or other encumbrance on its assets or any of its Subsidiaries pursuant to, any Contracts binding upon

Parent or any of its Subsidiaries or, assuming the filings, notices and/or approvals referred to in Section 5.2(e)(i) are made or obtained, any Law to which Parent or any of its Subsidiaries is subject or (C) any change in the rights or obligations of any party under any Contract binding upon Parent or any of its Subsidiaries, except, in the case of clause (B) or (C) above, for any such breach, violation, termination, default, creation acceleration or change that is not, individually or in the aggregate, reasonably likely to have a Parent Material Adverse Effect.

(f) <u>Parent Reports; Financial Statements</u>. (i) Parent has filed or furnished, as applicable, on a timely basis all forms, statements, reports and documents required to be filed or furnished by it with the SEC under the Exchange Act or the Securities Act since November 1, 2009 (the <u>Parent Applicable Date</u>) (collectively, the forms, statements, reports and documents filed with or furnished to the SEC subsequent to the date of this Agreement, including any amendments thereto, the <u>Parent Reports</u>). Each of the Parent Reports was prepared in all material respects in accordance with the applicable requirements of the Securities Act, the Exchange Act and the rules and regulations thereunder and complied in all material respects with then applicable accounting standards. Each of the Parent Reports, at the time of its filing or being furnished complied, or if not yet filed or furnished, will comply, in all material respects with the applicable to the Parent Reports. As of their respective dates (or, if amended prior to the date of this Agreement, as of the date of such amendment) the Parent Reports did not, and any Parent Reports filed with or furnished to the SEC subsequent to the date of this Agreement will not, contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements made therein, in light of the circumstances in which they were made, not misleading, except to the extent that the information in such Parent Report has been amended or superseded by a later Parent Report filed prior to the date of this Agreement.

(ii) Parent maintains effective disclosure controls and procedures as required by Rule 13a-15 or 15d-15 under the Exchange Act. Parent and its Subsidiaries maintain a system of internal accounting controls sufficient to provide reasonable assurance that (A) transactions are executed in accordance with management s general or specific authorizations, (B) transactions are recorded as necessary to permit preparation of financial statements in conformity with GAAP and to maintain asset accountability, (C) access to assets is permitted only in accordance with management s general or specific authorization, and (D) the recorded accountability of assets is compared with the existing assets at reasonable intervals and appropriate action is taken with respect to any material differences.

(iii) Each of the consolidated balance sheets included in or incorporated by reference into any Parent Reports (including the related notes and schedules) fairly presents, or, in the case of Parent Reports filed after the date hereof, will fairly present the consolidated financial position of Parent and its consolidated Subsidiaries as of its date and each of the consolidated statements of income, changes in stockholders equity (deficit) and cash flows included in or incorporated by reference into the Parent Reports (including any related notes and schedules) fairly presents, or in the case of Parent Reports filed after the date hereof, will fairly present the results of operations, retained earnings (loss) and changes in financial position, as the case may be, for the periods set forth therein (subject, in the case of unaudited statements, to notes and normal year-end audit adjustments that will not be material in amount or effect), in each case in accordance with GAAP consistently applied during the periods involved, except as may be noted therein.

(iv) Except as reflected or reserved against in Parent s most recent consolidated balance sheet (or the notes thereto) included in the Parent Reports, neither Parent nor any Subsidiary of Parent has any liabilities or obligations of any nature, whether or not accrued, contingent or otherwise, whether known or unknown and whether due or to become due, that would, individually or in the aggregate, have a Parent Material Adverse Effect.

(v) Parent s revenue recognition policies and practices are and have been in compliance with all rules, regulations and statements of the SEC with respect thereto, including SAB 101, <u>Revenue Recognition in Financial Statements</u>, and SAB 10<u>4</u>, <u>Revenue Recognition</u>, and, to the extent required, Parent recognizes

revenue in accordance with Accounting Standards Codification (ASC) 985, <u>Certain Revenue Arrangements That Contain Software Elements</u>, and ASC 605-25, <u>Revenue Recognition</u> <u>Multiple Element Arrangements</u>; and Parent maintains adequate controls over its revenue recognition policies and practices all of which are properly communicated to and applied by its sales organizations.

(g) <u>Absence of Certain Changes</u>. Since the Parent Applicable Date, there has not been any event, development or circumstance involving, or any change in the financial condition, assets, liabilities, business, or results of their operations or any circumstance, occurrence or development (including any adverse change with respect to any circumstance, occurrence or development existing on or prior to the most recent fiscal year end) which, individually or in the aggregate, has had or would reasonably be expected to have a Parent Material Adverse Effect.

(h) <u>Litigation and Liabilities</u>. There are no civil, criminal or administrative actions, suits, claims, hearings, arbitrations, investigations or other proceedings pending or, to Parent s knowledge, threatened against Parent or any of its Subsidiaries, except for those that would not, individually or in the aggregate, reasonably be expected to have a Parent Material Adverse Effect.

(i) <u>Compliance with Laws</u>. The businesses of each of Parent and its Subsidiaries have not been, and are not being, conducted in violation of any Laws, except for violations that individually or in the aggregate, have not had and would not reasonably be expected to have a Parent Material Adverse Effect. To the knowledge of Parent, no investigation or review by any Governmental Entity with respect to Parent or any of its Subsidiaries is pending or threatened, nor has any Governmental Entity indicated an intention to conduct the same.

(j) <u>Brokers and Finders</u>. Neither Parent nor any of its officers, directors or employees has employed any broker or finder or incurred any liability for any brokerage fees, commissions or finders fees in connection with the Merger or the other transactions contemplated by this Agreement, except that Parent has employed J.P. Morgan Securities LLC as its financial advisor.

(k) <u>Section 203</u>. None of Parent, Merger Sub or any of their Affiliates is an interested stockholder of the Company for purposes of Section 203 of the DGCL (other than as contemplated by this Agreement or the Support Agreement).

(1) <u>Reorganization Treatment</u>. Neither Parent, Merger Sub nor any other Affiliate of Parent has taken or agreed to take (or failed to take or agree to take) any action or knows of any facts or circumstances that would reasonably be expected to prevent the Merger from qualifying as a reorganization within the meaning of Section 368(a) of the Code.

(m) <u>No Other Representations and Warranties</u>. Except for the representations and warranties of Parent and Merger Sub contained in this Section 5.2, Parent and Merger Sub are not making and have not made, and no other Person is making or has made on behalf of Parent or Merger Sub, any express or implied representation or warranty in connection with this Agreement or the transactions contemplated hereby, and no Person is authorized to make any such representation or warranty on behalf of Parent or Merger Sub.

ARTICLE VI

COVENANTS

6.1 <u>Interim Operations</u>. (a) The Company covenants and agrees as to itself and its Subsidiaries that, after the date hereof and until the earlier of the termination of this Agreement pursuant to its terms or the Effective Time, except as otherwise expressly contemplated by this Agreement or as required by applicable Laws, without the prior written consent of Parent (which shall not be unreasonably withheld or delayed), the business of the Company and its Subsidiaries shall be conducted in the ordinary and usual course and, to the extent consistent

therewith, it and its Subsidiaries shall use their respective reasonable efforts to preserve their business organizations intact and maintain existing relations and goodwill with Governmental Entities, customers, suppliers, distributors, creditors, lessors, employees and business associates and keep available the services of its and its Subsidiaries present employees and agents. Without limiting the generality of the foregoing and in furtherance thereof, from the date of this Agreement until the earlier of the termination of this Agreement pursuant to its terms or the Effective Time, except (A) as otherwise expressly required by this Agreement or required by Law, (B) as Parent may approve in writing (which approval shall not be unreasonably withheld or delayed) or (C) as set forth in Section 6.1 of the Company Disclosure Letter, the Company will not and will not permit its Subsidiaries to:

(i) (A) amend its certificate of incorporation, by-laws or other applicable governing instruments; (B) split, combine, subdivide or reclassify its outstanding shares of capital stock; (C) declare, set aside or pay any dividend or distribution payable in cash, stock or property in respect of any capital stock; or (D) repurchase, redeem or otherwise acquire any shares of its capital stock or any securities convertible into or exchangeable or exercisable for any shares of its capital stock (other than repurchases or redemptions of restricted stock or other equity awards upon termination of employment, in accordance with the terms of such awards);

(ii) merge or consolidate itself or any of its Subsidiaries with any other Person, except for any such transactions among its wholly-owned Subsidiaries, or restructure, reorganize or completely or partially liquidate or otherwise enter into any agreements or arrangements imposing material changes or restrictions on its assets, operations or businesses;

(iii) spend in excess of \$6,500,000 individually or \$10,000,000 in the aggregate to acquire any business, whether by merger, consolidation, purchase of property or assets or otherwise (valuing any non-cash consideration at its fair market value as of the date of the agreement for such acquisition); <u>provided</u> that neither the Company nor any of its Subsidiaries shall make any acquisition that would, or would reasonably be likely to prevent, delay or impair the Company s ability to consummate the transactions contemplated by this Agreement; and <u>provided</u>, <u>further</u>, that the Company must also comply with the requirements set forth in Section 6.1(a)(iii) of the Company Disclosure Letter. For purposes of this clause (iii), the amount spent with respect to any acquisition shall be deemed to include the aggregate amount of capital expenditures that the Company or any of its Subsidiaries is obligated to make at any time or plans to make as a result of such acquisition within two (2) years after the date of acquisition;

(iv) incur any indebtedness for borrowed money or guarantee such indebtedness of another Person, or issue or sell any debt securities or warrants or other rights to acquire any of its debt securities or of any of its Subsidiaries, except for (A) drawings under and refinancings or replacements of the Company s or any of its Subsidiaries current credit agreements in an amount not to exceed \$25,000,000 in the aggregate at any one time outstanding, provided that the terms of any such replacement must satisfy the requirements set forth in Section 6.1(a)(iv) of the Company Disclosure Letter, (B) other indebtedness for borrowed money incurred in the ordinary course of business consistent with past practices not to exceed \$10,000,000 in the aggregate or (C) interest rate or currency swaps on customary commercial terms consistent with past practice and in compliance with its risk management policies in effect on the date of this Agreement and not to exceed \$5,000,000 of notional debt in the aggregate;

(v) make or commit to any capital expenditures other than in the ordinary course of business and in any event not in excess of the aggregate amount reflected in the Company s capital expenditure budget for the year in which such capital expenditures are made, a copy of which capital expenditure budget for 2010 and 2011 has been furnished to Parent;

(vi) transfer, sell, lease, license, mortgage, pledge, surrender, encumber, divest, cancel, abandon or allow to lapse or expire or otherwise dispose of any of its assets, product lines or businesses or of its Subsidiaries, including capital stock of any of its Subsidiaries and sales of obsolete assets, except for sales, leases, licenses or other dispositions of assets with a fair market value not in excess of \$10,000,000 in the aggregate, other than pursuant to Contracts in effect as of the date of this Agreement;

(vii) issue, sell, pledge, dispose of, grant, transfer, encumber, or authorize the issuance, sale, pledge, disposition, grant, transfer, lease, license, guarantee or encumbrance of, any shares of its capital stock or of any its Subsidiaries (other than the issuance of shares by a wholly-owned Subsidiary to it or another of its wholly-owned Subsidiaries), or securities convertible or exchangeable into or exercisable for any shares of such capital stock, or any options, warrants or other rights of any kind to acquire any shares of such capital stock or such convertible or exchangeable securities, other than (A) the issuance of Company Shares pursuant to existing Company Options, Company Warrants, and other Company Awards, and (B) the issuance of Company Options and Company Awards in amounts and at times consistent with the Company s historic hiring and compensation practices and not in excess of the amount set forth on Section 6.1(a)(vii) of the Company Disclosure Letter;

(viii) make any change with respect to accounting policies or procedures, except as required by changes in GAAP or by Law;

(ix) except as required by Law, (A) make any material Tax election or take any position on any Tax Return filed on or after the date of this Agreement or adopt any method therefor that is inconsistent with elections made, positions taken or methods used in preparing or filing similar Tax Returns in prior periods, (B) settle or resolve any material Tax controversy, claim or assessment, (C) enter into any material closing agreement, (D) waive or extend any statute of limitations with respect to Taxes, or (E) surrender any right to claim a refund for Taxes;

(x) make any loans, advances or capital contributions to or investments in any Person (other than the Company or any direct or indirect wholly-owned Subsidiary of the Company) in excess of \$1,000,000 individually or \$2,000,000 in the aggregate;

(xi) enter into (A) any non-competition Contract or other Contract that (x) purports to limit in any material respect either the type of business in which the Company or its Subsidiaries (or, after the Effective Time, Parent or its Affiliates) may engage or the manner or locations in which any of them may so engage in any business or (y) could require the disposition of any material assets or line of business of the Company or its Subsidiaries (or, after the Affiliates, (B) any other Contract that would have been a Material Contract had it been entered into prior to the date hereof;

(xii) except as required pursuant to Contracts in effect as of the date of this Agreement and set forth in Section 6.1(a)(xii) of the Company Disclosure Letter, or as otherwise required by applicable Law, (A) grant or provide any severance or termination payments or benefits to any of its directors, officers or employees, (B) increase the compensation, bonus or pension, welfare, severance or other benefits of, pay any bonus to, or make any new equity awards to any of its directors, officers or employees, except, in each case, for increases or awards made in the ordinary course of business and at times and in amounts consistent with the Company s historic hiring and compensation practices as described in Section 6.1(a)(xii) of the Company Disclosure Letter, (C) establish, adopt, amend or terminate any Company Compensation and Benefit Plan or amend the terms of any outstanding equity-based awards, (D) except as provided in Section 4.4(f), take any action to accelerate the vesting or payment, or fund or in any other way secure the payment, of compensation or benefits under any Company Compensation and Benefit Plans, to the extent not already provided in any such Company Compensation and Benefit Plan, (E) change any actuarial or other assumptions used to calculate funding obligations with respect to any Company Compensation and Benefit Plan or change the manner in which contributions to such plans are made or the basis on which such contributions are determined, except as may be required by GAAP, or (F) forgive any loans to any of its or of any of its Subsidiaries directors, officers or employees;

(xiii) amend, modify or terminate any Material Contract (excluding customer or supplier contracts entered into in the ordinary course of business), or cancel, modify or waive any debts or claims held by it or waive any rights other than in the ordinary course of business having a value in excess of \$1,000,000 in the aggregate;

(xiv) take any action or omit to take any action that is reasonably likely to result in any of the conditions to the Merger set forth in Article VII not being satisfied, including the acquisition of any business or assets reasonably likely to have such an effect;

(xv) settle any litigation or other proceedings before or threatened to be brought before a Governmental Entity if such settlements would exceed \$1,000,000 in the aggregate; or

(xvi) agree, authorize or commit to do any of the foregoing.

(b) After the date hereof and until the earlier of the termination of this Agreement pursuant to its terms or the Effective Time, Parent will not and will not permit its Subsidiaries knowingly to take any action or omit to take any action that is reasonably likely to result in any of the conditions to the Merger set forth in Article VII not being satisfied, including the acquisition of any business or assets reasonably likely to have such an effect. Notwithstanding the foregoing and for the avoidance of doubt, the foregoing shall not apply to any pending or planned (i) strategic engagement or investment, (ii) joint venture, (iii) acquisition or (iv) other similar transactions, that in the case of any transaction described in any of clauses (i) through (iv) of this Section 6.1(b), has been publicly announced prior to the execution of this Agreement.

6.2 Acquisition Proposals.

(a) <u>No Solicitation or Negotiation</u>. The Company agrees that neither it nor any of its Subsidiaries nor any of its or its Subsidiaries officers and directors shall, and that it shall instruct and cause its and its Subsidiaries directors, officers, employees, investment bankers, attorneys, accountants and other advisors or representatives (such directors, officers, employees, investment bankers, attorneys, accountants and other advisors or representatives) not to, directly or indirectly:

(i) initiate, solicit, or knowingly facilitate or encourage any inquiries or the making of any proposal or offer that constitutes, or would reasonably be expected to lead to, an Acquisition Proposal (as defined below) (including by amending, or granting any waiver under, the Rights Agreement not contemplated by Section 6.14);

(ii) engage in, continue or otherwise participate in any discussions or negotiations regarding, or provide any non-public information or data to any Person relating to, any Acquisition Proposal; or

(iii) otherwise knowingly facilitate any effort or attempt to make an Acquisition Proposal.

Notwithstanding anything in this Agreement to the contrary, prior to, but not after, the receipt of the Company Requisite Vote, the Company may (A) provide information in response to a request therefor by a Person who has made an unsolicited *bona fide* written Acquisition Proposal if the board of directors of the Company receives from the Person so requesting such information an executed confidentiality agreement on terms not less restrictive to such Person than those contained in the Confidentiality Agreement, dated June 25, 2009, between the Company and Parent (the <u>Confidentiality Agreement</u>): provided that such agreement need not contain a standstill provision; (B) engage or participate in any discussions or negotiations with any Person who has made such an unsolicited *bona fide* written Acquisition Proposal; and (C) after having complied with the terms of this Section 6.2, approve, recommend, or otherwise declare advisable or propose to approve, recommend or declare advisable (publicly or otherwise) such an Acquisition Proposal, if and only to the extent that (x) prior to taking any action referred to in clause (A), (B) or (C) above, the Board of Directors of the Company determines in good faith after consultation with its outside legal counsel and financial advisor that failure to take such action would be inconsistent with the directors fiduciary duties under applicable Law, (y) in each case referred to in clause (A) or (B) above, the Board of Directors of the Company has determined in good faith based on the information then available and after consultation with its financial advisor and outside legal counsel that such Acquisition Proposal constitutes, or is reasonably likely to result in, a Superior Proposal, and (z) in the case referred to in clause (C) above, the Board of Directors of the Company has determined in good faith based on the information then available and after consultation with its outside legal counsel that such Acquisition Proposal constitutes, or is reasonably likely to result in,

advisor that such Acquisition Proposal constitutes a Superior Proposal. The Company shall promptly disclose (and, if applicable, provide copies of) to Parent (to the extent not previously provided) any non-public information provided to any third party pursuant to clause (A) above to the fullest extent permitted by Law.

(b) <u>Definitions</u>. For purposes of this Agreement:

<u>Acquisition Proposal</u> means (i) any proposal or offer with respect to a merger, joint venture, consolidation, dissolution, liquidation, tender offer, recapitalization, reorganization, share exchange, business combination or similar transaction involving the Company or (ii) any acquisition by any person resulting in (or any proposal or offer to acquire in any manner which if consummated would result in) any person becoming the beneficial owner of, directly or indirectly, in one or a series of related transactions 15% or more of the total voting power or of any class of equity securities of the Company or any of its Subsidiaries, or 15% or more of the consolidated total assets (including equity securities of its Subsidiaries taken as a whole, in each case other than the transactions contemplated by this Agreement.

<u>Superior Proposal</u> means a *bona fide* written Acquisition Proposal not solicited in violation of this Section 6.2, executed by the Person making the proposal involving 80% or more of the assets (on a consolidated basis) or total voting power of the equity securities of the Company that the Company s Board of Directors has determined in its good faith judgment is reasonably capable of being consummated in accordance with its terms, and that, taking into account all legal, financial and regulatory aspects of the proposal and the Person making the proposal, if consummated, would result in a transaction more favorable to the Company s stockholders than the transaction contemplated by this Agreement (after taking into account any revisions to the terms of the transaction contemplated by Section 6.2(c) of this Agreement).

(c) No Change in Recommendation Agreement.

(i) The Board of Directors of the Company, and each committee thereof shall not:

(1) except as expressly permitted by this Section 6.2, withhold, withdraw, qualify or modify (or publicly propose or resolve to withhold, withdraw, qualify or modify), in a manner adverse to Parent, the Company Recommendation (<u>it being understood that</u> publicly taking a neutral position or no position with respect to an Acquisition Proposal at any time beyond ten (10) business days after the first public announcement of such Acquisition Proposal shall be deemed to be an adverse modification); or

(2) cause or permit the Company to enter into any letter of intent, term sheet, memorandum of understanding, agreement in principle, acquisition agreement, merger agreement or other agreement (other than a confidentiality agreement referred to in Section 6.2(a) entered into in the circumstances referred to in Section 6.2(a)) (an <u>Alternative Acquisition Agreement</u>) relating to any Acquisition Proposal.

(ii) Notwithstanding anything to the contrary set forth in this Agreement, prior to, but not after, the receipt of the Company Requisite Vote, the Company s Board of Directors may withhold, withdraw or modify the Company Recommendation in a manner adverse to Parent, or approve, recommend or otherwise declare advisable any Superior Proposal made after the date hereof and not solicited, initiated or encouraged in breach of this Agreement, if, but only if, subject to compliance with this Section 6.2, the Company s Board of Directors determines in good faith, after consultation with its outside counsel and financial advisors, that failure to take such action would be inconsistent with the directors fiduciary obligations under applicable Law (A) in connection with a Superior Proposal or (B) as a result of a fact, event, circumstance, development or a change in circumstances or facts that was not known (or, if known, the magnitude and material consequences of which were not known) by the Company s Board of Directors as of the date of this Agreement, (<u>a Company Change of Recommendation</u>); provided, however, that no Company change of Recommendation may be made until after at least two (2) business days following Parent s receipt of written notice from the Company advising that the Company s Board of Directors currently intends to take such action and the basis therefor. If such Company Change of Recommendation is

made in connection with a Superior Proposal, the notice described in the preceding sentence must attach the most current version of the proposed agreement under which a Superior Proposal is proposed to be consummated and disclose the identity of the third party making the Superior Proposal. The Company agrees that (i) during the two (2) business day period prior to its taking any action referred to in either clause (C) in Section 6.2(a) with respect to an Acquisition Proposal or in this Section 6.2(c)(ii), the Company and its Representatives shall, if requested by Parent, negotiate in good faith with Parent and its Representatives regarding any revisions to the terms of the transactions contemplated by this Agreement proposal continues to be a Superior Proposal in light of any revisions to the terms of the transaction contemplated by this Agreement proposed by Parent and any other information provided by Parent in response to such notice. Any material or substantive amendment to any Acquisition Proposal shall be deemed to be a new Acquisition Proposal for purposes of this Section 6.2, including with respect to the notice periods referred to herein.

(d) <u>Certain Permitted Disclosure</u>. Nothing contained in this Section 6.2 or otherwise contained in this Agreement shall be deemed to prohibit the Company from complying with its disclosure requirements under Rule 14d-9 or Rule 14e-2(a) promulgated under the Exchange Act or other provisions of U.S. federal or state law with regard to an Acquisition Proposal; <u>provided</u>, <u>however</u>, that if such disclosure has the substantive effect of withholding, withdrawing or modifying adversely to Parent the Company Recommendation, Parent shall have the right to terminate this Agreement as set forth in Section 8.4(a).

(e) <u>Existing Discussions</u>. The Company agrees that it will immediately cease and cause to be terminated any existing activities, discussions or negotiations with any Persons conducted heretofore with respect to any Acquisition Proposal. The Company also agrees that it will promptly request each Person that is a party to a confidentiality agreement in connection with any consideration of acquiring the Company or any of its Subsidiaries or business operations to return or destroy all confidential information heretofore furnished to such Person by or on behalf of the Company or any of its Subsidiaries.

(f) Notice. The Company agrees that it will promptly (and, in any event, within 24 hours) notify Parent if any Acquisition Proposal or any inquiry with respect to, or that would reasonably be expected to lead to, an Acquisition Proposal is received by the Company or its Representatives, any information is requested from the Company in connection with an Acquisition Proposal or an inquiry related thereto, or any discussions or negotiations are sought to be initiated or continued with the Company or its Representatives with respect to an Acquisition Proposal or an inquiry related thereto, indicating, in connection with such notice, the name of the Person making such Acquisition Proposal, inquiry or request and the material terms and conditions of any such Acquisition Proposal or inquiry (including, if applicable, copies of any written requests, proposals or offers, including proposed agreements) and thereafter shall keep Parent reasonably informed, on a reasonably current daily basis of the status of and any material changes to the terms of any such Acquisition Proposal or inquiry (including any amendments thereto). Contemporaneously with providing any non-public information concerning the Company to a third party in connection with any such Acquisition Proposal, the Company will provide a copy of such information to Parent, to the extent not previously provided.

6.3 <u>Information Supplied</u>. (a) Parent and the Company shall promptly prepare and Parent shall file with the SEC as promptly as practicable the Registration Statement on Form S-4 to be filed with the SEC by Parent in connection with the issuance of shares of Parent Common Stock in the Merger (including the proxy statement and prospectus (the <u>Prospectus/Proxy Statement</u>) constituting a part thereof) (the <u>S-4 Registration</u> <u>Statement</u>). Parent and the Company each shall use its reasonable best efforts to have the S-4 Registration Statement declared effective under the Securities Act as promptly as practicable after such filing, and promptly thereafter mail the Prospectus/Proxy Statement to the stockholders of the Company. The Company and Parent shall also use their respective best efforts to satisfy prior to the effective date of the S-4 Registration Statement all necessary state securities law or blue sky notice requirements in connection with the Merger and to consummate the other transactions contemplated by this Agreement and will pay all expenses incident thereto.

(b) The Company and Parent each agrees, as to itself and its Subsidiaries, that none of the information supplied or to be supplied by it or its Subsidiaries for inclusion or incorporation by reference in (i) the S-4 Registration Statement will, at the time the S-4 Registration Statement becomes effective under the Securities Act, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, and (ii) the Prospectus/Proxy Statement and any amendment or supplement thereto will, at the date of mailing to stockholders and at the time of the Company Stockholders Meeting, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. The Company and Parent will cause the Form S-4 to comply as to form in all material respects with the applicable provisions of (A) the Securities Act and the rules and regulations thereunder, (B) the Exchange Act and (C) the rules and regulations of the NYSE.

(c) No filing of, or amendment or supplement to, the S-4 Registration Statement will be made by Parent, and no filing of, or amendment or supplement to the Prospectus/Proxy Statement will made by the Company or Parent, in each case without providing the other party a reasonable opportunity to review and comment thereon. If at any time prior to the date of the Company Stockholders Meeting any information relating to the Company or Parent, or any of their respective Affiliates, directors or officers, should be discovered by the Company or Parent which should be set forth in an amendment or supplement to either the S-4 Registration Statement or the Prospectus/Proxy Statement, so that either such document would not include any misstatement of a material fact or omit to state any material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, the party which discovers such information shall promptly notify the other parties hereto and an appropriate amendment or supplement describing such information shall be promptly filed with the SEC and, to the extent required by Law, disseminated to the stockholders of the Company. The parties shall notify each other promptly of the receipt of any comments from the SEC or the staff of the SEC and of any request by the SEC or the staff of the SEC for amendments or supplements to the Prospectus/Proxy Statement or the S-4 Registration Statement or for additional information and shall supply each other with copies of (i) all correspondence between it or any of its Representatives, on the one hand, and the SEC or the staff of the SEC relating to the S-4 Registration Statement or the Merger and (ii) all orders of the SEC relating to the S-4 Registration Statement.

6.4 <u>Stockholders Meeting</u>. (a) The Company will take, in accordance with applicable Law and its certificate of incorporation and by-laws, all action necessary to convene a meeting of holders of Company Shares (the <u>Company Stockholders Meeting</u>) as promptly as practicable after the S-4 Registration Statement is declared effective to consider and vote upon the adoption of this Agreement and approval of the Merger and to cause such vote to be taken. Subject to the provisions of Section 6.2 hereof, the Company s Board of Directors shall recommend in the Prospectus/Proxy Statement and at the Company Stockholders Meeting that the holders of Company Stockholders Meeting if, as of the time for which the Company Stockholders Meeting is originally scheduled (as set forth in the Prospectus/Proxy Statement), (i) there are insufficient Company Shares represented (either in person or by proxy) to constitute a quorum necessary to conduct the business of the Company Stockholders Meeting, or to provide the Company Requisite Vote, or (ii) such adjournment or postponent would be required by Law. The Company s obligation to call, give notice of, convene and hold the Company Stockholders Meeting in accordance with this Section 6.4 shall not be limited to or otherwise affected by the commencement, disclosure, announcement or submission to the Company of any Acquisition Proposal.

(b) Notwithstanding any Company Change in Recommendation or otherwise (unless this Agreement has been terminated in accordance with Article VIII), the Company shall nonetheless submit this Agreement and the Merger to the holders of Company Shares for adoption and approval at the Company Stockholders Meeting.

6.5 <u>Filings: Other Actions: Notification</u>. (a) Parent and the Company shall each use reasonable best efforts to cause to be delivered to each party and the directors of each party a letter from its independent auditors, dated (i) the date on which the S-4 Registration Statement shall become effective and (ii) the Closing Date, and addressed to the parties and the directors of each party, in form and substance customary for comfort letters delivered by independent public accountants in connection with registration statements similar to the S-4 Registration Statement.

(b) The Company and Parent each shall, upon request by the other, furnish the other with all information concerning itself, its Subsidiaries, directors, officers and stockholders and such other matters as may be reasonably necessary or advisable in connection with the Prospectus/Proxy Statement, the S-4 Registration Statement or any other statement, filing, notice or application made by or on behalf of Parent, the Company or any of their respective Subsidiaries to any third party and/or any Governmental Entity in connection with the Merger and the transactions contemplated by this Agreement.

(c) Subject to the terms and conditions set forth in this Agreement, the Company and Parent shall cooperate with each other and use (and shall cause their respective Subsidiaries to use) their respective reasonable best efforts to take or cause to be taken all actions, and do or cause to be done all things, reasonably necessary, proper or advisable on its part under this Agreement and applicable Laws to consummate and make effective the Merger and the other transactions contemplated by this Agreement, including (i) preparing and filing all documentation to effect all necessary notices, reports and other filings (x) within forty-five (45) days of the date hereof (or as mutually agreed by the parties hereto), in the case of a filing under the HSR Act, but only if prior to the end of such forty-five (45) day period, the Company has not entered into a definitive agreement with respect to a transaction which, if consummated, would result in a filing under the HSR Act not being required with respect to the transactions contemplated hereby and (y) as promptly as reasonably practicable after the date hereof in the case of filings with non-U.S. Governmental Entities and (ii) obtaining all consents, registrations, approvals, permits and authorizations necessary or advisable to be obtained from any third party and/or any Governmental Entity in order to consummate the Merger or any of the other transactions contemplated by this Agreement; provided, however, that except as set forth in clause (e) below, nothing in this Agreement, including this Section 6.5, shall require, or be construed to require, Parent to proffer to, or agree to, sell, divest, lease, license, transfer, dispose of or otherwise encumber or hold separate and agree to sell, divest, lease, license, transfer, dispose of or otherwise encumber before or after the Effective Time, any assets, licenses, operations, rights, product lines, businesses or interest therein of Parent, the Company or any of their respective Affiliates (or to consent to any sale, divestiture, lease, license, transfer, disposition or other encumbrance by the Company of any of its assets, licenses, operations, rights, product lines, businesses or interest therein, loss of ability to set off for tax purposes in the future any and all losses accumulated by the Company or to any agreement by the Company to take any of the foregoing actions) or to agree to any material changes (including through a licensing arrangement) or restriction on, or other impairment of Parent s ability to own or operate, of any such assets, licenses, product lines, businesses or interests therein or Parent s ability to vote, transfer, receive dividends or otherwise exercise full ownership rights with respect to the stock of the Surviving Corporation. Subject to applicable Laws relating to the exchange of information, Parent shall have the right to direct all matters with any Governmental Entity consistent with its obligations hereunder; provided that Parent and the Company shall have the right to review in advance, and to the extent practicable each will consult with the other on and consider in good faith the views of the other in connection with, all of the information relating to Parent or the Company, as the case may be, and any of their respective Subsidiaries, that appears in any filing made with, or written materials submitted to, any third party and/or any Governmental Entity in connection with the Merger and the other transactions contemplated by this Agreement (including the Prospectus/Proxy Statement). In exercising the foregoing rights, each of the Company and Parent shall act reasonably and as promptly as practicable.

(d) In furtherance of, and not in limitation of the foregoing, the parties shall use their respective reasonable best efforts to respond promptly to any requests for additional information made by any Governmental Entity with respect to antitrust or competition matters, and to cause the waiting periods under the HSR Act and other similar Law applicable to the transactions contemplated hereby to terminate or expire at the earliest possible date

after the date of filing. The parties hereto agree not to extend directly or indirectly any waiting period under the HSR Act or any other similar Law or enter into any agreement with a Governmental Entity to delay or not to consummate the Merger, except with the prior written consent of the other parties hereto. If any administrative or judicial action or proceeding is instituted (or threatened to be instituted) challenging the Merger or the transactions contemplated by this Agreement as violative of any antitrust or competition Law, or if any antitrust or competition Order enacted, entered, promulgated or enforced by a Governmental Entity that would make the Merger or the other transactions contemplated by this Agreement illegal or would otherwise prohibit or materially impair or delay the consummation of the Merger or the other transactions contemplated by this Agreement, then, subject to Section 6.5(e), each of Parent and Merger Sub shall use its reasonable best efforts to contest and resist any such action or proceeding and shall use its reasonable best efforts to have vacated, lifted, reversed or overturned any antitrust or competition Order, whether temporary, preliminary or permanent, that is in effect and that prohibits, prevents or restricts consummation of the Merger or the other transactions contemplated by this Agreement and to have such Order repealed, rescinded or made inapplicable so as to permit consummation of the Merger and such other transactions.

(e) Notwithstanding anything to the contrary contained in this Agreement, the parties hereby agree and acknowledge that neither this Section 6.5 nor the reasonable best efforts standard nor any other provision of this Agreement shall require, or be construed to require, other than as set forth in the proviso below, Parent or any of its Subsidiaries or other Affiliates, in order to obtain any approvals or otherwise, to (i)(A) sell, lease, license, transfer, dispose of, divest or otherwise encumber, or hold separate pending any such action, or (B) propose, negotiate or offer to effect, or consent or commit to, any such sale, leasing, licensing, transfer, disposal, divestiture or other encumbrance, or holding separate, before or after the Effective Time, of any assets, licenses, operations, rights, product lines, businesses or interest therein of Parent, the Company or the Surviving Corporation (or any of their respective Subsidiaries or other Affiliates), or (ii) take or agree to take any other action or agree or consent to any limitations or restrictions on freedom of actions with respect to, or its ability to retain, or make changes in, any such assets, licenses, operations, rights, product lines, businesses or interest therein of Parent, the Company or the Surviving Corporation (or any of their respective Subsidiaries or other Affiliates); provided, however, that Parent s reasonable best efforts for purposes of Sections 6.5(c) and (d) shall include a requirement by Parent to agree to sell or otherwise dispose of, and to sell or otherwise dispose of, businesses or assets that, except as set forth in Schedule 6.5(e), in the aggregate, produced aggregate gross revenues in an amount not in excess of \$124 million (subject to adjustment as provided in Section 6.6) for Parent, the Company and their respective Subsidiaries during the 2009 fiscal year (the <u>Maximum</u> Divestiture Commitment), calculated in accordance with GAAP, on a basis consistent with the accounting principles used in preparing their respective 2009 financial statements included in the Company SEC Reports or Parent SEC Reports, as applicable, if such action should be reasonably necessary to (x) obtain any consent or approval required for the consummation of the Merger under Section 7.1(c) or (y) eliminate, terminate or remove the issuance of any Order described in Section 7.1(d), to the extent related to antitrust or competition matters, or (z) eliminate, resolve, settle or terminate any suit, action or proceeding related to antitrust or competition matters described in Section 7.2(c), in each case prior to August 20, 2011 (if the Termination Date is not extended as provided in Section 8.2) or November 21, 2011 (if the Termination Date is so extended). Notwithstanding the foregoing, however, Parent shall not be required to agree to any such dispositions (and shall be entitled in good faith to contest any threatened or pending litigation or proceeding brought by any Governmental Entity) unless failure to do so would result in the closing conditions not being met hereunder prior to the earlier of (1) the termination of this Agreement and (2) August 20, 2011 (if the Termination Date is not extended as provided in Section 8.2), or November 21, 2011 (if the Termination Date is so extended). For the avoidance of doubt, when calculating amounts under this Section 6.5(e), 6.5(d), 7.1(c) or 7.2(c), all dispositions or changes in scope of operations that have been or are obligated to be undertaken (or with respect to which written undertakings have been given to a Governmental Entity) or are required to be disregarded (in the case of Section 7.2(c)) with respect to the assets, licenses, operations, rights, product lines, business or interests of Parent, the Company or the Surviving Corporation (or any of their respective Subsidiaries or other Affiliates) for the purposes described in the preceding sentence shall be aggregated together to determine whether the dollar limits set forth herein have been met or exceeded.

(f) Subject to applicable Law and as required by any Governmental Entity, the Company and Parent each shall keep the other apprised of the status of matters relating to completion of the transactions contemplated hereby, including promptly furnishing the other with copies of notices or other communications received by Parent or the Company, as the case may be, or any of its Subsidiaries, from any third party and/or any Governmental Entity with respect to such transactions, and, subject to applicable Law, shall permit the other party to review in advance any proposed written communication to any such Governmental Entity and shall incorporate the other party s reasonable comments. Neither Parent nor the Company shall permit any of its officers or any other representatives or agents to participate in any meeting with any Governmental Entity in respect of any filings, investigation or other inquiry relating to the transactions contemplated hereby unless it consults with the other party in advance and, to the extent permitted by such Governmental Entity, gives the other party the opportunity to attend and participate thereat.

(g) Parent and the Company each shall give prompt notice to the other of any change, fact or condition that is reasonably likely to result in a Company Material Adverse Effect or a Parent Material Adverse Effect, respectively, or of any failure of any condition to the other party s obligations to effect the Merger.

6.6 Dispositions. (a) So long as Parent requests, and subject to the further limitations set forth below in Section 6.6(b), the Company shall use reasonable best efforts as promptly as practicable to effectuate the disposition of portions of its business identified by Parent as reasonably necessary to satisfy the conditions to closing set forth in Article VII (any applicable transaction document in connection with the foregoing being referred to as an <u>Asset Purchase Agreement</u>); provided, that the Company shall not be obligated to execute an Asset Purchase Agreement with respect to any business if the sum of (i) the 2009 revenues of such business, together with (ii) the 2009 revenues of all other businesses with respect to which the Company has executed an Asset Purchase Agreement and (iii) the 2009 revenues of all businesses which Parent has agreed with any Governmental Entity to divest or hold separate as contemplated by Section 6.5 (such sum, the <u>Aggregate Divestiture Commitment</u>) would exceed the Maximum Divestiture Commitment then in effect, unless Parent agrees in writing to increase the Maximum Divestiture Commitment to an amount not less than the Aggregate Divestiture Commitment. Subject to applicable Law, the Company shall keep Parent informed on a current basis of the status of its efforts and discussions in connection with the foregoing, and shall use reasonable best efforts to incorporate the reasonable comments and input of Parent in connection therewith. The Company shall comply with its obligations under any applicable Asset Purchase Agreements, and, subject to the terms and conditions set forth in any such Asset Purchase Agreement, the Company shall use (and shall cause its Subsidiaries to use) reasonable best efforts to take or cause to be taken all actions, and do or cause to be done all things, reasonably necessary, proper or advisable on its part under such Asset Purchase Agreements and applicable Laws to consummate and make effective the transactions contemplated thereby as soon as practicable and to enforce all of its rights thereunder. Subject to applicable Law, the Company shall keep Parent informed on a current basis as to the status of any applicable dispositions, including the status of all regulatory and other third-party approvals required in connection therewith, and as to any material communications with the applicable transferee, Governmental Entities or other third parties. For the avoidance of doubt, the terms of any Asset Purchase Agreement shall be subject to Parent s prior written consent.

(b) Notwithstanding any other provision of this Agreement, but without limiting the Company s obligations under Section 6.6(a), the parties acknowledge and agree that (i) if an Asset Purchase Agreement is executed, the Asset Purchase Agreement shall be expressly conditioned on the consummation of the Merger and shall provide that the closing under the Asset Purchase Agreement shall occur only if all conditions to Closing under this Agreement have been satisfied and the parties hereto are prepared to close the transactions contemplated by this Agreement immediately following the closing under the Asset Purchase Agreement, and (ii) any Asset Purchase Agreement shall provide that it may be terminated by the Company upon the termination of this Agreement without liability to the Company, and that any termination fee that may be payable shall be solely the obligation of Parent (and Parent will, if required, confirm in writing its agreement to pay any such termination fee).

6.7 <u>Access; Consultation</u>. Upon reasonable notice, and except as may otherwise be required by applicable Law, the Company shall (and shall cause its Subsidiaries to) afford Parent and its Representatives reasonable

access, during normal business hours throughout the period prior to the Effective Time, to its employees, properties, books, contracts and records and, during such period, the Company shall (and shall cause its Subsidiaries to) furnish promptly to Parent all information concerning its or any of its Subsidiaries business, properties and personnel as may reasonably be requested (including any financial statements, schedules or other data relating to the Company prepared by the Company in the ordinary course of business) and shall cooperate with Parent and its Representatives in preparing information and providing other assistance as reasonably requested by Parent and its Representatives in connection with Parent s information and reports filed with any Governmental Entity or provided to its current or prospective stockholders or other financial counterparties; provided that no investigation pursuant to this Section 6.7 shall affect or be deemed to modify any representation or warranty made by the Company hereunder; and provided further that the foregoing shall not require the Company to permit any inspection, or to disclose any information, that in the reasonable judgment of the Company would violate any of its obligations with respect to confidentiality if the Company shall have used all commercially reasonable efforts to obtain the consent of such third party to such inspection or disclosure. All requests for information made pursuant to this Section 6.7 shall be directed to an executive officer of the Company, or such Person as may be designated by any such executive officer, as the case may be. All such information shall be governed by the terms of the Confidentiality Agreement.

6.8 <u>Stock Exchange Listing and Delisting</u>. Parent shall use its reasonable best efforts to cause the shares of Parent Common Stock issuable in the Merger pursuant to Article IV hereof to be approved for listing on the NYSE, subject to official notice of issuance, prior to the Closing Date. Prior to the Closing Date, the Company shall cooperate with Parent and use reasonable best efforts to take, or cause to be taken, all actions, and do or cause to be done all things, reasonably necessary, proper or advisable on its part under applicable Laws and policies of the NYSE to enable the delisting by the Surviving Corporation of the Company Shares from the NYSE and the deregistration of the Company Shares under the Exchange Act as promptly as practicable after the Effective Time.

6.9 <u>Publicity</u>. The initial press release with respect to the Merger shall be a joint press release and thereafter the Company and Parent each shall consult with each other prior to issuing any press releases or otherwise making public announcements with respect to the Merger and the other transactions contemplated by this Agreement and prior to making any filings with any third party and/or any Governmental Entity (including any national securities exchange) with respect thereto, except as may be required by Law or by obligations pursuant to any listing agreement with or rules of any national securities exchange, and except any consultation that would not be reasonably practicable as a result of requirements of Law.

6.10 Expenses. Except as otherwise provided in Section 8.5, whether or not the Merger is consummated, all costs and expenses incurred in connection with this Agreement and the Merger and the other transactions contemplated by this Agreement shall be paid by the party incurring such expense, except that expenses incurred in connection with the filing fee for the S-4 Registration Statement and printing and mailing the Prospectus/Proxy Statement and the S-4 Registration Statement shall be shared equally by Parent and the Company.

6.11 Indemnification: Directors and Officers Insurance. (a) From and after the Effective Time, Parent shall, and shall cause the Surviving Corporation to, indemnify and hold harmless each present and former director and officer of the Company (when acting in such capacity) determined as of the Effective Time (the Indemnified Parties), against any costs or expenses (including reasonable attorneys fees), judgments, fines, losses, claims, damages or liabilities incurred in connection with any claim, action, suit, proceeding or investigation, whether civil, criminal, administrative or investigative, arising out of or pertaining to matters existing or occurring at or prior to the Effective Time, whether asserted or claimed prior to, at or after the Effective Time, to the fullest extent that the Company would have been obligated to indemnify such Indemnified Parties at the date of this Agreement under its certificate of incorporation or by-laws in effect on the date of this Agreement. Parent hereby assumes, and shall cause the Surviving Corporation to assume, all obligations of the Company to indemnify the Indemnified Parties for acts or omissions occurring at or prior to the Effective Time as provided in the respective certificate of incorporation or by-laws (or comparable organizational documents) of the Company or any of its Subsidiaries as now in effect, and any indemnification agreements or arrangements of

the Company or any of its Subsidiaries shall survive the Merger and shall continue in full force and effect in accordance with their terms. Such rights shall not be amended, or otherwise modified in any manner that would adversely affect the rights of the Indemnified Parties, unless such modification is required by Law. Parent or the Surviving Corporation shall also advance expenses as incurred to the fullest extent required under such instruments as in effect on the date of this Agreement; <u>provided</u> that the Person to whom expenses are advanced provides an undertaking to repay such advances if it is ultimately determined that such Person is not entitled to indemnification.

(b) Any Indemnified Party wishing to claim indemnification under paragraph (a) of this Section 6.11, upon learning of any such claim, action, suit, proceeding or investigation, shall promptly notify Parent and the Surviving Corporation thereof, but the failure to so notify shall not relieve Parent and the Surviving Corporation of any liability they may have to such Indemnified Party except to the extent failure prejudices Parent or the Surviving Corporation, as the case may be. In the event of any such claim, action, suit, proceeding or investigation (whether arising before or after the Effective Time), (i) Parent or the Surviving Corporation shall have the right to assume the defense thereof and Parent and the Surviving Corporation shall not be liable to such Indemnified Party for any legal expenses of other counsel or any other expenses subsequently incurred by such Indemnified Party in connection with the defense thereof, except that if Parent or the Surviving Corporation elects not to assume such defense or counsel for the Indemnified Party advises that there are issues which raise conflicts of interest between the Parent or the Surviving Corporation and the Indemnified Party, the Indemnified Party may retain counsel satisfactory to them, and Parent or the Surviving Corporation shall pay all reasonable fees and expenses of such counsel for the Indemnified Party promptly as statements therefor are received; provided, however, that Parent and the Surviving Corporation shall be obligated pursuant to this paragraph (b) to pay for no more than one firm of counsel (other than local counsel) for all Indemnified Parties, unless such counsel concludes in good faith that actual or potential conflicts of interest or differing claims against or potential defenses available to multiple Indemnified Parties make representation of all Indemnified Parties by a single counsel inappropriate, (ii) the Indemnified Party will cooperate in full in the defense of any such matter and (iii) Parent and the Surviving Corporation shall not be liable for any settlement effected without their prior written consent; and provided, further, that Parent and the Surviving Corporation shall not have any obligation hereunder to any Indemnified Party if and when a court of competent jurisdiction shall ultimately determine, and such determination shall have become final, that the indemnification of such Indemnified Party in the manner contemplated hereby is prohibited by applicable Law.

(c) Parent shall cause the Surviving Corporation to, and the Surviving Corporation shall, maintain a policy of officers and directors liability insurance for acts and omissions occurring prior to the Effective Time (<u>D&O Insurance</u>) with coverage in amount and scope not less extensive than the Company s existing directors and officers liability insurance coverage for a period of six (6) years after the Effective Time; <u>prov</u>ided, <u>however</u>, that, if the aggregate premium therefor exceeds 250% of the last annual premium paid prior to the date of this Agreement (such amount, as stated in Section 6.11(c) of the Company Disclosure Letter, the <u>Current Annual Premium</u>), the Surviving Corporation will use its commercially reasonable efforts to obtain D&O Insurance in such amount and scope as can be obtained for an aggregate premium not in excess of 250% of the Current Annual Premium; and <u>provided</u>, <u>further</u> that in lieu of such coverage, Parent may substitute a prepaid tail policy for such coverage meeting the requirements of this Section 6.11(c), which it may cause the Company to obtain prior to the Closing.

(d) Prior to the Effective Time, the Company may make any arrangements necessary to obtain or continue such D&O Insurance with coverage in amount and scope comparable to the Company s existing directors and officers liability insurance coverage for such six (6) year period, including the payment of a premium to the applicable insurance providers; provided, however, that the aggregate premium therefor may not exceed 250% of the Current Annual Premium and the Company shall use its commercially reasonable efforts to obtain the lowest aggregate premium that is reasonably obtainable for such coverage.

(e) If Parent or any of its successors or assigns (i) shall consolidate with or merge into any other corporation or entity and shall not be the continuing or surviving corporation or entity of such consolidation or merger or

(ii) shall transfer all or substantially all of its properties and assets to any individual, corporation or other entity, then and in each such case, proper provisions shall be made so that the successors and assigns of Parent shall assume all of the obligations set forth in this Section.

6.12 <u>Takeover Statute</u>. If any Takeover Statute is or may become applicable to the Merger or the other transactions contemplated by this Agreement, each of Parent and the Company and their respective Boards of Directors shall grant such approvals and take such actions as are necessary so that such transactions may be consummated as promptly as practicable on the terms contemplated by this Agreement or by the Merger and otherwise use reasonable best efforts to act to eliminate or minimize the effects of such statute or regulation on such transactions.

6.13 Section 16(b). The Board of Directors of each of the Company and Parent shall, prior to the Effective Time, take all such actions as may be necessary or appropriate pursuant to Rule 16b-3(d) and Rule 16b-3(e) under the Exchange Act to exempt from Section 16 of the Exchange Act (i) the disposition of Company Shares and derivative securities (as defined in Rule 16a-1(c) under the Exchange Act) with respect to Company Shares and directors of the Company subject to the reporting requirements of Section 16(a) of the Exchange Act or by employees or directors of the Company who may become an officer or director of Parent subject to the reporting requirements of Section 16(a) of the Exchange Act.

6.14 <u>Rights</u>. Prior to the Effective Time, the Board of Directors of the Company shall take any action (including, as necessary, amending or terminating (but with respect to termination, only as of immediately prior to the Effective Time) the Rights Agreement) necessary so that (a) none of the execution and delivery of this Agreement, the conversion of Company Shares into the right to receive the Merger Consideration in accordance with this Agreement, and the consummation of the Merger or any other transaction contemplated by this Agreement hereby will cause (i) the Rights to become exercisable under the Rights Agreement, (ii) Parent or any of its affiliates to be deemed an Acquiring Person (as defined in the Rights Agreement), (iii) the provisions of Section 11 or Section 13 of the Rights Agreement) to occur upon any such event or (iv) the Distribution Date or the Stock Acquisition Date (each as defined in the Rights Agreement) to occur upon any such event, and (b) the Final Expiration Date (as defined in the Rights Agreement) of the Rights will occur immediately prior to the Effective Time so that the Rights will expire immediately prior to the Effective Time. Without the prior written consent of Parent, neither the Board of Directors of the Company nor the Company shall take any other action to terminate the Rights Agreement, redeem the Rights, cause any person not to be or become an Acquiring Person or otherwise amend the Rights Agreement in a manner, or take any other action under the Rights Agreement, adverse to Parent or its affiliates.

6.15 <u>Stockholder Litigation</u>. The Company shall give Parent the opportunity to participate, at Parent s cost and expense, in the defense or settlement of any stockholder litigation that currently exists or arises after the date of this Agreement against the Company or its directors or officers, and no such settlement shall be agreed to without Parent s consent (which shall not be unreasonably withheld or delayed).

6.16 Employee Matters.

(a) For a period ending not earlier than 12 months following the Closing Date, employees of the Company and its Subsidiaries who continue their employment after the Effective Time (the <u>Affected Employees</u>) shall be provided with employee benefits under either (i) the employee benefit programs applicable to similarly situated employees of Parent or (ii) the employee benefit programs applicable to Affected Employees immediately prior to the Effective Time, as determined in the sole discretion of Parent. Parent shall, and shall cause the Surviving Corporation to, honor and pay from and after the Effective Time any bonus payment obligations of the Company and each Subsidiary of the Company properly accrued on the Company s financial statements in accordance with GAAP for the Company s fiscal year (or any portion thereof) ending with or before the Effective Time under bonus plans of the Company and each Subsidiary in accordance with their terms as in effect immediately prior to the Effective Time.

(b) With respect to any benefit plan, program, arrangement (including any employee benefit plan (as defined in Section 3(3) of ERISA) and any vacation program), Parent shall, and shall cause the Surviving Corporation to, recognize the service with the Company and its Subsidiaries (and any of their predecessors) prior to the Effective Time of Affected Employees for purposes of eligibility, vesting, level of benefits and benefit accruals under such plan or program, except with respect to any defined benefit pension plan and to the extent such recognition would result in a duplication of benefits.

(c) With respect to any welfare plan in which employees of the Company and its Subsidiaries and their covered dependents are eligible to participate after the Effective Time, Parent shall, and shall cause the Surviving Corporation to use reasonable best efforts to, (i) waive all limitations as to preexisting conditions, exclusions and waiting periods with respect to participation and coverage requirements applicable to such employees and covered dependents to the extent such conditions were satisfied, inapplicable or waived under the welfare plans of the Company and its Subsidiaries prior to the Effective Time, and (ii) provide each such employee and covered dependent with credit for any co-payments, deductibles and maximum out-of pocket requirements paid or incurred prior to the Effective Time in satisfying any analogous deductible or maximum out-of-pocket requirements to the extent applicable under any such plan.

(d) From and after the Effective Time, Parent shall, or shall cause the Surviving Corporation to, honor all obligations of the Company under the Company Compensation and Benefit Plans, Company Non-U.S. Compensation and Benefit Plans and compensation and severance arrangements and agreements in accordance with their terms as in effect immediately prior to the Effective Time; provided that, subject to the requirements of Section 6.16(a), nothing herein shall prohibit the Surviving Corporation from amending or terminating any particular Company Compensation and Benefit Plan or Company Non-U.S. Compensation and Benefit Plan to the extent permitted by its terms or applicable Laws. Notwithstanding any other provision of this Agreement to the contrary, Parent shall, or shall cause the Surviving Corporation to, provide Affected Employees whose employment terminates during the 12 month period following the Effective Time who are not (i) parties to individual agreements providing severance or termination benefits or (ii) entitled to severance or termination benefits in accordance with their individual agreements or the Company s 2010 Change of Control Plan (it being agreed that all Affected Employees described in clauses (i) and (ii) shall be entitled to severance or termination benefits at levels no less than and pursuant to the terms of either (i) the Company s severance policy described in Section 6.16(d) of the Company Disclosure Letter as in effect on the date hereof or (ii) under Parent s severance policy, as determined in the sole discretion of Parent, and such severance benefits shall be determined taking into account the service crediting provisions set forth in Section 6.16(b).

(e) Notwithstanding the foregoing, nothing in this Section 6.16 shall be deemed to modify any requirement under any federal, state, local or foreign law, statute or ordinance with respect to employment matters.

6.17 <u>Tax Treatment</u>. Parent, Merger Sub and Company shall each use commercially reasonable efforts to cause the Merger to be treated as a reorganization within the meaning of Section 368(a) of the Code and shall not knowingly take any action, or fail to take any action, which action or failure would prevent, or reasonably be expected to prevent, the Merger from qualifying as a reorganization under Section 368(a) of the Code.

6.18 <u>Certain Actions</u>. (a) Recognizing that until the Closing (and if this Agreement is terminated for any reason, thereafter), without the other party s consent, Parent and Company are competitors, notwithstanding anything to the contrary in this Agreement, the applicable party shall not, directly or indirectly, through the earlier of (i) the Effective Time (if the Closing occurs) and (ii) the period (if any) specified below following termination of this Agreement (if the Closing does not occur):

(A) for the term hereof and a period of six (6) months following any termination of this Agreement, directly or indirectly solicit for hire any officer or key employee set forth on Schedule 6.18A (Parent) or 6.18B (Company) or, except where this restriction is prohibited by law, hire any such employee; provided

that nothing shall prohibit soliciting and hiring by means of a general advertisement not directed at (x) any particular individual, or (y) the employees of the other party generally, or responding to and hiring an employee of the other party who initiates contact with the applicable party provided that such contact is not solicited by the applicable party; or

(B) seek to effect, announce or communicate, any potential reduction or elimination of the other party s products or services or change in the pricing thereof.

For the avoidance of doubt, it is agreed that internal planning conferences and related communications, in each case limited to senior executive of Parent and that are not in any way directly or indirectly communicated or disseminated to any Persons other than senior executives of Parent, shall not constitute communications prohibited by this Section 6.18.

(b) The parties agree that the term of the Confidentiality Agreement set forth in Section 19 thereof shall be extended through the earlier of (i) the Effective Time (if the Closing occurs) and (ii) one (1) year following termination of this Agreement (if the Closing does not occur). The parties also agree that notwithstanding the terms of the Confidentiality Agreement, any confidential information provided to a party in connection with the transactions contemplated by this Agreement may be disclosed to the extent required in connection with a regulatory filing or court order arising in connection with the transactions contemplated hereby in which case the parties shall use reasonable commercial efforts to preserve any confidential information so provided to the party, except as a party may agree in writing or as may be required in connection with a regulatory filing or court order arising in connection with the transactions contemplated hereby in which case the parties shall use reasonable commercial efforts to preserve any confidential information in connection with the transactions contemplated hereby in which case the parties shall use reasonable commercial efforts to preserve any confidential information so provided to the party, except as a party may agree in writing or as may be required in connection with a regulatory filing or court order arising in connection with the transactions contemplated hereby in which case the parties shall use reasonable commercial efforts to preserve any confidential information so provided.

(c) In addition, if the Closing occurs, Parent agrees to do the following (and not to announce, communicate or take any action to the contrary during the period through the Closing):

(i) to fulfill any Company customer agreements or purchase orders entered into in the ordinary course of business consistent with the Company s current business practices, unless the customer agrees otherwise;

(ii) to the extent requested by the applicable customer, provide warranty service on Company products required by contracts entered into in the ordinary course of business consistent with the Company s current business practices, as well as repair and maintenance for the longer of three (3) years from Closing or as required by law, at service levels consistent with the Company s current business practices; and

(iii) to the extent requested by the applicable customer, continue to support software applications, including the provision of bug fixes and updates for the longer of twelve (12) months from Closing or as required by law, at service levels consistent with the Company s current business practices.

(d) In the event of a breach of Section 6.18(a) prior to the termination of this Agreement,

(i) the non-breaching party shall provide notice of the breach as promptly as reasonably practicable following the date on which a Senior Officer of the non-breaching party obtains knowledge thereof;

(ii) with respect to the first such breach only, the breaching party shall have three (3) days in which to cure such breach, if curable;

(iii) if such initial breach is not cured within such three-day period, and in all cases with respect to any subsequent breach, the breaching party will pay the other party within three (3) business days of notice of the breach \$1 million by wire transfer,

provided, however, that such amounts shall only be payable in the event of a breach caused by (or at the instructions of or with the consent of or prior knowledge of (if the Senior Officer does not use reasonable efforts to prevent such breach and does not direct (or re-direct) the applicable Person to refrain from such activity after obtaining such knowledge) one or more of the breaching party s senior officers (the Senior Officers) identified

on Schedule 6.18AA (Parent) or Schedule 6.18BB (Company) or any successor. The parties agree that the amounts payable under this Section 6.18 are not intended to be and will not be punitive in effect and that they will compensate the non-breaching party for its losses (which may be difficult to ascertain) resulting from a breach of this Section 6.18. Nothing in this Section 6.18 shall limit any other right or remedy of any party under this Agreement.

ARTICLE VII

CONDITIONS

7.1 <u>Conditions to Each Party</u> s <u>Obligation to Effect the Merger</u>. The respective obligation of each party to effect the Merger is subject to the satisfaction or waiver at or prior to the Closing of each of the following conditions:

(a) <u>Stockholder Approval</u>. This Agreement shall have been duly adopted by, and the Merger shall have been duly approved by, holders of Company Shares constituting the Company Requisite Vote.

(b) <u>NYSE Listing</u>. The shares of Parent Common Stock issuable to the Company stockholders pursuant to the Merger as contemplated in Article IV shall have been authorized for listing on the NYSE upon official notice of issuance.

(c) <u>Regulatory Consents</u>. Unless the Company and Parent shall have agreed after consultation with their respective outside counsel that no filing under the HSR Act is required, the waiting period applicable to the consummation of the Merger under the HSR Act shall have expired or been terminated. All Governmental Consents set forth in Schedule 7.1(c) and any other notification, waiting period, or approval requirements under the comparable antitrust or competition laws of other applicable foreign jurisdictions in which either party has operations or from which either party derives revenues (which in either case are not *de minimis*) and which is legally required to be made or obtained at or prior to the Effective Time shall have been made or obtained without the imposition of any term, condition or consequence the acceptance of which would be reasonably likely to have a Company Material Adverse Effect or a Parent Material Adverse Effect (either before or after giving effect to the Merger, and in each case for purposes of this determination, disregarding clause (G) of the definitions of Company Material Adverse Effect and Parent Material Adverse Effect) or requiring the sale, lease, license, transfer, disposition of, divestment or other encumbrance, or the holding separate pending any such action of any assets, licenses, operations, rights, product lines, businesses or interest therein of Parent, the Company or the Surviving Corporation (or any of their respective Subsidiaries or other Affiliates) (other than as contemplated or required by Section 6.5(e)). For purposes of this Agreement, the term <u>Governmental Consents</u> shall mean all notices, reports, and other filings made or required to be made prior to the Effective Time by the Company or Parent or any of their respective Subsidiaries with, and all consents, registrations, approvals, permits, clearances and authorizations sought or required to be obtained prior to the Effective Time by the Company or Parent or any of their respective Subsidiaries from, any Governmental Entity in connection with the execution and delivery of this Agreement and the consummation of the Merger and the other transactions contemplated hereby.

(d) <u>No Injunction or Restraint</u>. No Governmental Entity of competent jurisdiction shall have enacted, issued, promulgated or entered any Law (whether temporary, preliminary or permanent) that is in effect and restrains, enjoins or otherwise prohibits consummation of the Merger or the other transactions contemplated by this Agreement (collectively, an <u>Order</u>).

(e) <u>S-4 Registration Statement</u>. The S-4 Registration Statement shall have become effective under the Securities Act. No stop order suspending the effectiveness of the S-4 Registration Statement shall have been issued, and no proceedings for that purpose shall have been initiated or be threatened by the SEC.

(f) <u>Blue Sky Approvals</u>. Parent shall have received all state securities and blue sky permits and approvals necessary to consummate the transactions contemplated hereby.

(g) <u>Dispositions</u>. If one or more Asset Purchase Agreements have been entered into, then (i) all of the conditions to the closing under such Asset Purchase Agreements shall have been satisfied or waived, and (ii) the Company and the applicable transferee shall have effectuated the applicable disposition and unless Parent otherwise consents (such consent not to be unreasonably withheld), without deviation from the terms set forth in each such Asset Purchase Agreement.

(h) <u>Opinion</u>. Each of Company and Parent shall have received a written opinion from its respective counsel, dated as of the Closing Date, to the effect that, for U.S. federal income tax purposes, the Merger will constitute a reorganization within the meaning of Section 368 of the Code. Parent, Merger Sub and the Company shall each execute and deliver to such counsel tax representation letters reasonably requested by such counsel, and upon which such counsel can rely, in connection with such written opinions.

7.2 <u>Conditions to Obligations of Parent and Merger Sub</u>. The obligations of Parent and Merger Sub to effect the Merger are also subject to the satisfaction or waiver by Parent at or prior to the Effective Time of the following conditions:

(a) <u>Representations and Warranties</u>. (i) The representations and warranties of the Company set forth in Section 5.1(b) relating to the fully-diluted capitalization of the Company shall be true and correct on the date of this Agreement and at the Effective Time as though made on and as of the Effective Time, other than (A) changes in capitalization due to any Permitted Capitalization Change and (B) other deviations in the Company s actual fully-diluted capitalization (including outstanding Company Shares, Company Preferred Shares, Company Options, Company Warrants, Company Awards and Common Stock Units) by an amount that does not exceed one percent (1.0%) of such fully diluted capitalization; (ii) the representations and warranties of the Company set forth in this Agreement that are qualified by Company Material Adverse Effect shall be true and correct at the Effective Time as though made on and as of the Effective Time (except to the extent that any such representation and warranty expressly speaks as of a particular date, in which case such representation and warranty shall be true and correct as of such earlier date); (iii) the other representations and warranties of the Company set forth in this Agreement shall be true and correct at the Effective Time as though made on and as of the Effective Time (except to the extent that any such representation and warranty speaks as of a particular date, in which case such representation and warranty shall be so true and correct as of such date); provided, however, that notwithstanding anything herein to the contrary, the condition set forth in this Section 7.2(a)(iii) shall be deemed to have been satisfied even if any representations and warranties of the Company are not so true and correct unless the failure of such representations and warranties of the Company to be so true and correct, individually or in the aggregate, has had or would reasonably be likely to have a Company Material Adverse Effect; and (iv) Parent shall have received at the Closing a certificate signed on behalf of the Company by the Chief Executive Officer and Chief Financial Officer of the Company to the effect that the conditions set forth in this Section 7.2(a) have been satisfied. As used herein, a _Permitted Capitalization Change shall mean any change in the equity capitalization of the Company made in compliance with Section 6.1(a)(vii).

(b) <u>Performance of Obligations of the Company</u>. The Company shall have performed in all material respects all obligations required to be performed by it under this Agreement at or prior to the Closing Date, and Parent shall have received a certificate signed on behalf of the Company by the Chief Executive Officer and Chief Financial Officer of the Company to such effect.

(c) <u>No Restraints</u>. There shall not be instituted or pending any suit, action or proceeding in which a Governmental Entity of competent jurisdiction is seeking (i) an Order or (ii) to (A) prohibit, limit, restrain or impair Parent s ability to own or operate or to retain or change all or a material portion of the assets, licenses, operations, rights, product lines, businesses or interest therein of the Company or any of its Subsidiaries or other

Affiliates from and after the Effective Time or any of the assets, licenses, operations, rights, product lines, businesses or interest therein of Parent or its Subsidiaries (including by requiring any sale, divestiture, transfer, license, lease, disposition of or encumbrance or hold separate arrangement with respect to any such assets, licenses, operations, rights, product lines, businesses or interest therein), other than with respect to the disposition of businesses or assets that, in the aggregate, taken together with any dispositions or changes in scope of business already undertaken by Parent pursuant to Section 6.5 or accepted by Parent pursuant to Section 7.1(c), exceed the thresholds set forth in the proviso set forth in Section 6.5(e) or (B) prohibit or limit in any respect Parent s ability to vote, transfer, receive dividends or otherwise exercise full ownership rights with respect to the stock of the Surviving Corporation, and no Governmental Entity of competent jurisdiction shall have enacted, issued, promulgated, enforced or entered any Law deemed applicable to the Merger individually or in the aggregate resulting in, or that is reasonably likely to result in, any of the foregoing; provided, however, this Section 7.2(c) shall have no effect on Parent s obligations under Section 6.5.

(d) <u>Material Adverse Effect</u>. After the date of this Agreement, there shall not have occurred any event, occurrence, discovery or development that, individually or in the aggregate, has resulted, or would reasonably be likely to result, in a Company Material Adverse Effect.

(e) <u>Receipt of Payoff Letter</u>. At or prior to the Effective Time, (x) Francisco Partners II, L.P. and FP Hypercom Holdco, LLC shall have provided the Company with a payoff letter in the form attached as an exhibit to the Support Agreement, (y) Bank of America, N.A. shall have provided the Company with a payoff letter acknowledging that, subject to the repayment of the aggregate principal amount outstanding under the Loan and Security Agreement, dated as of January 15, 2008, by and between Hypercom U.S.A., Inc. and Hypercom Manufacturing Resources, Inc., certain lenders thereunder and Bank of America, N.A. (the <u>Loan and Security Agreement</u>), together with all interest accrued and any other fees or expenses payable thereunder, (i) the Loan and Security Agreement shall be terminated, (ii) any and all Liens held by Bank of America, N.A. under the Loan and Security Agreement shall be released and (iii) the Company and its Subsidiaries shall be released from any and all liabilities and obligations (other than any indemnification obligations that survive termination) under the Loan and Security Agreement and (z) any other lender or administrative agent, as applicable, under any credit facility outstanding immediately prior to the Closing shall have provided the Company with a payoff letter acknowledging that, subject to the repayment of the aggregate principal amount outstanding under the applicable loan documentation, together with all interest accrued and any other fees or expenses payable thereunder, (i) the applicable loan documentation shall be released from any and ell Liens held by the lenders thereunder under the applicable loan documentation shall be released and (iii) the Company and its Subsidiaries shall be released from any and all Liens held by the lenders thereunder under the applicable loan documentation shall be released and (iii) the Company and its Subsidiaries shall be released from any and all liabilities and obligations (other than any indemnification obligations that survive termination) under

(f) <u>Resignations</u>. Each director of the Company shall have delivered a resignation letter effective as of the Closing.

7.3 <u>Conditions to Obligation of the Company</u>. The obligation of the Company to effect the Merger is also subject to the satisfaction or waiver by the Company at or prior to the Effective Time of the following conditions:

(a) <u>Representations and Warranties</u>. (i) The representations and warranties of Parent set forth in this Agreement which are qualified by a Parent Material Adverse Effect shall be true and correct in all respects as so qualified at and as of the Effective Time (except to the extent that any such representation and warranty expressly speaks as of a particular date, in which case such representation and warranty shall be true and correct as of such date), and (ii) the representations and warranties of Parent set forth in the Agreement which are not so qualified shall be true and correct at and as of the Effective Time as though made at and as of the Effective Time (except to the extent that any such representation and warranty expressly speaks as of a particular date, in which case such representation and warranty expressly speaks as of a particular date, in which case such representation and warranty expressly speaks as of a particular date, in which case such representation and warranty shall be true and correct at and as of the Effective Time (except to the extent that any such representation and warranty expressly speaks as of a particular date, in which case such representation and warranty shall be true and correct as of such date), <u>provided</u>, <u>however</u>, that notwithstanding anything herein to the contrary, the condition set forth in this Section 7.3(a)(ii) shall be deemed to have been satisfied even if any representations and warranties of Parent are not so true and correct unless the

failure of such representations and warranties of Parent to be so true and correct, individually or in the aggregate, has had or would reasonably be likely to have a Parent Material Adverse Effect. In addition, the Company shall have received at the Closing a certificate signed by the Chief Executive Officer and Chief Financial Officer of Parent to the effect that the conditions set forth in this Section 7.3(a) have been satisfied.

(b) <u>Performance of Obligations of Parent and Merger Sub</u>. Each of Parent and Merger Sub shall have performed in all material respects all obligations required to be performed by it under this Agreement at or prior to the Closing Date, and the Company shall have received a certificate signed on behalf of Parent and Merger Sub by the Chief Executive Officer and Chief Financial Officer of Parent and Merger Sub to such effect.

(c) <u>Parent Material Adverse Effect</u>. After the date of this Agreement, there shall not have occurred any event, occurrence, discovery or development that, individually or in the aggregate, has resulted, or would reasonably be likely to result, in a Parent Material Adverse Effect.

ARTICLE VIII

TERMINATION

8.1 <u>Termination by Mutual Consent</u>. This Agreement may be terminated and the Merger may be abandoned at any time prior to the Effective Time, whether before or after the adoption of this Agreement by stockholders of the Company referred to in Section 7.1(a), by mutual written consent of the Company and Parent.

8.2 <u>Termination by Either Parent or the Company</u>. This Agreement may be terminated and the Merger may be abandoned at any time prior to the Effective Time by action of either Parent or the Company if (a) the Merger shall not have been consummated by August 31, 2011 (the <u>Termination Date</u>), whether such date is before or after the date of adoption of this Agreement by the stockholders of the Company, provided, that, if, as of the Termination Date, all conditions to this Agreement shall have been satisfied or waived (other than those that are to be satisfied by action taken at the Closing) other than the condition set forth in Section 7.1(c), or Section 7.1(d) or Section 7.2(c), Parent may, by written notice to the Company, extend the Termination Date to November 30, 2011, (b) the adoption of this Agreement by the Company s stockholders required by Section 7.1(a) shall not have occurred at a meeting duly convened therefor or at any adjournment or postponement thereof at which a vote upon this Agreement was taken or (c) any Order permanently restraining, enjoining or otherwise prohibiting consummation of the Merger shall become final and non-appealable (whether before or after the adoption of this Agreement by the stockholders of the Company); provided that the right to terminate this Agreement pursuant to clause (a) above shall not be available to any party that has breached in any material respect its obligations under this Agreement in any manner that shall have been a principal cause of or resulted in the occurrence of the failure of the Merger to be consummated on or before such date.

8.3 <u>Termination by the Company</u>. This Agreement may be terminated and the Merger may be abandoned at any time prior to the Effective Time, whether before or after the adoption of this Agreement by the stockholders of the Company referred to in Section 7.1(a), by the Company if:

(a) at any time prior to the time the Company Requisite Vote is obtained, if (i) the Company s Board of Directors authorizes the Company, subject to complying with the terms of this Agreement (including Section 6.2(c)), to enter into an Alternative Acquisition Agreement with respect to a Superior Proposal not obtained in violation of Section 6.2(a), and (ii) the Company prior to such termination pays to Parent in immediately available funds any fees required to be paid pursuant to Section 8.5; or

(b) there has been a breach of any representation, warranty, covenant or agreement made by Parent or Merger Sub in this Agreement, or any such representation and warranty shall have become untrue after the date of this Agreement, in each case such that Section 7.3(a) or 7.3(b) would not be satisfied and such breach or failure to be true is not curable or, if curable, is not cured within 30 days after written notice thereof is given by the Company to Parent.

8.4 <u>Termination by Parent</u>. This Agreement may be terminated and the Merger may be abandoned at any time prior to the Effective Time, by action of the Board of Directors of Parent if:

(a)(A) at any time prior to the adoption of this Agreement by the Company s stockholders satisfying the condition set forth in Section 7.1(a), (i) the Board of Directors of the Company shall have made a Company Change of Recommendation or (ii) the Company s Board of Directors shall have failed to reconfirm the Company Recommendation following a 5% Holder s public announcement of opposition to the Merger or the other transactions contemplated hereby within the period of time beginning upon the Company s receipt of a written request by Parent to do so and ending on the date that is ten (10) business days after the Company s receipt of such request (unless Parent has previously made at least two (2) such written requests and the Board of Directors of the Company has complied with each of such requests), (B) either (i) the Company shall have failed to convene and hold the Company Stockholders Meeting prior to the later of the date that is forty-five (45) days after the date as of which the SEC staff has confirmed that it has no additional comments on the Proxy Statement (subject to adjournment from time to time as permitted by Section 6.4) or (ii) the minimum number of Company Shares necessary to constitute a quorum for the transaction of business shall have been represented in person or by proxy at the Company Stockholders Meeting and the Company shall have failed to take a vote of stockholders on the Merger and this Agreement at such meeting, (C) a tender offer or exchange offer for outstanding Company Shares shall have been publicly disclosed (other than by Parent or an Affiliate of Parent) and the Company s Board of Directors recommends that the stockholders of the Company tender their shares in such tender or exchange offer or, within ten (10) business days after the commencement of such tender or exchange offer, the Company s Board of Directors fails to recommend that the Company s stockholders not tender any shares into such tender or exchange offer, (D) the Company s Board of Directors shall have recommended to the stockholders of the Company any Acquisition Proposal or (E) the Company or any of its Subsidiaries shall have entered into a letter of intent, agreement in principle, merger agreement, share purchase agreement, asset purchase agreement, share exchange agreement, option agreement or other similar Contract relating to an Acquisition Proposal; or

(b) there has been a breach of any representation, warranty, covenant or agreement made by the Company in this Agreement, or any such representation and warranty shall have become untrue after the date of this Agreement, in each case such that Section 7.2(a) or 7.2(b) would not be satisfied and such breach or condition is not curable or, if curable, is not cured within 30 days after written notice thereof is given by Parent to the Company.

8.5 Effect of Termination and Abandonment.

(a) In the event of termination of this Agreement and the abandonment of the Merger pursuant to this Article VIII, this Agreement (other than as set forth in Section 6.10 and this Section 8.5) shall become void and of no effect with no liability on the part of any party hereto (or of any of its future, current or former directors, officers, employees, Affiliates, agents, legal or financial advisors or other representatives (<u>Related Persons</u>)); <u>provided</u>, <u>however</u>, that no such termination shall relieve any party hereto from any liability for damages to any other party resulting from any prior willful breach of this Agreement or from any obligation to pay, if applicable, the Termination Fee pursuant to Section 8.5(b) or the Parent Termination Fee pursuant to Section 8.5(c).

(b) If this Agreement is terminated (i) by the Company pursuant to Section 8.3(a) or (ii) by Parent pursuant to Section 8.4(a), then the Company shall, prior to such termination for a termination pursuant to clause (i) or promptly (but in no event later than two (2) days after the date of such termination) for a termination pursuant to clause (ii), pay Parent a fee equal to \$12.2 million (the <u>Termination Fee</u>), payable by wire transfer of same day funds. In addition, if this Agreement is terminated by Parent or the Company pursuant to clause (a) or (b) of Section 8.2 and (x) prior to such termination or at the time of the meeting of the Company stockholders referred to in Section 8.2(b), there shall have been publicly disclosed, announced, commenced, submitted or made by a third party (other than by Parent or an Affiliate of Parent) a *bona fide* Acquisition Proposal and (y) within twelve (12) months after the termination of this Agreement pursuant to Section 8.2, the Company shall enter into a

definitive agreement with respect to any Acquisition Proposal (which Acquisition Proposal is ultimately consummated) or an Acquisition Proposal shall have been consummated, then the Company shall promptly, but in no event later than two business days after the date of such consummation, pay Parent a fee equal to the Termination Fee, payable by wire transfer of same day funds (provided that for purposes of this sentence, the references to 15% in the definition of Acquisition Proposal shall be deemed to be references to 50%). For the avoidance of doubt, any payment to be made by any party under this Section 8.5(b) shall be payable only once to such other party with respect to this Section 8.5(b) and not in duplication even though such payment may be payable under one or more provisions hereof.

(c) So long as the Company has not breached in any material respect its obligations under Section 6.5, Section 6.6 or any other covenant in this Agreement (in each case to the extent such obligations or covenants relate to antitrust or competition Law matters (collectively, the <u>Antitrus</u>t Covenants), if this Agreement is terminated (i) pursuant to Section 8.2(c), based on the entry of any Order relating to antitrust or competition Law, or (ii) pursuant to Section 8.2(a), and at the time of such termination all conditions to Closing set forth in Article VII (other than the conditions set forth in Section 7.1(c), Section 7.1(d) (in the case of Section 7.1(d), solely to the extent such Order relates to antitrust or competition Law), Section 7.2(c) (in the case of Section 7.2(c), solely to the extent such suit, action or proceeding relates to antitrust or competition Law, Section 7.3(a) and Section 7.3(b)) have been satisfied or waived, then Parent shall promptly, but in no event later than two (2) days after the date of such termination, pay to the Company Parent a fee equal to \$28.4 million (the <u>Parent Termination Fee</u>), payable by wire transfer of same day funds; provided, however, that if Parent elects to extend the Termination Date pursuant to Section 8.2(a) and such termination occurs after August 31, 2011, then the Parent Termination Fee shall equal \$30.4 million; and provided, further, that the Parent Termination Fee shall not be payable if (1) Parent has waived in writing any condition related to antitrust or competition matters described in this Section 8.5(c) and is prepared to consummate the Merger notwithstanding the failure of such condition, and (2) the consummation of the Merger notwithstanding the failure of such condition would not reasonably be expected to subject any officer or director of the Company to personal liability. For the avoidance of doubt, any payment to be made by any party under this Section 8.5(c) shall be payable only once to such other party with respect to this Section 8.5(c) and not in duplication even though such payment may be payable under one or more provisions hereof. The parties hereto acknowledge that in any proceeding by the Company alleging breach of the Antitrust Covenants by Parent, the amount of the Parent Termination Fee paid by Parent shall be credited against any damages that the Company may be entitled to receive.

ARTICLE IX

MISCELLANEOUS AND GENERAL

9.1 <u>Survival</u>. None of the representations and warranties in this Agreement shall survive the consummation of the Merger. Nothing in this Section 9.1 shall limit any covenant or agreement of any party contained in this Agreement that by its terms contemplates performance after the Effective Time.

9.2 <u>Modification or Amendment</u>. Subject to the provisions of applicable Law, at any time prior to the Effective Time, the parties hereto may modify or amend this Agreement, by written agreement executed and delivered by duly authorized officers of the respective parties; <u>provided</u>, that after the Company Requisite Vote is obtained, any amendment which by Law requires further approval by the Company s stockholders shall be subject to the requisite stockholder approval.

9.3 <u>Waiver of Conditions</u>. (a) Any provision of this Agreement may be waived prior to the Effective Time if, and only if, such waiver is in writing and signed by the party against whom the waiver is to be effective.

(b) No failure or delay by any party in exercising any right, power or privilege hereunder shall operate as a waiver thereof nor shall any single or partial exercise thereof preclude any other or further exercise thereof or the exercise of any other right, power or privilege. Except as otherwise provided herein, the rights and remedies provided herein shall be cumulative and not exclusive of any rights or remedies provided by Law.

9.4 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, each such counterpart being deemed to be an original instrument, and all such counterparts shall together constitute the same agreement.

9.5 Governing Law and Venue; Waiver of Jury Trial.

(a) THIS AGREEMENT SHALL BE DEEMED TO BE MADE IN AND IN ALL RESPECTS SHALL BE INTERPRETED, CONSTRUED AND GOVERNED BY AND IN ACCORDANCE WITH THE LAW OF THE STATE OF DELAWARE WITHOUT REGARD TO THE CONFLICT OF LAW PRINCIPLES THEREOF TO THE EXTENT THAT SUCH PRINCIPLES WOULD DIRECT A MATTER TO ANOTHER JURISDICTION. The parties hereby irrevocably submit to the personal jurisdiction of the Chancery Court of the State of Delaware (or, if the Chancery Court declines to accept jurisdiction, and other courts of the State of Delaware and the Federal courts of the United States of America located in the State of Delaware) solely in respect of the interpretation and enforcement of this Agreement, and in respect of the transactions contemplated hereby, and hereby waive, and agree not to assert, as a defense in any action, suit or proceeding for the interpretation or enforcement of this Agreement or of any such document that it is not subject thereto or that such action, suit or proceeding may not be brought or is not maintainable in said courts or that the venue thereof may not be appropriate or that this Agreement or any such document may not be enforced in or by such courts, and the parties hereto irrevocably agree that all claims with respect to such action, proceeding or transactions shall be heard and determined in such a Delaware State or Federal court. The parties hereby consent to and grant any such court jurisdiction over the Person of such parties and, to the extent permitted by Law, over the subject matter of such dispute and agree that mailing of process or other papers in connection with any such action or proceeding in the manner provided in Section 9.6 or in such other manner as may be permitted by Law, shall be valid and sufficient service thereof.

(b) EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (i) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (ii) EACH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (iii) EACH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (iv) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 9.5.

9.6 <u>Notices</u>. Notices, requests, instructions or other documents to be given under this Agreement shall be in writing and shall be deemed given, (i) when delivered, if delivered personally to the intended recipient, and (ii) one (1) business day later, if sent by overnight delivery via a national courier service, and in each case, addressed to a party at the following address for such party:

if to Parent or Merger Sub

VeriFone Systems, Inc.

2099 Gateway Place, Suite 600

San Jose, California 95110

Attention: Chief Executive Officer

with a copy to:

Sullivan & Cromwell LLP

1870 Embarcadero Road

Palo Alto, California 94303

Attention: Scott D. Miller

Sarah P. Payne

if to the Company

Hypercom Corporation

8888 East Raintree Drive, Suite 300

Scottsdale, Arizona 85260

Attention: Chief Executive Officer

with a copy to:

DLA Piper LLP 2525

East Camelback Road

Suite 1000

Phoenix, AZ 85016-4232

Attention: Steven D. Pidgeon

and to:

DLA Piper LLP

2000 University Avenue

Suite 1000 East Palo Alto, CA 94303-2215

Attention: Richard S. Millard

or to such other persons or addresses as may be designated in writing by the party to receive such notice as provided above.

9.7 Entire Agreement. This Agreement (including any exhibits hereto), the Confidentiality Agreement and the Company Disclosure Letter constitute the entire agreement, and supersede all other prior agreements, understandings, representations and warranties both written and oral, among the parties, with respect to the subject matter hereof.

9.8 <u>No Third-Party Beneficiaries</u>. (a) Except as specifically set forth in this Section 9.8(a), Parent and the Company hereby agree that their respective representations, warranties and covenants set forth herein are solely for the benefit of the other party hereto, in accordance with and subject to the terms of this Agreement and this Agreement is not intended to, and does not, confer upon any Person other than the parties hereto

Edgar Filing: VERIFONE SYSTEMS, INC. - Form S-4

any rights or remedies hereunder, including the right to rely upon the representations and warranties set forth herein. Notwithstanding the foregoing, it is agreed that from and after the Effective Time each Indemnified Party is an intended third-party beneficiary of the provisions of Section 6.11. Each intended third-party beneficiary shall have the right to enforce by appropriate action (including suit for specific performance) the provisions of this Agreement intended to benefit such person.

(b) The representations and warranties in this Agreement are the product of negotiations among the parties hereto and are for the sole benefit of the parties hereto. Any inaccuracies in such representations and warranties are subject to waiver by the parties hereto in accordance with Section 9.3 without notice or liability to any other Person. In some instances, the representations and warranties in this Agreement may represent an allocation among the parties hereto of risks associated with particular matters regardless of the knowledge of any of the parties hereto. Consequently, Persons other than the parties hereto may not rely upon the representations and warranties in this Agreement as characterizations of actual facts or circumstances as of the date of this Agreement or as of any other date.

9.9 <u>Obligations of Parent and of the Company</u>. Whenever this Agreement requires a Subsidiary of Parent to take any action, such requirement shall be deemed to include an undertaking on the part of Parent to cause such Subsidiary to take such action. Whenever this Agreement requires a Subsidiary of the Company to take any action, such requirement shall be deemed to include an undertaking on the part of the Company to cause such Subsidiary to take such action and, after the Effective Time, on the part of the Surviving Corporation to cause such Subsidiary to take such action.

9.10 <u>Severability</u>. The provisions of this Agreement shall be deemed severable and the invalidity or unenforceability of any provision shall not affect the validity or enforceability or the other provisions of this Agreement. If any provision of this Agreement, or the application of such provision to any Person or any circumstance, is invalid or unenforceable, (a) a suitable and equitable provision shall be substituted therefor in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid or unenforceable provision and (b) the remainder of this Agreement and the application of such provision to other Persons or circumstances shall not, subject to clause (a), be affected by such invalidity or unenforceability, nor shall such invalidity or unenforceability affect the validity or enforceability of such provision, or the application of such provision, in any other jurisdiction.

9.11 Definitions. Each of the terms set forth in Annex A is defined in the Section of this Agreement set forth opposite such term.

9.12 Interpretation. (a) The table of contents and headings herein are for convenience of reference only, do not constitute part of this Agreement and shall not be deemed to limit or otherwise affect any of the provisions of this Agreement. Where a reference in this Agreement is made to a Section or Exhibit, such reference shall be to a Section of or Exhibit to this Agreement unless otherwise indicated. Whenever the words include, includes or including are used in this Agreement, they shall be deemed to be followed by the words without limitation.

(b) The parties have participated jointly in negotiating and drafting this Agreement. In the event that an ambiguity or a question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the parties, and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provision of this Agreement.

9.13 <u>Captions</u>. The Article, Section and paragraph captions herein are for convenience of reference only, do not constitute part of this Agreement and shall not be deemed to limit or otherwise affect any of the provisions hereof.

9.14 <u>Assignment</u>. This Agreement shall not be assignable by operation of Law or otherwise; <u>provided</u>, <u>however</u>, that Parent may designate prior to the Effective Time, by written notice to the Company, another wholly-owned direct or indirect Subsidiary to be a party to the Merger in lieu of Merger Sub, in which event all references herein to Merger Sub shall be deemed references to such other Subsidiary, except that all representations and warranties made herein with respect to Merger Sub as of the date of this Agreement shall be deemed representations and warranties made with respect to such other Subsidiary as of the date of such designation. Any purported assignment in violation of this Agreement is void.

9.15 <u>Enforcement</u>. Each party agrees that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached by the other parties hereto. It is accordingly agreed that the other parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement against each party in any court set forth in Section 9.5(a), this being in addition to any other remedy to which the company may be entitled at law or in equity.

[Signatures Pages to Follow]

IN WITNESS WHEREOF, this Agreement has been duly executed and delivered by the duly authorized officers of the parties hereto as of the date first written above.

HYPERCOM CORPORATION

By: /s/ Philippe Tartavull Name: Philippe Tartavull Title: Chief Executive Officer and President

VERIFONE SYSTEMS, INC.

By: /s/ Albert Liu Name: Albert Liu Title: Senior Vice President, General Counsel and Secretary

HONEY ACQUISITION CO.

By: /s/ Albert Liu Name: Albert Liu Title: Senior Vice President

ANNEX A

INDEX OF DEFINED TERMS

Defined Term 5% Holders	Section 5 1(d)(iii)
Acquisition Proposal	5.1(d)(iii) 6.2(b)
Affected Employees	6.16(a)
Affiliates	5.1(e)(ii)
Aggregate Divestiture Commitment	6.6(a)
Agreement	Preamble
Alternative Acquisition Agreement	6.2(c)(i)(2)
Anti-Bribery Laws	5.1(w)
Antitrust Covenants	8.5(c)
Applicable Date	5.1(e)(i)
Asset Purchase Agreement	6.6(a)
Audit Date	5.1(e)(i)
Average Closing Price	4.2(e)
Bankruptcy and Equity Exception	5.1(c)(i)
Book-Entry Company Share	4.1(a)(ii)
Book-Entry Parent Shares	4.2(b)(i)
business day	1.2
By-Laws	2.2
Card Organizations	5.1(v)(ii)
Card Regulations	5.1(v)(ii)
Certificate	4.1(a)(ii)
Certificate Letter of Transmittal	4.2(b)(i)
Certificate of Merger	1.3
Charter	2.1
Closing	1.2
Closing Date	1.2
Code	Recitals
Common Stock Unit	5.1(b)(i)
Company	Preamble
Company Awards	4.4(b)
Company Change of Recommendation	6.2(c)(ii)
Company Compensation and Benefit Plan	5.1(h)(i)
Company Disclosure Letter	5.1
Company Labor Agreements	5.1(m)
Company Material Adverse Effect	5.1(a)
Company Non-U.S. Compensation and Benefit Plans	5.1(h)(i)
Company Option	4.4(a)
Company Preferred Shares	5.1(b)(i)
Company Recommendation	5.1(c)(ii)
Company Reports	5.1(e)(i)
Company Requisite Vote	5.1(c)(i)
Company Share	4.1(a)(i)
Company Shares	4.1(a)(i)
Company Stock Plans	5.1(b)(i)
Company Stockholders Meeting	6.4(a)
Company Warrant	4.5
Confidentiality Agreement	6.2(a)(iii)

	Sa effer
Defined Term Contract	Section 5.1(d)(ii)
Current Annual Premium	6.11(c)
D&O Insurance	6.11(c)
DGCL	1.1
Effective Time	1.3
Environmental Law	5.1(k)
ERISA	5.1(k) 5.1(h)(i)
ERISA Affiliate	5.1(h)(iii)
Exchange Act	5.1(d)(i)
Exchange Agent	4.2(c)
Exchange Fund	4.2(a)
Exchange Ratio	4.1(a)(i)
Excluded Company Shares	4.1(a)(i)
GAAP	5.1(e)(iv)
Government Contract	5.1(q)
Governmental Consents	7.1(c)
Governmental Entity	5.1(d)(i)
Hazardous Substance	5.1(k)
HSR Act	5.1(b)(ii)
Indemnified Parties	6.11(a)
Insurance Policies	5.1(p)
Intellectual Property	5.1(n)(vii)(1)
Inventory	5.1(u)
IRS	5.1(h)(ii)
IT Assets	5.1(n)(vii)(2)
knowledge	5.1(b)(iv)
Laws	5.1(i)(i)
Licenses	5.1(v)(i)
Lien	5.1(b)(i)
Loan and Security Agreement	7.2(e)
Material Contracts	5.1(q)(i)(K)
Maximum Divestiture Commitment	6.5(e)
Merger	Recitals
Merger Consideration	4.1(a)(i)
Merger Sub	Preamble
NYSE	5.1(d)(i)
OFAC	5.1(x)
Order	7.1(d)
Parent	Preamble
Parent Applicable Date	5.2(f)(i)
Parent Common Stock	4.1(a)(i)
Parent Disclosure Letter	5.2
Parent Material Adverse Effect	5.2(a)
Parent Preferred Shares	5.2(c)
Parent Reports	5.2(f)(i)
Parent Termination Fee	8.5(c)
Permitted Capitalization Change	7.2(a)
person	4.2(b)
Person	4.2(b)
Proceeding	5.1(l)(iii)
Product	5.1(s)

Defined Term	Section
Product Certifications	5.1(t)
Prospectus/Proxy Statement	6.3(a)
Registered	5.1(n)(vii)(3)
Registered Letter of Transmittal	4.2(b)(i)
Related Persons	8.4(a)
Representatives	6.2(a)
Rights	5.1(aa)
Rights Agreement	5.1(b)(i)
RoHS Directive	5.1(k)(ii)
S-4 Registration Statement	6.3(a)
Sarbanes-Oxley	5.1(e)(i)
SEC	4.4(c)
Securities Act	4.4(c)
Senior Officers	6.18(d)
Series A Junior Participating Preferred Stock	5.1(b)(i)
Software	5.1(n)(vii)(4)
Software Products	5.1(n)(vi)
Subsidiary	5.1(a)
Superior Proposal	6.2(b)
Support Agreement	Recitals
Surviving Corporation	1.1
Takeover Statute	5.1(j)
Tax	5.1(l)
Tax Return	5.1(l)
Termination Date	8.2
Termination Fee	8.5(b)
Trade Secrets	5.1(n)(vii)(1)
WEEE Directive	5.1(k)(ii)

ANNEX B

SUPPORT AGREEMENT

SUPPORT AGREEMENT (this <u>Agreement</u>), dated as of November 17, 2010, by and among FP Hypercom Holdco, LL<u>C</u>(FP), VeriFone Systems, Inc., a Delaware corporation (<u>VeriFone</u>) and Francisco Partners II, L.P., a Delaware limited partnership (<u>FP</u>LP). Except as otherwise provided herein, capitalized terms that are used but not otherwise defined herein shall have the meaning assigned to such terms in the Merger Agreement (as defined below).

WHEREAS, contemporaneously with the execution of this Agreement, VeriFone, Hypercom Corporation (<u>Hypercom</u>) and Honey Acquisition Co. (<u>Merger Sub</u>), a Delaware corporation and wholly-owned subsidiary of VeriFone, have entered into an Agreement and Plan of Merger (the <u>Merger Agreement</u>), providing for, among other things, the merger of Merger Sub with and into Hypercom, with Hypercom continuing as the surviving corporation and as a wholly owned subsidiary of VeriFone (the <u>Merger</u>);

WHEREAS, in order to induce VeriFone to enter into the Merger Agreement, FP and FP LP wish to enter into this Agreement;

NOW, THEREFORE, for good and valuable consideration, the receipt, sufficiency and adequacy of which is hereby acknowledged, the parties hereto agree as follows:

1. <u>Representations of FP</u>. FP and FP LP (collectively, the <u>FP Parties</u>) each represent and warrant to VeriFone that:

(a) the FP Parties own beneficially (as such term is defined in Rule 13d-3 of the Securities Exchange Act of 1934, as amended (the <u>Exchange Act</u>)) or of record the Warrant to Purchase Common Stock (the <u>Warrant</u>), issued by Hypercom on April 1, 2008, and pursuant to which FP may purchase 10,544,000 fully paid and nonassessable shares (the <u>Shares</u>) of Hypercom s common stock, par value \$0.001 per share (<u>Hypercom Common Stock</u>), from Hypercom, at the exercise price of \$5.00 per share (subject to adjustment as provided in the Warrant) free and clear of all Liens,

(b) the FP Parties do not beneficially own (as such term is used in Rule 13d-3 of the Exchange Act) any shares of Hypercom Common Stock other than the Shares and, except for the Warrant, does not have any options, warrants or other rights to acquire any additional shares of capital stock of Hypercom or any security exercisable for or convertible into shares of capital stock of Hypercom,

(c) the FP Parties have full power and authority and have taken all actions necessary to enter into, execute and deliver this Agreement and to perform fully their obligations hereunder,

(d) this Agreement has been duly executed and delivered and constitutes the legal, valid and binding obligation of the FP Parties enforceable against the FP Parties in accordance with its terms,

(e) other than filings under the Exchange Act, no notices, reports or other filings are required to be made by the FP Parties with, nor are any consents, registrations, approvals, permits or authorizations required to be obtained by the FP Parties from, any Governmental Entity, in connection with the execution and delivery of this Agreement by the FP Parties, and

(f) the execution, delivery and performance of this Agreement by the FP Parties does not, and the consummation by the FP Parties of the transactions contemplated hereby will not, result in a violation or breach of, or constitute (with or without due notice or lapse of time or both) a default (or give rise to any right of termination, cancellation, modification or acceleration) (whether after the giving of or the passage of time of

both) under any contract, agreement, arrangement or commitment to which either of the FP Parties is a party or which is binding on it or its assets and will not result in the creation of any Lien on, or security interest in, any of the assets on properties of either of the FP Parties.

2. <u>Agreement to Deliver Proxy</u>. To the extent FP exercises the Warrant, in whole or in part, on or prior to any record date for any meeting of holders of Company Shares relating to the Merger or any other merger, consolidation, business combination, reorganization, recapitalization, liquidation or sale or transfer of any material assets of Hypercom or its Subsidiaries (each such meeting, a <u>Stockholder Meeting</u>), FP agrees to deliver to VeriFone promptly upon VeriFone s request an irrevocable proxy substantially in the form attached hereto as Schedule A to Vote the Shares at every such Stockholder Meeting and at every adjournment or postponement thereof:

(a) in favor of adoption of the Merger Agreement and approval of the Merger and the transactions contemplated thereby, and

(b) against any action or agreement that would compete with, or materially impede, or interfere with or that would reasonably be expected to discourage the Merger or inhibit the timely consummation of the Merger.

The proxy delivered by FP pursuant to this Section 2 shall be irrevocable during the term of this Agreement to the extent permitted under Delaware law. For purposes of this Agreement, <u>Vote</u> includes voting in person or by proxy in favor of or against any action, otherwise consenting or withholding consent in respect of any action (including, but not limited to, consenting in accordance with Section 228 of the Delaware General Corporation Law) or taking other action in favor of or against any action. <u>Voting</u> shall have a correlative meaning.

3. <u>No Voting Trusts</u>. The FP Parties agree that they will not, nor will they permit any entity under their control to, deposit any of their Shares or New Shares (as defined in Section 6 hereof) in a Voting trust or subject any of their Shares or New Shares to any arrangement with respect to the Voting of such Shares or New Shares other than agreements entered into with VeriFone.

4. No Proxy Solicitations. The FP Parties agree that they will not, nor will they permit any entity under their control to:

(a) solicit proxies or become a participant in a solicitation (as such terms are defined in Regulation 14A under the Exchange Act) in opposition to or competition with the consummation of the Merger or otherwise assist any party in taking or planning any action which would compete with, or materially impede, or interfere with or that would reasonably be expected to discourage the Merger or inhibit the timely consummation of the Merger in accordance with the terms of the Merger Agreement,

(b) directly or indirectly encourage, initiate or cooperate in a stockholders Vote or action by consent of Hypercom s stockholders in opposition to or in competition with the consummation of the Merger, or

(c) become a member of a group (as such term is used in Section 13(d) of the Exchange Act) with respect to any voting securities of Hypercom for the purpose of opposing or competing with the consummation of the Merger.

5. <u>Transfer and Encumbrance</u>. On or after the date hereof and during the term of this Agreement, the FP Parties agree not to transfer, sell, offer, exchange, pledge or otherwise dispose of or encumber the Warrant or any of their Shares or New Shares, unless the person to whom such Warrant, Shares or New Shares, as the case may be, has agreed to be bound by the terms hereof in writing; and provided, that the foregoing restriction shall not apply after the Company Requisite Vote has been obtained.

6. <u>Additional Purchases</u>. The FP Parties agree that, other than by exercise of the Warrant, they will not purchase or otherwise acquire beneficial ownership (as such term is used in Rule 13d-3 of the Exchange Act) of

any shares of Hypercom Common Stock after the execution of this Agreement (<u>New Shares</u>), nor will the FP Parties voluntarily acquire the right to Vote or share in the Voting of any shares of Hypercom Common Stock other than the Shares, unless the FP Parties deliver to VeriFone immediately after such purchase or acquisition an irrevocable proxy substantially in the form attached hereto as Schedule A with respect to such New Shares. The FP Parties also agree that any New Shares acquired or purchased by them shall be subject to the terms of this Agreement to the same extent as if they constituted Shares.

7. <u>Consent</u>. FP hereby (i) to the extent required or contemplated by the Warrant, consents to the Merger and its consummation pursuant to the terms of the Merger Agreement, and waives any event of default under the Credit Agreement, dated as of February 13, 2008, by and among Hypercom and FP as Lender and Administrative Agent, arising by virtue of the entry into the Merger Agreement or the consummation of the Merger and consummation of the other transactions contemplated by the Merger Agreement (subject to the repayment in full of all principal, interest, fees and expenses thereunder on the Closing), and (ii) consents to the treatment of the Warrant pursuant to the terms Section 4.5 of the Merger Agreement and confirms that other than as provided for in the Merger Agreement, no other notice regarding the Merger is required under the Warrant and the Related Agreements (as defined in the Warrant) in connection with the Merger; <u>provided</u>, <u>however</u>, that in no event shall Section 4.5 of the Merger Agreement in substantially the form attached hereto as Schedule B. FP LP hereby agrees to deliver a payoff letter on or prior to the Closing in substantially the form attached hereto as Schedule C.

8. <u>Stockholder Capacity</u>. No Person executing this Agreement who is or becomes during the term hereof a director or officer of the Company shall be deemed to make any agreement or understanding in this Agreement in such Person s capacity as a director or officer and nothing herein shall affect the ability of any Person to take any action as director of the Company permissible under the Merger Agreement.

9. <u>Specific Performance</u>. The parties acknowledge that there may be no adequate remedy at law for a breach of this Agreement and that money damages may not be an appropriate remedy for breach of this Agreement. Therefore, the parties agree that each party has the right to injunctive relief and specific performance of this Agreement in the event of any breach hereof in addition to any rights it may have for damages, which shall include out of pocket expenses, loss of business opportunities and any other damages, direct and indirect, consequential, punitive or otherwise. The remedies set forth in this Section 8 are cumulative and shall in no way limit any other remedy any party hereto has at law, in equity or pursuant hereto.

10. Entire Agreement; Amendment; Waiver. This Agreement (including the schedules hereto) constitutes the entire agreement, and supersedes all prior agreements, understanding, representations and warranties both written and oral, between the parties, with respect to the subject matter hereof. Subject to the provisions of applicable Law, the parties hereto may modify or amend this Agreement, by written agreement executed and delivered by duly authorized officers of the respective parties. Any provision of this Agreement may be waived if, and only if, such waiver is in writing and signed by the party against whom the waiver is to be effective. No failure or delay by any party in exercising any right, power or privilege hereunder shall operate as a waiver thereof nor shall any single or partial exercise thereof preclude any other or further exercise thereof or the exercise of any other right, power or privilege. Except as otherwise herein provided, the rights and remedies herein provided shall be cumulative and not exclusive of any rights or remedies provided by law.

11. <u>Notices</u>. Notices, requests, instructions or other documents to be given under this Agreement shall be in writing and shall be deemed given, (i) when delivered, if delivered personally to the intended recipient, and (ii) one business day later, if sent by overnight delivery via a national courier service, and in each case, addressed to a party at the following address for such party:

To VeriFone:

VeriFone Systems, Inc.

2099 Gateway Place

San Jose, California 95110

Telephone: 408-232-7800

Attention: Albert Liu

With a copy to:

Sullivan & Cromwell LLP

1870 Embarcadero Road

Palo Alto, California 94303

Telephone: 650-461-5600

Attention: Scott D. Miller

Sarah P. Payne

To FP or FP LP:

FP Hypercom Holdco, LLC

Francisco Partners II, L.P.

One Letterman Drive

Building C, Suite 410

San Francisco, CA 94129

Telephone: 415 418-2900

Attention: Keith Geeslin

With a copy to:

Shearman & Sterling

525 Market Street

San Francisco, CA 94104

Table of Contents

415 616-1248

Michael J. Kennedy

12. Miscellaneous.

(a) <u>Governing Law</u>. THIS AGREEMENT SHALL BE DEEMED TO BE MADE IN AND IN ALL RESPECTS SHALL BE INTERPRETED, CONSTRUED AND GOVERNED BY AND IN ACCORDANCE WITH THE LAW OF THE STATE OF DELAWARE WITHOUT REGARD TO THE CONFLICT OF LAW PRINCIPLES THEREOF TO THE EXTENT THAT SUCH PRINCIPLES WOULD DIRECT A MATTER TO ANOTHER JURISDICTION.

(b) <u>Venue</u>. The parties hereby irrevocably submit to the personal jurisdiction of the courts of the State of Delaware and the Federal courts of the United States of America located in the State of Delaware solely in respect of the interpretation and enforcement of this Agreement, and in respect of the transactions contemplated hereby, and hereby waive, and agree not to assert, as a defense in any action, suit or proceeding for the interpretation or enforcement of this Agreement or of any such document, that it is not subject thereto or that such action, suit or proceeding may not be brought or is not maintainable in said courts or that the venue thereof may not be appropriate or that this Agreement or any such document may not be enforced in or by such courts,

and the parties hereto irrevocably agree that all claims with respect to such action, proceeding or transactions shall be heard and determined in such a Delaware State or Federal court. The parties hereby consent to and grant any such court jurisdiction over the Person of such parties and, to the extent permitted by Law, over the subject matter of such dispute and agree that mailing of process or other papers in connection with any such action or proceeding in the manner provided in Section 10 or in such other manner as may be permitted by Law, shall be valid and sufficient service thereof.

(c) <u>WAIVER OF JURY TRIAL</u>. EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (i) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (ii) EACH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (iii) EACH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (iv) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 11(C).

(d) <u>Severability</u>. The provisions of this Agreement shall be deemed severable and the invalidity or unenforceability of any provision shall not affect the validity or enforceability of the other provisions of this Agreement. If any provision of this Agreement, or the application of such provision to any Person or any circumstance, is invalid or unenforceable, (a) a suitable and equitable provision shall be substituted therefor in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid or unenforceable provision and (b) the remainder of this Agreement and the application of such provision to other Persons or circumstances shall not, subject to clause (a), be affected by such invalidity or unenforceability, nor shall such invalidity or unenforceability affect the validity or enforceability of such provision, or the application of such provision, in any other jurisdiction.

(e) <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, each such counterpart being deemed to be an original instrument, and all such counterparts shall together constitute the same agreement.

(f) Termination. This Agreement shall terminate upon the earliest to occur of (i) the Closing and (ii) the termination of the Merger Agreement.

(g) <u>Headings</u>. The headings herein are for convenience of reference only, do not constitute part of this Agreement and shall not be deemed to limit or otherwise affect any of the provisions of this Agreement.

(h) <u>THIRD PARTY BENEFICIARIES</u>. NOTHING IN THIS AGREEMENT, EXPRESS OR IMPLIED, IS INTENDED TO CONFER UPON ANY PERSON OTHER THAN VERIFONE, FP AND THEIR RESPECTIVE SUCCESSORS AND ASSIGNS, ANY RIGHTS OR REMEDIES UNDER OR BY REASON OF THIS AGREEMENT.

IN WITNESS WHEREOF, the parties hereto have executed and delivered this Agreement as of the date first written above.

VeriFone Systems, Inc.

By: /s/ Albert Liu Name: Albert Liu Title: Senior Vice President, General Counsel and Secretary

FP Hypercom Holdco, LLC

- By: Francisco Partners II, L.P. Its Managing Member
- By: Francisco Partners GP II, L.P. Its General Partner
- By: Francisco Partners GP II Management, LLC Its General Partner

By: /s/ Tom Ludwig Name: Tom Ludwig Title: Authorized Representative

Francisco Partners II, L.P.

By: Francisco Partners GP II, L.P.

Its General Partner

- By: Francisco Partners GP II Management, LLC Its General Partner
- By: /s/ Tom Ludwig Name: Tom Ludwig Title: Authorized Representative

(SCHEDULE A)

FORM OF PROXY

The undersigned, for consideration received, hereby appoints Douglas Bergeron or another representative of VeriFone Systems, Inc. designated by him and each of them our proxies, with power of substitution and resubstitution, to Vote (as defined below) all shares of Common Stock, par value 0.001 per share, of Hypercom Corporation, a Delaware corporation (<u>Hypercom</u>), owned by the undersigned (the <u>Shares</u>) as of the date hereof as follows:

(a) FOR (i) adoption of the Agreement and Plan of Merger (the <u>Merger Agreement</u>), dated as of November 17, 2010, by and among Hypercom, VeriFone Systems, Inc. (<u>VeriFone</u>), a Delaware corporation, and Honey Acquisition Co. (<u>Merger Sub</u>), a Delaware corporation and wholly-owned subsidiary of VeriFone, and (ii) approval of the merger of Merger Sub with and into Hypercom, with Hypercom continuing as the surviving corporation and as a wholly owned subsidiary of VeriFone (the <u>Merger</u>); and

(b) AGAINST any action or agreement that would compete with, or materially impede, or interfere with or that would reasonably be expected to discourage the Merger or inhibit the timely consummation of the Merger

<u>Vote</u> means voting in person or by proxy in favor of or against any action, otherwise consenting or withholding consent in respect of any action (including, but not limited to, consenting in accordance with Section 228 of the Delaware General Corporation Law) or taking other action in favor of or against any action. This proxy applies to any Vote (i) at any meeting of the stockholders of Hypercom, and any adjournment or postponement thereof, at which the matters described above are considered, including the special meeting of stockholders of Hypercom to be held as soon as practicable after the date hereof or (ii) in connection with any written consent of stockholders of Hypercom.

This proxy is coupled with an interest, revokes all prior proxies granted by the undersigned and is irrevocable until such time as the Support Agreement (the <u>Support Agreement</u>), dated as of November 17, 2010, between the undersigned and VeriFone, terminates in accordance with its terms, at which time this proxy shall expire.

Except as otherwise provided herein, capitalized terms that are used but not otherwise defined herein shall have the meaning assigned to such terms pursuant to the Voting Agreement.

Dated November , 2010

[FP Hypercom Holdco, LLC][Francisco Partners II, L.P.]

By: Name: Title:

ANNEX C

UBS Securities LLC

555 California Street, 36th Floor

San Francisco, CA 94104

www.ubs.com

November 16, 2010

The Board of Directors

Hypercom Corporation

8888 East Raintree Drive, Suite 300

Scottsdale, AZ 85260

Dear Members of the Board

We understand that Hypercom Corporation, a Delaware corporation (the Company), is considering a transaction whereby VeriFone Systems, Inc., a Delaware corporation (VeriFone), will effect a merger involving the Company. Pursuant to the terms of an Agreement and Plan of Merger, draft dated as of November 13, 2010 (the Agreement), among VeriFone, the Company and a wholly owned subsidiary of VeriFone (Sub), Sub will undertake a series of transactions whereby the Company will become a wholly owned subsidiary of VeriFone (the Transaction). Pursuant to the terms of the Agreement all of the issued and outstanding shares of the common stock, par value of \$0.001 per share, of the Company (Company Common Stock) will be converted into the right to receive, for each outstanding share of Company Common Stock, 0.230 (the Exchange Ratio) shares of the common stock, par value \$0.01 per share, of VeriFone (VeriFone Common Stock).

The terms and conditions of the Transaction are more fully set forth in the Agreement.

You have requested our opinion as to the fairness, from a financial point of view, to the holders of Company Common Stock (other than FP Hypercom Holdco LLC and its affiliates, the Excluded Holders), of the Exchange Ratio provided for in the Transaction. Our opinion addresses the Exchange Ratio on a per share basis, without taking into consideration any individual holder s percentage ownership of Company Common Stock.

UBS Securities LLC (UBS) has acted as financial advisor to the Company in connection with the Transaction and will receive a fee for its services, a portion of which is payable in connection with this opinion and a significant portion of which is contingent upon consummation of the Transaction. In the past, UBS and its affiliates have provided investment banking services to the Company unrelated to the proposed Transaction, for which UBS and its affiliates received compensation, including having acted as financial advisor to the Company in 2007 and 2008 in its acquisition of the e-Transaction business line of Thales SA. In the ordinary course of business, UBS and its affiliates may hold or trade, for their own accounts and the accounts of their customers, securities of the Company and VeriFone and, accordingly, may at any time hold a long or short position in such securities. The issuance of this opinion was approved by an authorized committee of UBS.

Our opinion does not address the relative merits of the Transaction as compared to other business strategies or transactions that might be available with respect to the Company or the Company s underlying business decision to effect the Transaction. Our opinion does not constitute a recommendation to any shareholder as to how such shareholder should vote or act with respect to the Transaction. At your direction, we have not been asked to, nor do we, offer any opinion as to the terms, other than the Exchange Ratio to the extent expressly specified herein, of the Agreement or the form of the Transaction. In addition, we express no opinion as to the fairness of the amount or nature of any compensation to be received by any officers, directors or employees of any parties to the Transaction, or any class of such persons, relative to the Exchange Ratio. We express no opinion as to what the value of VeriFone Common Stock will be when issued pursuant to the Transaction or the UBS Investment Bank is a business group of UBS AG.

UBS Securities LLC is a subsidiary of UBS AG.

UBS Securities LLC

555 California Street, 36th Floor

San Francisco, CA 94104

www.ubs.com

price at which VeriFone Common Stock or Company Common Stock will trade at any time. In rendering this opinion, we have assumed, with your consent, that (i) the final executed form of the Agreement will not differ in any material respect from the draft that we have reviewed, (ii) the parties to the Agreement will comply with all material terms of the Agreement, and (iii) the Transaction will be consummated in accordance with the terms of the Agreement without any adverse waiver or amendment of any material term or condition thereof. We have also assumed that all governmental, regulatory or other consents and approvals necessary for the consummation of the Transaction will be obtained without any material adverse effect on the Company, VeriFone or the Transaction.

In arriving at our opinion, we have, among other things: (i) reviewed certain publicly available business and financial information relating to the Company and VeriFone; (ii) reviewed certain internal financial information and other data relating to the business and financial prospects of the Company that were not publicly available, including financial forecasts and estimates prepared by the management of the Company that you have directed us to utilize for purposes of our analysis; (iii) reviewed certain internal financial information and other data relating to the business and financial prospects of VeriFone that were not publicly available, including financial forecasts and estimates prepared by the management of the business and financial prospects of VeriFone that were not publicly available, including financial forecasts and estimates prepared by the management of VeriFone, as modified by the management of the Company, that you have directed us to utilize for purposes of our analysis; (iv) reviewed certain estimates of synergies prepared by the management of the Company that were not publicly available that you have directed us to utilize for purposes of our analysis; (v) conducted discussions with members of the senior managements of the Company and VeriFone concerning the business and financial prospects of the Company and VeriFone; (vi) reviewed publicly available financial and stock market data with respect to certain other companies we believe to be generally relevant; (vii) reviewed current and historical market prices of Company Common Stock and VeriFone Common Stock; (viii) reviewed the Agreement; and (ix) conducted such other financial studies, analyses and investigations, and considered such other information, as we deemed necessary or appropriate.

In connection with our review, with your consent, we have assumed and relied upon, without independent verification, the accuracy and completeness in all material respects of the information provided to or reviewed by us for the purpose of this opinion. In addition, with your consent, we have not made any independent evaluation or appraisal of any of the assets or liabilities (contingent or otherwise) of the Company or VeriFone, nor have we been furnished with any such evaluation or appraisal. With respect to the financial forecasts, estimates and synergies referred to above, we have assumed, at your direction, that they have been reasonably prepared on a basis reflecting the best currently available estimates and judgments of the management of the Company as to the future financial performance of the Company, VeriFone and such synergies. In addition, we have assumed with your approval that the financial forecasts and estimates, including synergies, referred to above will be achieved at the times and in the amounts projected. We also have assumed, with your consent, that the Transaction will qualify for U.S. federal income tax purposes as a reorganization within the meaning of Section 368(a) of the Internal Revenue Code of 1986, as amended. Our opinion is necessarily based on economic, monetary, market and other conditions as in effect on, and the information available to us as of, the date hereof.

Based upon and subject to the foregoing, it is our opinion that, as of the date hereof, the Exchange Ratio provided for in the Transaction is fair, from a financial point of view, to holders of Company Common Stock (other than the Excluded Holders, as to which we express no opinion).

UBS Investment Bank is a business group of UBS AG.

UBS Securities LLC is a subsidiary of UBS AG.

UBS Securities LLC

555 California Street, 36th Floor

San Francisco, CA 94104

www.ubs.com

This opinion is provided for the benefit of the Board of Directors, in their capacity as such, in connection with, and for the purpose of, its evaluation of the Exchange Ratio provided for in the Transaction.

Very truly yours,

UBS SECURITIES LLC

By: /s/ Gavin W. Slader Name: Gavin W. Slader Title: Executive Director

UBS Investment Bank is a business group of UBS AG.

UBS Securities LLC is a subsidiary of UBS AG.

By:/s/ Seth SunderlandName:Seth SunderlandTitle:Associate Director

C-3

PART II

INFORMATION NOT REQUIRED IN PROSPECTUS

ITEM 20. INDEMNIFICATION OF DIRECTORS AND OFFICERS

Section 102(b)(7) of the Delaware General Corporation Law, or DGCL, as amended, allows a corporation to provide in its certificate of incorporation that a director of the corporation will not be personally liable to the corporation or its stockholders for monetary damages for breach of fiduciary duty as a director, except where the director breached the duty of loyalty, failed to act in good faith, engaged in intentional misconduct or knowingly violated a law, authorized the payment of a dividend or approved a stock repurchase in violation of Delaware corporate law or obtained an improper personal benefit. VeriFone s certificate of incorporation provides for this limitation of liability.

Section 145 of the DGCL provides that a corporation may indemnify directors and officers as well as other employees and individuals against expenses (including attorneys fees), judgments, fines and amounts paid in settlement actually and reasonably incurred by such a person in connection with any threatened, pending or completed actions, suits or proceedings in which such a person is made a party by reason of being or having been a director, officer, employee or agent of the corporation, subject to certain limitations. The statute provides that it is not exclusive of other rights to which those seeking indemnification may be entitled under any bylaw, agreement, vote of stockholders or disinterested directors or otherwise. Article Eight of VeriFone s certificate of incorporation provides for indemnification by VeriFone of its directors, officers and employees to the fullest extent permitted by the DGCL.

The registrant expects to maintain standard policies of insurance that provide coverage (1) to its directors and officers against loss rising from claims made by reason of breach of duty or other wrongful act, and (2) to VeriFone with respect to indemnification payments that it may make to such directors and officers.

ITEM 21. EXHIBITS AND FINANCIAL STATEMENT SCHEDULES

Exhibit	
No.	Description
2.1	Agreement and Plan of Merger, dated as of November 17, 2010, among Hypercom Corporation, VeriFone Systems, Inc. and Honey Acquisition Co. (attached as Annex A to the proxy statement/prospectus contained in this registration statement).
2.2	Support Agreement, dated as of November 17, 2010, among VeriFone Systems, Inc., FP Hypercom Holdco, LLC and Francisco Partners II, L.P. (attached as Annex B to the proxy statement/prospectus contained in this registration statement).
3.1	Amended and Restated Certificate of Incorporation of VeriFone Systems, Inc. (incorporated herein by reference to Exhibit 3.1 to the Annual Report on Form 10-K of VeriFone Systems, Inc. filed on December 21, 2010).
3.3	Bylaws of VeriFone Holdings, Inc. (incorporated herein by reference to Exhibit 3.2 to the Registration Statement on Form S-1/A of VeriFone Holdings, Inc. filed on April 29, 2005).
3.4	Amendment No. 1 to the Bylaws of VeriFone Holdings, Inc. (incorporated herein by reference to Exhibit 3.3 to the Current Report on Form 8-K of VeriFone Holdings, Inc. filed on April 3, 2008).
5.1	Opinion of Sullivan & Cromwell LLP.
8.1	Tax Opinion of Sullivan & Cromwell LLP.
8.2	Tax Opinion of DLA Piper LLP (US).

Exhibit	
No.	

Description

23.1	Consent of Ernst & Young LLP relating to VeriFone s financial statements.
23.2	Consent of Ernst & Young LLP relating to Hypercom s financial statements.
23.3	Consent of Sullivan & Cromwell LLP (included in Exhibits 5.1 and 8.1).
23.4	Consent of DLA Piper LLP (US) (included in Exhibit 8.2).
24.1	Powers of Attorney (included on the signature page of this Registration Statement).
99.1	Form of Hypercom Proxy Card.
99.2	Consent of UBS Securities LLC.

ITEM 22. UNDERTAKINGS

(a) The undersigned registrant hereby undertakes:

(1) To file, during any period in which offers or sales are being made, a post-effective amendment to this registration statement:

(i) To include any prospectus required by section 10(a)(3) of the Securities Act of 1933.

(ii) To reflect in the prospectus any facts or events arising after the effective date of the registration statement (or the most recent post-effective amendment thereof) which, individually or in the aggregate, represent a fundamental change in the information set forth in the registration statement. Notwithstanding the foregoing, any increase or decrease in volume of securities offered (if the total dollar value of securities offered would not exceed that which was registered) and any deviation from the low or high end of the estimated maximum offering range may be reflected in the form of prospectus filed with the Commission pursuant to Rule 424(b) if, in the aggregate, the changes in volume and price represent no more than a 20 percent change in the maximum aggregate offering price set forth in the Calculation of Registration Fee table in the effective registration statement.

(iii) To include any material information with respect to the plan of distribution not previously disclosed in the registration statement or any material change to such information in the registration statement.

(2) That, for the purpose of determining any liability under the Securities Act of 1933, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.

(3) To remove from registration by means of a post-effective amendment any of the securities being registered which remain unsold at the termination of the offering.

(4) That, for the purpose of determining liability under the Securities Act of 1933 to any purchaser:

(i) If the registrant is relying on Rule 430B (§230.430B of this chapter):

A. Each prospectus filed by the registrant pursuant to Rule 424(b)(3) shall be deemed to be part of the registration statement as of the date the filed prospectus was deemed part of and included in the registration statement; and

B. Each prospectus required to be filed pursuant to Rule 424(b)(2), (b)(5), or (b)(7) as part of a registration statement in reliance on Rule 430B relating to an offering made pursuant to Rule 415(a)(1)(i), (vii), or (x) for the purpose of providing the information required by section 10(a) of the Securities Act of 1933 shall be deemed to be part of and included in the registration statement as of the earlier of the date such form of prospectus is first used after effectiveness or the date of

the first contract of sale of securities in the offering described in the prospectus. As provided in Rule 430B, for liability purposes of the issuer and any person that is at that date an underwriter, such date shall be deemed to be a new effective date of the registration statement relating to the securities in the registration statement to which that prospectus relates, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof. Provided, however, that no statement made in a registration statement or prospectus that is part of the registration statement or made in a document incorporated or deemed incorporated by reference into the registration statement or prospectus that is part of the registration statement will, as to a purchaser with a time of contract of sale prior to such effective date, supersede or modify any statement that was made in the registration statement or prospectus that was part of the registration statement or made in any such document immediately prior to such effective date; or

(ii) If the registrant is subject to Rule 430C, each prospectus filed pursuant to Rule 424(b) as part of a registration statement relating to an offering, other than registration statements relying on Rule 430B or other than prospectuses filed in reliance on Rule 430A, shall be deemed to be part of and included in the registration statement as of the date it is first used after effectiveness. Provided, however, that no statement made in a registration statement or prospectus that is part of the registration statement or made in a document incorporated or deemed incorporated by reference into the registration statement or prospectus that is part of the registration statement will, as to a purchaser with a time of contract of sale prior to such first use, supersede or modify any statement that was made in the registration statement or prospectus that was part of the registration statement or prospectus that was part of the registration statement or prospectus that was part of the registration statement or prospectus that was part of the registration statement or prospectus that was part of the registration statement or prospectus that was part of the registration statement or prospectus that was part of the registration statement or prospectus that was part of the registration statement or prospectus that was part of the registration statement or prospectus that was part of the registration statement or prospectus that was part of the registration statement or prospectus that was part of the registration statement or prospectus that was part of the registration statement or prospectus that is part of the registration statement or prospectus that was part of the registration statement or prospectus that was part of the registration statement or made in any such document immediately prior to such date of first use.

(5) That, for the purpose of determining any liability under the Securities Act of 1933 to any purchaser in the initial distribution of the securities, in a primary offering of securities of the undersigned registrant pursuant to this registration statement, regardless of the underwriting method used to sell the securities to the purchaser, if the securities are offered or sold to such purchaser by means of any of the following communications, the undersigned registrant will be a seller to the purchaser and will be considered to offer or sell such securities to such purchaser:

(i) Any preliminary prospectus or prospectus of the undersigned registrant relating to the offering required to be filed pursuant to Rule 424;

(ii) Any free writing prospectus relating to the offering prepared by or on behalf of the undersigned registrant or used or referred to by the undersigned registrant;

(iii) The portion of any other free writing prospectus relating to the offering containing material information about the undersigned registrant or its securities provided by or on behalf of the undersigned registrant; and

(iv) Any other communication that is an offer in the offering made by the undersigned registrant to the purchaser.

(b) The undersigned registrant hereby undertakes that, for purposes of determining any liability under the Securities Act of 1933, each filing of the registrant s annual report pursuant to Section 13(a) or Section 15(d) of the Securities Exchange Act (and, where applicable, each filing of an employee benefit plan s annual report pursuant to Section 15(d) of the Securities Exchange Act) that is incorporated by reference in the registration statement shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.

(c) The undersigned registrant hereby undertakes as follows: that prior to any public reoffering of the securities registered hereunder through use of a prospectus which is a part of this registration statement, by any person or party who is deemed to be an underwriter within the meaning of Rule 145(c), the issuer undertakes that such reoffering prospectus will contain the information called for by the applicable registration form with respect to reofferings by persons who may be deemed underwriters, in addition to the information called for by the other items of the applicable form.

(d) The registrant undertakes that every prospectus: (i) that is filed pursuant to paragraph (c) immediately preceding, or (ii) that purports to meet the requirements of Section 10(a)(3) of the Act and is used in connection with an offering of securities subject to Rule 415, will be filed as a part of an amendment to the registration statement and will not be used until such amendment is effective, and that, for purposes of determining any liability under the Securities Act of 1933, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.

(e) Insofar as indemnification for liabilities arising under the Securities Act of 1933 may be permitted to directors, officers, and controlling persons of the registrant pursuant to the foregoing provisions, or otherwise, the registrant has been advised that in the opinion of the Securities and Exchange Commission such indemnification is against public policy as expressed in the Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the registrant of expenses incurred or paid by a director, officer, or controlling person of the registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer, or controlling person in connection with the securities being registered, the registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Act and will be governed by the final adjudication of such issue.

(f) The undersigned registrant hereby undertakes to respond to requests for information that is incorporated by reference into the prospectus pursuant to Item 4, 10(b), 11, or 13 of this form, within one business day of receipt of such request, and to send the incorporated documents by first class mail or other equally prompt means. This includes information contained in documents filed subsequent to the effective date of the registration statement through the date of responding to the request.

(g) The undersigned registrant hereby undertakes to supply by means of a post-effective amendment all information concerning a transaction, and the company being acquired involved therein, that was not the subject of and included in the registration statement when it became effective.

SIGNATURES

Pursuant to the requirements of the Securities Act of 1933, the registrant has duly caused this Registration Statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of San Jose, State of California, on the 21st day of December, 2010.

VeriFone Systems, Inc.

By:/s/DOUGLAS G. BERGERONName:Douglas G. BergeronTitle:Chief Executive Officer

POWER OF ATTORNEY

KNOW ALL PERSONS BY THESE PRESENTS, that each person whose signature appears below hereby constitutes and appoints Douglas G. Bergeron, Robert Dykes and Albert Liu, and each of them acting individually, as his true and lawful attorneys-in-fact and agents, each with full power of substitution, for the undersigned in any and all capacities, to sign any and all amendments to this Registration Statement (including post-effective amendments or any abbreviated registration statement and any amendments thereto filed pursuant to Rule 462(b) increasing the number of securities for which registration is sought), and to file the same, with all exhibits thereto and other documents in connection therewith, with the Securities and Exchange Commission, granting unto said attorneys-in-fact and agents, with full power of each to act alone, full power and authority to do and perform each and every act and thing requisite and necessary to be done in connection therewith, as fully for all intents and purposes as the undersigned might or could do in person, hereby ratifying and confirming all that said attorneys-in-fact and agents, or his or their substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

Pursuant to the requirements of the Securities Act of 1933, this Registration Statement has been signed by the following persons in the capacities and on the dates indicated:

Signature	Title	Date
/s/ Douglas G. Bergeron	Chief Executive Officer and Director	December 21, 2010
Douglas G. Bergeron	(principal executive officer)	
/s/ Robert Dykes	Senior Vice President and	December 21, 2010
Robert Dykes	Chief Financial Officer	
	(principal financial and accounting officer)	
/s/ Robert W. Alspaugh	Director	December 21, 2010
Robert W. Alspaugh		
/s/ Dr. Leslie G. Denend	Director	December 21, 2010
Dr. Leslie G. Denend		
/s/ Alex W. Hart	Director	December 21, 2010
Alex W. Hart		
/s/ Robert B. Henske	Director	December 21, 2010
Robert B. Henske		
/s/ Richard A. McGinn	Director	December 21, 2010
Richard A. McGinn		
/s/ Eitan Raff	Director	December 21, 2010
Eitan Raff		
/s/ Charles R. Rinehart	Director	December 21, 2010

Edgar Filing: VERIFONE SYSTEMS, INC. - Form S-4

Charles R. Rinehart

/s/ Jeffrey E. Stiefler

Director

December 21, 2010

Jeffrey E. Stiefler

EXHIBIT INDEX

Exhibit

No.	Description
2.1	Agreement and Plan of Merger, dated as of November 17, 2010, among Hypercom Corporation, VeriFone Systems, Inc. and Honey Acquisition Co. (attached as Annex A to the proxy statement/prospectus contained in this registration statement).
2.2	Support Agreement, dated as of November 17, 2010, among VeriFone Systems, Inc., FP Hypercom Holdco, LLC and Francisco Partners II, L.P. (attached as Annex B to the proxy statement/prospectus contained in this registration statement).
3.1	Amended and Restated Certificate of Incorporation of VeriFone Systems, Inc. (incorporated herein by reference to Exhibit 3.1 to the Annual Report on Form 10-K of VeriFone Systems, Inc. filed on December 21, 2010).
3.3	Bylaws of VeriFone Holdings, Inc. (incorporated herein by reference to Exhibit 3.2 to the Registration Statement on Form S-1/A of VeriFone Holdings, Inc. filed on April 29, 2005).
3.4	Amendment No. 1 to the Bylaws of VeriFone Holdings, Inc. (incorporated herein by reference to Exhibit 3.3 to the Current Report on Form 8-K of VeriFone Holdings, Inc. filed on April 3, 2008).
5.1	Opinion of Sullivan & Cromwell LLP.
8.1	Tax Opinion of Sullivan & Cromwell LLP.
8.2	Tax Opinion of DLA Piper LLP (US).
23.1	Consent of Ernst & Young LLP relating to VeriFone s financial statements.
23.2	Consent of Ernst & Young LLP relating to Hypercom s financial statements.
23.3	Consent of Sullivan & Cromwell LLP (included in Exhibits 5.1 and 8.1).
23.4	Consent of DLA Piper LLP (US) (included in Exhibit 8.2).
24.1	Powers of Attorney (included on the signature page of this Registration Statement).
99.1	Form of Hypercom Proxy Card.

99.2 Consent of UBS Securities LLC.